

Swampy territory: The role of the palliative care social worker in safeguarding children of adults who are receiving specialist palliative care

**Carole Ruth Comben** 

Thesis submitted for the degree of Doctor of Philosophy University of Stirling

This study is dedicated to my daughter Harriet who has given me much needed support throughout my journey. I also thank my family and friends who have over the period maintained an interest in what I have been striving to achieve.

My supervisors Brigid Daniel and Sarah Wilson have provided encouragement, direction and support with patience and understanding, particularly during the analysis and write up, for which I am sincerely grateful. I would also like to thank Brenda Gillies who was a supervisor during the initial phase of my study and who took the time to comment on an early draft.

I am indebted to all those palliative care social workers who were interested enough in the research to participate in focus groups and interviews and to complete a questionnaire. Without their willingness to share their experiences and thoughts this study would not have been possible. My thanks also go to those who helped me with the practical arrangements for the fieldwork, both within the Association of Palliative Care Social Workers and the Department of Applied Social Science at Stirling University.

My fellow students have provided friendship and good humour which helped to keep me buoyant during the long process. Special thanks go to Helen Whincup for her thoughts on an earlier draft and particularly to Jane Robertson who proof-read the thesis prior to submission.

i

## Abstract

There is minimal research into families where a person who is receiving palliative care has dependent children. In particular, there is a lack of information about the work that palliative care social workers undertake with such families. This study examines the role of the palliative care social worker in safeguarding and promoting the welfare of children of adults receiving specialist palliative care. This is a qualitative study involving the collection of information from a total of 57 palliative care social workers in three ways: focus groups, survey and semi-structured interviews. The analysis demonstrates that palliative care social workers understand the term 'safeguarding children' to mean more than child protection alone. The analysis also focuses on the process of referral, institutional barriers, supports for their work with children, and inter-agency collaboration. An equally important aspect of the analysis, and one generally neglected in psycho-social studies, relates to the socio-economic context of the palliative care social workers' work, including the effects of the immigration status of families on children.

In the main, concerns about the present and future care of children come to the attention of palliative care social workers via members of the multidisciplinary team in which they are based. The numbers are reported to be relatively small in contrast to the total number of referrals received. The extent to which support to children is provided varies considerably; some palliative care agencies do not encourage palliative care social workers to work directly with children prior to bereavement and others restrict direct work with children post-bereavement. The main emphasis for all workers, however, is on support to the parents to help them understand their children's needs during the terminal phase of the illness. The well-being of children at this time of emotional stress is included in the palliative care social workers' definition of 'safeguarding children'. In addition, the

Abstract

wide-ranging examination of the palliative care social workers' involvement with the families illustrates the extent of the dedication involved in an under-resourced and little understood area of their work. Whilst palliative care social workers reported receiving basic training on child protection within their agency, with some themselves providing this training, further training on safeguarding children is not always available or known about. The amount and type of professional supervision and support is also varied, particularly in relation to child-care matters; not all have access to supervision from another social work professional.

There are a small number of examples given in the study where children are considered to be at risk in some way. Children cared for by lone parents are recognised as being especially vulnerable, particularly if future care plans are not in place before the death of the parent. Also identified as vulnerable are children of parents whose immigration status is in question as their future location may be in doubt, placing children and the surviving parent under additional stress. One of the major difficulties for palliative care social workers is securing support services from the local authority children's services. Whilst there are examples of collaboration and co-operation, the findings in this study echo those of others which examine the relationship between adult and children's services. In this study, palliative care social workers speak of frustration and disappointment in the responses they receive, and they are concerned that the needs of children of dying parents are not understood.

This study contributes to the debate about the focus of children in care services for adults and how staff may be supported in their task of safeguarding and promoting the welfare of children. The study also adds to the limited existing knowledge about palliative care social work practice.

iii

# **Table of Contents**

Acknowledgements	i
Abstract	ii
Table of Contents	iv
List of Tables	x
Chapter One: Introduction to the thesis	1
Rationale and context of the study	1
Aim of the study and development of research questions	6
Structure of the thesis	7
Notes	8
Chapter Two: Palliative care social work: context and role	9
Introduction	9
The context of palliative care social work	9
Role of the palliative care social worker	14
Work with children and their families	18
The palliative care social worker within the team	22
Knowledge and skill base of palliative care social work	25
Conclusion	26
Chapter Three: Children's experiences of living with a dying parent	28
Introduction	28
Incidence and prevalence	29
The nature of the literature on the impact of parental illness on	
children's lives	30
The child's experience of the terminal phase of a parent's illness	31
Coping with the terminal phase of a parent's illness	36

Family functioning	41
Conclusion	44
Chapter Four: Safeguarding children: implications for practitioners working	
with adults	45
Introduction	45
Background and context	45
Prevalence and incidence of abuse in the general population	49
Current practice	50
The role of adult services and children's needs	53
Inter-agency relationships	56
Parenting	59
Assessing children in need6	63
The child's voice6	65
Conclusion6	67
Chapter Five: The research journey 6	68
Introduction	68
Social work research6	68
Theoretical perspective6	69
Choosing the way to go 7	73
Starting out 7	75
Moving forward	76
Nearing the end, but still a long way to go	93
My 'self' in the research journey	97
Conclusion 1	103

experiences 1	05
Introduction 1	05
Description of practice examples 1	06
Effects of the parent's illness on children 1	07
Relationship context surrounding the illness 1	14
Broader social and economic context1	16
Discussion 1	18
Conclusion1	19
Chapter Seven: Palliative care social workers' perceptions of their role in	
safeguarding children1	21
Introduction 1	21
Safe in the widest sense 1	22
Putting 'a universal responsibility' into practice 1	24
Impact of the parent's illness on parenting capacity and meeting	
children's needs 1	27
How palliative care social workers perceived their role in safeguarding	
children 1	31
Emotional impact of the safeguarding role 1	33
Confidence in the safeguarding role 1	35
Palliative care social workers' thoughts on how they could be more	
effective in their role 1	36
Discussion 1	37
Conclusion 1	40

Chapter Six: The role of the palliative care social worker: the children's

to their attention 141
Introduction 141
Who makes the referrals? 141
Reasons for referrals 147
Incidence of referrals150
Types of referrals 151
Palliative care social workers' views about the referral process 154
Discussion 157
Conclusion 162
Chapter Nine: The role of the palliative care social worker: the work they
undertake with children and their parents 163
Introduction
Engagement with children and parents 163
Key features of the work with children and their parents
Practice examples 173
Process
Practice approaches196
Impact of the family's immigration status 200
Conclusion 202
Chapter Ten: The role of the palliative care social worker: liaising with local
authority children's services 205
Introduction 205
Amount and reasons for contact 205
Communication
Seeking advice and guidance 209

Chapter Eight: The role of the palliative care social worker: how children come

	Issues	around referrals	210
	Relatio	onships with local authority social workers	213
	Percep	otions of role	216
	Improv	ving relationships	220
	Discus	sion	221
	Conclu	usion	228
Chapte	er Eleve	en: The role of the palliative care social worker: what informs	
their	practice	e with children and their parents	230
	Introdu	uction	230
	Knowle	edge base	230
	Superv	vision and support	234
	Trainin	ng on safeguarding children	237
	Other i	influences on their practice	242
	Conclu	usion	243
Chapte	er Twel	ve: Conclusions	245
	Introdu	uction	245
	Main c	conclusions	245
	Key fin	ndings	246
	Reflec	tions on the systems perspective	251
	Streng	ths and limitations of the study	254
	Implica	ations of the findings for practice and future research	255
	Conclu	uding remarks	257
Refere	ences		259
Appen	dices		
	1	Focus group participant information sheet	294
	2	Focus group discussion guide	296
	3	Questionnaire	298

4	Questionnaire: accompanying letter and participant information		
	sheet	305	
5	Interview: accompanying letter and participant information		
	sheet	308	
6	Interview guide	311	
7	Focus group: informed consent form	315	
8	Interview: informed consent form	316	
9	Data collection and analysis: illustrations of process	317	
10	Glossary of terms: chapter nine	320	

# List of Tables

2.1	Number of adult in-patient units and beds 1	0
2.2	Number of community and hospital support services 1	0
4.1	Number of child protection referrals and registrations 2006/07 5	50
5.1	Profile of survey respondents by type of work-base	36
5.2	Profile of survey respondents by number of years in palliative care,	
	in current post and since qualification8	36
5.3	Profile of interviewees by type of work-base	39
5.4	Profile of interviewees by number of years in palliative care, in current	
	post and since qualification 8	39
6.1	Age and gender of children in practice examples 1	06
6.2	Country of origin of ill parents in practice examples1	06
6.3	Relationship of terminally ill carer to child and number of lone carers1	07
10.1	Palliative care social workers' potential involvement with local	
	authorities 2	206

# Introduction to the thesis

Every professional who meets a child in their working life must take time to understand the stories that children tell them and gain a meaningful picture of what is happening in their lives. Collier 2004

#### Rationale and context of the study

This thesis weaves together two strands of enquiry: safeguarding children where a parent is dying and specialist palliative care social work. It is not an exploration of the role of the workers in bereavement counselling but rather of their perceived role where there are concerns about the present and future care of the children. These concerns could have existed prior to, arisen because of, been exacerbated by, or become evident because of the parent's illness.

Since the end of the Second World War there have been changes in the care of people dying with cancer (Clark 2007). Advances in medicine have led to better pain control and pain relief and this, together with a clearer understanding of the impact of death and dying on all concerned, has helped to provide a more holistic approach to the dying person (Reith and Payne 2009). This approach has been developing since 1957 when Dame Cicely Saunders, herself a social worker, nurse and doctor, initiated the modern hospice movement by publishing her first paper on hospice care (Clark 1999; Clark 1998a) and in 1967 by opening the St Christopher's Hospice in south London (Clark 1998b). Her approach took account of the person's 'total pain' which highlighted not only the physical aspects of the pain but also the psychological, emotional, social and spiritual dimensions of distress which could contribute to their suffering (Saunders 1993). Indeed, the National Institute for Health and Clinical Excellence (NICE 2004:20) reflects this approach in their definition of palliative care which is:

...the active holistic care of patients with advanced, progressive illness. Management of pain and other symptoms and provision of psychological, social and spiritual support is paramount. The goal of palliative care is achievement of the best quality of life for patients and their families.

'Specialist palliative care' in the UK differentiates itself from the general palliative care provided in mainstream health settings. It is care delivered by a multi-professional team, where practitioners, the majority of whom are trained and acknowledged specialists in palliative care, are almost exclusively concerned with the needs and care of terminally ill patients (NCHSPCS<sup>1</sup> 1995; NHS Quality Improvement Scotland 2002). The core personnel are generally doctors, nurses, social workers, chaplains and, where funding allows, physiotherapists and occupational, art and complementary therapists (Firth 2003). One of the principles of the care provided is to help family members cope during the patient's illness and their own bereavement. Although most commonly associated with the hospice movement for people with cancer, specialist palliative care may be provided in a number of settings: at home, in independent (charitable) and NHS hospices and palliative care centres, in hospice day centres and in hospitals, and for people with other life-limiting illnesses. Not every patient who is terminally ill will require or receive specialist palliative care. This type of care is only usually needed when 'the problems of the patient are either so numerous, so severe, so difficult to relieve or so unusual that a team with expert and special skills be called in' (Doyle et al. 2004:2).

Social work is concerned with people across all age groups and with particular needs. Social workers have a dual role of care and control, particularly in work with offenders but also in other aspects of work, especially child care and child protection. They are mainly employed by local authorities, charitable and for-profit organisations. Since 1948, local authority social work has moved from the specialist (work with one group of

<sup>&</sup>lt;sup>1</sup> National Council for Hospice and Specialist Palliative Care Services

people) to the generic (work with a range of people across the age spectrum and with a range of needs) and back again to the specialist. Specialist palliative care social work itself has a relatively short history. It has been seen as having an important complementary role within the hospice movement (Dunlop and Hockley 1998) and has come to be considered a specialist area of social work (Sheldon 2000; Cadell et al. 2010). A small group of hospice social workers first met in 1982 at St Christopher's Hospice and in 1986, recognising the emerging specialisation of social work in both hospices and the developing community palliative care teams, formed the Association of Hospice Social Workers since renamed the Association of Palliative Care Social Workers (APCSW). Currently the Association has approximately 300 members. However, it is difficult to ascertain the total number of specialist palliative care social workers as there is a lack of information about the number of posts.

The Association defines specialist palliative care social work as follows:

Social work is seen as an integral part of the multidisciplinary team within palliative care offering an holistic service to patients and families. Specialist palliative care social workers offer a wide range of support to patients and families from practical help and advice around income maintenance, debt counselling, help with housing and accessing other services, through to advocacy, individual counselling and group support. This will include bereavement work with adults and children both as individuals and in group settings. Key to specialist palliative care social work is the desire and ability to see people as whole people and not as a set of problems, to understand the connections of their lives and to seek to act on, rather than ignore the constraints and discrimination they experience in society (APCSW 2006:6).

Specialist palliative care social work is therefore but one small area of social work and whilst there are palliative care social workers who work with children who have a lifethreatening illness and their families, the majority are involved with adults who are receiving specialist palliative care. This area of social work could be considered both specialist and generic in nature because, whilst the focus is on the dying adult, workers

may be involved with a variety of people of all ages in differing circumstances and with diverse needs.

Small (2001:961), in a critical commentary on the interface between social work and palliative care, commented that there was a paucity of literature in this area and noted that the amount published in peer reviewed journals was 'very small', a view also shared by Sheldon (2000). It would appear that this position is slowly changing. There are now some specific texts on palliative care social work (for example, Parker 2005; Beresford et al. 2007; Reith and Payne 2009); chapters in books on palliative care which refer to aspects of palliative care social work; and, articles in a range of palliative medicine, palliative care and social work journals. Some articles present conceptual frameworks, many are discussions often containing descriptions of social work activities. In 2005, the 'Journal of Social Work in End-of-Life and Palliative Care' was established in the United States; this continues to be the only journal concerned with this specialist area of social work. However, there are few empirical studies that investigate aspects of social work in palliative care (Brandsen 2005). Interestingly, mainstream social work literature contains little or no mention of specialist palliative care social work (Beresford et al. 2007:24). Indeed, a literature review undertaken to support the scrutiny of social work services and identify the key social work roles and tasks to meet the challenges of the 21<sup>st</sup> century (Blewett et al. 2007) contained no references to palliative care social work. Palliative care social work continues to be 'a service in the shadows' because of its low profile both within and outside the profession as a whole (Beresford et al. 2007:206).

Whilst there is a stress on the importance of the family in palliative care, reference in the literature generally refers to partners, adult relatives and carers. Despite the expansion of interest in children in other areas, there have been limited references to children under the age of 18 whose parents are dying and receiving palliative care. There is, however, more emphasis on bereavement counselling and support for children after the

parent's death. A reason for this, Chowns (2006:2) suggests, is that 'the territory that is children anticipating bereavement is unpredictable, uncharted and swampy underfoot'. Bereavement counselling on the other hand is based on an event, the death of a parent, which has already occurred and therefore may seem to be more 'manageable'. However, there are studies that show that the life threatening illness of a parent has a major impact on family life both emotionally and physically (Visser et al. 2004; Kennedy 2009). All family members will be affected but particularly children who are dependent on their parents for their physical and emotional well-being (Christ 2000). The ability of the dying parent to continue to meet the needs of their child is likely to be compromised due to the decline in their condition (Saldinger et al. 2003). Indeed, where a parent is already struggling to provide appropriate care or to protect their child from harm from others, a terminal illness is likely to exacerbate an already difficult situation for the child. The parent's terminal stage of illness exposes children to significant levels of stress (Thastum et al. 2008) although some children appear to fare better when faced with adversity whilst others suffer poor outcomes (Zaider and Kissane 2007). For any child the death of a parent, according to Worden (1996), is one of the most significant losses a child will ever have to face.

The wellbeing of every child is now the focus of the governments in England, Wales and Scotland (Statham and Chase 2010). There is an expectation that in endeavouring to achieve the shared vision of improving the outcomes for all children, key people and agencies should work well both individually and together to safeguard and promote the welfare of children (Scottish Executive 2002; HM Government 2007). All people who have contact with children and families in their everyday work, including those who do not have a specific role in relation to child protection, have a duty to safeguard and promote the welfare of children (Department of Health 2003; Welsh Assembly

Government 2006; Scottish Executive 2008<sup>2</sup>). Safeguarding has been a concept that has evolved in meaning (Walker and Thurston 2006). In 2007, in statutory guidance on Section 11 of the Children Act 2004, the Government described safeguarding and promoting the welfare of children as:

- protecting children from maltreatment
- preventing impairment of children's health or development
- ensuring that children are growing up in circumstances consistent with the provision of safe and effective care; and
- undertaking that role so as to enable those children to have optimum life chances and to enter adulthood successfully (HM Government 2007:11).

Illness and death will occur in families where relationships are not totally loving (Pardoe 1989; Hurdman 1995) and children may be vulnerable and at risk. Whilst there have been studies exploring the role of workers in adult mental health, domestic violence and substance misuse services in relation to child protection (Taylor and Kroll 2004; Darlington et al. 2005; Humphreys and Stanley 2006; Stanley et al. 2010), there is a gap in the literature concerning social workers in palliative care.

### Aim of the study and development of research questions

The aim of this study is to explore what is the role of the palliative care social worker in safeguarding and promoting the welfare of children of adults who are receiving specialist palliative care.

<sup>&</sup>lt;sup>2</sup> The wording in Scotland is slightly different: promoting the well-being of individual children and keeping children safe.

In order to achieve this aim, I developed the following key research questions:

- What do palliative care social workers understand by the term 'safeguarding children'?
- How do palliative care social workers identify situations where there are concerns about the care of a child and where a child may be at risk? What is the involvement of the palliative care team in the identification?
- How do palliative care social workers go about their work with children and their families?
- What is the palliative care social workers' perception of the quality of their relationship with the local authority children's services?
- What support and training do palliative care social workers receive about safeguarding children and from whom?
- What insights do the views of palliative care social workers offer to improve how children may be safeguarded within the palliative care setting?

#### Structure of the thesis

In this chapter I have outlined the rationale and context of the study together with the aim and research questions developed. The literature relevant to the study is reviewed in chapters two to four providing a platform on which to understand, develop and support the research. Chapter two considers specialist palliative care social work with specific emphasis on the role with children and families. Chapter three focuses on the experiences of children where a parent is dying and chapter four explores the literature regarding safeguarding children, particularly in the context of workers in adult care services. In chapter five the journey of the research is described. I explain the rationale for the methods used, the theoretical perspective and how the analysis was conducted, together with a self-reflection.

Chapter six uses the material from the stories of the palliative care social workers to describe some of the experiences of children and their parents as they journey through the last stages of a parent's life. In chapter seven the respondents' perceptions of their role in safeguarding children is examined. Chapter eight considers how the palliative care social workers become involved with families where there are concerns about the children, and in chapter nine I explore how palliative care social workers engage with these children and their parents. The relationship between the palliative care social worker and the local authority children's services is the focus of chapter ten. Chapter eleven analyses what informs the palliative care social workers' practice with children and families. Finally, in chapter twelve I present the key findings of the study, its strengths and limitations, and implications for practice and further research.

I have chosen to present my findings together with a discussion of their relevance to existing theoretical knowledge and empirical studies as an integrated whole in each chapter (Holloway 2005).

### Notes

1 The first year of this study was conducted when I was registered at the University of Dundee in the School of Education, Social Work and Community Education.

2 Throughout the thesis, the term 'children' should be read as shorthand for the more inclusive term 'children and young people' and 'palliative care social workers' as shorthand for 'specialist palliative care social workers'. These workers are referred to in the text in the female gender which reflects the composition of the current workforce and aims to protect the anonymity of the respondents.

# Palliative care social work: context and role

Caring, the palliative way, is to focus on the medical, nursing, spiritual, emotional and social needs of patients and their families and to seek to meet them through a variety of means. It is thus a style of caring: an acknowledgement of the need to approach patients holistically. Robbins 1997

## Introduction

In order to put palliative care social workers' involvement with children into perspective, I will explore in this chapter the literature that refers to the context and role of the palliative care social worker generally, their specific work with children and their families, their role in the multidisciplinary team, and their knowledge and skill base. As stated in chapter one, the role of the palliative care social worker has been poorly addressed in literature with few empirical studies, but over the past few years this position is slowly changing particularly with contributions from North America.

### The context of palliative care social work

The modern hospice movement and palliative care in the United Kingdom that began in 1957 has continued to develop both alongside and inside the NHS (Mathew et al. 2003; Small 2003). The NHS was primarily concerned with widespread acute and chronic health problems during the first two decades of its existence and little attention had been given to the care of the terminally ill (Clark 1999). Voluntary hospices provided the majority of in-patient care and continue to do so. The following tables show the provision of in-patient and support services at January 2009 (Help the Hospices 2009: 164-165).

Total	212	59	651	153	2572
Wales	15	9	65	6	78
Scotland	24	10	96	14	355
England	173	40	490	133	2139
Country	Total units	NHS units	NHS beds	Voluntary units	Vol. beds

 Table 2.1
 Number of adult in-patient units and beds

Country	Home care	'Hospice at home' <sup>3</sup>	Day care	Hospital support teams
England	226	95	225	228
Scotland	47	1	25	39
Wales	26	7	22	23
Total	299	103	272	290

These figures, however, do not reveal the disparity of services across the country. The Palliative Care Survey 1999 (NCHSPCS 2000:2) shows that 96% of health authorities described their palliative care services 'as not yet adequate across the board' and 15% felt they were 'struggling in developing all the service components'.

There appears to be a reduced access to, or uptake of, palliative care services within socio-economically deprived areas and by people from minority ethnic groups (Department of Health 2000; Payne 2010). Bowling (2009:35) points out that, compared to other social groups, poorer people tend to mistrust modern medicine, know less about how to access services and are reluctant to do so. Firth (2001) refers to the black and minority ethnic population in 2001 in England being estimated to be around 10% but figures from the Hospice Information Service show that 3% of recorded adult patients who used palliative care and hospice services in 1999-2000 were from these groups. Various reasons given in the literature for this low uptake include a lower incidence of certain types of cancer and other chronic diseases plus the younger socio-demographic

<sup>&</sup>lt;sup>3</sup> 'Hospice at home': teams that provide extended specialist palliative nursing, medical, social and emotional support and care in the patient's home.

profile. Another potential explanation is the 'culturally insensitive' nature and 'ethnocentric outlook' of the services provided (Randhawa and Owens 2004; Larkin 2008). In addition, a lack of knowledge about palliative care services and 'gatekeeping' by GPs leads to a lower referral rate (Koffman and Higginson 2001). Also reported is an ethos of mistrust that affects the decision-making of black and minority ethnic groups (Cort 2004) and in Bonifant's (2000) view there is no clarity about the cultural acceptance of palliative care. Oliviere (1999:55) writes about the need to respect and nurture cultural difference and offers clear advice on how to provide 'culturally safe practice' in palliative care.

There is evidence in one recent study of just over 100 service users that attitudes of people in receipt of palliative care are generally positive (Beresford et al. 2007). Many reported that despite initial feelings of 'horror and dread' when a referral to a hospice was mentioned, these views changed once contact had been made. Many service users reported that 'their health and sense of well-being actually improved after they entered specialist palliative care' (p.51). Indeed, there is no dispute in the literature that the patient and his or her needs are at the centre of specialist palliative care. This emphasis on the individual has always been a major feature. Saunders herself was clear that patients should be able to give their views and make their own choices; giving 'a voice for the voiceless' (Oliviere 2000).

Although specialist palliative care social work is provided in a range of settings, its provision, like palliative care, is not uniform throughout the UK. In a study by Lloyd-Williams et al. (1999) into the provision of social work support in hospices, out of the 97 hospices who responded, 73 stated that they had full or part time social work support, 18 had 'social work support available if required' and six had no social work support at all. Palliative care social workers are mainly employed by hospices, others by the local authority and a few by the NHS. Despite the NCHSPCS's (2000) recommendation that

social workers should be included as core members of the palliative care team there has been a reduction in provision in some adult services (Beresford et al. 2007). Many palliative social workers are lone workers or in a very small team of social workers, thus restricting the service they are able to provide.

However, gaining an understanding of the provision of palliative care social work is hampered by the problems of definition. Leading specialist social work practitioners (Sheldon 1997; Oliviere et al. 1998) make little reference to 'social work' in their writing referring rather to 'psychosocial care' although their work strongly reflects the values and principles of social work practice (Currer 2001). 'Psychosocial care' has come to be widely recognised as an integral part of holistic palliative care (Dix and Glickman 1997) providing a 'jigsaw of support' for the patient (Hearn et al. 2008). The definition offered by the National Council for Palliative Care's Working Party on Psychosocial Care is that:

Psychosocial care is concerned with the psychological and emotional well-being of the patient and family/carers, including issues of self-esteem, insight into adaptation to the illness and its consequences, communication, social functioning and relationships (Dix and Glickman 1997:6).

The Working Party recognised that staff of all disciplines in specialist palliative care services routinely provide psychosocial care in a 'broad sense'. Specialist psychosocial care, they concluded could be provided by 'appropriately trained and qualified staff' including clinical psychologists, social workers and counsellors.

Field (2000), however, is concerned that too much emphasis is being placed on the psychological aspects of psychosocial care and that 'social care' could become sidelined as 'practical help'. More focus, Field argues, should be placed on addressing and valuing the patient's family, cultural and social relationships, a view supported by Oliviere (2001). Hodgson (2005) suggests that social workers over-emphasise working with psychological issues rather than social needs because that fits in with the

'biomedical approach'. Sheldon (2008) also draws attention to 'social pain' which she sees as sitting alongside the physical, psychological and spiritual pain, supporting Field's view (2000) that it this the least understood aspect of 'total pain'. Oliviere (2001) describes the role held by palliative care social workers as an 'eccentric' one that is away from the centre of care on which the medical and nursing team focus. Social workers, he argues, advocate for a community and family perspective. The findings from a study of patients and carers (Clausen et al. 2005:283) support the view that both medical and nursing staff can often marginalise this 'social aspect'. The evidence, they argue, points to the need for a social model of illness. Napier (2003:151) also stresses the social dimensions and puts forward three basic aims for social workers:

- to address the social concerns and problems of seriously ill and dying people and their families
- to create and strengthen the social supportiveness of their environment and,
- to find ways of redressing the social inequalities that permeate the experience of illness and dying.

Psychosocial care then is generally considered a 'shared realm' with no one discipline having sole responsibility (Monroe 1998) and is 'poorly articulated' in the palliative care literature (Payne and Haines 2002). Palliative care social workers themselves add to the confusion. Some prefer to use the term 'family support worker' or 'counsellor' (Beresford et al. 2007). There does seem to be a reluctance on the part of some palliative care social workers to 'own' their profession (Currer 2001).

Social work itself has undergone a radical change in the wake of the NHS and Community Care Act 1990 with social workers adopting the role of care managers rather than working in the more traditional casework or therapeutic role. Although based in a specialist setting, palliative care social work is practised in a way which is consistent

with the traditional understanding of social work values, roles and responsibilities (Monroe 1998; Oliviere et al. 1998). It is also practised according to the current values of partnership, involvement, empowerment, respect and equality (Beresford et al. 2008:1405). In the view of Beresford and colleagues (2007:189) when a palliative care social worker has been available 'people's journey through their illness and palliative care system had been made more bearable'.

#### Role of the palliative care social worker

The role of the palliative care social worker is shaped by the attitudes and expectations of patients, their families, colleagues and society (Firth 2003). It has, Hodgson (2005) believes, been compromised by the dominance of the medical approaches to palliative care as well as a limited understanding and poor public image of social work (Oliviere 2001; Firth 2003; Raymer and Reese 2004; Beresford et al. 2007).

In each of the three editions of the *Oxford Textbook of Palliative Medicine*, Monroe has clearly and comprehensively described the role of social work in palliative care drawing on a wide literature base (Monroe 2004). She commences the introduction to each edition with the statement:

Social work is a necessary and appropriate part of palliative care. Palliative care starts with specific physical symptoms but it can only be completed by consideration of the patients' feelings, family and friendship networks, and social circumstances (2004:1007).

Monroe then describes three 'forces' which shape the social work role: the identification of non-medical social goals set by the palliative care team, the teamwork required to meet those goals, and the expectations and attitudes not only of the patients and families but society and colleagues. Once the physical pain has been relieved, the patient's non-medical needs she believes are threefold: a need to express emotional

pain, a need to explore spiritual pain and a need for practical help. Monroe (2004:1008) describes the core social work task as 'the social and psychological health of the patient, family, friends, and carers, before and after death'. However, she also cautions that palliative care social workers may have to 'balance individual needs and wishes against a social control and protection function as in child protection' (p.1107).

According to Monroe (2004) the social work task involves both assessment and intervention. Assessment includes that of the individual, the family and physical and social resources. Intervention includes: the provision of information; facilitating communication between patients and their families; assisting the patient and family to make realistic goals and giving them the confidence to achieve them; and bereavement counselling. Methods to achieve these interventions include working on a one-to-one basis, meetings with families or couples and group work. Earlier, Sheldon (2001:243) had concluded that it is the counselling aspect of the role 'which is inseparable from good palliative care social work, which makes this area of care so stimulating and challenging'.

Sheldon (2000), recognising the lack of research into the 'essence' of social work in palliative care, conducted a small qualitative study into the role. She acknowledged that there were limitations to her study as the conceptual framework was developed from the exploration of seven cases of only one palliative care social worker and then validated and further refined by another four workers. The framework centred around 'knowing and working with families'. The core values were seen as encouraging self-determination, being non-judgemental and affirming the client. The key elements identified were: having a family focus, being a team member, influencing the environment and managing anxiety in the family, the professional team and oneself. Of these, the social worker perceived having 'a family focus' was a key area of work (p.493). Interestingly, Sheldon speculates how far this study differentiates between the

specialist role of the social worker in palliative care to that in a multi-professional setting such as mental health. She concludes that it is the starkness of working with people who are dying, and the anxiety that this can cause the worker and the team, that sets it apart from other settings.

Sheldon (2000) recognised that one element was missing from her study: the views of the service users. She considered that this was an essential element in capturing a clearer picture of the complexity of the social work task in palliative care. This element has been recently addressed in a UK-wide study using a grounded theory approach involving 111 service users of specialist palliative care social work who were interviewed individually and involved in group discussions (Beresford et al. 2007:37). Out of the total number of service users 61 were bereaved and 52 had life-limiting illnesses and conditions (two of whom were both bereaved and patients); 39 were men and 72 were women; 10 described themselves as black and/or members of minority ethnic groups; and 15 had children of school age. Service users were involved in the design and management of the research. This in-depth study found that prior to meeting the palliative care social worker, service users held very negative views about social workers in general, based mainly on reports in the media but in some cases on direct experience. One service user is quoted as saying:

I thought that's it, they want to come in and take the kids because they know I'm not going to be able to look after them ... I was frightened (p.47).

However, once engaged with the social worker the study suggests that service users particularly valued the personal qualities of the social worker which included 'kindness, warmth, respect, compassion, caring, sensitivity, empathy and thoughtfulness' (p.92).

Service users also saw early contact with the palliative care social worker as 'very important' (Beresford et al. 2007:56). Generally referrals came to the palliative care social worker via other professionals, although some service users questioned how

much these professionals knew about the work of the palliative care social worker. Some palliative care social workers made an informal 'proactive approach' to all new patients and this personal approach was generally welcomed as it made people feel 'cared for and safe' (p.141). However, there seemed to be an inconsistency in the process and some families did not have any contact with a palliative care social worker until after the patient's death when they were contacted about bereavement services. There were few self referrals. In Zaider and Kissane's view (2007:73), because families may not be able to ask for or indeed recognise their need for psychosocial support, palliative care social workers and others in the multidisciplinary team need to be able to identify those families which 'carry the most vulnerability and distress'.

Evidence in the Beresford et al. study (2007) suggested that few people were referred to a hospice or a palliative care service specifically for social work support. A referral seemed to come about when the physical symptoms 'caught up with the emotional distress' rather than for any psychological need or indeed social problems (p.140). This, the authors argue, seems to go against the holistic approach to care and support.

The study further revealed that assessments were carried out informally and holistically. Service users valued this, feeling able to talk about the issues that were important to them. They were reported to like the use of 'everyday language' and saw the palliative care social worker 'as an ally' - someone who was 'on their side' (p.155). Service users also valued the time given to them by the palliative care social worker plus their accessibility, reliability and continuity of support. The reported interventions involved a range of issues including coping with change and loss; helping support the family; meeting children's present and future care needs; assisting with welfare benefits; and support needs. A key role identified by service users was the way palliative care social workers helped them to negotiate the health and welfare systems. The positive outcomes included an increase in self-esteem; increased support for family members

including children; a reduction of anxiety regarding practical matters; and, an increase in ability to cope with the medical aspects of the illness. Palliative care social workers were also involved in organising and facilitating bereavement groups both for adults and children.

In addition, Sheldon (1993), Egnew (2004) and Parker (2005) all stress the need for palliative care social workers to come to terms with their own fears of dying and death, including any previous losses, allowing them to feel more comfortable with the emotional aspects of death and dying.

### Work with children and their families

When a person is dying, Saunders (2004:xix) describes the family as:

finding themselves in a crisis situation with the joys and regrets of the past, the demands of the present and the fears of the future, all brought into stark focus. Help may be needed to deal with guilt, depression and family discord; may be the possibility of resolving old problems.

Indeed, the World Health Organisation (1990 and 2002) in its definitions of palliative care has identified the need to both support and address the needs of the patients' families. Placing the patients in the context of their families is seen an essential part of the task of the social worker (Egnew 2004; Beder 2006). This placing includes gathering information about how the family functions, its strengths and limitations as well as the existence of any vulnerable members such as children.

Palliative care social workers routinely assess symptoms of distress such as depression and anxiety in family members and, in addition, symptoms that indicate 'family system' distress (Walsh-Burke 2004). These symptoms might include anger, abuse or violence; substance abuse or dependence; and emotional cut-offs, including the isolation or scapegoating of a family member. Napier (2003) argues that in order that the family's

confidence be restored and strength be 'galvanized', the social worker may be most helpful by being the informer, coach, advocate and a reliable, safe person 'to steer them through' (p.160). She believes that thinking about families by focusing on their strengths and resilience does not ignore or minimise problems but rather 'lends respect to the idea of partnership' (p.160). Whilst each family member is at risk of distress due to the strong emotions and changes in the family structure and functioning caused by the impact of the illness, it is especially evident in the case of children facing a parent's death (Walsh-Burke 2004). Studies have shown the damaging effect that a lifethreatening illness of a parent may have on various aspects of a child's life (Vachon 1993; Rutter 1996) and the importance of children being kept informed about what is happening (Walter 1994).

There is literature that recognises how the needs of this community of children can impact on the role of the palliative care social worker (Oliviere et al. 1998; Monroe 2004). Monroe (2004:1011) believes that palliative care professionals 'have a responsibility to ensure that potential difficulties for children are minimized and social workers possess specialist skills in this area'. She considers that help for children should start with their parents. Describing these children as 'vulnerable', she reminds us that they may be unable to voice their needs and emotions; however, these may be displayed through their behaviour (Jewett 1997). Palliative care social workers, Zaider and Kissaine (2007:80) argue, may need to help a child develop a 'resilient mind-set' so that the child may better cope not only with the current circumstances of the dying parent but also with the bereavement process.

Sheldon (1997) drew up a framework to guide practice in supporting children when their parent is dying:

- find out what the child knows or believes about the situation
- help parents talk to their own children themselves

- use simple language, play and relevant books when engaging with children
- recognise the individual relationship of the child and the dying person
- help parents to consider what resources there might be within their family or community that could also be helpful to their children while they themselves have so much to cope with (pp.83-86).

Sheldon, therefore recognised the importance of communicating with children. Whilst Luckock et al. (2007) agree that all social workers should be able to communicate with children, they argue that evidence shows that there is no guarantee that on qualification social work students have been prepared to do so.

In her study of the role of social workers in palliative care, Sheldon (2000) found that it was apparent that palliative care social workers are often in the position of hearing about risk to vulnerable children and adults and therefore need to make judgements about how to 'manage risk'. She also raises the point that because social workers have a legal duty to protect vulnerable children 'there may be a preventive purpose alongside counselling which may therefore bring in advice and information giving' (Sheldon 2001:243). Sheldon (2001) and Monroe (2004) both consider that a core part of the role of the social worker is to assist patients and families to communicate. If family members are not open with each other then 'social adjustment' is unlikely to take place, since everyone is putting up a wall of secrecy' (Faulkner and Maguire 1994:76). Family meetings 'can be a powerful tool for change and resolution of problems' (Monroe 2004:1010) and for helping individual members to communicate more openly (Firth and Anderson 1994; Fineberg 2010).

There is one important aspect that Sheldon (1997) believes a palliative care social worker should keep in mind: the recognition of the individual relationship of the child to the dying person. Indeed, Monroe (2004) considers that learning about the family's

strengths and difficulties, finding out about the normal patterns of communication, identifying any areas of conflict and support prior to the illness will go some way to understand the family's current and future ability to cope. The illness can be the 'final burden that topples a delicately balanced system of nurture and support' (p.1008).

There is, however, a lack of literature analysing the palliative care social worker's role in relation to children. More specifically, whilst Beresford and colleagues (2007) did not interview children and young people in their study, there was evidence from parents that palliative care social workers were carrying out some 'intensive individual and/or group work' with them which parents and children were reported to value highly (p.107). However, it appeared that palliative care social workers were only working directly with a small proportion of the children of parents interviewed in the study; a small number of those parents where this did not happen reported they would have welcomed some direct support for their children.

A particularly sensitive area of practice identified in the literature is regarding the future care arrangements of children, particularly where they are being cared for by lone parents who are receiving specialist palliative care. Indeed, these parents may be frightened of agreeing to see the palliative care social worker because they fear the risk of their child being taken into care (Beresford et al. 2007). This can cause uncertainty and anxiety for both the parent and the children. Who has parental responsibility after death is not always a straightforward matter particularly now that family relationships are becoming more complex (Sheldon 2008). Willis et al. (2001) found in their retrospective review of ten oncology patients who were single parents that the making of care plans, either verbally or formally prior to the parent's death, was no guarantee that they would be put in place. In a substantial minority of these cases, a person who the dying parent had not wanted to care for the child ultimately did so. In one case, the plan made was felt to be unacceptable by the children and later resulted in a suicidal crisis for one of

them. Nadimi and Currow (2008:171), as a result of their own small study of three families, suggest that 'social workers' multifaceted involvement' should include a sensitive exploration of the available options. Assessment by social workers as soon as 'sole parenthood' is identified would, they believe, allow more time for the legal, emotional and social issues to be addressed.

In addition to the stresses of working with people who are dying, Firth and Anderson (1994) describe the pain and anxiety that is borne by workers where a patient has young dependent children. Firth (2003) also draws attention to the powerful emotions that can be released in staff members, for example anger, fear and anxiety. These emotions, they argue, may be projected on to other members of the team, the organisation, worker's own family and sometimes the patient and their family. A team of reliable professionals who are communicating well with each provides patients with a sense of security, consistency and comfort, which according to Jeffrey (2004:112) is essential to meet the needs of patients and their families.

### The palliative care social worker within the team

Another important aspect of literature relates to the multi-professional team in specialist palliative care. The team model was very much part of Saunders' vision of palliative care; she saw the strengths in bringing together professionals who could respond to the patient's 'total pain'. However, she also saw the need for them to overcome professional rivalries, the temptation to compete and learn how to share and when to hand over (Saunders 1990). Both Ajemian (1993) and Firth (2003) identify the importance for staff to have a strong professional identity before joining a palliative care team because of the need to be secure enough within their professional role to allow other staff to sometimes share some aspect of it without being threatened.

Different palliative care services organise their teams in different ways in terms of size and personnel depending on local circumstances (Speck 2006). Whilst there is much debate in the literature about whether palliative care teams are multidisciplinary or interdisciplinary in nature (Meier and Beresford 2008) the consensus appears to be that they should be inter-disciplinary. Raymer and Reese (2004:150) argue that a distinguishing core value of hospice care is that care is provided by an interdisciplinary team as opposed to a multidisciplinary one. In the view of Farber et al. (2004) and Egnew (2004) interdisciplinary teams, where people are openly communicating with each other, share common goals and expectations of care can support the patient and family more effectively than working on an individual basis. Corless and Nicholas (2004) describe the focus of the interdisciplinary team being the knowledge and expertise possessed by each of the team members as well as personal talents. They consider the hallmark of the interdisciplinary team is the process of interaction based on respect, information sharing and shared decision-making. They describe 'role blending' as an indication of the degree to which professionals do what needs to be done with less attention to which discipline is responsible for what, a view supported by Hill (1998), Payne (2000), Sheldon (2000) and Molyneux (2001). Bronstein (2003) points out that unless social workers collaborate with other professionals it is virtually impossible for them to serve clients effectively.

The potential barriers to multidisciplinary working are numerous, however, and include language, communication, values, professional loyalty and role ambiguity (Hill 1998; Payne 2000). In Firth's view (2003:118), subtle differences in the value base of specialist palliative care professionals can be a problem and a patient's voice might be 'lost' if there is conflict in a team (p.110). Issues of confidentiality may also cause some difficulties in terms of what should be shared with whom within the team (Egnew 2004).

Dunlop and Hockley (1998) argue that having a social worker as a core member of the multidisciplinary team can lead to a more creative service for patients and their families, as well as providing for the individual and collective needs of the team. However, NICE guidelines (2004) distinguish between core members and others, and it appears that social workers are optional core members which Hodgson (2005:53) considers to reflect a 'hierarchy of care'. Social workers therefore are not always seen as an equal member of the team but supporting the medical and nursing staff as an 'internal consultant' or 'resource' (O'Connor et al. 2006). A social worker, Monroe (2004) believes, can remind the team to keep 'the whole person' in mind whilst interventions are planned. This view is supported by Raymer and Reese (2004) who argue that historically social work has differed from other disciplines in health care; social workers, they argue, 'advocate for the individual and family within the system' (p.155), ensuring that psychosocial care is a core activity (Cadell et al. 2010). However, Firth and Anderson (1994:158) believe that a singleton palliative care social worker in a team has an 'impossible task' that may lead to defensive adaptations in the individual worker and the team. Power issues may be present if people from different disciplines assert their perspective is superior (O'Connor et al. 2006). Payne (2006:149) argues that palliative care social workers should practise jointly with other professional colleagues, and Dunlop and Hockley (1998) and Monroe (2004) advocate engaging in teaching activities with other members of the team. In doing so palliative care social workers may demonstrate their worth, gain acceptance and help build a 'community of practice' (Wenger 1998).

However, Berzoff and Silverman (2004) put forward the view that social workers are generally not socialised to practise as part of a team. In their opinion social workers take on a subordinate role particularly in a medical setting, a view supported by Firth (2003). Reese and Sontag (2001) identify various potential barriers to the full use of social workers in a hospice team. These include:

• a lack of knowledge of social workers' expertise

- role blurring an overlap of roles which may mean other professionals take on roles that are more appropriate for the social worker
- conflicts arising from differences between professions in values and theory base
- negative team norms
- lack of commitment to the team process
- power differentials in the team
- client stereotyping clients may refuse to see a social worker due to misperceptions held by the client (pp.168-173).

In contrast, Blacker and Deveau (2010:243) argue that social workers bring a useful understanding of group dynamics and communication that is invaluable to the team's development.

#### Knowledge and skill base of palliative care social work

As suggested by the above studies, the breadth of theoretical information informing palliative care social work is therefore 'formidable' (Egnew 2004: xxi). The literature describes the various theories that specialist palliative care social workers should understand in order to be effective in their work (Sheldon 1993; Quinn 1998; Egnew 2004; Monroe 2004; Parker, 2005; Blacker and Deveau 2010). This knowledge base, it is considered, should build on that already gained from basic social work training. Areas identified include: theories of loss and grief; death, dying and bereavement; historical and sociological perspectives on death and bereavement; the psychosocial context of illness; attachment theory; ethical issues in palliative care; communication skills; bereavement counselling; advocacy and concepts of collaborative practice. How a family may be affected by legislation and what community services and welfare provision might be available is also considered to be important. There is, however, an

Chapter Two

apparent lack of reference to knowledge about the immigration system and to child care legislation, policies and guidance.

Having knowledge and skills in various therapeutic techniques, for example cognitivebehavioural work, is seen to be an advantage (Reith and Payne 2009) and considered to be a valid model for end of life care (Moorey and Greer 2002). The importance of being knowledgeable about spirituality is also mentioned (Lloyd 1997; Everard 2005) although many social workers in Lloyd's study (1997) did not see their role as including any spiritual care element. In Thomson's (2005:41) view, having an awareness of people's different ways of life, beliefs and traditions is essential for a sensitive delivery of palliative care together with the concepts of 'cultural competence' and 'culturally safe practice' (p42).

In addition, highly developed communication skills are perceived to be crucial. Egnew (2004:xii) refers to the 'sensitive in-depth end of life discussions that accompany a respectful death' highlighting the importance of the social worker being an 'extremely skilled interviewer'. Reith and Payne (2009:65) advocate palliative care social workers using a 'narrative approach'. Since good interpersonal skills are essential to 'listen and hear the person, carer and family' (Cullen 2005: 93). The use of counselling skills in communication or counselling as an intervention have been well established in palliative care (Vachon 1993). Skills in communication and work with children in particular have received less attention in the literature in the past but there is now a growing recognition of their importance (Kennedy 2008; Fearnley 2010).

# Conclusion

The recent investigation of palliative care social work from the perspective of service users (Beresford et al. 2007) provides a new dimension and important insight into how

Chapter Two

palliative care social work is understood and valued by those receiving the service. Whilst there is literature concerning preparation of children for the death of a parent, there appears to be a gap in how palliative care social workers work specifically with families where there are concerns about the present and future care of children. This is even though Sheldon and Monroe, the two main British contributors to this body of literature, believe that palliative care social workers are in a position to be alert and respond to children who might be at risk. The next chapter moves on to explore this aspect of their role further through the existing literature about children's experiences during the terminal phase of a parent's illness.

# Children's experiences of living with a dying parent

Coming to terms with death and loss is the most difficult challenge a family must confront in life. Walsh and McGoldrick 1998

# Introduction

Since the 1990s there has been a gradual increase in interest in the experiences of families with children where a parent is seriously ill or dying. Most of the research is North American in origin with a limited amount from Britain and Australia. In addition, there are a small number of studies from Northern Europe in the English language literature. The majority of the research has been conducted by health care academics for health care professionals, although the findings are very relevant to social workers in general and to palliative care social workers in particular. A large proportion of the research has been qualitative in nature or mixed method producing rich data from small samples. Contradictions and non-uniformity in findings may be attributable to differing reporting mechanisms, methodologies, sample sizes and illness-related characteristics. There are also some practice-based papers that describe the reactions of children and parents to the terminal illness and death of a parent.

The chapter will point to gaps in the literature as there are few studies that explore the nature of the relationship between the dying parent and the child and the impact of that relationship on the child's emotional and physical well-being during this time of disruption and uncertainty. In contrast, the focus rather is on what happens after the parent's death and the literature on child bereavement is extensive. There is also more research on the period after diagnosis and during initial treatment which concentrates mainly on mothers with cancer with a limited exploration of the experience of fathers. Most of the studies are psychosocial in approach and pay less attention to the social context. Further, a very large proportion of studies consider families where there are two

parents in the household and most studies do not include families where there is a history of emotional and behavioural problems concerning the child prior to the illness or where the child is vulnerable. Similarly, the relationship between parents and between parents and children before the onset of the illness is not usually explored. Whilst there are studies reporting on children of various ages, there is little reference to pre-school age children. Finally, the context is mainly white Caucasian, middle and upper class families and generally studies do not include families living in poverty or trans-national families.

In this part of the literature review I will first report on the incidence and prevalence of children living with a dying parent, before moving on to consider the nature of the literature on the impact of this experience on children's lives. Consideration will then be given to the literature on how children cope with the terminal phase of the parent's illness and finally on the impact of the illness on family functioning. The aim is to extrapolate lessons from this body of literature that may be applied to the practice of palliative care social workers in safeguarding children.

# Incidence and prevalence

In the UK it is estimated that approximately 1 in 10 of all cancer cases occur in adults aged 25-49 (Cancer Research UK 2009) and in 2007, more females (4,015) died from cancer than males (2,929) in this age group. The most common cancers included breast, malignant melanoma, bowel and cervical cancer. It is likely that many of these people will have been caring for children under the age of 18. Based on data gathered from a large sample of adolescents in the north of England, it is estimated that the risk of experiencing parental death by 16 years is 6% (Harrison and Harrington 2001). In the United States, the National Cancer Institute estimates that 24% of patients receiving treatment for cancer have children under 18 (Rauch and Muriel 2004). It is to be

expected therefore that many children live much of their childhood with a sick parent and can often become young carers<sup>4</sup>. Information from the 2001 Census indicated that there were approximately 175,000 young carers in the UK (HM Government 2008:123). A UK survey conducted in 1993 found that 56% of young carers are living in lone parent families and that it was mothers who were cared for in 70% of lone parent families (Dearden and Becker 2004).

# The nature of the literature on the impact of parental terminal illness on children's lives

Studies regarding the terminal phase of the illness tend to be retrospective in nature and based on the responses of the remaining parent and other bereaved family members. The remembering of past events and feelings will be influenced by the individual's emotional state and perspective both at the time and at recall (Addington-Hall and McPherson 2001; McPherson and Addington-Hall 2004). Whilst it is important to view incidents from a 'historical' time frame, it is also important to consider the contemporaneous experience of participants, particularly the children, whilst their journey is taking place. Whilst there has been an expansion of user involvement in research and in participatory approaches in research generally, there is still a reluctance to involve patients and their families, particularly children, during a time of such uncertainty and vulnerability. Issues of 'gatekeeping' by service providers and the 'protecting' of vulnerable service users by 'paternalistic cultures' are evident (Beresford et al. 2007) including bureaucratic barriers imposed by various ethics approval processes (Peck and Jones 2004; Balen et al. 2006; Part and Comben 2007; Kendal et al. 2009). However, feedback from participants, particularly children, in prospective studies has indicated that involvement in such research, although emotional at times,

<sup>&</sup>lt;sup>4</sup> Young carers: those under 18 who are carrying out 'significant caring tasks and assuming a level of responsibility for another (adult) person which would usually be taken by an adult' (Department of Health 1999a).

was positive and helpful (Chowns 2005; Sheehan 2007; Kennedy 2008).

Few studies therefore include direct information from the children. Chowns (2005:131) suggests that this area of research is both 'scary' and 'swampy' in nature. This is because the uncertainty of the situation might lead both the researcher and practitioner to question their abilities and skills; there may be an instinct to protect children from painful issues by not talking about them; and being involved might mean confronting adults' own memories of childhood distress and mortality (Monroe 1990). Given the larger numbers of research studies into the effects of bereavement, the journey would appear to be more difficult to manage than the destination, death and bereavement. However, where studies have gathered direct information from children it has been found that parents and children did not always perceive child functioning in the same way (Osborn 2007). There is often more agreement between the responses of parents and children concerning behavioural problems than emotional ones, maybe because these are easier for the parent to recognise (Visser et al. 2004). The child's ability to self-reflect will depend on their stage of development and confidence in describing how they feel (Lewis 2007).

#### The child's experience of the terminal phase of a parent's illness

The literature concerning the terminal phase of a parent's illness, has identified several factors that may influence the child's experience particularly the age and gender of the child and practices of communication and sharing information with the child. Whilst this study concentrates on the terminal phase of a parent's illness, it is important to acknowledge that the experiences of children and their parents at the diagnosis and early treatment phase of the illness are also likely to have implications for the family and

the professionals involved at the later stage<sup>5</sup>. However, Visser et al. (2004) argue that longitudinal studies are needed to test this hypothesis.

#### Age and gender

Children's understanding of the concept of death will shape their thoughts, feelings and reactions to a parent's terminal illness (Rauch and Muriel 2004; Dent 2005; Christ and Christ 2006; Diareme et al. 2007; Himebauch et al. 2008; Fisher 2009). Parents may not realise that a child's experience of the disease process and the losses that happen along the way are different from those of an adult. Christ (2000) undertook a landmark piece of qualitative and quantitative research over several years with 88 families and their 157 children (aged 3-17 years) who coped with the terminal illness and death of a parent. Christ divides her findings regarding the children into five age groupings.

In the first group (3-5 years) Christ found that children were distressed not by the illness, but by the changes in and the preoccupations of the care-giving parent as well as by the separations caused by the ill parent's hospital admissions. Any psychological symptoms or misbehaviour, which were few, were dependent on the adequacy of care arrangements, the number of separations and emotional climate generally. Some children's behaviour regressed and others became aggressive. Children in the second group (6-8 years) showed strong emotions of sadness, anger, fear and anxiety and were much troubled by what was happening around them, particularly any separations and restrictions on their activities. Most were found to be reluctant to talk about their fears.

<sup>&</sup>lt;sup>5</sup> A comprehensive discussion on the range of literature regarding the early stages of parental illness and its impact on children is given by Visser et al. (2004); Diareme et al. (2007); Grabiak et al. (2007); Osborn (2007); Semple and McCance (2010).

The children in the third group (9-11 years) showed a need for information during all the stages of the illness. Some exhibited few reactions whilst others became angry and most had problems concentrating at school. In the fourth group (12-14 years), although able to cognitively to understand what was happening, children could still have misunderstandings due to lack of experience and avoidance of facts. They were typically argumentative and demanding; their distress symptoms included sleep problems, anger and sadness. Girls seemed to experience more emotional distress than boys. Where there were pre-existing family conflicts and communication problems, the children's anxiety was often displaced, causing angry and provocative interactions that could become out of control. Children in the final group (15-17 years) were found to be more realistic and flexible in their thinking about their parent's illness. Unless they had pre-existing emotional problems, their behaviour appeared more constrained than younger adolescents although most showed irritability.

Whilst Christ's findings are important due to the nature of the study and sample size, there are limitations. The sample only included families who were white Caucasian, middle class, emotionally stable, economically productive, affluent and articulate: 'an idealised population' (Chowns 2006). Christ herself recognised that other contextual factors such as poverty, illness-induced separation and difficult family relationships may influence the outcomes.

#### Communication and sharing information

Communicating with and providing information to children have been found to be important ways of supporting children when their parent is ill (Christ 2000; Christ and Christ 2006; Thastum et al. 2008). Dunning (2006), writing on her practice in a hospice in North America, describes children whose acting out behaviour caused referrals to her service. When investigated she found that the basis of the behaviour was in misconceptions and fantasies about the parent's illness due to a lack of communication.

However, informing children that their ill parent is dying is a challenging and daunting task. A retrospective qualitative study of nine surviving parents highlights some of the issues and difficulties involved (Macpherson 2005). Parents' reluctance to tell children stemmed from their own feelings of powerlessness and vulnerability. Parents experienced confusion and paradoxical struggles: knowing that the death would happen but denying it. They were struggling to look after themselves, their partners and their children and felt torn between the conflicting needs, leaving them feeling isolated. All the well parents respected the wishes of the dying parent with regard to what information was given to the children. Siegel et al. (1996) in a larger study with 90 children aged 6-16 years found that even in families that have good communication practices, parents need guidance and encouragement to discuss the probable death of an ill parent. The study suggested that well mothers were more likely to be perceived by adolescents as 'open communicators' regarding illness and death than well fathers.

In a qualitative study involving interviews with parents and children, Kennedy (2008) explored the experiences of 12 families in order to identify the children's main underlying need/s. Children described wanting honest information and access to both emotional and practical support within and outside the family. This support was often described as lacking. Parents focused on difficulties in telling the children they were terminally ill and would have liked further information and support in this area especially in relation to younger children and those children with special needs. Communication was felt by both children and parents to be difficult due to each not wanting to upset the other. However, children who had open communication within the family were described as having better access to information and support both within and outside the family, leading to decreased concerns and increased coping.

Whilst for some children visiting a parent in a hospice may be a positive experience (Thastum et al. 2008), for others it has been found to be a forbidding and frightening

experience. They do not generally ask questions, possibly because they are intimidated by the environment and lack of opportunity (Thompson and Payne 2000). Giving children information by way of a workbook designed to facilitate communication between children, their parents and nursing staff was found in one small study to help children to become more involved in their visits to the hospice (Macpherson and Cooke 2003).

One small study of three mothers with cancer (Barnes et al. 1998) considered their communication strategies with their children who had learning disabilities and behavioural problems. The findings show that mothers chose to protect these children by withholding information; siblings, however, received more detailed and explicit information. The children with learning difficulties had particular difficulties coping with changes in routine, the observable changes in their mother and her unavailability. There were also reported behavioural changes in the children.

In researching the experiences of children facing the life-threatening illness of a parent, Chowns (2006) approached her study in a groundbreaking way. Aware that children's voices are not generally centre-stage, Chowns used a collaborative inquiry approach with nine children aged 7-15 years whose parents had cancer. The children, together with four adults, co-researched their experience through the making of a video for public use. The children came from five families, four with an ill mother and one an ill father. A major finding was that the children were clear that they preferred to be informed and involved, wishing to be told the truth as fully and as soon as possible. They saw delay and deception, whatever the motives, as a breach of trust; whilst they could understand why parents might not be open and honest with them, it did not, in their minds, justify their actions. Having to face the life threatening illness of a parent brought with it a sense of isolation. The uncertainties about their parent's future and their own were ever present. Children sought support and understanding rather than 'protection and

prevarication'. A key finding was that the children demonstrated sophisticated communication strategies, distinguishing between talking, telling and foretelling, and made careful choices as to when and with whom to 'not-talk'. Landry-Dattee and Delaigue-Cosset (2001) reporting on the benefits of support programmes for children noted similar findings.

These studies suggest therefore that children both want and need to be informed and involved. However, parents may not know how to prepare the children, particularly as they feel they need to come to terms with what is happening first.

#### Coping with the terminal phase of a parent's illness

Saldinger et al. (2003) found that the anticipation of death may provide one of the biggest strains of all for parents and their children. These findings are from a retrospective, qualitative study regarding 32 parental anticipated deaths involving 58 bereaved children, aged 6-16 at the time of initial contact, and their surviving parents. Saldinger and colleagues concluded that when a parent is terminally ill, the child's relationship with that parent is compromised by 'two mutually-reinforcing factors'. One factor is the parent's inability, both physically and emotionally, to reach out to the child, and the other is how able the child is to tolerate the parent's physical, mental and emotional deterioration. For those children who have a low threshold of tolerance, a guilty tension develops between the wanting and not wanting to connect with the dying parent. It is also easier for parents to overlook the plight of children who internalise their distress but children can also be protective of parents by not showing how they feel. The evidence from Sheehan's small study (2007) echoed these findings. However, none of the participants revealed any significant family dysfunction which may alter the process.

Christ (2000) also suggests that the terminal phase of a parent's illness may be a period of greater psychological stress for children than the period after the actual loss and that anxiety experienced during the course of the illness 'waxes and wanes' (p.24). This is supported by Siegel et al. (1996) who in a quantitative study showed that children reported significantly higher levels of depression and anxiety in the pre-death period but 7-12 months post death, their reports were similar to children in the community who had not experienced such a loss. Heiney et al. (1997) suggest that children, particularly aged 6-12 years, experience a higher level of anxiety themselves if their parents also display a high level of anxiety.

Siegel et al. (1992) compared 62 school-aged children with a terminally ill parent with a community sample using a variety of standardised rating scales. Children of a terminally ill parent self-reported higher levels of depression and anxiety and lower levels of selfesteem. Parents who were terminally ill reported higher levels of behaviour problems in their children than those in the community sample. Whilst able to describe behavioural problems, for example aggression (externalised behaviour), the parents were unable to describe anxiety and depression states (internalised behaviours). It has been suggested that the ill parents may not have recognised their children's emotional state and needs due to their own high anxiety levels or because behavioural problems are easier to recognise (Visser et al. 2004). Siegel and colleagues (1992) suggest that such distress among children may be seen as an important aspect of mourning. Such mourning might be temporary and helpful (anticipatory mourning) during the terminal phase, or something which might lead to pathological mourning and depression and anxiety in later life. In a later study Siegel et al. (2000) found that low levels of self-esteem in children correlated to being younger, having high levels of anxiety and also to the child's perceptions of the well parent's parenting. What was not explored was the quality of parenting as a variable in these studies.

Similarly, Beale et al. (2004) found that children with dying parents manifest significant distress as well as a greater understanding of their parent's illness than is usually suspected. They interviewed 28 children, who were aged 3-18 years, and had been referred to their Palliative Care and Symptom Service and their parents both together and alone. This was a group of children who had been referred to clinicians at the parents' request. They were seen in a playroom where they were free to choose a play activity. The results showed that all the children had some understanding of their parent's condition but were frightened about the outcome. Even small children demonstrated 'a remarkable awareness' of the illness and its implications. The children displayed high levels of aggressive behaviour and separation anxiety. The authors concluded that discussions between parents, children and other adult caretakers are important when the patient is first aware of the terminal condition. Timely intervention, particularly with children, they believe, should take place to encourage better communication and possible resolution of conflicts before death is imminent.

Other studies have found that children perceive very little opportunity for control over their parent's cancer and the more serious the cancer the more children try to avoid it (Compas et al. 1996). The children in Chowns' (2006) study found that a key strategy for coping was to focus on the positives in life. Another was a need to experience a safe place in order to both hold on to and let go of the strong emotions generated by the experience of living with a seriously ill parent. Siblings, it was found, could be both sources of support and stress, as were grandparents and friends. School was seen as another source of stress even though individual members of staff might be experienced as being highly supportive.

A small, retrospective, qualitative study of five young people who had experienced the chronic illness and death of a father showed that whilst a parent was ill they had felt powerless and isolated (Semmens and Peric 1995). These feelings of isolation and

alienation were echoed in another retrospective, qualitative study conducted by Dehlin and Martensson (2009), also with five adolescents who were aged 14-17 when their parent died. The adolescents described the illness and death as a great threat, and in spite of all the difficulties retained a hope that the parent would recover. They had developed different managing strategies to cope with the threat, including wanting and trying to make their lives function normally even though they were aware they were not. They saw their friends as a 'protected zone' where they did not have to think about the illness, and unlike responses within the Chowns' study above, saw school as a place where they could get away and be normal. Another strategy was to avoid talking about the illness. Again, the variability of parenting was not considered in these studies.

Children have consistent styles of coping with transitions and these will have an impact on their reaction to the parent's terminal illness (Rauch and Muriel 2004). Children living with a dying parent may be described as living in adversity, yet many will be able to cope and achieve positive outcomes (Spira and Kenemore 2000). This will be due to the existence of protective factors, 'resilience', that can help explain why one child may cope better with adverse life events than others (Alvord and Grados 2005; Daniel et al. 2010). Gilligan (1997) identified three bases that underpin the factors associated with resilience: a secure base, good self-esteem, and, a sense of self-efficacy. In a small, qualitative study of a Family Support Programme for children aged from 5 to 18 years, Bugge et al. (2008) found that children need help to understand themselves and to learn about how to deal with a range of difficult and confusing feelings, thoughts and reactions. They need to talk in private about the imminent death of a parent but dialogue with other family members is important too. Adolescents, for example, may have more people they can rely on than younger children to discuss matters with, for example, a teacher, school nurse, friends (Christ et al. 1994). From Dunning's (2006) experience, children require help to identify and label their feelings. This includes feelings about the current situation, losses already experienced and worries about their own future and

what will happen to them. If such help is offered this may act as a buffer against the effects of the adversity. Children however may be experiencing other adversities in life such as racism, marital violence, abuse and neglect (Daniel et al. 2010) prior to the additional adversity of the terminal illness of a parent to which palliative care social workers should be alert.

Particular difficulties have been identified in the literature regarding child carers. Aldridge and Becker (1999) and Dearden and Aldridge (2010) provide a detailed overview of research regarding children as carers. Disability and illness are often associated with poverty (Dearden and Becker 2000) and children are more likely to become young carers if their family lacks sufficient personal or family support (SCIE 2005). 'Children as carers' is explored specifically by Gates and Lackey (1998) in a small study of 11 children aged 10-19 years. Whilst the study mainly considers children's care of older relatives in the household, rather than parents who had cancer, it does highlight certain key points. Caring responsibilities frequently interfered with doing homework and spending time with friends. Feelings of stress, fear and fatigue were described by the children, and observations made during the study provided evidence of the effect of the stress and strain of the illness on the family. The children expressed anger about not being informed about the diagnosis, the current status of the illness and changes in treatment (Lackey and Gates 2001).

In one study, Aldridge and Becker (1993) found that young carers were often reluctant to identify themselves or make it known how much caring they do. They were frightened of professional intervention that might lead to separation and the disintegration of the family as well as a fear of bullying from other children. As such, young carers may often feel isolated (Underdown 2002). They may have fewer opportunities for social activities and caring duties which set them apart from the friends that they do have and inhibit them forming new friendships (Thomas et al. 2003). More than a fifth of all young carers

in a UK study experience educational problems and this is more marked in the 11 to 15 age group (Dearden and Becker 2004). Being a young carer may put additional pressures on a child or young person at a time in their life that is potentially already stressful and support may not always be available (Cree 2003). However, whilst there might be detrimental effects for children there are also positive aspects too, with children being provided the opportunity to develop and mature in terms of confidence levels and organisational abilities (Aldridge and Becker 2003).

# **Family functioning**

Family functioning has been found to deteriorate when a parent has cancer (Visser et al. 2004). In a review of earlier research, Lewis (1990) came to the view that it was probable that a small number of families might be at high risk for functional breakdown but that studies to date had focused on identifying high risk patients rather than high risk families. Since then more research has been conducted, although there are more studies that consider family functioning during the initial phase of illness rather than the terminal phase (Huizinga et al. 2003; Zaider and Kissane 2009).

A parent with a terminal illness presents a challenge for the family and often requires changes in parenting roles, responsibilities and goals (Helseth and Ulfsaet, 2005). Parents describe themselves living in a state of emergency. The illness consumes their energy, physically and emotionally but they strive to be good parents. The prospect of the death of a parent may revive memories of past losses and hurts (Kissane and Bloch 2002) for children as well as adults, for example due to unfinished business through remarriage. This may cause tension and conflict in the family, therefore affecting its functioning. There is also evidence that maternal depression can impair family functioning, particularly in domains of problem solving and role functioning, and a sense of coherence in the family can improve it (Schmitt et al. 2008). The quality of marital

adjustment can further be affected by the level of depression in women and the demands experienced by the illness (Lewis and Hammond 1992). The depression of the mother has been found to be more significant than that of the ill parent, which suggests that if a father is ill it is the mental health of the mother that remains the important factor in maintaining positive family functioning. However, there is little research on how much the terminal stage of the illness of one parent impacts on family functioning, on parenting generally, or about how men's experience of their own illness affects their role as fathers. In her study of six families facing the death of a parent, Kennedy (2008) describes the participating family members as being profoundly distressed. The physical and emotional manifestations of the illness were all-consuming for parents which led to their children's needs taking a 'back seat'. For the well parent there was a struggle to reconcile the competing claims being made upon them from the sick parent and their children.

Family functioning during palliative care is the focus of a research study by Kissane and Bloch (2002) into the effectiveness of family-focused grief therapy in preventing pathological grief during which they screened families of patients dying of cancer. Whilst the focus of this research is on a family-centred model of care, the screening process used validated measures to identify dysfunctional families who were then invited to participate in the trial. Although this study did not solely concentrate on families with children under the age of 18 (the mean age of the children was 29), it is helpful here as it considers descriptors of family behaviour that may indicate where there could be concerns about children in a family due to poor family functioning.

Kissane and Bloch (2002) describe three key dimensions of family functioning: cohesiveness, conflict and expressiveness, and allocated names for five classes of

families that formed their typology of family functioning<sup>6</sup>. Two showed characteristics of well-functioning: 'supportive' and 'conflict resolving'. 'Supportive' families are highly cohesive, have good expressiveness and are low in conflict. 'Conflict resolving' families have moderate conflict characteristics, high cohesiveness and above average expressiveness. They are able to tolerate a degree of difference of opinion with respect. In the authors' view, neither of these classes need specialised psychological help and in the clinical work it is estimated that a third of families are 'supportive' and a fifth are 'conflict resolving'. Two classes were clearly dysfunctional, however. 'Hostile' families (about 6% of the families) are described as tending to reject help and are distinguished by high conflict levels, poor expressiveness and poor cohesion. They are fractured and chaotic, structure and order are absent, and children, primarily adolescents, may become scapegoats for unresolved tensions elsewhere, including those in the parental relationship. 'Sullen' families (about 9% of the families) are described as having moderate levels in each area. They have the highest rates of depression and exercise the highest level of control over family life. The final class was 'intermediate' (a third of the families) because of their position between well-functioning and dysfunctional types and are characterised by moderate cohesiveness. Therapy by those trained in family focused grief therapy is offered to families thought to be at risk before bereavement and takes a systemic approach which includes children. This study is included here because it demonstrates that a proportion of families with children who are likely to come to the attention of palliative care social workers will be affected by poor family functioning. Whilst not advocating undertaking family therapy, Lichtenthal and Kissane (2008:45) are of the opinion that applying a family-centred approach to assessment and intervention increases the likelihood of successful resolution of family strain.

<sup>&</sup>lt;sup>6</sup> The authors stress that the names are not intended as clinical descriptors to be shared with the family but as a 'shorthand' in communicating about the model (Kissane and Bloch 2002:37).

#### Conclusion

Having a parent who is dying is a stressful life experience for a child. Much of the literature on the experiences of this community of children concentrates on the importance of open communication between them and their parents. Others examine the psychological effects on the children. The research provides evidence that children have higher levels of anxiety and distress than the norm population and parents may not recognise the symptoms. The quality of parenting was rarely a variable in the reported studies, however.

There are lessons from the literature that may be applied to palliative care social workers in safeguarding and promoting the welfare of children. These include: recognising the importance of the age and stage of development of the children; parents and children 'seeing' things differently; and parents needing assistance to recognise and understand their children's needs as children, including those with special needs, who want and need to be kept informed about what is happening. Young carers may be particularly vulnerable and children generally may feel isolated and look for support from people they can trust. Maternal depression is likely to have an influence on family functioning; children, particularly adolescents, whose behaviour is causing concern may be reacting to poor family functioning. There is a need to consider what protective factors may be put in place for children already affected by other adversities.

In the following chapter, I will explore the literature on safeguarding and promoting the welfare of children with particular reference to how it relates to palliative care social workers based in an adult-focused service.

# Chapter Four Safeguarding children: implications for practitioners working with adults

Safeguarding children is everyone's responsibility. HM Government 2007

# Introduction

The literature on safeguarding children is fairly extensive and so I will draw on the most relevant for the study: the implications of safeguarding children for those practitioners working primarily with adults as their focus. In doing so I will consider the role of adult services and children's needs; interagency relationships; parenting, with reference to the potential impact of serious illness on its effectiveness; assessment of children in need; and last, but not least, the child's voice. The literature reviewed consists mainly of legal, policy and guidance documents, practice literature and statements of opinion. Reference is made to empirical research studies where available, the majority of which relate to practice in England. The first section briefly considers the background and context as well as the prevalence and incidence of abuse in the general population before moving on to explore the current practice regarding safeguarding children in England, Wales and Scotland.

# **Background and context**

Since the establishment of Children's Departments by the Children Act 1948, there have, according to Corby (2006a), been three phases in the provision of services to children and their families<sup>7</sup>. The first phase was from 1948 to the early 1970s which was characterised by 'a family breakdown approach'. The second phase, from mid-1970s to

<sup>&</sup>lt;sup>7</sup> For a detailed description of the developments in England and Wales see Ball (1998); Holman (1998); Stevenson (1998); Hendrick (2003); Ferguson (2004); Corby (2006a, 2006b), and for those in Scotland see Murphy (1992); Hothersall (2006); Stafford and Vincent (2008).

the mid-1990s, was a 'period dominated by the goal of child protection'. The third current phase, from the mid-1990s, is one of 'support to families'. Child welfare practice in Scotland is underpinned by the Children (Scotland) Act 1995 and in England and Wales by the Children Acts 1989 and 2004. The need to improve the quality of life for all children has informed the governments' planning and guidance. Initiatives have been introduced in all three countries to target children and families in need of support as well as those defined as children in need and those at 'risk of harm' (Horwath 2010:30).

Whilst the philosophy of the Children Act 1989 was to provide support to families, workers were found to be much influenced by the public inquiries of the 1970s and 1980s (Aldgate and Tunstill 1996). The findings of 24 studies commissioned by the Government were published in Child Protection: Messages from Research (Department of Health 1995). The research found that professionals had not achieved a satisfactory balance between child protection, family support and child welfare. There was too much stress on child investigations rather than enquiries, and a failure to provide family support to meet the needs of children and their families. Only a small number of referrals actually led to child protection registrations. Further, the assessment tended to be one of risk rather than one of an assessment of need (Thoburn et al. 1997). There was also a concern that children were being routed inappropriately into the child protection system as a means of gaining access to services (Platt 2006a). Calls were made for a 'refocusing' of child protection work which 'prioritised' section 17 and Part 3 of the Act in order to help support families with children 'in need' (Parton 2007:19). In response, government guidance (Department of Health 1999b) expected that child care referrals would receive initial assessments, using a framework to assess needs unless the situation clearly called for a child protection core assessment. Policy had moved from concern about 'the family' to 'the welfare' of the children, their vulnerability and well-being, whilst at the same time upholding parental responsibility (Parton 2006).

Lord Laming's inquiry into the tragic death of 8 year old Victoria Climbié (Laming 2003) highlighted poor practice within and between the agencies. The resulting Green Paper Every Child Matters (DfES 2003) led to the introduction of the Every Child Matters: Change for Children programme (HM Government 2004). The Government's aim was for every child, whatever their background or their circumstances, to have the support they needed to be healthy; stay safe; enjoy and achieve; make a positive contribution; and, achieve economic well-being. The Children Act 2004 provides the underpinning for the programme. It requires local authorities to lead on integrated delivery through multiagency children's trusts, to develop a children and young people's plan, and to set up a shared database of children, containing information relevant to their welfare. A raft of statutory and other guidance supporting the programme has been produced (HM Government 2005). Area Child Protection Committees have been replaced in England (HM Government 2010) and Wales (Welsh Assembly Government 2006) with Local Safeguarding Children's Boards (LSCB) with stronger statutory powers. Children and young people are now to have far more say about issues that affect them, individually and collectively (HM Government 2010).

Following devolution in 1999, the Scottish Parliament made children and young people a key priority in the shaping of its policies. The findings of a multidisciplinary audit and review into the child protection arrangements in Scotland *'It's Everyone's Job to Make Sure I'm Alright'* (Scottish Executive 2002) drew attention to the serious weaknesses in the system and, despite examples of good practice with children, identified a significant number of children left unprotected or with their needs not met (Daniel 2004:247). A three-year Child Protection Reform Programme '*Protecting Children and Young People*' was introduced with an action plan designed to provide early support to children and families, better direct help to children in need of protection and effective continuing support. A review assessing the progress of the Programme provided evidence that the programme, according to practitioners from different disciplines, had had a positive

impact and that there were improvements in child protection practice (Daniel et al. 2007).

In 2001, the Scottish Executive published For Scotland's Children: Better Integrated Services (Scottish Executive 2001) highlighting the problems caused by a lack of joint working across children's services and making recommendations for integrated children's services. The Local Government Act (2003) places a duty on local councils and their partners (particularly police and health services) to develop Community Plans that bring together the delivery of local services. The findings of a major review into the effectiveness of the children's hearings system (Scottish Executive 2004) led Scottish Ministers to look at children's services more widely and published proposals in 'Getting It Right For Every Child' (Scottish Executive 2005a). The resulting Getting it Right For Every Child programme (GIRFEC) aims to 'place children at the centre of service provision in Scotland' (Scottish Executive 2006a:3) and to refocus children's services towards prevention, early intervention and family support. In particular, the agency of first contact must aim to offer help; the child should not automatically be passed to another agency. The aim is to ensure that every child is safe, healthy, active, nurtured, achieving, respected, responsible and included. It seeks to do this by providing a framework for all services and agencies working with children and families with a view of achieving a co-ordinated approach. It is a long-term programme which recognises that changes in systems, practices and professional cultures take time to develop. There are various projects in selected local authorities that have specific remits to test the implementation of the GIRFEC approach (The Scottish Government 2010). Child Protection Committees replaced Area Child Protection Committees with a clearer remit and duties (Scottish Executive 2005b).

Whilst Scotland has not moved so clearly from 'protecting children' to 'safeguarding children', the philosophy is similar and the resulting policy and guidance carry similar

implications for children, families, agencies and their staff as their counterparts in England and Wales.

#### Prevalence and incidence of abuse in the general population

A study of the prevalence of child maltreatment (Cawson et al. 2000) estimated that 7% of children under 18 years had experienced serious physical abuse at the hands of parents during childhood. Another 6% had frequent and severe emotional maltreatment. Serious absence of care<sup>8</sup> was experienced by 6% of children. Of those children under 16 years who had experienced sexual abuse, 1% was by a parent or carer, 14% by another relative or someone known to them and 4% by an adult stranger or someone they had just met. Bullying by other children was a feature of childhood experience of almost a third of the sample. Another study which examined abuse and neglect within the family (Cawson 2002) found that 16% of children (1 in 6) experienced serious maltreatment by parents with one third experiencing more than one type. Maltreatment was more common in families where there was social disadvantage, stress, health problems, domestic violence and family breakdown.

Incidence refers to the number of new cases identified as occurring in any one year. Statistics gathered by local authorities include those for the number of child protection referrals<sup>9</sup> made to the local authority in any one year as well as the number of children placed on the child protection register or in England are subject to a child protection plan (see Table 4.1). However, they can only be an indicator of the size of the problem, as abuse and neglect are both under-reported and under-registered. At 31 March 2007, the rate of registration per 10,000 children under 18 was 25 in England, 36 in Wales and

<sup>&</sup>lt;sup>8</sup> Serious absence of care at home covers aspects of physical neglect e.g. frequently going hungry; not being taken to doctor when ill; regularly having to take care of themselves when parent had problems with alcohol and/or drugs.

A referral is the first stage of the child protection process in all areas of the UK.

28 in Scotland. Child neglect accounted for the largest number of registrations: 44% in England, 52% in Wales and 49% in Scotland (Vincent 2008).

Table 4.1 Number of child protection referrals and registrations 2000/07			
	England	Wales	Scotland
Number of referrals in year ending 2006/07	545,000	49,561	11,960
Total number of registrations as at 31 March 2007	27,900	2,295	2,593

Table 4.1 Number of child protection referrals and registrations 2006/07

Whilst it is difficult to give an accurate figure for the prevalence of child maltreatment, the above figures would indicate that it is likely that palliative care social workers will be working with some families where child maltreatment has occurred or is occurring.

# **Current practice**

The policy of 'safeguarding' has a much wider remit than 'protecting children from abuse and neglect' (Munro and Parton 2007:14). The governments' plans therefore are to mainstream services so that they can support all children, including those at risk, preferably in their home environment and to use family support as a primary mechanism to protect children from harm (Axford and Little 2004:168). This approach is no longer just the responsibility of social services but of all professionals who might come into contact with children (Axford and Little 2006).

However, a lack of resources both in terms of personnel (staff recruitment and retention) and money for a wide range of support services creates difficulties. Social workers are struggling to deliver both child protection services and family support services (Spratt 2001). Because of the inadequate services available, families with low-level problems are often turned away without receiving the help they seek (Munro 2007). From a child protection file study of 205 children in one local authority, Thorpe and Bilson (1998) suggest that once a referral was judged to be one that required investigation, any welfare or needs issues that were beyond this focus were frequently ignored, and where

child protection issues were not substantiated, the cases were closed and no services were provided.

Although a case might be classed as one of family support it does not in Spratt's view (2001) negate the social worker having to manage the tension between support and evaluating risk. Platt (2006a:278), in a qualitative study conducted in two local authorities between 1999 and 2001 on the impact of the refocusing initiative, found that initial assessments on 'borderline' cases enabled social workers to focus on both helping and protecting children. However, the focus on the event that triggered the referral was often the deciding factor on the choice between an investigation and initial assessment together with the worker's interpretation of its seriousness (Platt 2006b:13).

Parton et al. (1997) have argued that social workers mainly carry out their duties as defined by their employing organisation and both social workers and organisations are anxious not to be in a position where they are blamed if things go wrong. Recent child deaths, for example, Victoria Climbié, Baby Peter and Kyra Ishaq, can only heighten this anxiety and so issues of child protection take priority (Aldgate and Tunstill 1995).

A child 'in need' refers to a child's state of development and health, both of which are defined in s.93 of the Children (Scotland) Act 1995 and s.17 of the Children Act, 1989. However, no definitions are provided for the legal terms 'a reasonable standard of health and development' or 'significantly impaired'. Because it is only a duty for authorities to 'take reasonable steps' and 'as they consider appropriate' it depends on how each local authority (and social worker) defines and prioritises the concept of 'a child in need' (Colton et al. 1995). 'Vulnerable children' are defined more broadly than 'children in need' and are described as those who have acquired or encountered some difficulty that requires additional help from public agencies if their life chances are to be optimised (Preston-Shoot and Wigley 2005). Out of a total of 11 million children in

England, the government estimated in 2004 that 3.66 million were 'vulnerable children' and 388,200 were 'children in need' (DfES 2005).

Local authorities throughout England, Wales and Scotland are legally required to determine the nature and extent of children's needs in their areas in order to provide a needs-led service. Axford et al. (2009) conducted an analysis of reports on children's needs produced by two local authorities in England between 1997-2007. They drew particular attention to one local authority that investigated the need for a support service for young people aged 11-18 years who were struggling to come to terms with parental separation, divorce or other issues related to family change (p. 247), a definition which might include the death of a parent. It is not clear whether local authorities generally might consider that children who are risk of being, or are bereaved, might be 'children in need'.

In a study of eight local authorities in Wales and a comparable eight in England, social workers and managers considered that ideally there should be more emphasis on family support work but resources for doing this were scarce (Colton et al. 1995). Social workers were found to be interpreting the concept 'a child in need' in a wide variety of different ways with little agreement as to how it should be defined. In the absence of any substantial guidance social workers were relying on their own experience plus material that was primarily designed for child protection work. The Social Services Inspectorate (1997:1) in their report on the inspection of management and practice of eight local authorities noted the imbalance: 'Departments continue to respond to child protection and looked after children to the exclusion of other families of children in need'. However, Spratt (2000) in a small study found that Inspectors themselves were not able to agree about whether situations described in a number of vignettes were child protection investigations or child-care problems. This highlights the general problem that there is no agreement regarding definitions. Jack (1997) has argued for a more fundamental

change in attitudes towards social welfare provision for families involving a 'deconstruction of child protection discourse and the development of a new child welfare discourse' (p.670). If a lack of clarity exists within children and family services about 'children in need', then this is even more likely to be the case among workers in agencies for adults, such as palliative care social workers.

#### The role of adult services and children's needs

The current discourse that 'it is everyone's job' or 'it's everyone's responsibility' to support children has significant implications for those working primarily with adults who may also be parents. This is particularly pertinent in situations, like palliative care, where there is the potential for children's lives to be significantly affected in different ways, for example by changes to parenting capacity, family dynamics, and living conditions (Barnard 1999; Kroll and Taylor 2003).

As there were no studies found that relate specifically to palliative care, it is possible to draw analogies from those in other fields of adult services. For example, in their qualitative study of practitioners who encountered parental substance misuse in both adult-focused and child and family settings within the voluntary and statutory sector, Taylor and Kroll (2004) report that child and family workers expressed concerns about the division between services for children and those for adults. Not only was there a different agenda at times but a 'polarization' that significantly affected the assessment process. Taylor and Kroll consider that mental health professionals appeared to avoid asking their adult patients or clients crucial questions about their children. Specialist drug workers were similarly too exclusively focused on the needs of their adult clients, and concerns about sustaining trust and breaching confidentiality appeared at times to override the principle that the child's welfare was the paramount consideration. Reder and Duncan (1999:56) similarly write about professionals getting so caught up in

assessing parents' mental health that 'it interfered with thinking about the needs of the child'.

During the mid-1990s social work practitioners 'rapidly perceived themselves as being deskilled as generic social workers' (Pearce 2003:114). She describes the separation of adult care, mental health and child care services as leading to 'a distance of workers from each other, with a consequent lack of understanding about both the work that is carried out and the issues that inform practice' (p.115). Drug social workers in Taylor and Kroll's (2004) study felt uncertainty in assessing parenting and child development, whilst child care workers felt uncertainty about the impact of substance misuse. All the staff from non-child care agencies felt they lacked experience and training in child protection issues and the majority of child care workers felt vulnerable in relation to knowledge about drugs and alcohol. There were anxieties and insecurities therefore for both child care workers and adult based professionals about their levels of knowledge and the degree to which they felt equipped to assess behaviours outside their immediate realm of expertise. Further, all practice regarding children raised dilemmas about access, confidentiality and the distress caused by encountering children living in damaging environments. The same issues could apply both to palliative care social workers, who may feel unsure about assessing parenting capacity and child development, and to child care workers who may not understand the impact of terminal illness on parenting and on children in the family. Indeed, some workers may not feel able to admit their lack of understanding and others may not appear to value the knowledge that other workers may have about the family (Stevenson 2007).

D'Cruz and Stagnitti (2008:158) draw attention to the fact that parents and professionals may unintentionally become 'collusive'; if professionals identify so closely with the parents' experiences they may excuse or minimise problematic parenting. Where a parent is dying, the concerns of the child, according to Segal and Simkins (1996), can

very often take second place. In addition, there may be feelings of betrayal: a worker avoids raising concerns with parents and child protection agencies to protect their relationship with the parents. Professionals may also wish to avoid raising difficult issues with ill patients because doing so may cause distress and this may be particularly the case where medical advice is against doing so (Altschuler 1997). Interestingly, Fitzgerald and Kay (2008) take another view. They consider that safeguarding may best be achieved through services to adults in terms of parenting support, for example, drug and alcohol services, although palliative care services would also be relevant here. This is because many services to children have high thresholds for access and families may not otherwise receive support.

Children as carers can also present challenges to workers whose focus is the parent. I drew attention to the stress that children caring for an ill parent may experience in chapter three. Young carers have only recently been recognised as children who may need support and services in their own right (Dearden and Becker 2001). However, significant factors such as lone parenthood, low income or poverty, lack of adequate parental and family services and informal support may lead to a child to undertake 'the type of caring responsibilities that may be inappropriate to their age and level of maturity' (Aldridge 2008:254), in turn making them vulnerable. Often young carers are hidden from children's and adults' services or may be treated as a low priority until there is a crisis (The Princess Royal Trust for Carers 2006). In addition, many children and families are not aware of their right to an assessment either under s.17 Children Act 1989 and s.93 Children (Scotland) Act 1995 or the Carer's Recognition and Services Act 1995 (Underdown 2002). Subsequently, young carers are not often assessed as children in need. Dearden and Becker (2001) suggest four reasons for this:

 a lack of awareness among health and social services staff about young carers' needs and rights

- families' fear of professional interventions that the children will be removed from home
- a lack of consensus regarding which department, community care or child and family services, should be responsible for assessing need with children falling through the net and their needs not being met by either service
- whether young carers should be assessed as children in need, at risk or as carers (p.227).

Young carers themselves have emphasised that for an effective assessment to take place, staff in children's and adults' services should communicate with each other (Underdown 2002). Using a whole family approach to assessment under the above specified legislation would ensure that the needs of all the family members are taken into account rather than a concentration on either the parent or child.

# Inter-agency relationships

The concept of inter-agency working in safeguarding children and providing family support services is at the core of governments' policies in this area. Guidance was first issued in 1988 by the Department of Health and Social Security. The latest edition in England is *Working Together to Safeguard Children* (HM Government 2010) and in Wales is *Safeguarding Children: Working Together Under the Children Act 2004* (Welsh Assembly Government 2006). The current document in Scotland is *Protecting Children - a Shared Responsibility: Guidance on Inter-agency Co-operation* (The Scottish Office 1998) and is under review. However, interagency working is something that has proved to be very difficult thing to achieve and continues to be a constant theme in inquiries, inspections and serious case reviews as a contributing factor as to why children have not been protected (Laming 2003; Ofsted et al. 2008; Rose and Barnes 2008).

to get right (Walker and Thurston 2006). Reder and Duncan (2003:82) comment that 'problems of communication seem to haunt professional practice'. They put forward various factors that can have an effect on inter-agency relationships including status and power; stereotypes; value systems about children and families; disrespect for the other's expertise; resource pressures; and, mutual unfulfilled expectations. The literature describes how different professional groupings have their own knowledge and practice base that they will draw on which will influence the way they consider issues of safeguarding children and child protection (Hallett and Stevenson 1980) and may inhibit a development of positive inter-agency relationships. There have been some reported improvements (Joint Chief Inspectors 2005, 2008), however. In Scotland, findings from the evaluation of the Highland Pathfinder<sup>10</sup> (The Scottish Government 2009:135) have shown the 'emergence of an inter-professional culture to support multi-agency working across children's services'. This has been partly attributed to collaborative working according to a set of agreed principles and values. There was evidence of more interagency trust and a shared understanding of children's needs. Whilst the language of tariffs and thresholds was found to be still present, it was less apparent in inter-agency and inter-professional language than at the outset of the project. The engagement of frontline workers in adult services has been identified as a target for the next phase.

Whilst the literature on multidisciplinary working in child welfare is growing (Frost and Robinson 2007) the emphasis seems mainly to be on professionals from different disciplines working together. Such literature has notably explored how the construction of their identities, including their value base, influences their way of interpreting situations and seeking solutions. This aspect was explored in chapter two when considering the palliative care social worker as a member of a multidisciplinary team.

<sup>&</sup>lt;sup>10</sup> Pathfinder projects have been established 'to help shape, develop and test the practice tools and training materials and to inform the development of national guidance'. The Highland Pathfinder was launched in September 2006 with a remit to address all aspects of children's needs from birth to 18 years (The Scottish Government 2009).

What appears to be lacking is an in-depth exploration of the experience and relationship between social workers from agencies or teams with an adult focus and social workers in local authority child and family teams. These 'adult service' social workers may be employed in voluntary or independent organisations as well as the local authority. This raises the question of whether there is a common understanding from a social work profession perspective about the child's welfare or whether the focus on the adults and their needs clouds this view.

Horwath and Saunders (2005) undertook an in-depth study into the identification, referral and assessment of child neglect. Although based in the Republic of Ireland, the policies are similar to those in England, Wales and Scotland. This study involved a range of 794 professionals using postal questionnaires and 85 in focus groups who worked directly with children or parents/carers in one Health Board area. The study highlighted the following:

- no agreement among professional groups or between the same professionals about the types of neglect that should be referred to social work teams
- no shared common language or baseline e.g. 'good enough parenting' was interpreted in different ways
- lack of an evidence based approach towards practice
- a mixed understanding by professionals referring cases about the operation of the child protection system
- although there were examples of positive experiences of working with social workers, many professionals focused on the negative experiences
- the referring of children was seen as a last resort (pp.108-109).

Harlow and Shardlow (2006), in a small study of core group practice in a northern metropolitan borough of England, found that the turnover and shortage of social workers

in local authority departments can lead to a disruption of relationships between staff in other agencies. This lack of availability can lead to gaps in service, and difficulties in building trust. Laming (2003) also drew attention to the number of social work vacancies and the difficulties for new and inexperienced social work staff joining a department. In their study Harlow and Shardlow (2006:69) found that social workers expressed the view that other professionals had a lower tolerance of risk and that sometimes 'other professionals were worrying unnecessarily and that referrals would be made about relatively minor matters about which they could do little'. Those professionals with adults as their focus were criticised for only seeing the situation from the parent's perspective and there were issues about how much information could be shared.

# Parenting

The structure of families has changed in the United Kingdom. There has been an increase in remarriage and cohabitation leading to reconstituted families and complicated family ties. Pugh (2005:45) draws attention to these changes in family life: two-thirds of children are born outside marriage, one in five live in a lone parent family and one in four will experience their parents' divorce with a large majority losing contact with their fathers. Families are becoming smaller and more geographically dispersed which can lead to a sense of isolation for both parents and children. These factors may lead to complications regarding the present and future care of a child when a parent is dying.

Studies of parenting in poverty agree that living in poverty can affect all involved (Tuck 2000). Poverty brings with it a sense of apathy and feelings of powerlessness and helplessness (Blackburn 1991). Poverty and health are linked in terms of intertwined, physiological, psychological and behavioural processes 'by which poverty may influence a person's susceptibility to a disease or condition' (p.44). Blackburn provides evidence

to show direct connections between poor diet and ill health and, poor housing conditions and ill health. Physical symptoms can affect emotional behaviours. Parents in poverty may 'have to make health choices that serve to protect one aspect of health or the health of another family member whilst undermining another aspect of health or the health of another person' (p.46). However, Stevenson (2007) points out that the physical health of parents is rarely raised as an issue in Serious Case Reviews<sup>11</sup>.

The capacity of parents to care for their children is now one of the three key areas in the assessment of a child's needs and vulnerability. However, up until the refocusing debate, research had focused on the concepts of abuse, risk and need and not on the detail of parenting skills (Woodcock 2003). Now that parenting has become a matter of public concern, recent research has looked at family support and a response to need. Although parenting has been an aspect of these research studies, Woodcock puts forward the view that social workers' construction of parenting is generally a missing dimension. In a small exploratory study of social workers in one local authority department, Woodcock's (2003:100) analysis revealed four types of expectations that underlie judgements of what was or what was not good enough parenting. The social workers' expectations of parenting were to prevent harm; to provide routinized and consistent physical care; to know and to meet appropriate developmental levels; and, to be emotionally available and sensitive. The overriding concern was the capacity of the parent to prevent harm.

Hoghughi and Speight (1998) consider that parenting can be referred to as a relationship, a process, a group of activities and anyone concerned with the care of a child may be seen as part of the parenting process. They describe parenting as a

<sup>&</sup>lt;sup>11</sup> Serious Case Reviews are undertaken by LSCBs in England and Wales and Child Protection Committees in Scotland when a child dies and abuse or neglect is known or suspected to be a factor in the death (Welsh Assembly Government 2006; Scottish Executive 2007; HM Government 2010).

'process that adequately meets the child's needs, according to prevailing cultural standards' (p.294). Donald and Jureidini (2004:8) include in their definition of the quality of parenting the 'ability of the adult to recognise and adequately provide for, in a developmentally and emotionally appropriate manner, a child's current and anticipated needs'. Factors in the child and environment shape the quality of the parenting but the determinant of the experience resides with the 'parenting capacity'. This capacity they argue 'is not a measure of how people go about the task of parenting but the psychological qualities they bring to those tasks'. An example given is whether parents give priority to the child's needs over their own (p.12). When caring for children in the family, some parents have reported that losing some aspect of their health 'tips the balance between managing to be a good enough parent and not managing at all' (Segal and Simkins 1996:17). This resonates with findings in the literature discussed in the previous chapter regarding parents who are terminally ill.

The concept of 'good enough parenting' was first used by Winnicott (1965) to describe an environment facilitated by parents that enables a child's needs to be met. He recognised that the vast majority of parents are 'good enough' to meet their child's needs. In *'Child Protection: Messages from Research'* (Department of Health 1995) various studies drew attention to the fact that parenting which is lacking warmth and consistency, and which is excessively critical and punitive, is particularly harmful to children. The bonding between a parent and child and the subsequent attachment is now recognised as an important element in our understanding of parenting (Daniel and Rioch 2007) and secure attachment relationships are considered to be key (Howe 1999, 2005). Assessment of parenting capacity needs to take into account the power structures within families (Stanley et al. 2003) and not to concentrate solely on the mother's role; the role of fathers should not be overlooked (Daniel et al. 2005). A parent's own experiences of abuse and neglect may shape their capacity to parent (Woodcock 2003). As a result of an audit of cases, Macpherson (2009:483) surprisingly

discovered that of the 59 families with a terminally ill or deceased parent who had direct responsibility for a dependant child under 20 years in their home, in 49% one or both parents had experienced sexual abuse in childhood. This may mean that this group of parents are particularly vulnerable during the terminal phase of their illness.

Being alert to current and previous behaviours in the family can be important when shaping a picture of the quality of parental functioning, including how children and parents have related to each other, sources and types of stress and disadvantage, as well as sources of support (Woodcock 2003:91). The impact of domestic violence for example, can reduce a mother's self esteem and undermine her confidence (Mullender et al. 2002).

A diagnosis of terminal illness of one partner in a long-term relationship, however, may not change a negative relationship overnight (Altschuler 1997). Altschuler goes further by stating that although illness may alter experiences of abuse, neglect and marital disharmony it may also 'determine the availability of family members to one another' (p.77). She also differentiates between 'parenting' which she describes as the emotional responsibility parents have for children and 'parental activities' which is what people do as parents. She believes that it is important for a parent to remain 'emotionally available' for the child for as long as possible as this will contribute 'most powerfully' to the child's well-being. When a parent can no longer provide this support, particularly in the case of one parent families, it is important that there are other adults either within or outwith the family who are able to take this on.

Segal and Simkins (1996), writing from their experience as counsellors for people with multiple sclerosis, describe how each family member has to deal with feelings of aggression and control. In their experience disabilities and illness can exacerbate these feelings and also the need or wish to control those around them. Children and adults

may be a danger to each other: 'destructive and self-destructive behaviour always poses multiple problems for professionals, especially when children are involved' (p.89). They describe how an illness may cause extremes of parental behaviour. The parent may just wish to withdraw from their responsibilities leading perhaps to unintentional cruelty and neglect, or they may attack their children in some way in order to cause suffering as they are suffering. A parent may continue to provoke a child in order to elicit behaviour so that 'punishment' can be claimed to be 'justifiable'. Certain types of illhealth and treatments can also change people's behaviour, personality and characteristics which may result in a variety of negative responses towards their children, putting them at risk. Segal and Simkins (1996:99) also describe a 'more concealed kind of cruelty' occurring when attempts are made to punish and control children by 'intentionally exaggerating' the effect of the behaviour by saying for example, 'You'll kill your father if you go on like that'.

The existing literature on the potential impact of terminal illness on parenting, coupled with the prevalence of maltreatment in the general population, suggest the need for palliative care social workers to be alert to the indicators of the need for action in relation to children.

## Assessing children in need

I criticised the literature reviewed in the previous chapter for either not considering or not considering enough the wider social context of families affected by the terminal illness of a parent. Whilst the psychosocial aspects of the impact of the illness on family members are important, so are the other factors which will have an influence on the way the family functions.

In England and Wales, the Framework for the Assessment of Children in Need and their Families (Assessment Framework) (Department of Health et al 2000) is a conceptual framework for assessing and identifying vulnerable children and children 'in need' and their families. It draws on an ecological model (Bronfenbrenner 1979) and provides a systematic way of looking at and analysing what is happening to children and young people within the context of their families and the wider community. This framework, Horwath (2001:33) believes 'should ensure that practitioners, managers and policy makers maintain a child focus, irrespective of how the world of the child changes and develops over time'. In Scotland, the Integrated Assessment Framework (IAF) follows similar principles to the Assessment Framework, setting them within the Scottish context, and is seen as the cornerstone of Scottish children's services policy (The Scottish Government 2005). The frameworks are conceptualised as a triangle with three domains: the child's developmental needs; the capacities of the parents or caregivers to support their child's development and respond appropriately; and, wider family and environmental factors which may impact on development and parental capacity. Each domain is divided into dimensions that describe the key aspects that should be assessed under each domain (Adcock 2001). Houston (2002) describes the Assessment Framework as a 'watershed' in three ways. It emphasises the connection between poverty and child developmental delay, advocates a systemic approach and moves away from the preoccupation with child abuse as opposed to children's needs.

The frameworks are considered to be the most used ones in the UK (Crisp et al. 2005:42), mostly within the statutory services, to locate risk within the broader context of children's needs (Helm 2010). In keeping with the move towards a more multidisciplinary approach to the welfare system, the frameworks have been designed to be used by any practitioner, irrespective of discipline, at the first sign of emerging vulnerability about a child (Horwath 2007a:1286). The guides and supporting tools (HM Government 2006a; 2006b; 2006c) make clear that the frameworks are a generic

assessment of needs which aims to help early identification of need and promote coordinated service provision. Horwath (2010b:57) describes them as maps to help practitioners across all disciplines organise their thinking about what is happening to a vulnerable child in order to assess and meet their needs. Indeed, they would be helpful to palliative care social workers, particularly as they may not work with families on a regular basis where there are concerns about a child.

#### The child's voice

In whatever setting they work, professionals are likely to miss signs that something is wrong if they are not able to hear the child's voice or consider their view point. Again, the literature from child protection and safeguarding children research provides useful pointers for palliative care social workers both for being alert to concerns about a child's welfare and more generally in relation to providing support to the child and parents.

Earlier, I drew attention to children not having agency. Finding out the child's wishes and feelings is a significant aspect of the Children Act 1989 and Children (Scotland) Act 1995. These Acts draw on Article 12 of the United Nations Convention on the Rights of the Child (United Nations 1989) laid down that:

States Parties shall assure to a child who is capable of forming his or own views the right to express those views freely in all matters affecting the child, the views of the child being given due weight in accordance with the age and maturity of the child.

It is not only what children say, but their behaviour, demeanour and general appearance also need to be taken into account when listening to them (Adcock 2001). Observation is therefore an important tool for gathering information and this is particularly true for younger children (Alderson 2008). However, Archard and Skivenes (2009), based on findings from a small research study show how there can be a gap between the ideal and reality in the actual practice of social workers. Although the social workers

demonstrated that they use child-friendly approaches when gathering the views of children, there was little evidence of workers seeing the views making a 'real difference to what happens' (p.397). In Alderson's view (1992:175) 'treating children with respect can markedly increase their competence'. Whilst studies (Butler and Williamson 1994; Hallett et al. 2003) have found that confidentiality and trust are of the utmost importance for young people when deciding whether or not to tell someone about a problem, informal networks, particularly friends, could usually be relied upon to provide confidentiality more than formal agencies like social work.

A paternalistic perspective (Fox-Harding 1991) evolves where adults speak for the children, giving meaning to their experiences and making decisions regarding their future without necessarily consulting them (Smith et al. 2003). Holland (2010:115) claims that studies suggest that whilst adults tend to focus on the 'bigger picture', for children the 'everyday' is more likely to be their priority. Workers may have limited training (Luckock et al. 2007) and experience of talking to children and be anxious about doing so at such an emotional time (Fearnley 2010). When parents and professionals collaborate together regarding the well-being of children, the children may be seen as vulnerable and unable to put forward their own interests (Wattam and Parton 1999); it being easier and less discomforting to concentrate on the needs of adults rather than children (Butler et al. 2002:98). A review of Serious Case Reviews 2001-03 (Rose and Barnes 2008) confirms this view.

I have already described research findings in the previous chapter about children wishing to be informed about the state of their parents' health. Children in a study into children's involvement in their parents' divorce (Butler et al. 2002) echoed this wish and further stated that they wanted to be consulted on the key decisions that were to be made regarding their future. This involvement would be particularly applicable to

children being cared for by a lone parent who is terminally ill when decisions are being made about their future care arrangements.

There is, therefore, useful literature regarding listening to children and hearing their voice that is relevant for palliative care social workers. However, there was no research found which explores palliative care social workers' views about, or skills in, listening to children and this gap is addressed in this thesis.

## Conclusion

The emphasis in England, Wales and Scotland is that all children should have the best life chances in order to achieve their potential. The literature explored in this chapter has highlighted the broad range of issues that are involved in safeguarding and promoting the welfare of children that may be of particular relevance to the role of the palliative care social worker in adult services. However, there was minimal literature found that related specifically to them.

The literature does not imply that workers in adult services should act as 'child protection' workers. Rather they should be able to recognise indications when a child's needs are not being met and also recognise aspects of parental behaviour that can reasonably be expected to have an impact on children. They then should consider how these needs can best be met. Whether palliative care social workers consider the welfare of children to be part of their role, and if so in what way and how they go about it, is the focus of this study. The next chapter describes how the study was undertaken.

## The research journey

The excitement resides not so much in reaching the destination, for we can never completely enter the world of the other, but in the voyage and what might be found on the way. Darlington and Scott 2002

## Introduction

In palliative care it is not uncommon for the experiences of a service user to be referred to as a journey. The metaphor is widely used in literature and on occasion by service users themselves (Kristjanson and Ashcroft 1994; Beresford et al. 2007). It therefore seems appropriate to also consider this piece of research into an aspect of palliative care as a journey. It certainly felt like a rather long journey with its ups and downs, twists and turns, stops and starts. In this chapter I will consider the various stages from the selection of methodology, the research design, research methods used, the ethical issues and, the conduct of the analysis. Finally, I will consider my 'self' in the research journey. However, as 'a journey begins before the travellers depart' (Charmaz 2006:1), I will briefly locate this research project in the context of social work research and describe the theoretical perspective taken.

## Social work research

The primary purpose of social work research is generally considered to 'promote the development and improvement of social work practice' (Powell 2002: 21). Social work research has endeavoured to become more relevant to practice, mainly through the inclusion of, and collaborative inquiry with, service users (Powell 2002; Beresford et al. 2007). Butler's (2002:24) view is that social work research is about 'social workers, what they think, what they believe, what knowledge they claim and what they do with it'. What makes it distinctive, he believes, is 'its relationship with the lived experiences and practical concerns of those engaged in social work' (Butler 2006:8). This study sets out

to do just that: to find out what are the experiences of a particular group of social workers in their work with children and their families in a particular set of circumstances with the aim of improving services.

## **Theoretical perspective**

It seemed appropriate to use a systems approach to explore the role of the palliative care social worker in safeguarding children for three reasons. First, palliative care social workers operate within a team, an organisation and wider societal systems. The systems approach allows for the study of horizontal interactions between agencies as well as vertical interactions within them (Fish et al. 2009). Second, in exploring the workers' role with families, systemic thinking can help in the consideration of the ill person in all their significant relationships and networks and in the wider context of society (Oliviere et al. 1998:55). Third, an ecological perspective can assist when considering the needs of the child (Rose 2010:40). Whilst the systems approach is not without its critics (Houston 2002; Payne 2005), I could see strengths in using it, not as an underpinning theory, but as a broad framework to help me explore the various 'layers' involved and their impact on one another.

Systems theory has had a major impact on social work since the 1970s (Payne 1991:134). General systems theory is a biological theory proposed by Bertalanffy (1971), and also used in engineering, in which all organisms are systems, composed of sub-systems and in turn part of supra-systems. Bodies and engines are seen as either open systems (influenced by factors outside themselves) or closed systems (self-contained and impervious). Pincus and Minahan (1973) have been the most influential interpreters of the application of systems theory to social work practice based on an understanding that social systems are open systems. The significance of open systems is that all parts of the system are connected and what happens in one part will have an

effect on others. The system requires to be kept in a steady state, adjusting itself to try to maintain that state and there is a feedback loop which provides the capacity for change (Coulshed and Orme 2006:55).

People, as Pincus and Minahan (1973:4) argue, depend on three kinds of systems: informal or natural resource systems such as family, friends, work colleagues; formal systems such as community groups; and societal systems such as hospitals and schools. The focus of the work does not have to be on bringing about change in the individual; other parts of the system may be the target for change. The approach allows workers to be innovative in the way they assess situations. Pincus and Minahan (1973:63) identify four systems within social work:

- Change agent system: social workers and people in the agency in which they work.
- *Client system:* people who seek help from the change agent and are the expected beneficiaries of the service. *Actual clients* are those who have agreed to receive help; *potential clients* are those whom the worker is trying to engage.
- *Target system:* people with whom the change agent needs to change or influence in order to achieve the goals.
- Action system: people with whom the change agent works to achieve the goals.

The phases of planned change require identified practice skills<sup>12</sup> and different social work methods may be used at various points, for example counselling or task centred work. They also consider it important to analyse the type of relationships the worker forms with the systems encountered and identify three types: collaborative, bargaining and conflictual.

<sup>&</sup>lt;sup>12</sup> Eight practice skills identified: assessing problems; collecting data; making initial contacts; negotiating contracts; forming action systems; maintaining and co-ordinating action systems; exercising influence; terminating the change effort (Pincus and Minahan 1973).

Hudson and Payne (2009:Preface) describe dying as 'a family affair, its impact ripples out to affect close friends, neighbours and the wider world' and palliative care is generally seen to be concerned, not only with the patient, but also with the patient's family (World Health Organisation 2002). The family is considered to be who the patient says it is and therefore can include not only blood relationships, but others including friendship, care and duty (Kissane and Bloch 2002). A systems approach to working with families has been recognised as a helpful way to see the family as a 'set of interacting parts with a particular purpose' (Coulshed and Orme 2006). Families have sub-systems: marital, children, sibling, mother-daughter, father-son and so on (Barker 1981). However, families today are becoming more complex due to divorce and remarriage and so the family system is redefined, creating new sub-systems. There are also supra-systems to which families may belong including the extended family, neighbourhood, and societal systems such as health care. Families are open systems and are constantly changing in relation to internal and external forces. Each system and sub-system is defined and understood by its boundaries (Barker 1981) and hierarchies (Mehta et al. 2009). The boundary defines the system and highlights the type and amount of communication that occurs between them. Mehta et al. (2009:237) argue that it is an understanding of the family's boundaries and the degree to which they are 'permeable' that allows professionals to 'make an impact on the family unit'. The reasons behind the boundaries' existence are also important: culture may be an influence as well as previous negative experiences with health and social work professionals. It is also advocated that the family must be understood as a whole, rather than examining individual members or sub-systems in isolation from each other, or from the supra-systems of which the family is a part (Wright and Leahey 2005).

A family systems approach is promoted by several authors in the palliative care arena (Knapp and DelCampo 1995; King and Quill 2006; Mehta et al. 2009; Munroe and Oliviere 2009). A significant change such as a diagnosis of a terminal illness means that

a family has to reorganise itself and individual members may not function as they did previously. Rolland (2005:2585) has devised a conceptual framework (Family Systems-Illness Model) for evaluation, formulation and intervention with families coping with chronic illness or disability based on a family systems framework. This model is based on a strength-orientated perspective which views family and relationships as a potential resource and emphasises the possibilities for resilience and growth. Rolland believes it is important to consider the type and stage of the illness together with the individual and family life cycles as well as the family's belief systems. He points out that when a parent develops cancer during the child-bearing phase of life, a family's ability to 'stay on course is severely taxed' (p.2591). As well as helping with social and relationship problems, palliative care social workers endeavour to develop the capacity of families to manage future adversities (Payne 2007). They enhance the resilience of patients and their families by not focusing on them as individuals, but as members of a wider system using their knowledge to mobilise resources and support (Monroe and Oliviere 2007).

The ecological model applied to child development was first outlined by Bronfenbrenner (1979) where the development of children is seen to be influenced by the factors and interacting processes at individual (micro), family (meso), community (exo) and societal levels (macro). He focuses on the 'balance of stresses and supports in the family environment and the interactions between them' (Jack 2000:704). The development of children and their life chances is seen to be influenced by the factors and interacting processes at individual, family, community and societal levels. For example, poor health and disability are significant factors that can contribute to a family experiencing poverty, which in turns impacts negatively on family functioning. As described in chapter four, this model is the basis of the Assessment Framework and is also seen to be 'consistent with a strength's perspective that looks to the whole community as a resource' (Ward and Rose 2002:215). The child's attachment network also needs to be considered in an

ecological approach; this includes the nature of a child's attachments to adults and to peers and siblings (Daniel et al. 2010:38).

Reflecting on his work, Bronfenbrenner (1992) drew attention to an earlier lack of a time dimension. The 'chronosystem' model focuses particularly on developmental stages triggered by life events or experiences. These may have their origins either in external environment, such as going to school, or within the organism, such as puberty and severe illness. Whatever their origin, the critical feature of such events, he argues, is that the existing relationship between the person and environment is altered, thus creating a dynamic that may prompt developmental change. Therefore, this model is helpful for the understanding of situations where children are living with a dying parent. These children may face two or more experiences at the one time, for example going through puberty, changing schools, as well as coping with the terminal illness of a parent. It is this accumulation of adversities that is the threat to healthy child development (Garmezy 1994). However, many children and families display degrees of resilience despite the overwhelming situation, the components of which, Jack (2000:708) argues, should feature in any ecological assessment made.

## Choosing the way to go

This study adopts a qualitative research methodology. There are various ways of defining qualitative research (Silverman 2000; Bryman 2001) and no real consensus has been reached. Denzin and Lincoln (2000:3) describe it as involving:

an interpretive, naturalistic approach to the world. This means that qualitative researchers study things in their natural settings, attempting to make sense of, or interpret, phenomena in terms of the meanings people bring to them.

Taking these and similar understandings (Whittaker 2009) of the primary focus of qualitative research, I considered it to be the most beneficial methodology to adopt. In

order for me to take the 'interpretive, naturalistic' approach, it was important to select data collection methods that would allow me to 'hear' what the palliative care social workers had to say about safeguarding children from their own experiences. The gathering of specific practice examples, I considered, would help me to understand their interpretation of this aspect of their role as well as some of the difficulties that they may encounter in carrying it out. Questions would require to be framed to allow this to happen. Methods needed to include interactive opportunities so that workers would be able to explain their approaches and I could ask follow-up questions as required. During the subsequent analysis of findings, I would need to ensure that themes emerged from what the workers had said rather than starting with a preset range of issues.

Taking an inductive approach and looking at specific instances of palliative care social work practice would help me to understand how the findings might be applied to the palliative care social work community as a whole. The use of small, purposively selected samples and interactive data collection methods would, I hoped, provide close contact between the participants and me as the researcher (Snape and Spencer 2003), as well as generating rich data providing an in-depth understanding of the 'world' of the palliative care social worker. As was shown in the literature review, little is currently known about palliative care social workers' involvement in safeguarding children and an advantage of qualitative research is that it can be useful in 'opening up' a new area in order to identify notable issues (Fitzpatrick and Boulton 1994). I did, however, consider two other approaches: case study research and the Delphi technique.

A case study is an empirical inquiry that investigates a phenomenon within its real-life context (Yin 2003:13). This approach is appropriate when context is particularly important and where multiple perspectives need to be taken into account (Walshe et al. 2004). It is an intensive investigation of a single 'unit' that may be a person or a setting (Gilgun 1994; Bowling 2009). Whilst there were many aspects of the case study

approach which could be applicable in this study, case studies require multiple sources of information, and as revisiting cases is a desired feature of the approach, the geographical, time and resource constraints made this difficult to achieve. Also, as this was a new area of study, it was my intention to gain a breadth of views from many 'units'.

The Delphi technique seeks to obtain consensus on the opinions of 'experts' or 'participants' through the administration of a series of structured questionnaires which is continued until a group consensus is reached (Hasson et al. 2000; Powell 2003; Keeney et al. 2006; Bowling 2009). It combines qualitative and quantitative methods (Stewart 2001). I was attracted to this technique after reading how it was applied to an aspect of multidisciplinary child protection research (Gabb et al. 2006). However, after a closer examination of the technique and potential results, I decided that the in-depth and rich data I was seeking could only be provided though a variety of other approaches.

## Starting out

In order to discover some of the possible issues that might arise, frame specific research questions, and challenge some of my assumptions, I conducted a small exploratory study (Mason 2002; Sarantakos 2005) after undertaking an initial literature review (Hart 1997). I was also interested in whether the proposed research would be of value and interest to palliative care social workers (Lewis 2003).

This initial exploratory study involved a single interview and a paired interview with palliative care social workers. I had previous contact with all three workers, two through APCSW as I had become an associate member, and one through previous research. Thus, they were 'hand-picked' (O'Leary 2004). One worker was employed by a local authority but working within a NHS setting; one of the pair was employed by a combined

in-patient and day hospice for adults, and the other by a children's hospice. The pair of workers already knew each other, which allowed on this occasion for a stimulating and safe environment (Lewis 2003). Both interviews produced interesting and helpful information that highlighted some of the dilemmas they faced when working with families. The palliative care social worker from the children's hospice also identified aspects of concerns about children's welfare in her engagement with families. Whilst this was an important area, incorporating it into this study would have made it lose focus and become unwieldy. However, I believe it is a topic that should be explored. The fact that there was interest in the proposed study and a range of identified issues led me to feel confident about taking the research forward.

## Moving forward

Having refined the research questions as a result of the initial literature review and exploratory study, the next stage was to finalise the research strategy itself, bearing in mind that a good design should be sensitive, flexible and adaptive as well as manageable both in terms of time and resources (Flick, 2007a). In this section, the research design is explained, with information on the methods of data collection and analysis selected, the reasons why they were chosen, the sampling used and the ethical issues that shaped the process.

In order to answer the questions posed by this study, it was important to understand the complexity of the factors and processes that influenced the actions of the palliative care social workers in safeguarding children. I was seeking to deepen understanding about the different experiences of palliative care social workers rather than establish fact. I was also aware that my research would be one-dimensional. This research study was an initial investigation which, depending on the outcomes, could be further investigated

in different ways by seeking the views of service users and other professionals with whom palliative care social workers are involved.

I was mindful of Charmaz's (2006:15) opinion that 'methods wield no magic'; rather, what is important is how a researcher uses the methods chosen. The challenge for me was to understand the reality from the 'inside' (Alston and Bowles 2003). At the outset, I considered that a mix of qualitative approaches would provide both breadth and depth and the process would be iterative (Froggatt 2001). Using focus groups at the initial stage would help confirm some of the issues identified during the exploratory study and also to help identify new ones (Lewis 2003). The findings would inform the survey stage: the administration of a questionnaire using mainly open-ended questions would gather a breadth of opinion. The final stage: a series of in-depth, semi structured interviews to obtain more rich data, would be informed by the findings of the previous two. Thus I embarked on a cyclical process of data collection, analysis, further data collection and re-analysis. Using these three methods also provided a 'within methods' triangulation (Flick 2007b).

## Stage one: focus groups

The strength of focus groups is seen to lie in their ability 'to explore topics and generate hypotheses' via group interaction (Morgan 1988:21) and they are increasingly used in social work research (Linhorst 2002). Indeed, Kitzinger (2005) refers to this group interaction as a 'defining feature'. Morgan (1997) sees focus groups being used in three ways: as a 'self contained' method in studies; as a supplementary source of data; and in multi-method studies. In this research study, focus groups with palliative care social workers were used as both a supplementary source to inform the development of a questionnaire and to aid in the planning of semi-structured individual interviews, as well as in their own right as part of a multi-method approach.

The specific purpose of the focus groups was to seek views of palliative care social workers on what, if any, role they considered themselves to have in relation to safeguarding children and to analyse the way they spoke about the issues. In setting up the focus groups, I was influenced by Krueger and Casey's (2000:5) definition of a focus group: 'a carefully planned series of discussions designed to obtain perceptions on a defined area of interest in a permissive, non-threatening environment'. I was mindful too of the quality factors identified by Krueger (1993)<sup>13</sup> in all aspects of the process, as well as recommendations made by authors who have undertaken health-related research (Sim 1998; Kitzinger 2005). Views appear to vary on optimum focus group sizes (Morgan 1997; Krueger and Casey 2000). The reasons given are that it might be difficult to stimulate discussion with under six people and to control with over ten (Bloor et al. 2001). Four to eight is usually considered ideal (Kitzinger 2005). I aimed for six members.

## Recruitment and sampling of participants

Recruitment of participants and the focus group fieldwork took place over a three month period from July to September 2007. An initial contact was made with the Chairperson of the APCSW to inform her of the research study and to ask permission to contact the membership (271 members at January 2008). This was granted and interest was expressed in the study.

The first focus group was held in a hospice in England and the second and third groups during the APCSW Annual Conference. For the first focus group, the recruitment was facilitated by an intermediary who was a palliative care social worker based in England. I had contacted this worker to arrange a meeting to discuss her

<sup>&</sup>lt;sup>13</sup> Krueger's ten quality factors in focus group research include: clarity of purpose; appropriate environment; sufficient resources; appropriate participants; skilful moderator; effective questions; careful data handling; systematic and verifiable analysis; appropriate presentation; honouring the participant, client and method (Krueger 1993:67).

extensive experience of work with dying parents and their children. It was her suggestion that she invite other workers in her geographical area to form a focus group; an offer that I accepted. This approach, Patton (2002) would identify as 'opportunistic'. I sent her information leaflets (Appendix 1) to be forwarded to potential members of the group. The intermediary kindly made all the arrangements and 12 members were expected and attended the group. Although this was double my target number, I was pleased that so many workers were interested and did not wish to miss an opportunity of gathering views from as many people as possible. Bloor et al. (2001) advocate the use of pre-existing groups and an intermediary as a way of reducing the recruitment effort, although they recognise that this may lead to a loss of control. In the event, the latter did not prove to be the case.

Recruitment for the second and third focus groups was different. I had previously contacted the Conference Organiser to ask permission to hold focus groups and this was readily given. Two time slots on the second day were identified in the programme. On the first day, I gave a short presentation about the research, explained the purpose of the focus groups and made a general invitation for people to attend. I was aware that an issue of 'volunteer' sampling is that only those most interested are likely to attend (O'Leary 2004) and I was competing with 'free time'. Ten people attended in total, five in each group. 'Piggy-back' focus groups, such as these, may erode quality as they might be influenced by the climate of the conference and responses might be shaped by its stimulus (Krueger 1993). However, there is also a view that they can be productive with meetings of professional associations (Krueger and Casey 2000). When comparing the content of these two focus groups against the first, it did not appear that the quality was eroded or shaped by the Conference.

#### Profile of research participants

A total of 22 palliative care social workers participated in the three focus groups. All were qualified social workers, although their work titles varied. In the first group, all 12 people worked in England and were female. The second group of five people comprised two male and three female members, all working in England and the third group, also of five people comprising one male and four females, included people based in England, Wales and Scotland. The high female membership of the groups reflected the overall membership of the Association: at March 2010, 94% of the membership was female.

There are advantages and disadvantages in acquaintances being members of a focus group (Morgan 1997). In the first group, most members already knew each other. Acquaintances are generally seen to talk more easily to each other but can rely on 'taken for granted assumptions' (Morgan 1997:37) which the researcher is hoping to investigate. Having strangers in a group avoids a 'polluting' and an inhibiting effect (Kitzinger 2005), but Morgan (1997) believes that the important issue is to recognise that each mix produces different group dynamics. All that is required according to Hudson (2003:203) 'is a group of people with sufficient interest in the issue or question to contribute to the discussion'. This proved to be the case.

## Conduct of focus groups

Groups usually last between one and two hours (Brown 1999; Bowling 2009) and Morgan (1997) suggests the optimum length of a focus group discussion is 90 minutes. The first focus group lasted 1 hour and 40 minutes; the other two groups for an hour each due to size and time restrictions. Members in all groups, however, did not appear to be flagging nor need any encouragement to contribute.

A discussion guide (Appendix 2) was developed as a result of the exploratory study and the literature review (Stewart and Shamdasani 1990; Morgan 1997) and I endeavoured to bear in mind that 'quality focus groups depend on quality questions' (Krueger 1993:76). At the first group, I did not use all the topics on the schedule due to lack of time. The questions I did use were not always in the prepared format as the discussion was free flowing, providing information on the areas that the questions were designed to cover. After this first group, I reviewed the topic areas and decided not to make any changes, considering that they were broad enough to allow for the issues to be covered in a naturalistic way. If necessary, more probing questions (Litosseliti 2003) could be used to refocus the discussion.

Despite the size of the first group, the group members were courteous to each other, rarely interrupting each other and not holding side conversations. Group interaction was relaxed and insightful. Throughout there was humour in the group and members supported each other by non-verbal cues and by verbal agreement. They spoke to each other rather than focusing on me as the moderator, indicating my low level of involvement (Morgan 1997). All members made contributions. The 'host' member, who was a line manager for two other members, contributed slightly more than the others but this did not seem to inhibit the general discussion. The other two groups were just under half the size of the first, the group atmosphere and process were similar, with all members making more or less equal contributions. In the second group two members were from the same hospice and one was the line manager of the other. However, nobody appeared to be intimidated by other members of the group and all seemed to be interested in what others had to say.

An assistant moderator, if present, would have been able to take notes during the discussion, with particular reference to the non-verbal content of the discussion (Krueger and Casey 2000). Resources, however, did not allow for this. All of the

groups were audio-recorded (with the permission of all the participants) and I made field notes as soon as possible after each event that included my thoughts and observations as well as a reflection on the atmosphere and interaction. I personally transcribed the recordings and the findings from my initial analysis helped me to move on to the next stage of the research, the development of the questionnaire.

#### Stage two: survey questionnaires

Brace (2004:5) usefully describes questionnaires as 'A medium of remote conversation between researcher and respondent'. As this was an emergent qualitative research study, a postal questionnaire was considered to be the best option to ascertain the views of as large a sample of palliative care social workers as possible. It was hoped that this would provide a range of practice examples from a variety of work settings in different parts of the UK (Darlington and Scott 2002).

Although questionnaires are a popular method in quantitative research, it is possible for them to be used as part of a qualitative study, if a semi-structured interview schedule is used (Bowling 2009). However, I recognised that there are particular difficulties in using postal questionnaires in an explorative study (Foddy 1993; Birchall and Hallett 1995; Graham 2000) and these needed to be taken into account when planning this part of the fieldwork. The first was obtaining a high enough response rate. People's commitment to completing a questionnaire is known to vary for a range of reasons (Bowling 2009). However, one of the features of qualitative research is the use of a relatively small sample and whatever number is received, the researcher cannot consider that respondents' responses are typical of the whole population and make generalisations. This led to the second difficulty: depth of views. One of the reasons for using a qualitative approach in the questionnaires was to obtain as much depth of information as possible from the respondents in order to understand any variations in terms of views and experiences. Although Birchall and Hallett (1995:10) consider that the 'subtleties of

people's experience, understanding and motivations are unlikely to be conveyed even in open-ended responses', I did hope that palliative care social workers would be sufficiently motivated and feel able to share anonymously as much information as possible. The third difficulty was designing a questionnaire that was clear and precise for every respondent at every point, as people can have different preconceptions as they interpret the questions (Foddy 1993). The fourth was structuring the whole questionnaire in terms of length versus adequate specificity, for, if it was too long, the likelihood was that people may choose not to complete it. These difficulties were taken into account when planning this second stage.

#### Development of survey

The purpose of the questionnaire was to gather information that would help clarify the views of palliative care social workers about their role where an adult is receiving specialist palliative care and there are concerns about the children in their care. In developing the questionnaire I took account of the various good practice guides, for example Denscombe (1998); Thomas (1999); Graham (2000). Consideration was given to the use of vignettes,<sup>14</sup> which are believed to engage participants and elicit more meaningful and considered answers than the typical questionnaire (Birchall and Hallett 1995). This might possibly be seen as less personally threatening, and allow sensitive topics to be discussed in a more neutral way (Lee 1993). Vignettes have been successfully used in social work research about child protection (Birchall and Hallett 1995; Horwath and Saunders 2005). However, as there is currently little literature on the work of palliative care social workers with children and families, I chose to seek examples of actual experiences, and therefore decided not to use vignettes in this study.

<sup>&</sup>lt;sup>14</sup> Vignettes are a simulation of real events depicting hypothetical situations (Wilks 2004:80) and come in various forms, from two lines devoid of context to several paragraphs of a story unfolding over time (Giovanni and Becerra 1979).

In keeping with the iterative nature of the research, I devised a draft questionnaire based on the findings from the three focus groups. For example, issues of supervision and training had been raised in the groups and I wished to gain a clearer picture of their availability. Also, examples had been given in the groups about how children first came to the attention of palliative care social workers, and I sought to examine this in more detail. I conducted a pre-test of the questionnaire with a local palliative care social worker. This included a 'think-aloud' testing (Bowling 2009:301). A revised draft was then read by that worker, plus my two supervisors and a past PhD supervisor, who were all non-palliative care social work specialists. They specifically checked that there was no apparent confusion, assumptions or ambiguities (Thomas 1999). In the light of feedback the questionnaire was further revised. It was then piloted by a small number of people who represented the target group of palliative care social workers (Thomas 1999; Graham 2000; Brace 2004). Four members of the APCSW, whom I had met at different events and who were aware of my research, agreed to take part. Three completed the questionnaire and returned the material as requested. The fourth chose to give her responses and views by phone.

The final anonymous questionnaire (Appendix 3) consisted of 25 questions, nearly all being open-ended questions with two Likert scale questions using a visual numerical scale. The questionnaire was divided into five sections. The questions were designed to elicit information about the respondents' perceived role in safeguarding children including brief examples from practice. On reflection perhaps the questionnaire was too intimidating to complete in terms of the amount of time and effort. On the other hand, those who did respond supplied detailed, thoughtful information about their practice. Once all the questionnaires were received, I commenced the data analysis. The information gathered from the focus groups and survey helped me to formulate the interview schedule for the next stage and focus on areas that were least well explored previously. Areas included aspects of the palliative care social worker's place in the

multidisciplinary team, relationships with the local authority children and family social workers, and the influence of the parents' illness on decisions regarding their care of their children.

#### Recruitment and sampling of participants

Recruitment of participants, preparation and fieldwork took place over a four month period from November 2007 to March 2008. I had already gained permission from the APCSW to administer a questionnaire to the membership. This was a purposive sample (Mason 2002:124). A total of 271 questionnaires, together with a covering letter, participant information sheet (Appendix 4), and pre-paid return envelope, were sent out in January 2008, using pre-printed labels provided by the Association.

#### Profile of research participants

A total of 57 palliative care social workers returned 56 questionnaires<sup>15</sup>. Seven returned the questionnaire indicating that they did not work in palliative care for adults. Although there is no agreed standard of an acceptable response rate it appears that a response rate of 50% from a sample not known to the researcher is generally considered to be satisfactory (Graham 2000). The response rate of 21%, therefore, is a much lower rate than the usual expectations of postal questionnaires. However, similar qualitative surveys of health and social service professionals have produced a comparable response rate (Darlington and Feeney 2008).

Fifty respondents were female and seven were male, reflecting the composition of the total membership. Fifty were based in England, two in Wales and five in Scotland. As shown in Table 5.1 there was a range of work locations with the most common being combined day/in-patient hospices. There was a spread of palliative care social work experience with four respondents having 20 plus years experience as Table 5.2 shows.

Type of work-base	Number of respondents
In-patient hospice	2
Day hospice	1
Combined day/in-patient hospice	31
Community service	5
Hospital	7
Hospital/community service	2
In-patient hospice/community service	3
In-patient/day hospice/community service	2
Other	3

#### Table 5.1 Profile of survey respondents by type of work-base

# Table 5.2 Profile of survey respondents by number of years in palliative care, in currentpost and since social work qualification

Number of years	Number of respondents in palliative care	Number of respondents in current post	Number of respondents since qualification
0 - 4	22	33	3
5 - 9	18	15	9
10 - 14	8	6	5
15 - 19	5	1	13
20 – 29	4	2	11
0ver 30	0	0	14

#### Stage three: semi-structured interviews

Interviews are the most favoured method in qualitative research (Bryman 2004; Sarantakos 2005). A semi-structured, semi-standardised interview was the chosen method in this study because this allowed some flexibility over the direction of the conversation during the interview as well as the opportunity of gathering rich data (Mason 2002; Patton 2002). It would help me to achieve 'a depth and roundness of understanding' (Mason 2002:65). A feature of in-depth interviews is their focus on the individual and their personal context, as well as being a vehicle for expressing feelings and understanding motivations and decisions (Lewis 2003). They would allow me to

<sup>&</sup>lt;sup>15</sup> Two workers at the same hospice returned a combined response on one questionnaire.

explore the meanings workers attach to their role and their perceptions of the processes involved in, for example, their relationship with the multidisciplinary team and other professionals (Mason 2002). Such interviews enable people to talk about past events and ensure that the information is not idealised or general but a description of what was actually done (Arthur and Nazroo 2003). In addition, they are a way of helping interviewees to think, consider and make explicit matters under discussion (Robson 2002). As I was particularly interested in hearing about examples from practice, this method seemed to be particularly relevant and would build on knowledge already gained. Using this combination of opening up areas and exploring details has been referred to as 'content mapping' and 'content mining' (Legard et al. 2003:148).

In order to hear about practice examples, it seemed appropriate to incorporate a narrative approach within the interviews (Kvale 2007). This is an event centred approach which focuses on 'the spoken recounting of particular past events that happened to the narrator' (Andrews et al. 2008:5). Hall (1997:177) points out that social work accounts make use of narrative methods of plot, character and point of view. Workers are able to demonstrate that their work is in accord with responsible and justifiable professional activity. Narratives are also a common genre from which to retell or come to terms with a sensitive time or event (Coffey and Atkinson 1996:56) such as work with families where a parent is dying. They are subject to change and re-interpretation over time (Riessman 1993:64). A central element of the approach is to ask the participant to present narratives of situations which can be complemented by questions that ask for the person's 'subjective definitions' (Flick 2007b:57). This combination of question and answer sequences, together with narratives of situations that the interviewe has experienced, Flick describes as an 'episodic interview' (Flick 2007b:58).

#### Recruitment and sampling of participants

Recruitment of participants, preparation and fieldwork for the interview stage took place over an 11 month period from January to November 2008. Palliative care social workers were invited to volunteer to take part in the interviews and information about the process was included in the letter accompanying the questionnaire sent to all members. A prepaid reply post card was also included which allowed workers to express an interest in taking part in the interviews and could be returned separately from the questionnaire, thus ensuring that anonymity would not be compromised. Thirty-two post cards were returned. A participant information sheet was sent out with an accompanying letter (Appendix 5) and a form asking for information about their place of work. Twenty forms were received (one later withdrew). I was concerned that some potential interviewees, especially those who were relatively inexperienced, might have perceived an interview with me as being similar to a supervision session and thus were reluctant to take part.

The overall selection was based on geographical location and mix of work settings in an endeavour to ensure that the sample population was as diverse as possible. Plans were made for the first rounds of interviews based on geographical proximity and a total of 13 interviews were held between June and September 2008. My initial thought was that I would interview about 12 workers. However, after consideration of the material gathered and discussion with my supervisors, a further round of four interviews was planned to secure saturation and a further spread of geographical locations. The selection of the sample in the second round was pragmatic, based on geographical, time and financial constraints. This final round of interviews was conducted in November 2008.

In return for their assistance, time and thought the participants were informed that they would receive a copy of an executive summary of the thesis when the research was completed.

#### Profile of research participants

A total of 17 palliative care social workers participated in this stage of the research. Two participants were male and fifteen were female<sup>16</sup>. All three countries were represented; the majority being based in England. They were located in a range of work settings (see Table 5.3). Twelve served in urban communities, two in rural and three in mixed rural and urban communities. Again there was a spread of palliative care experience as Table 5.4 shows.

Type of work-base	Number of interviewees	
Combined day/in-patient hospice	6	
In-patient hospice/community service	2	
In-patient/day/community service	4	
Hospital	2	
Community service	3	

#### Table 5.3 Profile of interviewees by type of work-base

## Table 5.4 Profile of interviewees by number of years in palliative care, in current postand since social work qualification

Number of years	Number of interviewees in palliative care	Number of interviewees in current post	Number of interviewees since qualification
0 - 4 years	6	7	2
5 - 9 years	7	8	2
10 - 14 years	2	0	3
15 - 19 years	1	1	3
over 20 years	1	1	7

#### Conduct of interviews

The interviews lasted between 54 and 120 minutes and the average length was 96 minutes. The length of the interview appeared to depend on the length of the narration of the practice examples that the participants chose to share. There were those participants who appeared to have limited experiences to talk about and others who spoke at length, with very little prompting about their chosen practice examples.

All the interviews were held at the participant's work base (during work hours for all but one) and conducted in rooms that were private. The aim of the interview was to ensure that the overall research question would be answered, whilst encouraging a flow of conversation and allowing the participant to talk in a relaxed and open manner (Whittaker 2009). A fairly flexible approach was adopted regarding the phrasing of the questions but they were asked roughly in the same sequence to facilitate the organisation and analysis of data (Patton 1990). At the heart of the interview was the recounting of the examples from practice. Questions about feelings and attitudes then followed (Arthur and Nazroo 2003). But before asking participants to disclose examples from their practice, it was first necessary for me to gain their trust (Holloway and Freshwater 2007). This was taken account of in the planning of the interview guide (Appendix 6) by asking for more general information at the beginning. The topics were similar to those covered in the questionnaire, but the vehicle of the interview allowed for a more in-depth discussion. All the participants were willing to answer all the questions.

Establishing rapport and developing trust was therefore an important feature of the interview. However, I recognised that too strong a connection between a researcher and participant could cause difficulties (Darlington and Scott 2002). The participant may take it for granted that the interviewer understands what they are speaking about and, as a result, miss aspects from their narrative. There was an additional risk of my making assumptions due to my previous research and not checking them out. In addition, I had to maintain my role as a researcher during the interview and be aware of when I was slipping into a previous social work tutor role (Alston and Bowles 2003). On the occasions when this did occur, any advice sought was given after the interview was completed. I saw the interview as a collaboration (Oakley 1981; Legard et al. 2003). I

<sup>&</sup>lt;sup>16</sup> The two male participants were the only two who agreed to be interviewed.

participants' account. Some interviews evoked sensitive issues and I acknowledged any distress displayed by participants.

Whilst there does not appear to be any research that shows the 'Hawthorne' effect, i.e. participants benefiting simply by taking part in the research project (Allmark 2002; Payne and Payne 2004), feedback gathered from various research studies indicate that participants frequently acknowledge how positive they found the experience to be (Beresford et al. 2007; Kennedy 2008; Brooks 2009). This study was no exception. Whilst I did not ask for any specific feedback after the interviews (or the focus groups), there were participants who voluntarily commented that the interview had been enjoyable and mentally stimulating.

All of the interviews were audio-recorded (with the permission of the participants) and I made field notes shortly after each interview which described the physical context of the interview, non-verbal responses and my thoughts about the interaction together with my immediate reactions to the content. I transcribed the recordings as soon as possible after each interview, giving me an early opportunity to reflect both on the process and content, with particular reference to my interactions with the participants and their perspectives.

## Ethical issues

As I commenced this study at the University of Dundee, a research protocol was submitted to the University Research Ethics Committee (UREC) of the School of Education, Social Work and Community Education in June 2007. This included the design of the study, recruitment strategy, methods to be used and data protection measures. Overall approval and full approval for stage one (focus groups) was granted by UREC in August 2007. The University of Stirling Department of Applied Social Science (DASS) Ethics Committee granted approval for stage two (survey) and

recruitment for stage three (semi-structured interviews) in December 2007. Approval for the outstanding aspects of stage three was granted in February 2008.

Care was taken to ensure that informed consent was obtained from all the participants. For those taking part in the focus groups and interviews, this was done both in writing (Appendices 7 and 8) and verbally before the group or interview commenced. Completion of the questionnaire was taken as consent being given. Prior to giving their consent, all participants had received a participant information sheet outlining the nature and purpose of the study and the interview or focus group. Participants were also given the opportunity to request further information but no one did so. All were informed that they could withdraw from the study at any time without giving a reason for doing so. None did so.

The researcher as a 'whistle-blower' is a particular issue when conducting social work research in the field of child protection (Darlington and Scott 2002). Confidentiality is overridden when researchers have a legal or moral duty to report incidents of harm (Social Policy Association 2009:4). Interview participants were informed that, if during the course of the interview, it appeared that a child under discussion might be at risk and the situation seemed to require intervention, then the researcher and participant would discuss what action should be taken. If any participant, as a result of the interview, wished to make a complaint about any aspect of the interview stage, the contact details of a named, senior person at the University were also given in the information sheet.

Confidentiality and anonymity were addressed at all stages of the research and focus group participants were reminded to keep the discussions confidential (Litosseliti 2003). There was no identifying data requested in the questionnaire. I did however struggle, as have other researchers (Macpherson 2010), with how best to anonymise the rich

information provided about the children, their circumstances and the work of the palliative care social workers. The stories, often poignant, I considered should be included in some way to highlight the diversity of the work and it took several painstaking attempts before I achieved a satisfactory outcome. Details have therefore been removed or altered in order to preserve confidentiality, but not I hope at the cost of losing important information about process. I also decided not to provide social details of the sources of the case and practice examples in order to protect the identity of this small community of palliative care social workers.

Data collected has been stored securely during the study in compliance with the Data Protection Act (1998). Upon completion of the study, data will be destroyed or kept in secure DASS archive facilities for ten years.

## Nearing the end, but still a long way to go

#### Data analysis

In many respects, analysing your data has similarities with social work practice. In both roles, you are faced with rich and complex accounts and your task is to make sense of them. This involves seeing patterns and meaning in the accounts and it is inevitable that this involves you exercising your judgement (Whittaker 2009:89).

Within and after each stage of the data collection, a preliminary analysis of the data of one stage helped to inform the next, and progress within the current one. The discussions in the focus groups, for example, were wide-ranging and the interesting looking themes from the focus groups were tested in the survey with specific questions. If the answers seemed to require further clarification, then they were followed up in more detail during the interviews. The findings also provided useful background information on different aspects of palliative care and palliative care social work, giving me a firm platform on which to base my approaches to the potential interviewees. There

were many areas that I had initially considered exploring during the semi-structured interviews. However, the findings from the preliminary analyses helped me to focus on those that could most productively be achieved within an interview and its restricted time-frame: examples from practice and a clarification of particular issues to help me achieve a better understanding of the workers' role. Illustrations of this process are provided in Appendix 9. The interview process was also a developmental one; later interviews explored more fully those aspects that seemed to be germane to themes produced by the preliminary analysis of early interview data.

At the end of each stage, I immersed myself in the data and began a more detailed analysis. This was conducted manually as the numbers were manageable. Also, I had no experience of computer packages and did not wish to become anxious about learning a new process to the detriment of the analysis.

For the focus groups, I used the 'The Long Table Approach' described by Krueger and Casey (2000:132) in my search for themes and sub-themes within and across the questions. The core elements of the approach are cutting, sorting and analysing through comparing and contrasting the data (p.137). For the questionnaires, again a thematic approach was taken for the responses to the open-ended questions, taking one question at a time across all the questionnaires (Graham 2000). When categories for each question had been identified they were then compared across questions leading to a greater consistency (Fielding 2008).

The analysis of the interviews presented me with some particular challenges. During the interviews, and particularly during the transcription and subsequent readings, I was aware of the richness of the information contained in the practice examples: the stories. I decided to use a non-cross-sectional data organisation (Spencer et al. 2003:203) of this narrative part of the interviews as they highlighted some of the difficulties faced by

children and their families, as well as the dilemmas for the palliative care social workers. I was anxious that the stories about the children were not lost. However, I did recognise that the perspectives of the children and family members were second hand and may be seen as paternalistic in nature (Fox-Harding 1991). Their circumstances had been described and interpreted by palliative care social workers, based on what they remembered, and then further interpreted by myself. It took several attempts before I was able to produce thematic summaries about the stories, as there were many ways I could categorise this material. Eventually, I decided that as children were a focus of the whole study, I would consider the material first from their perspective, then from that of the parents' before moving on to the perspective of the palliative care social workers. Re-reading the practice examples from these perspectives provided a more coherent approach from which to progress. I compared these examples by the use of tables (Gibbs 2007), which seemed particularly suited to the narrative part of the interviews as it seemed to 'ground perspectives in the situations of clients and social work practice' (Woodcock 2003:93). In many ways my approach had the characteristics of the 'immersion/crystallisation method' (Borkan 1999; Crabtree and Miller 1999). This is when the researcher, after a 'prolonged immersion into and experience of the text ... and concerned reflection' enables the researcher to emerge with 'an intuitive crystallisation of the data' (Crabtree and Miller 1999:23).

I took an idiographic approach to the remainder of the analysis of the interviews (Smith et al. 1999:220) by looking at individual cases in some depth. I began with three transcripts chosen at random. I read one a number of times in order to become 'intimate' with the account, noting in one margin aspects that appeared to me interesting and significant. Then I re-read the transcript this time noting in the other margin emerging themes using 'in vivo' coding (Buff 2005:154). Using participants' own words as labels kept me grounded in what the participants had said. In searching for themes, I was mindful of capturing something important about the data in relation to the research

question (Braun and Clarke 2006). Having listed the themes, I looked for connections between them. I repeated this exercise with the other two transcripts and consolidated a list of themes. This process was continued for all the transcripts. I then moved on to use the 'Framework' analysis method (Ritchie et al. 2003:220), which suited my way of thinking and working, to compare instances and individuals. This is a matrix-based approach which allows data to be synthesised and charted so that the researcher can read across data without losing sight of the individual participants. In addition, I referred to my field notes in order to take account of the non-verbal aspects of interview.

Once the data from all three stages of the data collection had been analysed, it was possible to make cross-comparisons, looking for similarities and differences in the findings. Writing memos (Robson 2002; Fielding 2008) helped me to develop my thinking further. I then discovered the voice-centred relational method of data analysis (Mauthner and Doucet 1998) and was excited by the possibilities it afforded (Deery 2005; Paliadelis 2005; Fairtlough 2007; Byrne et al. 2009). It is 'voice-centred' in that it explores the individuals' narrative account of experience and 'relational' as it recognises that these accounts are situated within a wider network of relationships and social structures.

The method involves reading each transcript four times (Mauthner and Doucet 1998; Fairtlough 2007). At the first reading I reflected on the story: the plot, the people, the words and metaphors used as well as my reaction to what had been said. During the second reading, I read for the voice of 'I'; how the workers spoke about themselves, their experiences and feelings to gain insight into their identity and role. At the third, I was looking at interpersonal relationships, how they described their interactions with others. The final reading concentrated on the broader context: the social, political, cultural and organisational aspects that impacted on their role. Returning to the transcripts in this way helped me to refocus my attention on the individual workers in

their situations. Whilst I have not reported my findings using this method specifically, it did enhance my thinking about the language used, the relationships and the context. I have been able to weave the information gathered into my writing.

As I talked to more palliative care social workers, my impression was that I was hearing about palliative care social workers playing an important role in children's and their parents' lives. Those I spoke to cared about the children, their current situation and their future. Many spoke fluently and at length about the children, their families and the work undertaken. It was as if this was the first opportunity they had to share, and even unburden themselves, about the issues involved. I regret not asking why the workers had selected the particular families to share with me as it would have provided an interesting dimension. One of the difficulties in using a questionnaire is that the results are 'one-dimensional'. The printed word does not always carry the passion, although there were instances when it did shine through. However, this is not the same as having the opportunity to observe a person's reaction to the questions and the way in which they tell their 'stories'. Whilst it was possible to discern some of this passion in the focus groups, I was aware that some people may feel uncomfortable and therefore be reticent in showing their emotions in front of others.

## My 'self' in the research journey

'Research design should be a reflexive process which operates throughout every stage of a project' (Hammersley and Atkinson 1995:24). Qualitative research particularly requires a high level of reflexivity about the phenomenon under study (Darlington and Scott 2002:18) due to the immersion of the researcher in the research. Self-reflection has become increasingly significant in social work, particularly in social work education. Its roots are to be found in the early psychotherapist approach, where an understanding and use of self were integral to practice learning (Schön 1983). The term 'reflectivity'

has emerged from the professional practitioner and educational discourse: the process of reflecting upon practice. 'Reflexivity' comes more from social science researcher discourse, from qualitative and ethnographic researchers. It has, according to Finlay and Gough (2003:ix), 'exploded' into academic consciousness.

Being reflexive appears to be more complicated than being reflective: not merely searching for implicit theory, but searching for an understanding of the many ways in which one's own presence and perspective influence the knowledge and actions created. There are many definitions of reflexivity in research (Fook and Gardner 2007) but, at its simplest it is 'to have an ongoing conversation about experiences whilst living in the moment' (Hertz 1997:vii). Birch (1998) also sees research as a journey, particularly in terms of reflexivity. She divides the journey into three phases: going there; being there; and, being here (p.162). *Going there* refers to what is happening before the field of study: entering into the research, defining what to study, and gaining access into the field. *Being there* relates to being in the field: the actual taking part in the research. This includes the participation, the involvement and relationships made. *Being here* is the creation of the text: the final stages of analysis and the writing up. I will use these phases to share my aspects of my inner dialogue during the research process.

#### Going there

I came to this research study after a long career in social work education together with practice experience in child care and particular interests in medical social work. I was excited about this new phase in my life, determined to do well, but also anxious about proving my capabilities, not only to the academic world but also to myself. In addition, I had to cope with a role reversal, now I was the student. The first supervision sessions felt strange, particularly as they were with staff who had previously been work colleagues, but this strangeness soon disappeared and the sessions felt supportive and challenging. Reading is a passion and undertaking the initial literature review was

conducted with enthusiasm. As my experience of conducting research was limited, I also read widely about the various methods of conducting qualitative research and attended relevant lectures on methodology. At times, all this seemed rather bewildering as I grappled with some of the philosophies and concepts, but gradually an understanding emerged.

I felt confident about gaining access to palliative care social workers after attending the launch of *Palliative Care, Social Work and Service Users: Making Life Possible* (Beresford et al. 2007) where many people expressed an interest in what I was proposing to do. I was delighted when the Chair of the APCSW gave me permission to contact the Association members. However, I now felt that I had also to prove my capabilities to palliative care social workers since they were granting me the privilege to enter into 'their world'. Being a qualified social worker, I hoped, would be an asset in conducting this research; there could be a shared understanding about the values of social work. However, I recognised it might also be hindrance as it might lead to my making assumptions and introduce bias and this was something I needed to take into account in my contacts and also later during the analysis.

From the beginning, I documented my journey keeping a personal diary of events and thoughts (Moch 2000). This was not only in order to keep an audit trail (Huberman and Miles 1998) but to remember and understand the decisions made along the way that might otherwise be lost in memory as other events took over. I saw this as a positive tool (Payne and Payne 2004). Sometimes, there were no entries for days or weeks on end. However, when a particular difficulty (for example, the move of my principal supervisor to another university) or momentous event (for example, gaining approval from the Ethics Committee) occurred, I took the opportunity to reflect back over the time since the last entry. Events that happened in my personal life were also noted and it

was interesting to look back and see how these had impacted on my 'research based self' (Reinharz 1997).

#### Being there

The first focus group was both exciting and daunting. Exciting, because I felt that I was at last beginning the research 'proper'. Daunting, because of the large size of the group and wanting to ensure that the time would be productively spent, not only for me but also for the palliative care social workers who had taken the trouble to attend. In the event, the group functioned well, the discussion was stimulating, and workers spent time informally afterwards talking about their experiences and interest in the research. The final arrangements for the second and third focus groups were more daunting because of the 'unknowns': would the participants at the Conference be interested enough to give up their free time to attend; would there be enough to hold two focus groups as planned? I was relieved that five people attended each group and that once again the discussions flowed well. Being a moderator was a relatively new experience but my years of experience running seminar groups gave me some confidence.

Just after the completion of the focus groups, I moved from the University of Dundee to the University of Stirling to follow my principal supervisor. This entailed adjusting to a new location, new 'rules', a new student group and new second supervisor. People were very welcoming and I soon felt settled. Having a new supervisor, who was not a social worker, brought a different perspective to the research. Her challenges in particular about social work issues that I had perhaps taken for granted, opened up new areas of thinking.

The questionnaire was planned to reach a large audience but its design was not as straightforward as I had originally thought it might be and I had to weigh the advantage

of obtaining detailed information against a possible low return rate. The richness of data received was more than enough compensation for a low return.

I looked forward to the interview stage: getting out, meeting palliative care social workers at their work base, and hearing their stories. After talking to each participant to make the final arrangements for the interviews, I felt confident about what lay ahead. All were very welcoming and positive about the research. I had been anxious to demonstrate that I had an understanding of social work issues and did so by stating in the participant information sheets that I was a qualified social worker. I hoped that this would reassure the potential participants as well as reduce the power imbalance (Skinner et al. 2005). I was aware that my role was to be a researcher, not a social worker and certainly not a social work educator! The latter particularly, as mentioned earlier, was something that I had to check myself on many occasions to ensure that they were being asked for the 'right' reasons; that they were important for the research study. Most participants did not appear to have any difficulty in telling stories with detail and colour and I frequently felt humbled and often very saddened by what was told to me.

#### Being here

Analysis is a challenging and exciting stage of the qualitative research process. It requires a mix of creativity and systematic searching, a blend of inspiration and diligent detection (Spencer et al. 2003:199).

After trying various ways of looking at the data, writing up my ideas, and having discussions with my supervisors, a picture began to emerge that I felt represented the findings in a meaningful way. I had been heartened and reassured when I read the words of Mauthner and Doucet (1998:122) regarding the early phases of data analysis:

they can 'feel messy, confusing and uncertain because we are at a stage where we simply do not know what to think yet'. I had feelings of panic and wondered whether themes would ever crystallise (Borkan 1999). I realised too that the participants' words could be interpreted in many ways and felt a strong sense of responsibility in what I was undertaking. I also recognised that there were occasions during the interviews when I failed to ask follow-up questions to clarify points which meant I might be making assumptions.

During the various analysis stages, I wondered how necessary it was to go through all the steps with each transcript. How thorough did I have to be? Is there a 'right' way to use a particular method? The words of Froggatt (2001: 434) were consoling: 'There is no one right way to work with the data, and it is a process best learnt by doing'. I recognised that my interpretation of the data would be influenced by my belief that children should be 'looked out for' and their voice heard. I felt a great responsibility when interpreting the data and writing up my findings. I was not only presenting it for academic scrutiny but to the social work community. I did not want to let myself down and particularly not the children whose lives I had heard about, nor all the palliative care social workers who had contributed at various points along the way. Whose 'voice' (Hertz 1997) was I selecting to put forward in the write-up? How would I chose which extracts to quote to illustrate my themes (Humphries and Martin 2000)? I was also conscious of the fact that I could be judging the reality of others in relation to my own (O'Leary 2004). Supervision sessions and feedback on my writings were extremely constructive and helped me to take a fresh look at my findings. Being here was, at times, a lonely existence but also a heartening and rewarding one.

#### Conclusion

This chapter has explored my journey through the research, charting the stages and phases within it, together with my inner dialogues. In chapters seven to eleven I present the analysis of my findings together with a discussion as an integrated whole (Holloway 2005). As I saw children to be at the centre of the study, it seemed important to first present a picture of their experiences at this difficult point of their lives, as described to me by the palliative care social workers. This picture acts as a backcloth to the subsequent chapters: a reminder of the children, their circumstances, difficulties, challenges and needs. The palliative care social workers' perceptions of their role in safeguarding these children follow in chapter seven, with an examination of what safeguarding means to them generally and specifically in their own work context. It seemed important to present these findings early in order to gain a picture of how this understanding impacts on their work with children, their families and also their colleagues both in their own and other agencies. Chapter eight importantly considers how children come to the attention of the palliative care social workers; in particular, who makes referrals and why, as this process is likely to determine which children may be safeguarded. Chapter nine then concentrates on the findings that relate to the work palliative care social workers undertake with children and their parents. It contains examples of their safeguarding practice and shows the range and diversity of work undertaken. These examples highlight the relationship between palliative care social workers and social workers in local authority children's services and these findings are examined in chapter ten. The final findings chapter concentrates on what informs the palliative care social workers' practice in safeguarding children, including information on their knowledge base, any supervision and training received and the influence of previous work and life experiences on their practice.

The following chapter describes the experiences of children where a parent is dying as told by the palliative care social workers during the interviews.

### Chapter Six The role of the palliative care social worker: the children's experiences

Each child is an individual and each child's story is unique. Daniel et al. 2010

#### Introduction

The literature review drew attention to the many difficulties and issues which can face children and their families when a parent is dying from a terminal illness. This chapter aims to illustrate the variety and complexity of the issues that, in turn, face palliative care social workers working with adults in their involvement with families. It also highlights those aspects that did not feature in research studies. The aim is to put children at the centre, ensuring that their voices are heard, albeit second hand and as interpreted by the palliative care social workers and then by me.

During the individual interviews, palliative care social workers were invited to describe at least two practice examples of their involvement with families where there had been concerns about the present and future care of the children. These examples provided a richness of material that allowed me to consider issues from the perspectives of both children and their parents, helping me to more fully understand some of the apparent needs of children. This understanding, I hoped, would also help me keep the child at the centre whilst reflecting and promoting reflection on the role of the palliative care social worker. Many examples were described in depth, others less so. This seemed to depend mainly on the amount and duration of contact between the family and the palliative care social worker. Whilst three main themes emerged: the effects of the parent's illness on the children, the relationship context around the illness, and the broader social and economic context, there was not a clear subdivision between them.

As will be seen there was often an overlap, demonstrating the complexity of the circumstances faced by children, their parents and palliative care social workers.

First I will provide a brief description of some of the characteristics of the practice examples. I will then consider each of the three themes with some short extracts from the practice examples to illustrate particular issues. Details have been removed or altered to ensure anonymity and preserve confidentiality. I conclude with a brief discussion on what the examples appear to tell us about the needs of children and the issues facing palliative care social workers.

#### **Description of practice examples**

There were a total of 45 practice examples which included palliative care social work involvement with 29 families pre-bereavement, 12 both pre- and post-bereavement and 4 post-bereavement. All age groups of children were represented in the sample as shown in Table 6.1. Information about the parents' country of origin was not specifically sought; some details were gained from the narratives (Table 6.2) but to ensure anonymity specific countries where known have been grouped together. There were more mothers than fathers who were dying or had died and approximately one third of parents were lone carers (Table 6.3).

Age in years	Boys	Girls	N/K	Total
0 - 4	2	8	2	12
5 - 10	14	14	5	33
11 - 15	13	16	7	36
16 - 18	4	3	0	7
Total	33	41	14	88

 Table 6.1
 Age and gender of children in practice examples

ents in practice examples

Nationality	Number	Nationality	Number	Nationality	Number
Britain	21	Asia	3	Europe	2
Africa	5	South America	2	West Indies	1
Not specified	11				

Relationship of terminally ill carer to child	Number	Number of lone carers
Mother	28	15
Father	12	1
Grandmother	4	0
Grandfather	1	0
Not specified	1	0
Total	46*	16

Table 6.3 Relationship of terminally ill carer to child and number of lone carers

\* In one example, a child was sequentially cared for by a mother and grandmother.

#### Effects of the parent's illness on their children

The research studies in the literature review identified various ways the terminal illness of a parent may affect the children in the family. The practice examples provided by the palliative care social workers particularly illustrated children's awareness of the illness; how they were informed about it; its impact on them; and, how parents' coping abilities influenced the care they provided to their children.

#### Child's awareness of the illness

A child's awareness of the parent's illness seemed to depend on the age and level of understanding of the child and also whether, or how well, the parents hid the fact that they were ill. In general, children were not encouraged to ask questions about the illness, a cloak of secrecy prevailed. There were examples of children who had lived with their parent's illness all their lives, whereas for others the illness represented a sudden and unexpected intrusion. There were children who had witnessed or experienced a variety of incidents related to the illness. Examples included children coming home from school and finding their ill mother in bed on a regular basis; a parent haemorrhaging or collapsing in their presence; a parent behaving in an uncharacteristic and perhaps frightening way due to either the illness itself or the medication; and, children reading letters from hospital for a parent who was not able to understand English. Children were often present when palliative care staff visited the home and might have overheard or misunderstood conversations.

A young mother, who had come to the UK on a work permit together with her husband and young children, was diagnosed with cancer and later developed brain metastases. The mother had continued to work for as long as possible and her husband had to continue working when she became ill, as he was not entitled to benefits due to his immigration status. The mother, either because of her illness or medication, was exhibiting confused behaviour and possibly hallucinating whilst the youngest child aged under five years, was in the house with her. This caused the youngest child to be extremely frightened and distressed.

There was a variety of examples of children who were very aware of the illness, wishing to be involved in what was happening. This included being with the ill parent whilst they died; seeing the parent's body immediately after the death; giving advice regarding the preparation of the body and helping with the rituals; and after the death, welcoming people to the house to view the body, and being part of the funeral.

#### Informing children about the illness

Children were usually informed about the impending death of a parent by the parents themselves. Generally, all the children in the family were told, but young children were frequently excluded in the belief that they were not of an age to understand. Accounts suggested that the communication process was often difficult, ad hoc and buffeted by events. Some children were prepared for the fact that a parent was dying months before hand and others a matter of weeks or days before. This latter time-scale might be determined by circumstances where a parent's condition suddenly deteriorated leaving very little, if any, opportunity for the children to be prepared and for the family to make plans. There were accounts of parents intending to tell children about the illness but delaying doing so as they were unsure how to go about it or when would be the best time to do so. There were examples of parents refusing or being unable to tell their

children right up to the last moment and, in rare circumstances, not even by the surviving parent after the death. In many instances, the children were reported to know that their parent was ill but not the extent of the illness.

Whilst the mother was working, the father who had cancer was caring for their children of various ages. The father's condition deteriorated very rapidly and he died shortly after a referral to the palliative care services. One of the children attended a special school and had severe communication and learning difficulties. The parents had wanted to help all their children know what was happening but were unable to find a way to let this child know or help her understand what was happening to her father.

#### Parents' coping abilities

There were suggestions in the accounts that the ill parents adopted either an active or passive style of coping with the illness, particularly in relation to their children. In the latter group not all were able to come to terms with the likely outcome of their illness and some refused to accept how ill they were. These parents tended to be unable to make plans for themselves or for the future care of their children, on occasion refusing to tell people about their illness. Ill parents who felt physically weak and tired tended to opt out and had no strength to care for their children either physically or emotionally, raising the question 'should I be a parent first or a patient?'. Parents in receipt of personal care services were known to ask the individual care staff coming to the house to undertake tasks with their children instead. This was either to allow them some respite or to meet the children's needs, assistance for the children not otherwise being provided by the local authority child and family services.

A mother, a lone carer, had made care plans for her two children. The youngest was to be adopted by a friend and the older child, a teenager, was already in foster care due to an earlier breakdown in their relationship. The mother was now an in-patient in a hospice and feeling that she had done all she could in regard to the future care of her children, did not wish to see the local authority social workers anymore to discuss their care. She felt very tired and needed to have it recognised that she was dying.

There were other examples of ill parents who, whilst accepting what was happening to them, were determined to remain alive for as long as possible in order to 'be there' for their children. This was particularly apparent in situations where there were difficulties surrounding the family's immigration status and when the ill parent's survival meant that the children and well parent could remain in the UK. There were also reports of parents who were isolated due to limited access to family members, again this was particularly significant for those families whose country of origin was not the UK.

There were accounts where children were a support to the ill parent who might be encouraging their child not to attend school but remain at home to be with them. Feelings of inadequacy and helplessness were described with ill parents being unable to cope with their own needs let alone those of their children. A late diagnosis, a rapid development of the illness, a late referral to the palliative care service all influenced how much time parents had to adjust themselves to the situation let alone consider the needs of their children.

A mother with advanced cancer lived with her teenage children. The father's whereabouts were not known. The children knew she had cancer but she was only referred to palliative care services at a very late stage when her condition was deteriorating very rapidly. She died two weeks later before the concerns about the

behaviour of the children and their future care could be assessed and plans made as a family.

Different coping styles were described in relation to the surviving parent and the ill parent, often affected by gender. These included a father returning to his native country for a while as he could not cope with his wife's illness as well as the children; a husband giving up work to look after his sick wife and children; parents attempting to work for as long as possible despite their illness.

#### Impact of a parent's illness on the children's behaviour

The examples highlighted a range of behaviour which included descriptions of children who were appearing to cope extremely well, being supportive and continuing to enjoy school to those who were bed wetting, not attending school, being suspended from school, not speaking at school, sleeping badly, drug taking, becoming 'naughty' and, being withdrawn, uncommunicative, hyperactive and confrontational.

Two primary school aged children were living with both their parents. Their father had cancer and had been ill for several years. His condition was deteriorating and he had a tendency to be aggressive. The younger child had become confrontational and preoccupied with thoughts of violence. Parents found the behaviour of both children difficult to manage.

There were also implications for the children as a result of the nature or the speed of the progress of the illness. If the illness was developing slowly, the family and the children were aware of the changes and either adjusted to them or resented them. There would be times too, when a parent was admitted to hospital or hospice and the child, when visiting, would be aware of the treatments being given to the parent as well as to other

patients. This could cause the child distress. One young boy was reported to be unable to visit his father because he was frightened of the change in his father's physical appearance.

#### Care arrangements for children following the parent's death

Respondents described parents making decisions about the children's future, both with and without their involvement. The age of the child and their perceived understanding appeared to influence whether or not adults considered they had a contribution to make. On occasion children took an active part in helping to sort out their own future care arrangements. However, sometimes a parent was unable to decide what was best for the child and died before reaching a decision; sometimes the decision was made in the very last days of the parent's life. There were descriptions of children moving to live with grandparents, aunts and uncles, older brothers and sisters, foster carers and friends, sometimes prior to the death. There was an example of a very young child being taken to another country by the father to live with his relatives after her mother had died. In some cases, no arrangements were made before the death of the parent.

A lone mother, dying of cancer, was caring for her pre-teenage son. The mother died before making any arrangements for him, as she could not decide with whom he should live. In the event, the boy went to live with the grandparents who had previously been involved in his upbringing and lived locally, despite previous concerns with respect to the quality of the relationship between the grandmother and grandson. The grandparents began to complain about the boy's behaviour - they were unable to control him and he was described as causing problems. The grandmother became ill also with cancer, deteriorated quite quickly and died. The grandfather accused his grandson of causing the cancer because of his behaviour. He no longer wished, or was able, to continue caring for his grandson. As no other family members were able or willing to take responsibility for the boy, the local authority placed him with a foster carer.

In about a third of the practice examples, the parent who was dying was a lone mother caring for her children and in some instances there was little or no family support during the illness. The father might no longer be in contact with the family. However there were instances where lone mothers decided to re-involve the estranged father; or fathers, on hearing of the illness, themselves took the initiative to re-engage with the family and offer future care for the children. Sometimes the father returned to live in the family home before the mother died. One lone mother refused to tell her children about her terminal illness, neither would she inform their father because of her bitterness about the ending of their marriage. On occasion, the current or future carers were thought by various parties, including the palliative care social worker, to be unsuitable to look after the children.

A mother with young sons was an in-patient in a hospice and her health was deteriorating. She had an unhappy childhood, brought up with her siblings by her father in a very deprived area and had previous partners who were physically abusive to her. A family member disclosed to the palliative care social worker that the mother had been sexually abused as a child by her father who was now caring for her boys.

Lone parents of children between the ages of 16 and 18 years, where there were no family members available to provide care, faced particular difficulties in attempting to sort out future living arrangements. Local authorities were generally not receptive to the young person taking over tenancy agreements and there were concerns that such a young person might therefore become homeless on the death of the parent.

#### Relationship context surrounding the illness

Changes to family functioning and the relationships between children and their parents were a feature of many of the research studies discussed in the literature review. As can already be seen from some of the above practice examples, the quality of relationships in the family had significant implications for the current and future care of the children and role changes within the family.

#### Impact of relationships on child's care

The relationships that the well parent in particular had with their children, depended on their ability to cope with the changes that the impending and actual death of the ill parent might bring. The accounts suggested positive outcomes with support from friends, family and agency staff and less successful outcomes if the relationship between the child and the surviving parent was already poor, or the parent was unable to meet or did not understand the child's needs, or considered their own needs as more important. There were examples of families where there was a history of domestic violence or alcohol problems. The changing nature of the illness could lead to aggression or the withdrawal of a parent or worsen an already difficult relationship between a parent and a child.

A teenage daughter lived with her father who had cancer with brain metastases. Her mother had died some time previously. There were angry clashes between father and daughter regarding the daughter's social life of which he did not approve. The daughter admitted she was scared of her father and the relationship had not been good prior to his illness. The father accused his daughter of being the cause of his illness and threatened to kill her.

#### Role changes in the family

Changes to the roles of nuclear and extended family members occurred both during the illness and after the death of a parent. During the illness, particularly in the situation where the mother was the ill parent, there were examples of children having to take more or full responsibility for themselves and for their siblings, and perhaps providing physical care for the ill parent.

In one family, a father returned to his native country leaving his dying wife and children for a period of time. The adolescent daughter had to cope with the care of her mother and younger siblings. On one occasion when the mother's condition deteriorated, necessitating an emergency admission to hospital, the ten year old was left to care for the frightened younger siblings whilst the older daughter accompanied the mother in the ambulance.

A commonly reported feature was children being looked after by others whilst the ill parent was an in-patient in a hospital or hospice. This care was usually provided by relatives or friends. During the illness, the well parent might take on new responsibilities for the children as well as the caring role for the ill partner. However, this was not always an easy transition to make for all parties. There were also examples of families struggling with a lack of support.

A mother had cancer and had been ill for some time. The relationship between the parents had broken down before the diagnosis but they remained together. There were four children of varying ages. The mother had returned to work after her treatment but when the illness returned the father gave up work to look after his wife and youngest child leading to difficulties. The mother felt 'so flat, so wretched' and was disengaged from what was going on. The atmosphere in the home was described as one of 'absolute gloom and despair'.

After the death of a parent, there were families where children took on the responsibility of providing practical and/or emotional support to the surviving parent. There were instances of older children, sometimes themselves under the age of 18 years, taking over the parental duties for their younger siblings, there being no surviving parent. This on occasion caused role conflict. The re-introduction of a parent, usually the father to his children, meant the making of new relationships. How well families adjusted to their new circumstances was reported to be influenced by various factors. These included to what degree the surviving parent was 'caught up' in their own grief; the previous family history; the desire to respect the dead parent's wishes; their determination to succeed; the amount of support provided by family, friends and other agencies; the bureaucracy of various agencies; the gender of the surviving parent; and, the parent's ability to take on new roles. It was easier to cope and make plans if the parent felt secure in terms of employment and finance and there were not too many practical obstacles to overcome.

#### Broader social and economic context

As discussed in the literature review most of the existing studies into the child's experiences of living with a dying parent are 'psycho-social' in approach but appear to pay little attention to the social context. Three sub-themes relating to a broader social and economic context were identified in the examples: poverty, gender roles, and immigration status. These are inter-linked and references have already been made to aspects of these issues in earlier cited examples.

#### Poverty

Being concerned about finances was seen as an extra burden carried by parents which in turn impacted on the children. There were children who had experienced financial hardship in the family as a result of the parent being unable to work due to their own illness or to looking after the ill partner and children. There were families who were described as living in areas of deprivation and poverty and severe financial difficulties were identified as an issue for many families. A number were reported to live in poor quality housing and the home conditions of one family were described as 'appalling'.

A very isolated lone mother with a poor prognosis lived in a deprived part of the city with her pre-teenage child. Both were victims of anti-social behaviour: damage had been caused to the house and there were also incidents of name-calling. As this was so frightening they kept the curtains closed at all times. The mother had a very limited amount of money and refused to go into debt, preferring to do without; she had not realised that she was entitled to Disability Living Allowance.

#### Gender roles

In various examples there were fathers who were described as taking on roles previously undertaken by the ill mothers. Attitudes to gender roles impacted on the provision of practical support given to a family. There were reports of instances where local authority staff would not provide resources for families where a mother was dying and the father was working. Fathers were considered to be 'responsible adults' and it was thought they should 'be there' for their children even if this meant taking time off from their work or working reduced hours. This might have meant a loss of income or even the job itself for the fathers concerned and some, because of their immigration status, would not have been entitled to receive benefits.

#### Immigration status

There were practice examples given of families where the immigration status of the family was dependent on the dying parent. There were reports of the Home Office allowing the well parent and children to remain in this country only until the ill parent died, causing great distress and uncertainty. Families had come to this country with a view of obtaining a 'better life' for their children and the children had generally settled well, making friends and wishing to remain in the UK. There was sometimes a great uncertainty about what a return would mean and the notice of departure often appeared to be very last minute allowing no time for preparation and goodbyes. Not only had the child lost a parent but also friends and a familiar way of life.

A mother had come to the UK with her pre-teenage children. She was determined to make a life in this country primarily for them. She was diagnosed with cancer shortly after her arrival, which was treated. She continued to work and remained well for a period but then the cancer returned. The children's father arrived in the UK before she died to look after the children. The mother had been determined that the children should remain in UK and the father was happy to comply with her wishes, but the Home Office refused to extend his visa, which he appealed. The children were very distressed about the death of their mother and the uncertainty surrounding their future. The behaviour of the younger children in particular was a cause of concern. There was no financial support available and the family had received an eviction notice.

#### Discussion

The practice examples provide a wealth of information about the apparent needs of the children. The palliative care social workers' accounts suggest that it would be reasonable to believe that, to varying degrees, the children would be feeling bewildered, frightened, unhappy, threatened, lost, guilty, depressed, scared, unsupported and

perhaps even unloved. It would perhaps be appropriate to speculate that children in the circumstances described needed consistency and continuity, to be informed, supported, listened to, protected, taken account of, respected and to be prepared for what is and might be happening to them and the people around them.

The circumstances of a parent's terminal illness clearly 'rippled out' (Hudson and Payne 2009) to affect many people, but particularly children, and it is evident from the practice examples that the family system changes as a result. From an ecological perspective (Jack 2000), the development of children and their life chances may well be affected by the changes in the socio-economic circumstances that illness can bring. There was evidence that children's development and well-being were already being compromised prior to the illness; the illness compounded it.

Whilst the examples highlight many aspects featured in the literature discussed in chapter three, they also highlight those that were either not featured or very minimally so in research studies. These aspects include children who were living in neglected or abusive situations; families where there was existing conflict between parents, including situations of domestic violence; families suffering from financial hardship and those living in poverty either prior to or as a result of the illness; and, the uncertain immigration status of the families which impacted on every aspect of decision making pre- and post the parent's death.

#### Conclusion

Within this chapter there have been illustrations of some of the complex issues that can come to light after the involvement of a palliative care social worker in families where there is a child under the age of 18 years. These issues will be explored in future

chapters. First, I will examine how safeguarding children is perceived by the palliative care social workers in this study.

### Palliative care social workers' perceptions of their role in safeguarding children

All those who come into contact with children and families in their everyday work, including people who do not have a specific role in relation to child protection, have a duty to safeguard and promote the welfare of children. DfES 2006

#### Introduction

As indicated in chapter four, safeguarding children encompasses a wide spectrum of concepts and activities from the broad promotion of welfare to the sensitive and extreme aspects of child protection. Government guidance makes it clear that anyone who works with adults who are parents or carers of children has responsibilities towards those children (The Scottish Government 2005; HM Government 2010). In this chapter I will focus on what the palliative care social workers understand about safeguarding children and how they see it impacting on their work. This chapter will therefore act as a back-cloth to the subsequent chapters that will both describe and analyse the palliative care social workers' involvement with children and their parents and also with the local authority children's services.

The first part of the chapter concentrates on the specific responses of all participants about what safeguarding means to them, how the interviewees viewed their agencies' general attitude to children and all the participants' views on how a parent's illness may impact on their capacity to meet children's needs. The second part considers what all the participants in the study understand their role to be in safeguarding children, its emotional impact on them, how confident they consider themselves in the role and how they might become more effective. In the third part, there is a discussion on the emerging issues.

#### Safe in the widest sense

When asked to describe what safeguarding meant to them, the overwhelming response from the palliative care social workers in the survey, focus groups and interviews was that safeguarding children carried a much broader remit than child protection. One worker in a focus group described it as:

Ensuring children are safe in the widest possible way and not just at risk from a particular person.

Another described safeguarding as 'less of a word than protecting' which I took to mean that safeguarding was a more general term; protecting was more specific. A similarly broad view was taken by one interviewee who said that:

I think there is a subtle difference between child protection and safeguarding children. We talk very much about safeguarding, literally keeping children safe and I think of being much more aware of the impact of things like domestic violence, or a parent with mental health issues, and I think having a parent who is dying increases the child's sort of risk of emotional abuse. The practicality of keeping a safe environment for them.

The prevailing view was that emotional harm was more frequent than other types of harm and that it was generally 'unwitting' emotional harm brought about by the impact of the illness on the family. The broad brush of safeguarding was described by one worker in an interview, which included having regard to the emotional needs of a child:

... safeguarding them in a general way, even just the right to feel safe and have a routine and not to be moved from pillar to post. Absolutely crucial if you work with the family. It is an absolute. You need to be making sure that their child is looked after and safeguarded when they are gone and also be aware of helping the child emotionally to get through the difficult experience.

Palliative care social workers generally saw the family as their priority with the focus being on the children, being alert to their needs and ensuring that they were not 'forgotten'. One interviewee, for example, specified that safeguarding children was about

protecting a child from being harmed and working with the family as a whole system. Another considered that protecting children from harm was a 'tiny, tiny piece of the puzzle'. She was more concerned about the impact of loss on a child's development and in her experience was aware that a difference could be made if some 'deliberate preparation' was made in partnership with the parents.

Neglect, as a particular aspect of harm, was rarely referred to except in a general sense, something to be aware of. 'Unwitting emotional neglect' was mentioned by one interviewee, where parents might be so caught up with their own emotions and difficulties that they were not aware of their children's emotional needs. However, there was evidence that palliative care social workers recognised when children were emotionally vulnerable but did not seem to rush to describe or perhaps even see them as being 'neglected'. There were two examples in the survey where young children were considered to be emotionally neglected and at risk. However, the local authorities concerned refused to become involved, as there was 'no tangible proof' for them to do an assessment in one situation and 'nothing specific had happened' in the other. Relationships with the local authority will be explored in chapter ten.

Whilst child protection, in terms of physical, emotional and sexual abuse, was seen to be an important part of safeguarding, it was considered to be a rare occurrence in the experience of the palliative care social workers. One respondent in the survey gave the following opinion:

Coming from a child protection background I am amazed how infrequently we 'come across' or are 'exposed' to issues of protection except in the form of emotional support, readiness for the dying, and my suspicion is whether issues are submerged by the situation [dying] the family are experiencing.

This view was echoed by an interviewee who also wondered why this should be the case. Another palliative care social worker kept a record of referrals to the local authority from her hospice and in 2007, five referrals were made and in 2008, one.

#### Putting 'a universal responsibility' into practice

The widespread view of the palliative care social workers in the interviews was that inpatient hospices were generally 'child-friendly'. They illustrated this claim by describing how children were encouraged to visit their dying parent and a room was provided for children with a range of toys, DVDs, videos and games. A selection of information leaflets and books on aspects of bereavement were also available that could be shared with children, including culturally specific resources. There were four examples given of children being invited to live in the hospice so that they could spend more time with their dying parent. For one hospice, this resulted in a large financial cost and an emotional experience for the staff. However, as the palliative care social worker involved explained in a focus group:

It was very draining but that has actually set a benchmark for us and it has made us less frightened of it and it really was a very positive experience and actually now we are more child-friendly. ... we are much more confident about preventative work.

Attitudes have changed over time as one worker based in a hospice explained in her interview:

Certainly when I came here 21 years ago nobody thought about the children the impact on them. There was no service. Very much forgotten.

There were hospices where a specialist children's worker was in post. The main emphasis of their work appeared to be in relation to the provision of support to children post-bereavement by way of group activities or individual work. There were fewer examples of some individual pre-bereavement work jointly planned with palliative care

social workers. The specialist workers also provided support and advice to any hospice staff as required.

Safeguarding children was seen by the participants as a universal, collective responsibility. One palliative care social worker summed this up by writing:

Everyone who has contact/knowledge of children has an important role to play in ensuring their safety. We must prioritise this and be vigilant at all times. However, in a small minority of interviews there was a view that the response of the medical and nursing staff, and the workplace as a whole, to children was poor. Nurses, one hospice based worker considered, would be 'shocked' if they realised that the child's welfare should be a primary consideration. Another palliative care social worker spoke of the medical and nursing staff feeling that children were 'difficult' and that contact with them required someone with special qualifications; another, of the fact that there was no strategy to meet children's emotional needs. Indeed, there was one hospital based palliative care social worker who recognised that not much work was done with the children by any member of staff and that it 'spooked' the consultant. At one hospice it was recognised that the needs of children were not well met and although there was a plan to appoint a specialist worker progress was too slow in the view of the

palliative care social worker. There had been an effort to make general information available to children by use of a display board but the management considered this inappropriate, saying information should be put into children's hands via the parent.

This universal responsibility was seen to extend to all who had a role in the workplace. In order to raise awareness within hospices, the general practice described by the interviewees was that all staff, including voluntary workers, received training on safeguarding children. The five palliative care social workers in the interviews and focus groups who identified themselves as the designated child protection officers for their agency had been responsible for drawing up the safeguarding policy and procedures for

all the staff. They, and other palliative care social workers, were also involved in the training on awareness, policy and procedures. Such training often provoked discussion about standards and one worker specifically mentioned that, as a result, staff had come to understand that it was acceptable to discuss a family's situation where they had concerns.

There was a discussion in one focus group about how much palliative care social workers should say to families at the initial stages of contact with regard to their responsibility about 'looking out for' the needs of children. There was a range of views expressed: highlighting this aspect of their responsibilities at the start was useful if a concerning situation were to arise later; explaining was an honest approach to take; doing so raised a barrier and it was better not to explain this aspect at the outset; and it would depend on how it was done. Generally it appeared that there were also clear agency procedures about making a referral where child protection was an issue, although two workers in one focus group were unsure whether there was a policy in place in their agency. One interviewee reported that despite one being in place, she received no advice or direction from the hospice in which she worked about whether she should report some specific information about children's safety to the local authority. The line manager was unavailable and there seemed to be a feeling amongst the medical and nursing staff that making a referral would be too upsetting for the patient; the thought of doing so, according to the worker, 'disturbed them immensely'. In the end, the worker drew on her own experience in local authority work and made a report, but wondered afterwards whether she had delayed too long. There was an echo of this for another worker who was not sure in hindsight whether she should have acted sooner with one family and commented:

I think sometimes on your own in a situation you can feel very isolated.

This highlights the particular difficulties where there are lone workers, or a lack of personnel who can give appropriate guidance, particularly in situations where decisions are needed to be made guickly.

## Impact of the parent's illness on parenting capacity and meeting children's needs

The concept of 'good enough parenting' discussed earlier might be challenging to apply with families when a parent is dying. Indeed, there was no clear response to the question 'What is good enough parenting?' asked during the interviews. There was a recognition that standards differed. There were those who avoided giving a definition saying it was 'hard to do', 'subjective', 'a terrible value laden issue', 'never an expression I have used kind of regularly or think of'. There were those workers who accepted that there were families who operated in different ways and perhaps not in a way that they would want for a child. One thought it important to consider whether the parenting was good enough for that child and another similarly that it needed to be 'contextualised', it being difficult to decide what was the minimum level particularly in relation to families living in very deprived areas. The provision of emotional warmth and support as well as physical care and protection from harm were specifically mentioned by three people. There were those who were more specific in terms of situations where a parent was terminally ill, saying that being 'good enough' meant communicating with and appropriately involving a child; ensuring a child still had a feeling of belonging and being loved.

There was an acknowledgement that illness might prevent parents from parenting in the way they had previously. Workers had witnessed young mothers who were dying withdraw emotionally from their children in order to cope with their own emotional needs. Supporting parents in order to help them provide emotional warmth was

therefore considered to be an important element of safeguarding children. A worker, who had prior experience in local authority child protection work, spoke of the differences in attitudes in relation to parenting:

In local authority it is usually parents who can't parent and then you get involved and assess risk but in these kind of cases it is parents who can parent but the illness prevents them from doing so. They are still good parents but they need support.

This apparent split between 'our' families and 'their' families is interesting as it highlights a crucial dilemma raised in many of the interviews: the balance between the needs of the ill parent and the child in terms of safeguarding the children. Discretion, it seemed to be suggested, was sometimes required. The importance of not allowing the illness to be the primary focus was mentioned, as well as the need to look at the strengths within and outside the family. One palliative care social worker in a focus group described the dilemma by saying:

Have compassion with the person who is dying but if you take it too far you can miss the child who is at risk – need to protect the child. We have to hold the balance – not be judgemental and protecting children.

There was a view that parents often struggled particularly as they came near the end of their life, because of the burden they felt they had placed on their children. The impact on teenage children and their behaviour was considered by one palliative care social worker to be particularly difficult as teenagers often struggled in any case, trying to deal with their own understanding of themselves as individuals. Trying to untangle what was general teenage behaviour and what behaviour was in relation to the parent's illness was immensely difficult. The general view was that children were able to understand a parent's changed behaviour if explanations were given together with reassurances that they were not to blame for what was happening.

However, specific medical conditions could cause extra dilemmas for the worker. Sometimes a child could be in danger because of a change in personality in the ill parent due, for example, to a brain tumour. The resulting change in behaviour might cause the child to be at risk. The prevailing view of the interviewees was that this is not a straightforward matter and the context of the situation needed to be taken into account. In coming to a decision, one worker described how there were times when she had 'dithered and hovered' about whether to make a referral because she was aware that such action might destroy family relationships and alienate the patient from using the hospice services in the future. If the situation were to last for some time then the impact of this on the children would need to be taken into account. One worker stated this starkly in an interview:

If you have a troubled parent that is likely to abuse for the next ten years, it is very different if you have one that might be dead in three weeks.

Where these circumstances occurred, palliative care social workers explained that they tried to ensure that the children would not be left alone with the ill parent. This might be discussed with the well parent, or at a family meeting, and care packages organised as necessary to cover crucial times of the day to ensure the child was not solely in the care of the ill parent.

Whilst about a quarter of the survey respondents clearly stated that the child's welfare was paramount regardless of the parent's illness, others were less categorical. The overall impression was that the palliative care social workers would be attempting to understand what was happening and endeavouring to provide appropriate support to the parents. Here too there was recognition of the dilemma of balancing the needs of parents and children in what was known to be a short-term situation. One palliative care social worker wrote:

The illness would have no influence on the decision made. We would be sensitive to the needs of the family at the time but our concerns would be raised

with social services irrespective if the parent was ill. We would work proactively and preventatively with such a family but recognising that part of the prevention might involve social services.

Safeguarding and promoting the welfare of children is therefore not a straightforward matter. By being part of an adult service, a few palliative care social workers saw themselves looking primarily at the needs of the patient with children as 'add-ons'. Others were clear that children's needs should be the primary focus. There could be a conflict of interest with parents wanting their life to continue as normal as possible and with children needing to know what was happening. The worker might on occasions be advocating for the child with the parents, which one worker described as being 'a delicate line to walk'. One palliative care social worker described how, for her, the child's welfare was something to hang on to when everything else was in a muddle and staff were emphasising the difficulties the family were in. The worker would then point out that for this child the standard of life was not good and the welfare of the child was paramount. There was another view that this was too simplistic - the whole family's welfare had to be paramount and adults had to feel strong and capable and resilient in order to provide parenting for children. However, it was apparent from many of the practice examples shared in the interviews that this was something that parents often struggled to achieve.

Other workers found that standards sometimes varied between other professionals in the team. There were some who were more forgiving of parents' behaviour because of the nature of the illness which could sometimes lead to conflict between them and the palliative care social worker.

# How palliative care social workers perceived their role in safeguarding children

When asked about how they saw their role in safeguarding children, the responses generally reflected the palliative care social workers' descriptions of specific work with children and families which will be considered in chapter nine. About half of the respondents in the survey described their role, or part of their role, in safeguarding children as offering support to parents and undertaking preventive work. One worker wrote:

The safeguarding of children is one element of many in supporting a parent with a life limiting condition.

Being in a special position to look out for children came through many of the responses and was summed up by one worker:

I feel I have a unique opportunity to assess a family situation and offer input to improve things if there are concerns about a child's welfare. However, I also see my responsibility as including a need to report a situation if abuse is suspected.

Being alert to the needs of children, their safety and behaviour, and undertaking a risk assessment were other, though less often mentioned aspects. One worker described it by writing:

I think it is an extremely important role in that as a palliative care social worker I often come to know the family very well. This means we, as social workers, are often placed to recognise any child protection issues or to recognise when a child may be vulnerable ... We may also realise when a bereaved parent may be vulnerable to letting other (abusive) adults become involved in their family life.

However, there was one worker who appeared to stand back a little from children when she wrote:

I don't make decisions about child care, but I can provide advice to child care professionals.

Liaising with other agencies, especially social services, and making a referral to child protection agencies if there was any suspicion, evidence or concerns about a child being at risk also featured in about half the descriptions provided in the survey. The need to follow agency procedures and guidance was included here. However, one worker wrote:

My primary role is to support patients and their families through their illness and on into bereavement as necessary. I have a duty to report concerns, which I do, but I am not a key worker in child protection. I think the role can easily become blurred, particularly when the child protection concerns are low key.

It is not entirely clear what this worker meant by the 'role can easily become blurred' and as this was a survey response, I was unable to ask for clarity. One interpretation could be that she saw a clear separation of roles, one as a supporter and the other as a protector and that there should not be an overlap. Another could be more straightforward in that she saw herself as primarily a worker for adults rather than children. In either case, the blurring was something that was uncomfortable for her.

In the focus groups, palliative care social workers also described the various ways they were able to safeguard children by undertaking preventative work with a whole family perspective. This included supporting and enabling parents to understand their children's needs, mediating and advocating on behalf of parents and children in order to obtain services, and counselling. Working with aspects of risk was generally recognised, as well as being aware of the need for children to be prepared for the death of a parent. Educating other members of the multi-professional team about safeguarding children through training, discussion both in meetings and on an individual basis, were also seen as part of their role.

In the interviews, whilst protecting and ensuring children were safe from any sort of abuse or harm were mentioned, protecting them from emotional harm was seen as an important dimension of the role. One worker went on to say:

To see that their emotional needs are met and not neglected I think that is the hardest part really.

Looking after the child's needs, alleviating emotional distress, making things clearer, keeping them informed, helping them say their goodbyes and supporting them afterwards were all aspects of this protection. Having direct contact with children varied depending on agency policy and circumstances and will be discussed in chapter nine. Again, the role of an educator to other members of staff was also seen as an important task as well as being a supporter to enable them to undertake work with children.

#### Emotional impact of the safeguarding role

Caring for people with cancer and their families comes at an emotional cost (Faulkner and Maguire 1994). Children facing the death of a parent can arouse strong feelings in the health care and other professionals involved, as they may be reminded of themselves as children or of children to whom they are close (Monroe 1990). This can cause a tentativeness in approaching families by not wishing to add to the stress and a concern 'not to transgress family values and child rearing practices' (Macpherson and Cook 2003:398). The dilemma of having compassion for the dying parent, whilst at the same time being alert to a child possibly being at risk, was highlighted by one worker who said: 'having to hold the two bits takes a lot out of us'.

During interviews and focus groups workers displayed warmth and compassion when they spoke about the families and occasionally came close to tears: 'I can feel myself welling up' remarked one interviewee. There was much detail in the descriptions, which indicated how workers would often go 'the extra mile' for the families. One worker

movingly described how, when a mother appeared near to dying in a hospice and wanted to write something for her children:

I dashed down to the gift shop - and all they had were get well soon cards but I found a Monet water lily sort of thing.

Although workers felt able to 'switch off' there were some circumstances that made it more difficult to do so. One person summed it up by saying:

Some people touch you and you can't let everybody or you would just be a wreck.

Another worker described feeling 'raw' after the Home Office refused the application for the father and his children to remain in the country after the mother had died. Other examples of the difficulties regarding the immigration status of families also resulted in feelings of frustration and heartbreak for palliative care social workers who, despite their efforts to advocate on the family's behalf, failed to secure the remaining family members' right to stay in the UK.

Workers did not often tell children themselves that a parent was dying or had died, preferring to support the parent to do so. However, when they have been asked to do so they found it extremely emotional; one worker described it as 'harrowing' telling a little boy that his father had died. Palliative care social workers rarely attended funerals but there were examples of where they had decided that this was in the best interest of the child for them to do so. One worker explained her feelings about attending funerals when describing her long involvement with a mother who had died:

I don't have a blanket rule but you know - if I feel I need to - I get a feeling I need to go and I didn't feel that I needed to go but in a way it was weird because sometimes even I can't believe she has gone after four years so may be I needed some closure and should have gone.

The general impression though was that despite the high emotional level of their work, palliative care social workers were able to find ways of coping with the distress and gain from it.

# Confidence in their safeguarding role

The survey asked a specific question about how confident the respondents felt about working in situations regarding safeguarding children. Workers ranked themselves and two thirds were confident or very confident and one third less confident or not at all confident. Eighteen workers did not respond when asked what they considered would make them more confident. Of the remainder, more training scored the highest response (n=20). Interestingly, having more experience was thought to be beneficial for five people. Other responses included having more time to spend with families, having a more appropriate response from the local authority and receiving regular updates on local working practices with liaison available. One heartfelt plea came from one worker:

I need to be in a context (organisation) that is confident – it can't be left to me or any other one / individual professional to keep instilling such confidence. Regular training seems key but also a reflection for colleagues how much they are doing already and how much they do know.

There was no mention of supervision giving confidence or more or different supervision being required to provide it.

During the interviews and focus groups, the general impression was that workers appeared confident as they talked about their work with children. Those workers who had previous experience of child-care commented that they felt more confident about working in families where there were children, comparing themselves to those workers whose background was with adults: One interviewee explained:

I think because of my background I am probably more geared up than a lot of people to deal with this, because a lot of people are adult social work trained and I do feel that is why I am more confident than a lot of people.

One worker without a child care background spoke of going into situations feeling she had a lack of appropriate skills:

I think it is very frightening for any of us to go into a situation when we don't feel very skilled and especially if we think that our role should be this and it feels like because nobody else is responding that we are being expected to also do this. In that area that we may not be comfortable - it is easy to move on to a level of panic that may not be just for us as social workers but across the multidisciplinary team.

Also mentioned was the need to keep up-to-date in regard to safeguarding children by attending training, where this was available, or by reading. This is the dilemma: how much training and experience should workers in adult services, and palliative care social work in particular, have in regard to safeguarding and promoting the welfare of children?

# Palliative care social workers' thoughts on how they could be more effective in their role

The interviewees were asked what would help them to be more effective in their role with children and families. The responses echoed those given in the survey regarding confidence; there was not a majority view and training interestingly was not seen as a panacea. The range could be grouped into three areas: issues concerning their own practice, their workplace and other agencies particularly social services. In relation to their own practice, having more time to get to know the families and to follow up; having more experience of working with children; and, receiving more training in child welfare matters in order to keep up to date were all thought to be beneficial. Within their workplace, aspects considered to be important were having the social work role and expectations clarified in general, with their role with children in particular affirmed, and having a more child friendly environment. The ways that other agency practice could help them be more effective included more multidisciplinary and 'joined-up' working, especially with social services; all professionals speaking 'the same language'; having a

named person in the local authority child and family team to contact for advice; obtaining a response from social services when contact was made; and being informed about available resources.

# Discussion

The findings show that being alert to the needs of children in families where a parent was dying was perceived to be an important part of palliative care social workers' role. Personal feelings can influence a professional's practice in matters of child protection such as fear, guilt and sympathy (Pollak and Levy 1989; Horwath 2007a). Fear of physical aggression was mentioned in this study, where a parent had threatened to kill his daughter but in the event he was not aggressive. The main fear expressed was that a dying parent might withdraw from the palliative care service as a whole or from the palliative care social work service in particular. This fear could be linked to feelings of guilt and shame, which, according to Pollak and Levy (1989), centre on professionals wishing to be seen as trusting and supportive by service users, something which the palliative care social workers strove to achieve. Workers stressed the need to gain the parents' permission to make a referral whenever this was possible. Indeed, in the few examples given where workers did decide to refer without permission because of their concern for the child, the ill parents did subsequently withdraw from their service and even from the palliative care service as a whole.

Feelings of sympathy, compassion, or over-empathy with the dying parent might also be reasons for non-referral. These could be linked to not wishing to put a parent under the additional stress and anxiety that a referral to the local authority might bring. Parents may not be able to provide the best care for their children because of circumstances beyond their control, including poverty (Srivastava et al. 2005). Stevenson (2007:11) argues that these struggles with 'major environmental deficits' combined with an

uncertainty about thresholds may lead to a 'kind of passivity' on the part of workers. Horwath (2007a) also found that professionals made excuses for parents' behaviour in order to avoid making a referral.

This issue of parental capacity was one that obviously challenged workers in the study. A child's health and development rests with the ability of parents and caregivers to meet his or her needs appropriately (Department of Health 2000b:20). For every social worker, the construction of parenting will influence their practice (Daniel 2000; Woodcock 2003). However, in palliative care, it could be argued, there is an extra dimension: how you expect a parent who is dying to parent. What is considered to be 'good enough' or reasonable in these circumstances? Parents may lack the physical or emotional reserves to engage with their children (Rauch and Muriel 2004; Morrison 2010). If the general view is that 'parents need to be able to prioritise the child's needs above their own' (Sidebotham and Weeks 2010:96), parents who are dying are clearly at a disadvantage. Moyers and Mason (1995) demonstrated that the greater the combination of circumstances that seemed to weaken parents' capacity, the more the children were likely to be 'in need'. The literature suggests therefore that there is a need to discover what other social supports are available from the wider family and friends, the quality of the environmental factors, as well as factors that influence culture including ethnicity and religious beliefs (Rolland 2005; Rose 2010). Indeed the findings in this study point to a range of supports that some children called on outside the family including school friends, teachers and palliative care social workers. All will have an impact on the parents' capacity to respond to their children's needs. The findings here show that this exploration was something that workers endeavoured to undertake, albeit often within a tight time-frame.

Whilst child neglect was rarely specifically referred to, workers were very alert to the emotional vulnerabilities of children. It appeared that they, unlike local authority social

workers (Horwath 2005; Buckley 2003; Horwath 2007a), did not focus on specific incidents but on the overall impact of neglect on the socio-emotional development of the child. However, there is very little written about the link between neglect and physical health of parents, but it has been recognised that physical illness can have an impact on their capacity to act effectively as parents (Stevenson 2007), especially when associated with finance and partnership problems (Berry et al. 2003). This means that there could be competing priorities as well as an ignorance of a child's needs. This neglect is not deliberate and the parent is unaware of the damage being caused (Golden et al. 2003). The dilemma for workers here appears to be whose needs should take priority: those of the dying parent or the child? There is a tension of accepting less than adequate care for the child, knowing that it is likely to be short-term, against taking action and alienating the parent. Unlike social workers in other settings, e.g. mental health (Taylor and Kroll 2004), palliative care social workers are not balancing a parent's capacity to change with the risk to the child, as in palliative care it is expected that the parent's medical condition will deteriorate. Rather, short-term measures need to be found to assist the parents in the care of their children and this was the view here. The ill parent's needs may fluctuate so that support might be required at different times of the day as well as week to week. However, it is not just the ill parent's availability, physically and emotionally, that should be considered but also that of the well parent, if present (Morrison 2010). A matter of future concern could be the ability of the well parent to undertake the full parenting responsibilities. The dynamics in the family will change without the presence of an adult carer previously present.

As palliative care social workers are part of a palliative care team, how that team looks at thresholds of concern around parenting and child-care might influence an individual worker. Workers may adjust their own standards to meet those of the team, the 'group think' (Janis 1972 cited in Horwath 2005:93). However, there were workers in the interviews who appeared able to challenge their professional colleagues about family

situations and were prepared to do so; others less so. Palliative care social workers therefore depend on information from other professionals in their team about children and also from parents themselves. However, as noted earlier, parents may be frightened of their children being removed if they disclose any problems (Stanley et al 2003; Beresford et al. 2007); disclosure being perceived as a threat rather than a potential avenue of support.

## Conclusion

This chapter reveals a multitude of complex issues and dilemmas for palliative care social workers in their duty to safeguard and promote the welfare of children. The general view was that safeguarding children embraced a wider range of aspects than child protection alone, reflecting current government policies. Safeguarding was also considered to be an agency-wide responsibility and safeguarding training was reported to be in place. Generally, agencies were considered to be 'child friendly' and 'child aware', although a minority of staff were thought to be less aware than others. Whilst all workers in this study accepted that safeguarding children was part of their responsibilities, there was a difference in how it was embraced: some did so with reluctance, but most with confidence. Although there are dilemmas for all workers in any adult care service regarding priority of needs, for palliative care social workers there is the added dimension of the imminent death of a parent and the emotions that this brings to the family and the workers involved.

The next chapter explores how children come to the attention of palliative care social workers.

# The role of the palliative care social worker: how children come to their attention

New patients bring a new set of stories, often emotive and horrific. Oliviere et al. 1998

# Introduction

This chapter explores how children and families, such as those described in chapter six, are referred to palliative care social workers. From the range of responses gathered from the survey, focus groups and interviews it was apparent that children of service users came to the attention of palliative care social workers for a variety of reasons and in a variety of ways. I will describe the processes involved, including information about who makes referrals, the reasons for them and the rate of occurrence. This is followed by a consideration of the views of the interviewed palliative care social workers about the referral process. Finally, I will discuss these findings in the light of relevant literature.

# Who makes the referrals?

The vast majority of referrals to the palliative care social work service for general social work support and child care concerns were reported to come from a member of the palliative care team. Nursing staff were usually the first people to pick up concerns about children and young people. Generally, they were the ones who gathered information about patients on admission to the palliative care service and had ongoing contact with them. Doctors too were involved in admissions to in-patient and day hospices and would sometimes make referrals.

#### The multidisciplinary team and referrals

As is normally the case, all the palliative care social workers interviewed were members of a multidisciplinary team. The interviewees described how new and current patients were discussed at team meetings generally held once a week. These discussions could lead to a palliative care social worker becoming involved with families where there were children. The composition and numbers of the staff attending the team meetings varied, but they were usually comprised of representatives of the medical and nursing staff, and sometimes included occupational therapists, physiotherapists, complementary therapists and chaplains. The numbers attending ranged from 6-17. There was a commonly held view that these meetings were medically orientated, frequently chaired by the consultant or a senior member of the nursing staff, although there were examples of the chairing role being rotated around the members. There was also mention of the tensions that the medical model could create within the team meetings; one worker observed in an interview: 'power lies in the medical hierarchy'. There was a general feeling that palliative care social workers considered themselves to be undervalued, low in status, and having to 'fit in' and work hard to prove their worth.

Further, there was a range of opinions as to how family-focused the palliative care social workers considered the multidisciplinary meetings to be. At the one extreme, just under half the interviewees described their team colleagues as being sensitive, family-focused and 'holistic', with patients being seen in their context. One palliative care social worker commented 'even the doctors will be asking about the family'. At the other extreme, and less frequently, there were descriptions of how teams sometimes needed 'reminding' and to be 'nudged' to move from the medical side to social aspects in discussions about patients. Palliative care social workers found themselves routinely asking questions in meetings about the family composition and future care needs of children, seeing this need to remind their professional colleagues as an important task. There was a feeling, one worker thought, that working with children was a specialism,

something she disagreed with. There was a less commonly held view that some staff, particularly doctors, tended as one interviewee remarked to 'get a bit panicky where there are children involved'.

The composition of the team was found to make a difference to the discussions about family matters. One interviewee positively described the team:

It is consultant-led. She will always go through psychosocial needs and there is a lot of staff support in there and I give support to the community nurses, reassuring them they are doing the right thing, advising them when to refer on. Another explained how a change of team leader in a community team to someone with a health visiting background had led in her view to a positive difference in the language used and the type of information about the family raised. There were examples too of how teams had become more aware of the needs of children over time. In-house child protection training was highlighted as a way of making a positive difference to the staff's awareness. Listening to discussions at the multidisciplinary team meetings about new and existing service users also provided an opportunity for team members to hear a variety of perspectives about family matters, helping them to understand that there were different standards regarding the care of children. Such interactions raised the team's awareness about when it would be appropriate to talk to a palliative care social worker regarding any concerns about a family.

# The perceived understanding of the role of the social worker in the multidisciplinary team in relation to children

In order for team members to make an appropriate referral to the palliative care social worker, it might reasonably be assumed that they had an understanding of the worker's role, particularly in relation to families where there are children under the age of 18 years. However, there was a range of views from the interviewees about whether this was in fact the case, and very often difficulties were reported. For example, there was

sometimes a lack of clarity as to responsibilities within the team and to what social work could bring. There were reports of team members sometimes specifying what they thought the social worker should be doing; one worker ruefully saying 'everybody thinks they can do social work'. Lone workers spoke of feeling particularly vulnerable. As a focus group member commented:

As a single worker within a medical team, it's sometimes a bit lonely when you have tricky situations.

One palliative care social worker based in a hospital was described as an 'extended' member rather than a core member of the team. She consequently considered herself to be at the bottom of the hierarchy, leaving her feeling devalued.

Although there were three participants who specifically stated that they did not believe their role was understood, the impression from the interviewees and focus group members was that generally palliative care social workers felt comfortable about their place within the palliative care team. The workers' expertise in childcare matters, at whatever level, was recognised by the other members, even if there was some lack of clarity about social work per se. About a third of the interviewees volunteered that they felt confident they had been able to make their role clear, especially two who had set up the palliative care social work service in their agencies. They considered they had worked hard to build up their reputation, helping the team understand what they could offer service users and their families. One of these two interviewees commented:

I feel that the role of social workers within the team is probably respected more and probably understood better than it used to. And that is something that has happened only over time.

This view was echoed by a focus group member who commented that to begin with in a multidisciplinary team 'you do have to actually prove your worth and fight for your position' and 'earn trust', but she believed she was seen as the 'expert' concerning child protection issues.

#### Referrals by nursing staff

There were a small number of less than favourable comments in the interviews about the ability of the nursing staff on in-patient units to pick up issues regarding children. Comments included doubts about the nurses' awareness and also their anxiety about certain situations: 'some nurses get a bit panicked by certain families to begin with'. One palliative care social worker had felt it necessary to remind staff at one in-patient unit to pass on information if there were children involved. Two interviewees described nursing staff who saw the patients as the sole focus of their work, and whilst they recognised the welfare of the family was important, did not consider children unless they appeared on the ward. The emotional impact of seeing the child seemed to trigger an emotional reaction that something should be done.

The characteristics that palliative care social workers used in the interviews to describe nurses working in the community (generally referred to as clinical nurse specialists, home care nurses or Macmillan nurses) varied from highly positive to very negative. At the positive end, just under half the interviewees gave praise for the initial assessments that usually included genograms as well as the patients' social circumstances. Trust was expressed in the nurses' assessment skills and there was a feeling that the clinical nurse specialists were family focused, picking things up from what parents had said or what they had observed during their visits. One palliative care worker particularly commented that nurses saw themselves as providing a holistic service not just a medical one. Examples were also given of where nurses would phone the palliative care social worker from outside a house for advice if they had immediate concerns or would later approach the palliative care social worker if they needed some help with family dynamics. There were also reports of nursing staff being 'very good' at family work but still needing some encouragement to undertake work with children.

At the negative end, concerns were expressed by a small minority that community nurses concentrated only on the needs of adults and, that the standard of assessment skills was poor, particularly where small children were concerned, showing a noticeable lack of knowledge about child development. One comment was that assessment could be 'a bit hit and miss'. Another worker explained that she sometimes looked at the files before nurses made their visits 'to see if I could pick up anything'. This appeared to be a checking up process and it would be interesting to speculate how the nurses viewed this action. There was an opinion that some nursing staff tended to link into the emotions of the child's mother and prior judgements about parents' capabilities might have been made based on where people lived, their backgrounds, and any mental health or dependency issues. There were reports too of nurses who had visited homes many times but never been able to see the children, although it was recognised that this was not always possible because of the timing of the visits during school hours. There was also the view that some nursing staff seemed to act as gate-keepers and filtered families away from the palliative care social worker.

#### Referrals from other sources

There were examples of referrals in the interviews from other ancillary staff within the workplace as well as from other agencies in the community. In one hospice, for example, a palliative care social worker described how she had received a referral from a hospice shop driver regarding a family to whom he had made a delivery. Mention was also made of referrals from staff in oncology departments, district nurses and general practitioners. One palliative care social worker based in a hospice covering a large rural area received phone referrals from a variety of sources which often led to the worker offering advice and support to workers and/or families over the phone rather than face to face. There were also instances where relatives or friends of service users made a referrals regarding their concerns about the care of children. Less common were referrals

from schools. Similarly, there was little mention of local authority social work staff making referrals to a palliative care social worker regarding children.

#### Self referrals

The number of service users referring themselves to the palliative care social worker was low. This reflects the findings in the Beresford et al. study (2007) where it was considered that this was possibly due to service users' ignorance about social work and that 'it had not occurred to them that there was such a thing as a palliative care social work service' (p.53). It was suggested by one interviewee in this study that those who did self-refer had heard about the social work service from someone who had already received help, or they already had a positive experience of social work. The literature about the specialist palliative care service made available to new service users was reported to describe the family care support or social work service but no mention was made during the study about self referrals from this source.

# **Reasons for referrals**

There were various circumstances that prompted a referral to a palliative care social worker. However, once a palliative care social worker engaged with a family, further areas of concern might well become apparent relating to the progression of the illness, ongoing assessment, the ability of the family to share their concerns, and/or the development of trust in the worker. Overall, reasons for referrals fell into four main categories which are described below and illustrated with examples from the survey and interviews.

# Advice and support to parents about informing children about a parent's illness and imminent death

There were parents who were uncertain about how and when to tell their children about what was happening and likely to happen and sought advice.

A clinical nurse referred a husband and wife who were asking for advice about how to tell their two primary school aged children that their father was dying. The parents were also concerned about the children's behaviour. *Interview* 

A clinical nurse was concerned about a family where the mother was dying and there were four young children in the family. The mother had been ill for some time, but the parents were unable to tell the children that their mother was now very ill and dying.

Interview

# The daily care of children during the period of the illness and beyond

This broad category applied to situations where there were concerns about parents' capabilities to care for their children that might result in them either being in need or at risk in some way. A decreased lack of ability may have been due to the nature and progress of the illness itself, including the effects of treatment; the emotional impact of the illness on the ill and/or surviving parent; poor parenting skills; domestic violence; or, a lack of understanding of the parents regarding their children's needs. There were children with learning difficulties or special needs who required particular care which parents were no longer so able to provide. Referrals were received regarding the inappropriate physical chastisement of children, parents with mental health issues and, parents who had alcohol or drug related problems. There were also requests for benefit advice due to financial difficulties impacting on the daily care of children.

A nurse referred a young mother with a terminal diagnosis who had three young children. The mother was terrified of the implications of dying and leaving a young family. The father was coping practically, but emotionally struggling. Survey A community nurse expressed concern about two small daughters of a patient, a single mother with cancer. There was evidence of very poor parenting skills, a chaotic home and use of non-prescribed drugs. Survey A doctor raised concerns that a mother who had a young son was arriving at the

A doctor raised concerns that a mother who had a young son was arriving at the hospice drunk. Survey

# Planning for the future care of children following a parent's death

This was particularly relevant in situations where a parent was a lone carer and there were no future care plans in place or, plans already made were being contested. Also included here are families whose immigration status was in question.

A single mother, who was terminally ill, expressed concern about the care of her daughter when she died. Her ex-partner did not have parental responsibility but wanted to maintain contact. The maternal grandmother did not want the ex-partner involved.

Survey

A community nurse specialist expressed concern about a single mother with a poor prognosis who had a son of 10 years who was autistic. The mother had a poor relationship with the child's father and the contact was not good. The nurse believed the mother was 'not facing issues' about the child's future care. *Survey* 

The home care service referred a lone parent with a short prognosis who had a 16 year old boy with a poor school attendance record. There was no involvement with the father and no other family members were available to help. *Interview* 

## Advice and support to parents regarding children's behaviour

Included here are situations of children refusing to attend school, being bullied at school, and being beyond their parents' control.

School and hospice staff expressed concerns about a 16 year old whose father was very ill. The relationship with the mother was 'very stormy'. The child was not attending school because of relationship issues with peers. *Survey* 

A team member identified that a 13 year old daughter was providing a lot of care for her father who had a long standing degenerative disease. The mother worked night shifts to finance the home. The daughter was unhappy and bullied at school. *Survey* 

# **Incidence of referrals**

The incidence of families with children under the age of 18 being referred to palliative care social workers varied widely. There was no mention made of any audit of these types of referrals and the numbers given were impressionistic and generally low: five or under a year. One interviewee described setting up a working group within the hospice to audit how much contact the various professionals in the team have with children in order to advise the management group about the need for appropriate resources and perhaps a specialist children's worker. Another, who reported low numbers, was surprised that the referral rate was not higher due to the increase in the number of younger patients with cancer. Another worker commented that her referral numbers of families with children had steadily increased to about 15 a year, which she too had linked to the changes in the pattern of illness, particularly a rise in the incidence of

breast cancer. There being a younger population in the catchment area was also considered to lead to a higher referral rate. Overall, the general view among those interviewed was that the number of families with children coming to the attention of palliative care social workers was increasing. However, there was a small minority within the survey who commented that it was rare to have referrals about how children were cared for and they were not able to give examples of any recent involvement. The number of incidents where palliative care social workers were involved with major concerns of neglect or abuse was described as being particularly small in proportion to the total number of people referred to them.

## Types of referrals

#### Formal referrals

Included here are those referrals made in writing, in multidisciplinary team meetings and by telephone. An example of a formal referral approach was described by a palliative care social worker attached to a day hospice which included a multidisciplinary assessment undertaken by a doctor, nurse, physiotherapist, complementary therapist, occupational therapist and chaplain working in pairs, and the palliative care social worker individually. The prospective patient and main carer were therefore interviewed four times during the course of an afternoon. A pro-forma was then completed leading to a comprehensive care plan. Feedback from both patients and staff was reported to be positive.

There was a variety of examples given in the interviews of written referral systems. In one hospice any worker could complete a referral card or pass it on to someone who wished to self-refer. The card asked for basic contact details and the reason for the referral, for example practical advice like housing, benefits or for an emotional or counselling assistance. These referral cards were dealt with as they came in, although

there was a formal allocation session once a week. One hospice had moved from a paper referral system to an electronic system whereby all new referrals came to the palliative care social worker.

One community based palliative care service team comprised nurses, occupational therapist and the palliative care social worker employed a duty system whereby each member of the team took it in turn to spend a day receiving phone referrals from various sources including general practitioners and hospitals. It was believed that this ensured a good quality of information on referral. Approximately 4-8 referrals were received daily. The referrals were allocated the next day at a team meeting according to the identified need and issues regarding complex family dynamics, children and benefits were passed to the palliative care social worker. One palliative care social worker based in a hospital visited the appropriate wards twice a week to discuss all the patients with the senior ward staff.

There was also an example of 'care review meetings' at an in-patient hospice (which had previously been called 'family meetings'). On admission, patients would be invited to nominate one or more family members to attend such a meeting to be held a few days later, the purpose being to discuss aspects of care important to the patient and family. These meetings were seen as low key and would normally be attended by the patient, a nurse, sometimes a doctor and facilitated by the palliative care social worker. If there were children under 18 years in the family, then issues about their care could be discussed.

Referrals from other health professionals external to the agency, such as district nurses, general practitioners and outreach workers, came mainly by phone.

#### Informal referrals

One palliative care social worker in a focus group described the informal approach used in a hospice:

Most of the work I pick up in this unit is when the nurses will say [to a patient] we have someone from the family support team do you want to say hello to? And I go along, or I meet people in the kitchen and I set up formal situations, but sometimes it is a bit tricky.

Here I understood 'tricky' to mean that not all people welcomed the worker making such a direct approach. Spending time on the wards and routinely meeting families that come through the hospice were recognised as positive ways to proactively engage with people but time was not always available for this to take place. This was easier within in-patient hospices or day hospices, less so in community services. There were examples of patients and families being referred to the palliative care social worker 'in passing' by a doctor or nurse. This apparent 'hit and miss' approach seemed to be an acceptable process. Informal meetings with the family of an in-patient were also described where a palliative care social worker and nurse together explained what the hospice could offer.

# Information received on referral by palliative care social worker

The quality of information received varied due to the formal or informal nature of the referral. Genograms appeared to be widely used by nursing staff during the admission process and were considered to be a helpful tool. However, their lack of completeness in terms of names and ages of children was occasionally criticised by palliative care social workers.

## Referrals post-bereavement

Whilst this was not a focus of the study, there was evidence of a more pro-active service being offered post-bereavement. Families were contacted and provided with information

about the counselling services available, which in some agencies included those specifically for children and young people.

## Palliative care social workers' views about the referral process

The general view was that palliative care social workers were clear about their role in the referral process. Their comments about the appropriateness and timing of referrals, as well as what might influence service users in accepting or refusing help, are analysed below.

#### Appropriateness of referrals

The widespread view was that referrals were generally appropriate and it was clear that families were not referred to palliative care social workers solely because there were children under 18 years. One worker explained:

I'm not going to push myself to get involved unless it is clear that there are needs.

Workloads varied due to the nature of the referral process. It was rare for palliative care social workers not to accept a referral and it appeared that there were no waiting lists in operation. In those agencies where there was more than one palliative care social worker, referrals were allocated at their own team meeting, or earlier if the referral was judged to be urgent.

# Timing of referrals

One worker commented on the difficulties where, for example, referrals were not made to the specialist palliative care service in 'good time'. This in turn tended to leave a referral to the palliative care social worker almost too late for work to be productively undertaken with the family. There were reports about nurses who, because of their ability to hold on to the 'difficult emotional stuff', tended not to seek advice until matters had come to a crisis point. This again could mean leaving a referral too late for the

palliative care social worker to help sort out child care concerns, especially legal matters such as guardianship.

#### Service users accepting or refusing help

There was a recognition that the stress parents were under might influence how they perceived the possible involvement of a social worker. It was the general experience of those interviewed and in the focus groups that many parents appeared unhappy, worried, anxious and fearful about the thought of seeing a social worker, and some 'reeled back in horror' at the suggestion of doing so. Often, parents were described as 'resistant', refusing any help offered by a social worker because they were anxious that, as a result, their children would be taken away. This resistance was reported to be particularly noticeable in families where there were current problems or a history of abuse or domestic violence. One palliative care social worker explained her way of managing such a situation:

I have a title 'Macmillan social worker' and sometimes I follow up by saying 'I haven't come to take the children away'. And I will do this quite light-heartedly because some people have had awful experiences.

A worker in a focus group gave her explanation for the reluctance to take up the offer of help:

The most complex and vulnerable families are the families that are the least able to take your services, not always because you are a social worker but because they are so overwhelmed with other issues in their lives.

Others considered that perhaps parents might be concerned about 'letting the floodgates open' or they were feeling that they were coping and doing the best for their children.

There were also comments on the issue of parents 'gate-keeping' their children's accessibility to services in relation to childhood bereavement because of their fear that

social workers may be disapproving of aspects of family life which might be revealed. However, if a palliative care social worker had concerns about a family situation, one route was to keep in touch with other team members who did have contact until such time as the family might be prepared to accept help.

In contrast, there were those palliative care social workers who believed that if they were 'taken along' by the nursing staff, introduced to the patient and then explained what they could offer, they were not seen as a threat and their services accepted. Indeed, one focus group member explained her experience:

I've seen parents who are relieved when we know things are wrong regarding the children and we have stepped in - they have not had the capacity to do anything. The relief is on the parents' faces when someone has had the courage to say - 'I'm concerned about your children'.

There were also those who considered that their services were often 'sold' to the family by nurses in particular who would say to family members: 'Our social workers are different', compared, presumably, to the negative public image of local authority social workers.

## Use of the title 'social worker'

There was a difference of opinion between palliative care social workers in the focus groups about how they should describe themselves to family members, particularly at the point of referral. There was a recognition that although it may be a barrier, referring to themselves as a 'social worker' was a more honest approach that defined their profession. For example, one palliative care social worker commented:

I do stick resolutely to the title of social worker. We are not going to be ashamed of it. We will work through that prejudice and it is often the nurses who are introducing the idea of myself to families and they do struggle and they say 'can't we call you something else?'. And I say 'no, that's what we are and it's deceitful to go in on any other platform.

In saying this, the palliative care social worker demonstrated a very strong sense of pride in her profession. Another supporting view was that if the title social worker was not being used it was somehow 'fudging' the issue, misleading people and denying their social work profession. One worker explained that when using the title 'social worker', she would make it clear that she was employed by the hospice rather than the local authority because in doing so 'they trust you that bit more'. Another described the dangers of being 'too pally' with the family and the need to define the boundaries, a view supported by others.

In contrast, there were workers who, because of the many negative first reactions of families to the title 'social worker' and the refusal to take up services, had changed their title to 'family support worker' finding that this was a more neutral term that appeared to relax family members. However, parents had on occasion been found to see this title as threatening too. One person described how they might 'emphasise the counsellor bit and sometimes emphasise the social worker bit' or 'play down the social worker bit' depending on the circumstances. Overall, across all three focus groups there was a mix of titles used, with the title 'social worker' being used by about two thirds. In the interviews two of the seventeen had job titles of 'family support worker' and in the survey approximately 90% had social worker in their job title. However, one interviewee whose work title was 'social worker' explained that she introduced herself to families as a 'family support worker' and another whose title was a 'family support worker' who made clear she was a social worker when contacting outside agencies believing that this made her status more clear.

# Discussion

There is minimal research regarding referrals to palliative care social workers in general (Beresford et al. 2007). This study highlights the complexity of such processes where

there are concerns about children and the influence of, and the dependence on professional values, personal beliefs, intuition and knowledge, as in child protection generally (Dent and Cocker 2005:153). The findings in this study show that palliative care social workers are very dependent on others, mainly health professionals, for referrals to their service. This gatekeeping role was also identified by Beresford et al. (2007) in their study of palliative care service users. They also found that social work support was offered in an inconsistent and non-systematic way before a person died compared to a more systematic and proactive way post-bereavement. Whilst being dependent on others for referrals is true for social workers generally, this study highlights particular issues for palliative care social workers.

Record keeping itself was not a specific topic in the interviews. There was no mention of an audit of referrals regarding child care concerns or the involvement of palliative care social workers with families with children. It is therefore difficult to estimate with any accuracy the amount of involvement that occurs. There was also no mention of numbers of users of the specialist palliative care service with children to know the proportion of those referred to palliative care social workers. Overall, there was no evidence of social work support being offered consistently or systematically, but neither was there evidence to suggest that workers turned down the referrals that were made to them.

Palliative care social workers are part of a system: the change agent system (Pincus and Minahan 1973:54). As already stated, whilst this study did not explicitly examine the workings of palliative care teams, it became apparent that how they function can impact on the referral process. The differences between multidisciplinary and inter-disciplinary teams were explored in chapter two. The findings here suggest how the role of the palliative care social worker may be compromised by the dominance of the medical approaches to palliative care and to a limited understanding of the social work role within the team as suggested by various authors (Oliviere 2001; Firth 2003; Raymer and

Reese 2004; Beresford et al. 2007). Whilst the differences in professional cultures and values impacting on inter-professional working described by Fitzsimmons and White (1997) were evident in the current study, there were also descriptions of workers displaying 'professional adulthood' (Laidler 1991 cited in Molyneux 2001:33). Palliative care workers in Beder's study (2006:53) described themselves as the 'emotional arm' of the team. The findings here showed that workers who regularly raised questions in the team about the presence of children saw it not only as their duty but also as a way of helping the other professionals to recognise the needs of children. In these situations, palliative care social workers spoke of being sufficiently confident in their own roles and professional identity to work productively with other team members, sharing ideas and expertise, particularly in situations that involved children. They were therefore viewing their own agency as targets of change and trying to change the system from the inside. They are not, however, able to stand apart from their own system and have to recognise that their ongoing relationships with colleagues and status may affect, and be affected by, their efforts to bring about change.

In Sheldon's (2000) opinion, working with people who are dying can bring emotional distress and anxiety that has been described as 'hard work' for all the people involved, including professionals (James 1993). Adding the dimension of children to this already stressful situation could be experienced as an extra burden to the staff involved. One way perhaps for staff to deal with this would be to focus on the patient rather than consider who else may be affected within the family. However, this approach would be contrary to the ethos of palliative care, which is clear about the place of the family within the care umbrella (WHO 1990), and does not appear to recognise the concept of family systems. Nevertheless, how each palliative care team interprets the 'total care' ethos can be expected to influence how the individual professionals practise. The findings here reflect a range of interpretations and practice in the referral process.

Confidentiality as a barrier to sharing information within the team (Egnew 2004) was not raised as an issue in this study. This is a little surprising in comparison with other adult services, for example in drug services (Taylor and Kroll 2004), but perhaps the holistic approach of palliative care allows for a sharing of information.

Of all the staff involved, nurses were the main source of referrals regarding child care concerns because of their initial and ongoing contacts with the patients, both in the hospice and in the community. Nursing staff, therefore carry a responsibility for ensuring that children do not lose out on appropriate support at a critical time in their lives. Their ability to look beyond the patient and pick up issues about children may be dependent on several factors: awareness, knowledge and confidence. A study conducted by Heaven and Maguire (1997) found that hospice nurses failed to pick up or register correctly both emotional concerns and those concerns that were important to the patient. It is perhaps more understandable in the hospice setting for nurses to be less aware of children where it might be a question of 'out of sight, out of mind'. However, for those in the community, it would be reasonable to expect that there would be tangible reminders of children in the home, either through their direct presence or through toys and photos. There was evidence in this study that nurses were 'child aware'. The completion of a genogram, something that appeared to be a regular part of the assessment process (to be discussed in chapter nine), can raise awareness of the existence of children and used as a starting point for discussion with the patient (Oliviere et al. 1998). Research would suggest however, that medical practitioners do not always feel equipped or competent to engage in conversations with children themselves nor with the parents about how they might communicate with their children (Turner et al. 2007; Fearnley 2010).

In order to work productively together, professionals should understand not only their own role but that of the other professionals involved. Barriers to making full use of the

palliative care social worker by medical and nursing staff were discussed in chapter three (Reese and Sontag 2001). The debate around the title of social worker is interesting in this context. There is evidence from other studies that the social work role is not always well understood by health professionals (Bywaters 1991; Davies and Connelly 1995; Bliss 1998; Bliss et al. 2000). However, these studies were generally considering professionals who worked for different agencies such as district nurses and social workers based in social service departments. Here the opposite was true; the professionals were working for the same agency and yet there were still difficulties. This lack of understanding might lead to some professionals not always being helpful or knowledgeable about referring people on to a palliative care social worker. There may be 'competition' between nurses and social workers; nurses believing that they can take on the psychosocial roles and social workers perceiving that they were taking over their duties (Reese and Sontag 2001; Hudson 2002). There was evidence in this study that rather than being on equal terms palliative care social workers believed that they were perceived as an 'internal consultant' or 'resource' (Monroe 1998; O'Connor et al. 2006). Nurses, especially those in the community, were seen to have overlapping competences and shared responsibilities, sometimes welcomed though sometimes not. Such overlaps are referred to as 'role blurring' (Falk 1977 cited by Hall 2005:192; Sheldon 2000; Reese and Sontag 2001) which was evident here; this may be positive where the focus is rather on the needs of the patient and the family (O'Connor et al. 2006).

The value that palliative care social workers themselves place on the significance of safeguarding the needs of children can also be expected to affect whether or how they influence and encourage team members to raise concerns. In addition, how much confidence palliative care social workers have in their own level of knowledge and expertise in working with and for children may also have a bearing on their

encouragement of such referrals in this area. These aspects will be discussed in future chapters.

# Conclusion

This chapter has examined the issues relating to the referral of families to the palliative care social worker where there are concerns about the children. The findings revealed a complex process, frequently affected by professional roles and team functioning which has not previously been addressed in the literature. Nursing staff were the main source of referrals and although the numbers fluctuated they were generally low. However, just because there is a child in the family, it should not be assumed that there is a need for a social worker to be involved. What is perhaps required is for professionals to be alert to the needs of children and for there to be an enhanced understanding of the role and expertise of the palliative care social worker so that families and children who do require support do not 'fall through the net'. There is evidence in this study that palliative care social workers do take it upon themselves to remind other team members to be 'child aware'. The next chapter will examine the process of the palliative care social workers' involvement once a referral has been made.

# The role of the palliative care social worker: the work they undertake with children and their parents

Far more people are affected by disease than are infected by it. Anderson 1995

# Introduction

In the previous two chapters I explored how palliative care social workers perceived their role in safeguarding children and how children came to their attention. The first part of this chapter explores how workers viewed their engagement with children and their parents and identifies the key themes of the work they described in the interviews, focus groups and survey. The second part illustrates the process of their involvement through the presentation of five practice examples taken from the individual interviews. These examples illustrate the key themes as well as the breadth and depth of their work and highlight some of the dilemmas and tensions that it brings. The third part contains an analysis and discussion of the process of the work undertaken by palliative care social workers with children and their parents. It also focuses on the practice approaches to this work as well as on the organisational constraints, ending with a section on how the immigration status of families can impact on families and the work.

## Engagement with children and parents

The palliative care social workers considered that the starting point for their work should always be with the parents. There was a predominant view that it was important to support the parents, working with them first rather than immediately becoming involved with the children, if indeed at all. It was recognised however, that parents, in dealing with their own grief, could be struggling to help children cope with theirs. Finding out about the parents' awareness of the children's needs, how much the children knew, what they might be experiencing, any changes in their behaviour and, what supports

were already in place were all considered to be part of the palliative care social worker's initial assessment of the situation regarding their children.

As discussed in chapter seven, this capacity 'to juggle different needs and vulnerabilities' was widely considered to be an important aspect of the work. This meant that when working with the family as a whole there was a balance to be struck when considering the needs of the individuals within the family whilst, at the same time, respecting the wishes of the dying parent. This was considered to be particularly evident in circumstances where either a child did not know about the illness, or knew about the illness but not its implications. One worker ruefully commented that she knew 'she will only get one shot at it' when visiting some families, as it would be too painful for them to ask her back.

## Involvement with children

As discussed in chapter four there has been a growing emphasis in policy documents, literature and research regarding the need to hear the voice of the child. Whilst many palliative care social workers said obtaining children's views was important, there was a limited amount of reported direct work undertaken by palliative care social workers with children and young people. This came as a surprise, as it did to Beresford et al. (2007) in their study because, over the years, palliative care social workers have been reported as developing services for children (Monroe 1998; Oliviere et al. 1998; Chowns 2006). The amount of direct work seemed to depend on agency policy, general circumstances and the preference and confidence of the worker, as well as the willingness of the parents and children for the work to take place.

There were barriers to direct work being undertaken. In one hospice, the worker felt that she was not encouraged to work with children; in another, the worker was told that she was a social worker for adults and should not undertake work with children. A few

workers commented that they tended not to even see children unless they came to visit their parent in the hospital or hospice during the day and that there was little opportunity to meet children of day patients unless a home visit was specifically made. Communitybased workers were generally only able to see school children if home visits were made during holidays or after school hours. No one claimed that they were uncomfortable about working with children, although there were some who stated they preferred certain age groups. It could be that workers did not wish to become too involved because of the emotional distress it might cause them, preferring to only work with the parents although this was not stated directly.

One worker commented that although children were not her 'special interest', she was happy to work with the family but would pass any direct work with children to the children's worker in the hospice. Another worker, based in a hospital, who was not so confident about work in this area, referred a father to a palliative care social worker in a nearby hospice for advice about how to tell his children that their mother was dying.

Most of the work with children would best be described as informal; for example, seeing children, particularly young ones, when visiting the family home or showing children round the hospice. Observing a child in their home allows a worker to see that child in context, providing the opportunity 'to enter the child's world' (Helm 2010:56) and there were many examples given in the interviews and focus groups where observation had been an important part of the worker's assessment of the situation. One practice example shared in a focus group is a clear illustration of an observation process:

I think I would be looking at the way the children in the family are behaving and picking up clues about their behaviour to see if something is wrong and also observing interactions between parents and children. I can think of one case ... where it's the grandmother who is ill. ... She has responsibility for caring for her two grandchildren ... and she is having tremendous problems with the behaviour of the younger child, and those problems seem to be around the lack of her

parenting skills. And we have come to the conclusion there are problems there just by observing the interactions between the grandmother and the child. In the opinion of Sidebotham and Weeks (2010:96) all professionals who have contact with parents are in such a position to comment on their observations of parents' behaviour and interaction with a child. This should include positive as well as negative behaviour or interaction (Glaser 2002).

More formal engagement with children ranged from a single meeting to sustained work on an individual basis as well as joint work with their parents. There were examples of workers giving advice to children, advocating on their behalf, assisting them to reengage with school, as well as helping children talk about their feelings. Workers stressed that it was important for parental agreement to be given before they became involved in any direct work with children. One worker remarked:

I tend to prefer to work with the family rather than an individual child anyway because it tends to be about the family rather than the individual child.

A minority of palliative care social workers said they preferred to work indirectly with the children, through the parents. As one worker explained:

I don't dive into direct work immediately (with children). I don't reach for the crayons and the paper. I reach for the adults in the situation. That is who the child knows best, that is the people who should be communicating with that child. The adult closest to that child should be doing that work and I will do it if they can't or won't or along side but empowering them is where I am at.

Another echoed this by saying:

I feel strongly that you support the parent - no good doing games and interesting things with the child if the mum is sitting in the corner completely depressed because that is going to be the issue, that is going to make the difference for that young person.

Working with children was described by one interviewee as: 'hard work and time consuming work', referring in particular to the time it took to get to know the children. Another worker who welcomed the opportunity to do so, summed up her thoughts about working with children by saying:

I think children are amazing - when you do get to talk to them their understanding surprises us and their ability to cope surprises you as well.

However, as stated earlier it is not suggested that every child of a dying parent will require professional help and support (Worden 1996) or is at risk of significant harm. Indeed, legislation emphasises that the rights and responsibilities of parents should be respected. What perhaps is required is an understanding about the lived experience of the children in the family, so that workers can feel confident that every assistance available has been offered. Palliative care social workers therefore should be alert to the cues and clues provided by parents, by the children themselves if seen, and by the environment in which they live, as well as professionals who have had access to the family. As Helm (2010:56) points out:

It is not the child's responsibility to communicate their needs; it is the practitioner's duty to listen and use the means of communication and engagement that best allow the individual child to express their lived experience and needs.

#### Continued work post-bereavement

This study focuses on the work of the palliative care social worker prior to the death of a parent, but there were examples of practice where engagement, initiated before the death, continued afterwards. Sometimes the young person initiated contact, on occasion after a gap of several years. This was welcomed by the workers and was not seen as a dependency issue. Indeed, it highlights the value the children had placed on their previous contact and the importance of consistency and availability. However, whilst some palliative care social workers reported that it was possible for them to engage with

children directly prior to the death of a parent, not all were encouraged or able to do so afterwards due to agency policy. Rather, the workers' role was perceived as offering support to the surviving parent and other adult family members in order for them in turn to support the child in their grief. However, parents might not be in a position to do this due to a previously poor, neglectful or abusive relationship or they might be incapable for other reasons. Such support tended to be offered on a one-to-one basis and available for a period of up to one year, although some workers were restricted to a lesser period of weeks or months after the death. In some hospices there was a dedicated children's worker or specialist bereavement counsellors who could take on post-bereavement work with children. Children might be invited to join group activities with other bereaved children and in 'Remembrance Days'. Workers' views about organisational constraints on engaging in this work will be examined later in the chapter.

## Key features of the work with children and their parents

Weaving through the descriptions of the work undertaken by the palliative care social workers with children and their parents were the following five key themes.

- 1 Children need to know what is happening
- 2 Planning for the future care needs of children is essential
- 3 Practical support is crucial
- 4 A family does not exist in a vacuum
- 5 Joint work with others can be beneficial.

#### 1 Children need to know what is happening

An underlying premise to the work as described by the respondents was that children whose parents are dying should be kept informed about what is happening; a view widely supported in literature (Christ 2000; Chowns 2006; Christ and Christ 2006; Kennedy 2008). The general view among the respondents was that it was better if the

news about the illness and its expected outcome came from the parents themselves. However, if parents felt unable to do so, and studies show that parents do struggle to do so (Houldin et al. 2006; Kennedy 2008), then palliative care social workers explained that they would either be present to support the parents or, more rarely, would do the explaining in the parent's presence. Those workers who had been asked to tell a child about a death or impending death of a parent spoke of the emotional challenge involved.

The literature points to two main reasons why parents do not tell children: their belief that children are not aware of what is happening or are too young to understand (Beale et al. 2004; Christ and Christ 2006). This was supported by the respondents in this study. One interviewee described how she tried to encourage the parents to tell their children about their mother's illness and prognosis:

They kind of accepted that the older boys needed to know but they just thought it was hilarious that I would suggest that the three year old should. They actually laughed out loud.

Workers recognised that for a parent to tell a child that they were dying was not an easy task and one described it as 'the most major thing for them to do as parents'. They saw themselves as facilitating these difficult conversations by finding out from parents what they believed the children already knew and understood, as well as the reasons for their reluctance to share information with the children. There were many reports from the participants of parents who were often unsure about how to explain what was happening to their children and therefore needed guidance. Both Barnes et al. (1998) and Turner et al. (2007) highlight these difficulties, particularly where a child has learning needs and behavioural problems. Palliative care social workers provided leaflets and other written material, including activity books and directed parents to relevant information on the internet. In an example where there had been a late referral

to the hospice, the children had been told that their mother was ill but not that she was dying. The palliative care social worker commented:

...and this is very often the way. At every point up until the last treatment was stopped and she knew she was going to die, I think the family said things like 'the doctors are doing what they can'. You know those adult truths but they are avoiding the central truth which is that we all know that mum is going to die. Parents don't intend to put their children under this - they are trying to protect them from the awful news.

## 2 Planning for the future care needs of children is essential

There were children whose future care after the death of a parent, usually the mother, was uncertain, not known, not considered safe or, in dispute. These situations were usually where the mother was a lone carer. Palliative care social workers encouraged these parents to discuss the options in order to help them come to a decision, ensuring where necessary that it was legally secured. This might involve the palliative care social worker in negotiations with estranged parents. In one family, the worker, at the request of the mother who was dying, contacted the father who had not seen his children for two years. After many discussions he contacted the family, resumed contact and took over the care of the children when their mother died.

Palliative care social workers, however, may not always understand the importance of such planning or lack the confidence to engage in it. Not every palliative care social worker wished to initiate a discussion with parents about the future plans for their children. One worker stated:

I think in a way I wait and see if people begin to drop clues about that [plans for children after parent dies] rather than being too proactive.

However, not engaging in such discussions could cause difficulties. It might be the case that a parent was putting off making a difficult decision but postponing it could cause uncertainty for the child and lead to a lack of preparation for all parties concerned and

there were examples where this had occurred. Nurses who had been working with the family might leave a referral to the palliative care social worker regarding legal matters too late for wills to be discussed and completed, particularly in relation to the future care of children. A worker wrote:

At times my colleagues and I feel a great sense of <u>urgency</u> [respondent's emphasis], especially re future planning/guardianship, as we are keenly aware of short prognosis for some clients, yet a significant number are not able to make plans.

## 3 Practical support is crucial

A further significant aspect of the work concerned the family's living circumstances. Many families were described as living in poor accommodation and struggling financially, which could lead to their sense of security becoming even more threatened (Walsh 2005). Parents who were no longer able to remain in employment due to their illness or caring responsibilities often had extra financial burdens as a result of treatment. Mcleod and Bywaters (1999) draw attention to the health inequalities and how already marginalised groups face extra difficulties in the event of ill health. Palliative care social workers recognised that these difficulties could inhibit a family from engaging with the more emotional aspects of care resulting from the illness. This in turn could impact on the parents' care of the children. One worker summed it up by saying:

I am a firm believer that you can't get anyone to sit down to talk about their feelings if they have no roof over their head or no food.

Children's lives are not only affected by the immediate financial position of the family but also by other resources to which the family has access such as housing and community resources (Jack and Gill 2010).

Palliative care social workers generally do not hold budgets and therefore could not offer direct financial support but were able to apply for funding from relevant charities.

They also offered advice and practical support which included assistance related to care packages, housing, benefits, education and immigration status. A worker described her role as:

A Jack of all trades - filling in a form one minute and doing intensive therapeutic counselling the next.

## 4 A family does not exist in a vacuum

Chowns (2006) and Rolland (2005) observed that children and their parents were not only part of a family system but other systems too, and these may impact on the way they cope during the period of illness and post bereavement. This is evidenced here through the workers' explanation of the process of individual pieces of work or by direct general reference, as one interviewee commented:

It is very much looking at individuals as part of a system and the roles that they play, which is why I wouldn't put a false demarcation around family members - they are all part of the patient's system - and that would include the school, the church, just that sense of network.

Palliative care social workers helped parents to negotiate the health and welfare systems upon which they depended. Indeed, workers saw liaison with other agencies to be an important part of their task. Within the family system, other relatives may be involved in the current and future care of the children as well as family friends. These relationships were generally positive, but on occasions were seen to be unsuitable for children or causing difficulties. Mention was also made of a family's cultural heritage and religion having an influence on family functioning as well as creating challenges for the worker in finding appropriate resources to meet the needs of parents and children. There were also examples of families who faced particular difficulties in ongoing negotiations with and appeals to the Home Office because of their immigration status and these will be explored later.

### 5 Joint work with others can be beneficial

Palliative care social workers provided many examples of productive joint work where they and others became the action system to achieve the goals (Pincus and Minahan 1973). A common area was work in conjunction with other members of the multidisciplinary team. Such work included visits with nurses to families in the community, the running of family meetings together with medical and/or nursing staff, participating in a joint assessment process and, participating in case discussions at the team meetings. These activities allowed team members to draw on each other's knowledge and skills and to keep a wider perspective. The occasions when children were invited to live in the hospice enabling them to spend more time with their dying parent was very much a combined venture for all the staff in the hospice. There were examples too of successful joint work with local authority child and family social workers and education staff that enabled children's needs to be met in a satisfactory way. The relationship between palliative care social workers and local authority child and family teams will be explored in chapter ten.

## **Practice examples**

In chapter six, I illustrated the children's experiences of living with a terminally ill parent with brief scenarios from the workers' stories. Here, in order to illustrate the key features of the work undertaken by palliative care social workers with children and their families such as those described in that earlier chapter, I have chosen to present five practice examples. However, selecting five out of the forty five was not an easy task because all the stories were important in different ways. The examples here are selected because they include children in differing circumstances, illustrate different approaches and a range of practice. Several involve direct work with children, which is slightly unrepresentative as just under half of all the practice examples involved working with children directly. They also highlight some of the complexities, dilemmas and tensions in

the work. I will explain why each has been chosen before giving an account of the work undertaken as described to me by the palliative care social worker during the interview. Names have not been used and details have been removed or altered to ensure anonymity and confidentiality. Care has been taken to ensure that this does not detract from the process as described.

## Family A

This example was chosen because it exemplifies a palliative care social worker taking a family systems approach, gaining an understanding of the boundaries and hierarchies (Mehta et al. 2009) by working with parents, with children and parents together, and also with children on their own. The palliative care social worker here interpreted her role to be a supporter to the parents, listening to their concerns about their children, observing the children's behaviour, and forming and carrying out a plan of work that involved herself and other professionals. It is an example of contact maintained over a longer period of time than perhaps is usually the case and of a palliative care social worker exhibiting confidence in her work with children.

A community nurse specialist had been visiting a family where the father had cancer and had been undergoing treatment. He lived with his wife and two primary aged children. The parents had asked for advice and guidance about what to tell the children and so the nurse passed the request on to the palliative care social worker.

The social worker's initial impression of the children was that they were polite but exhibited hyperactive behaviour which the worker described as chaotic. The father explained that he could be aggressive at times due to a reaction to his medication. Both parents were very worried about the children and felt that they could not do

anything to manage their behaviour and consequently had lost control of the situation.

During her visits, the social worker set up a pattern of seeing the couple first, then the family as a whole and then the children individually or together and finally back to the family. Her real concern was the younger child who was 'acting out', being very confrontational, trying to shock by use of language and outrageous behaviour. The social worker was very worried about the child's mental health. She asked the parents' permission to contact the psychological services, which was given, and the social worker prepared reports. The child was offered help and the social worker continued to meet with the family. After several months, the child had calmed down and was performing well in school.

The social worker continued to work with the family, reducing her visits to about every 5-6 weeks. Family members were now more able to talk to each other and managing to cope with their problems. Both children had responded well to the social worker throughout her contact with them and they very much wanted her to continue seeing them.

When reviewing her work with the family, the social worker explained:

I was feeling a bit overwhelmed in there at one point and was really glad that [the child] had somewhere else to go. ... I know I was the catalyst for getting that moved so much quicker.

## Family B

This example shows a palliative care social worker involved in the future planning for a young boy normally cared for by his grandparents. The palliative care social worker

engaged with the adults involved both individually and as a group in a family meeting. The example illustrates collaborative working, on this occasion with a doctor who was part of the multidisciplinary team. It shows how a family's previous negative experience of social work can act as a barrier to engagement with a palliative care social worker, and how engaging in memory work can ensure that a child receives a more rounded family history.

A grandmother, the main carer for her grandson, aged 9 years, was a hospice inpatient. The ward staff had referred the grandmother to the social worker because they were aware that she was concerned about the future well-being of her grandson. When the social worker began to talk to the grandmother it transpired that there had been many past difficulties. He was four years old when he was moved from his mother because of her lack of care. The mother had since died and the whereabouts of the father were not known. The local authority were involved and did not want the boy to stay with the grandfather but to live with an aunt and uncle, and for this to happen before the grandmother died. The timing of this planned move had made the family angry and distressed.

The social worker first had to gain the trust of the grandparents because they associated her with the local authority. The social worker considered it was not her place to tell the grandparents what to do but to encourage them to look at the issues. The grandmother was concerned that the family would not in future talk about the child's mother in her absence. The grandmother explained to the social worker about her wants for her grandson and her concerns about her husband. She agreed her husband was not able to look after their grandson.

The social worker described how both she and the doctor spent time with the grandmother talking over her concerns. The social worker helped the grandmother make memory boxes for her grandson that included material about the boy's mother. The idea of holding a family meeting at the hospice was raised at a multidisciplinary team meeting. It was proposed that the social worker and the doctor would facilitate the meeting to help the grandmother talk about her concerns. The grandmother was in favour of this idea. Both grandparents and the aunt and uncle attended. The grandson had not been invited but knew the meeting was being held and the basis of what was being discussed. The social worker had briefly met the grandson and was aware that he too had negative views about social workers.

At the meeting, it was agreed that the grandmother should return home, the grandson spend increased hours with the aunt and uncle and the grandfather collect him from school. Social services were to be informed of the plan. However, the grandmother's health deteriorated very quickly and she died the following day. The social worker remained in contact with the family. The grandson had moved to live with the aunt and uncle, remained in the same primary school and had daily contact with his grandfather who continued to play a major part in his grandson's life.

When I asked the social worker what she thought of the outcome she said:

... disappointment - in that we were not able to facilitate the plan we had put in place ... I think it was probably successful and I think the family meeting had joined together not a fractured family but a family where there were issues ... so there was stability there for the child. That was our main concern.

## Family C

This example is again one where direct work was undertaken with the children before their mother died. It illustrates how the palliative care social worker, restricted by agency

policy regarding 'post-bereavement' support to children, chose to be flexible in her approach. The emphasis here was very much on the needs of children in a family where there has been a history of domestic violence.

The social worker first met the family when the mother was dying at home. She had been asked to visit by the community nurse because of child protection concerns. There were two teenagers living at home. A single older son lived nearby. The family had been known to social services because of a history of domestic violence.

Shortly after the initial contact, the mother was admitted to the hospice as an inpatient. The social worker came to know the children quite well and talked to them about their relationship with their parents. The youngest was particularly close to her mother but hated her father because of his violent behaviour.

In the light of this, the social worker met regularly with the father trying to balance her own working relationship with him and with the children. The youngest child very much resented this. The mother had previously refused to believe how ill she was and had not made provision for the children and there were no relatives who could help. The mother became very withdrawn and when she died the children were very distraught. They invited the social worker to attend the funeral, something she did not do on a regular basis, but did on this occasion to support them. As the hospice did not offer bereavement support to children under the age of 18 years, the social worker was unable to continue supporting them directly; something she felt very unhappy about particularly in this instance. However, she was able to provide bereavement support to the father and chose to visit him at his home so that she was better able to monitor the situation.

A little while after, the social worker received a distraught phone call from the older child who said his father had punched him and he wanted to run away from home. The social worker contacted the local authority duty team and also the older son who agreed to take his brother for the night. Shortly afterwards, he returned home saying he did not have much choice but it was better than being in care. The social worker again had to explain to the children that she was no longer able to work with them and liaised with the local authority social worker.

Some time later, the younger child called the social worker to say that her father had physically hurt her. The social worker contacted social services and asked them to follow this through. According to the procedures of the local authority concerned, the social worker should have been kept informed of the outcome; she felt let down when this did not happen.

There was further contact when the social worker later received a phone call from the younger child's school. Apparently the child was in a distressed state, had become quite depressed and did not want to go home. Once again, regretfully the social worker said, she had to direct them to the local authority and reflected:

I find that a dilemma sometimes, getting children to work with me closely, to trust me, knowing that I can only take things so far.

## Family D

This example highlights how a late referral to palliative care can create difficulties in making future care plans for children. In this situation the parent was a lone carer and the family's situation illustrates the complexities of negotiating and liaising with various local authority services to secure supportive services for the children. It also shows the

importance of considering the children's views as well as those of the parent and demonstrates how a family's resilience can be enhanced once support is provided.

A lone mother had been receiving treatment for cancer. Her condition had deteriorated and the hospital had consequently referred her to specialist palliative care services. The community nurses visited and discovered that the mother was very ill indeed and in their opinion had only a few weeks to live. The family consisted of a teenage daughter who had not attended school for several months having stayed at home to look after her mother. An older son was still at school. The nurses were so concerned that the family was not prepared for what was happening that they asked the palliative care social worker to visit with them.

The social worker first ensured that all the necessary practical arrangements including benefits were in place. As the social worker came to know the family she learnt that the mother had become withdrawn over the past few years and had left the children to their own devices. The worker also realised that there was no father involved or any other family member who could help. The son appeared to be mature for his age, but she became particularly concerned about the daughter's future care.

The doctor explained to the mother that she was terminally ill and had only a very short time to live. Once the son was made aware of this he said that he wished to look after his sister. The social worker talked to the mother about her wishes for her children. The mother said that she also wanted her son to take responsibility for his sister. The mother had a few weeks to spend time with the children and to write down her wishes for them. The mother died a few weeks after the social worker became involved.

After many phone calls the social worker managed to involve social services. Education services organised specialist remedial input for the daughter. She attended school regularly and made good progress. Her brother continued to attend school and became the guardian of his sister with social services monitoring the situation. The palliative care social worker ensured that he received all the benefits available and negotiated with the housing department that he take over the tenancy at age 18 with the proviso that social services continue to monitor the situation in the meantime. 'Children in need meetings' were held after the mother's death, which the social worker attended together with the children. Bereavement support was provided for both children.

When asked what she now thought about the situation, the social worker explained: The son is so amazing, I have a lot of admiration for him and so determined from the beginning. I didn't know the mum that well and the kids, I think they are wonderful, just fantastic.

## Family E

I selected this example because it illustrates a family coping with the threat of having to leave the UK once a parent has died. It highlights the importance of rituals surrounding a death and how a hospice was able to meet the family's requirements with the support of the palliative care social worker. There is evidence too of collaborative working among the hospice staff when the children were invited to stay in the hospice with their mother.

A terminally ill mother was living with her husband and two mid-teenage daughters. They had come to the UK seeking asylum and had been in the country for several years. The Home Office had refused their asylum application and as the appeal

procedure had been exhausted the family were to return to their native country. However, because of the mother's terminal illness, the Home Office had agreed that the family could remain until after her death.

When the social worker first visited the family with the community nurse, she was unsure how much the girls knew about their mother's illness. They encouraged the mother to tell her daughters but the mother refused. The girls were able to understand English but the social worker needed to use the services of a translator when talking to their mother. The social worker again explained to the mother that the girls needed to know and to have the opportunity to talk things through. The mother agreed, and when told, the children said that they had known all along because they had been reading the letters to their mother sent from the hospital.

The social worker visited the family regularly. The girls were attending school and enjoyed doing so. The social worker described the father as being very quiet; the children described him as a loving father. The mother's condition deteriorated and she was admitted to the hospice which, being some distance from the family home, meant the girls were not able to visit very often. When their mother was close to dying, the girls were invited to stay in the hospice. The social worker helped them to prepare memory boxes and the mother wrote letters for the girls. In addition, the children talked to the social worker about how they wanted to stay in the UK as they had come to see it as 'their country'. They said there were scared about returning to their relatives who had not been supportive to the family, although they were looking forward to seeing their grandparents.

The social worker linked with the school. The girls' friends came to visit them at the hospice. The social worker spent time with the girls talking and understanding about the cultural needs and requirements surrounding death. The social worker contacted

the Home Office asking for information about how long the girls could remain in the UK after their mother's death so that they could plan some bereavement work. The Home Office could not give a date and explained that decisions were made on a daily basis. The girls also wanted to write to the Home Office, asking for help from the social worker, which was given.

The mother died and the girls helped with the rituals involved in the preparation of her body. The social worker talked regularly to the girls on the phone and when she was unable to get in touch with them anymore assumed that they had left the country. She had attempted to find out if there were any resources or services that the girls could be referred to in their country of origin, but found it impossible to do so. About a month or two later she received a letter from the girls saying that they were all right and there was no need to worry about them.

When asked, the social worker explained how she had felt about her work with this family:

I feel good about it because I think we achieved a lot. It felt like we could do work and we could prepare and the children were very involved and we could do things according to what their needs were. If the Home Office could just give us an indication, if we had a week or two in terms of planning. They could not bend the system and I think that is the frustrating part. With death and dying, life goes on without realising the impact and the additional needs for children at that time just kind of seems it is not being acknowledged.

## Process

What came through all the narratives concerning work with children and their parents was the importance of process. Process can be seen as 'a systematic series of actions directed toward some purpose' (Pincus and Minahan 1973:85). Process and purpose according to Pincus and Minahan are related concepts. To understand the process it is

important to know what determines the worker's purpose and how and why purposes change. The start of the process is the referral and initial engagement but what directs it subsequently is not only the relationships developed and the assessments undertaken, but the illness itself.

#### Impact of the illness on time available

Life expectation is not predictable, and for some their condition may worsen and they die sooner than was expected. This may mean that plans for the future care of their children have not been made or finalised or the wishes of the parent are not known, as in Family D. There were other examples in the study where this was the case. In one family, the mother unexpectedly died two weeks after the referral leaving a complicated family situation with no time for any assessment or planning which involved the mother. Palliative care social workers recognised that time was needed to assess the potential strengths of the family and its ability to change (Jackson 2000b). Oliviere et al. (1998:25) are also of the opinion that it is important to take time when assessing to ensure that 'one is not drawn into the strong emotion surrounding the ill person'. Time, however, was something that palliative care social workers frequently mentioned as a rare commodity in their work subsequently hampering their assessment of a family situation. This was graphically described by one worker:

I think that one of the things that I notice about working in this environment, you have to think on your feet a lot of the time because you don't have the luxury of being able to do a thorough in depth assessment. And when somebody in the family is dying, then you know there are kind of much bigger issues that they are trying to deal with and so you are very much standing on the perimeter and just using all your senses in trying to make sense of what is going on. So rather than being able to do several visits, and talk about this and play with the children and see what kind of level of development they have - I don't think you always have the time to do that.

Consequently, this is particularly true for palliative care social workers who may be going into situations where they do not have the luxury of postponing decisions until they have a wealth of information; they 'must often act on the basis of incomplete information' (Pincus and Minahan 1973:103). A sudden deterioration in the patient's condition can lead to minimal information being gathered. Assessment and work then tends to be intensive (Trevithick 2005).

If there was an early referral to the palliative care service and then on to the social worker, there was more opportunity for the worker to be involved with the family as in Family A. This was regarded as positive, particularly for those workers who wished to encourage parents to be involved in making memory boxes (to be discussed later). It also allowed the worker opportunities to help the parent or parents make plans for the children both in the preparation for the bereavement itself and care arrangements postbereavement.

An earlier than expected death can also have an impact on the process of the engagement. For example, in Family B, the process of holding the family meeting was important in terms of clarifying what was to happen, when and why. Although the grandmother died the next day and all aspects of the agreed plan could not be put in place, this did not detract from the importance of the meeting itself. The lady died knowing that the future of her grandson was clear, her views about the importance of family history heard and respected and, family relationships repaired. In Family D, where the mother was very ill when the worker first became involved, the emphasis of the involvement, after immediate practical help was given, was to ascertain the wishes of the mother and together with her and the children begin to make plans for their future.

#### Building relationships and trust

Having a good relationship with a client in social work has historically been important (Beistek 1961; Harlow 2003) and essential to good practice (Trevithick 2003). According to Sudbery (2002:151), the primary resource in social work practice is the social worker and 'their use of relationship'. This 'relationship' was seen as 'the primary tool' in social work in the 1970s, but increasingly and particularly within the statutory services, building relationships and trust is now less possible due to the ways in which services are delivered (Parton 2008:258). This was highlighted in Family B where the family felt let down by the local authority worker and in Family C in regard to the children's fractured contact with the child and family team. Howe (1998:45) believes that the value of building relationships has become 'confused and ambivalent'. However, Cree and Davis (2007) found in their study that carers and service users wanted workers who would listen, treat them with respect, as well as seeing them in the context of their families, as the palliative care social worker demonstrated in Family B. Indeed, in palliative care there continues to be an emphasis on the service user being at the centre and the need to have a holistic approach to practice (Oliviere 2000). Service users in palliative care themselves place great emphasis on the relationship element of practice (Beresford et al. 2007). Mullard (2005:55) writes about the importance of 'presence' - the unseen therapeutic relationship in palliative care that requires an openness of all the participants involved. Other studies (Howe 1998; Leigh and Miller 2004) have found that service users generally value empathy, good listening skills, humour, respect, warmth and honesty in their social workers. From the way the palliative care social workers in this study spoke of their relationships with the families it was clear that they strove to achieve these qualities and is particularly evidenced in the practice examples described above.

One of the essential features in each of the practice examples given in the interviews was the importance that the workers themselves placed on building relationships and

trust as opposed to going in to a family situation and being directive. There was, for example, clear evidence of the social worker working in partnership with the parents in Family A and after time with the grandmother in Family B.

Getting to know families during such a time of stress and distress could mean that palliative care social workers might become very close and emotionally involved. It became apparent across all the practice examples that in some circumstances, often in their encounters with children, the worker was viewed as a friend, as in Family A and particularly in Family C. This finding resonates with those in the Beresford et al. (2007) study. Where this was raised specifically, workers stressed the importance of reminding people of their role and maintaining boundaries. One worker summed it up by saying:

In the past some people have said you are kind of a real friend and I am always careful to say that is really kind of you. It's nice in some ways that you can see me in that way but actually I am a social worker and I am coming from the hospice and this is what I do - this is my supportive role and try to put that boundary back on it because I just worry when kids do say - oh you are not going to go away are you - you will always be around and I am sure what they are doing is stating their fears for the future.

This may also demonstrate the particular value children place on having someone outside the family for support as research in chapter three indicated.

Literature supports the view that working in partnership with parents is generally considered to be beneficial (Thoburn et al. 1995) because of the link to better outcomes for children (Department of Health 1995; Cleaver 2000). This was evident here, for example for the children in Family A. However, the picture of partnership is less clear in Family C where there were issues of domestic violence within the family. Here, it appears that the worker endeavoured to work alongside the father in an effort to obtain better relationships within the family but there is no sense of a partnership. Indeed, in this situation a directive approach with the father might, from the outside, seem to be

warranted in order to safeguard the children. In Family D, the mother died a very short time after the social worker became involved and there was no father present, but a partnership evolved instead with the adolescent children. In Family E too, there was more of a partnership between the worker and the children than with the worker and the parents. For the mother, there were perhaps language difficulties as it is not clear how this was managed in the hospice. The father is portrayed as a caring person but there is little evidence of a working relationship between him and the social worker. This could have been due to cultural or language issues, or the lack of opportunities for the worker to see the father, rather than to a low value being placed on the need for engagement.

#### Involvement with fathers

There were proportionately more examples given in the interviews, focus groups and survey of situations where mothers were dying. In the interviews, for example, the ratio of fathers to mothers dying was 1 to 2.3. Views were expressed by the palliative care social workers about the different losses there might be if it was the mother, rather than the father, who was dying. One worker in an interview commented:

This may be sexist, but if the mother is dying, I just think it is a bigger impact on perhaps the emotional, not necessarily the financial. I just tend to think that children grieve very differently for a mum than a dad.

Mothers seemed to be regarded as the lynch pin of the family and whilst there might be financial difficulties after the death of a father, it was considered that there were more chances of the emotional and practical concerns being managed. Studies have shown that fathers may be less likely to be aware of their children's emotional state and children may be less likely to talk to their father (Kennedy 2008). This could be because mothers are generally seen as the expressive leaders of the family (Cohen et al. 1997) and children generally behave more obediently towards their fathers and hide their emotional state from them (Visser et al. 2005). Studies have also shown that fathers are

less likely to offer closeness and support (Cawson et al. 2000). Three workers in this study specifically described how, when a mother was dying, they would be looking to see what the relationship was like between the father and the children. One gave her reason for doing this:

... as fathers fall roughly into two categories - either they are really going to rise to the challenge and be brilliant or, they are useless.

Mothers, it was assumed, were used to the caring side; fathers, it was thought, struggled with this aspect and tended to put their own needs above those of the children. These views seem to reflect the more traditional views of the roles of mothers and fathers, men and women (Featherstone 1997; Turney 2000; Christie 2001; Featherstone 2003). They also support the view expressed by Daniel et al. (2005) that gender divisions in parenting are still present and that gender assumptions about parenting roles appear to continue (Daniel and Taylor 2006).

Daniel and Taylor (2005) wrote about a recent 'explosion' of interest in the role of fathers together with a recognition that mothers tend to be the focus in child care and child protection work. Indeed it is mothers, not fathers, who are considered to be the perpetrators of child neglect based on the breakdown of nurturing, historically attributed to mothers (Scourfield 2003; Daniel and Taylor 2006). There is also evidence of fathers not being involved in assessments and a general 'failure of policies, procedures and agencies to find ways of engaging with fathers effectively' (Gilligan and Manby 2008:183). Some men avoid social workers and health visitors which, Scourfield (2006:445) speculates, they see as 'women's business' and social workers and health visitors are criticised for concentrating on the mother (Daniel and Taylor 2001). If social workers in general see women as ultimately responsible for children and men 'as no use, irrelevant and as absent' (Scourfield 2006:443), was this also true of the palliative care social workers in this study? To what extent were fathers involved in the process?

Indeed, the telling of the stories rarely focused on the fathers unless they were the sole carers of children.

The five practice examples show different levels of involvement of workers with the fathers of the children. In Family E, for example, the social worker described him as 'very absent'. There was little evidence that the worker discussed the emotional needs of the children with the father, or in fact, the father's own emotional state and I failed to pursue the issue. The worker made reference to the fact that 'we involved Dad at some point' but did not make it clear why or what the outcome was. The emphasis of the work appeared to be with the dying mother and the children who were more readily accessible during the day and appeared to wish to have contact with the worker. In Family A, the father was at home during the day and so was available to meet with the worker. In Family C, the worker offered bereavement support to the father but there is no clear information about whether his behaviour towards the children was questioned or challenged in any way by the palliative care social worker. In Families B and D, there were no fathers to be involved, although there was limited involvement with the grandfather and uncle in Family B.

## Assessment

Assessment has been described as being a 'crucial feature of social work' (Jackson 2000a:20) and 'at the heart of all good social work practice' (Watson and West 2006:30). In the Beresford et al. study (2007:62) the assessment process, as described by service users, was presented as a picture of 'a subtle process where the emphasis was placed as much on building the relationship between the social worker and the service user as on finding out the facts'. Service users had welcomed this informality as it made them feel at ease and gave them a sense of being listened to. From listening to how the workers described their practice in this study, this informality appeared to be present and they approached assessment with an emphasis on relationship-building.

There was, for example, little mention of the completion of checklists or forms as part of their own assessment. The informal approach was illustrated by a hospice based worker:

I would have talked to the patient - both the patient and partner, and wider family or friends and then perhaps talk to the children if they are around - come and get a biscuit in the kitchen ... and then ask if anyone is involved with them in the community, nurse-wise. I would ask them how things are functioning at home. In this way, palliative care social workers are constructing their 'own narrative' about the situation (Fook 2002:118).

Whilst there are many definitions and ways of undertaking assessments described in social work literature, the accepted view is that assessment should be an ongoing process (Crisp et al. 2005), particularly in the case of dying people (Beder 2006). It should be a 'video' not a 'snapshot' (Oliviere et al. 1998). Pincus and Minahan (1973) developed a process model of assessment in their systemic approach to practice that recognised that the worker must continually assess situations to inform their decision-making. This would include the context as well as the individual, as described by one survey respondent:

There should always be an assessment of the individual and of the context to avoid assumptions. Such an assessment invariably includes issues such as history of the child care offered prior to diagnosis/symptoms of illness, levels of mental capacity/likelihood of this being impaired (temporarily/permanently).

This approach was generally reflected in the study. One palliative care social worker particularly remarked that assessments needed to be continually reviewed because 'not everybody reveals everything to begin with'. There were, however, depressing comments made, both relating to hospital rather than hospice care. One focus group participant commented:

The danger is that you become an assessment machine to enable a discharge if you work in a hospital setting and you lose the wider dimensions - thinking about children and their needs is one of them really. The holistic bit.

For this worker, the demands of the medical setting appeared to take precedence, which meant that concerns about children might be missed. Another palliative care social worker in a focus group made a similar observation with emphasis on the difference between hospital and hospice care:

The one great advantage to me is that if you are in a hospice you have more time, but having moved to the national health recently it has struck me that in hospitals you have no time. Hardly getting to know anybody to pick up concerns and worries and suddenly they are out again.

As Buckley (2003) found in her study, it was not always easy to gain a clear picture from the information gathered across the data about how palliative care social workers undertook their assessments in relation to children and their parents, and therefore what informed their practice in this aspect. One interviewee, for example, explained that she did not use any assessment tool but relied on her training and experience. Sometimes how the worker approached the task was implied in the narrative, for example:

I talk to the patient and partner, wider family and friends, children. Ask anyone else involved.

Overall, the data suggests that workers tended to use the 'exchange model' of assessment (Smale and Tuson 1993). This model views everyone as 'experts in their own lives' and there is an emphasis on exchanging information. The aim is to identify people's internal resources and potential and discover how best to use these as well as identified external resources to meet needs and achieve goals (Milner and O'Byrne 2002). Indeed, from the way the palliative care social workers spoke about their contact with families, it would also be reasonable to assume that a narrative approach (Fook 2002) was often used in gaining a picture of what was happening in the family. Here

again, the emphasis is on natural exchange, encouraging people to talk about their experiences and any questions used should be open to help build a reflective story (Parton and O'Byrne 2000). This recognises that service users have 'agency': they have responsibility for making decisions and being involved in 'the sense making activities of assessment' (Coulshed and Orme 2006:35). This links back to the concept of working in partnership described earlier.

There were, however, workers who were specific. One interviewee spoke of the strengths perspective (Trevithick 2005:72) in her explanation of how she approached a family:

These are capable people who are dealing with an extremely difficult situation. I am working with people's strengths, resilience, their capacities to cope.

Another spoke of how she used a 'solution-focused' approach to assessment and she also commented that she liked to look at possibilities and strengths in situations. There was also mention of the use of psychodynamic assessment and a family-focused assessment. Another referred to having a systemic approach, looking at people's networks and family dynamics - how people actually communicated with each other. A more detailed list was provided by another worker who described herself as acting as a change agent, listening, being flexible and having a systemic non-judgemental approach. She said:

It is very much looking at individuals as part of a system and the roles that they play, which is why I wouldn't put a false demarcation around family members they are all part of the patient's system - and that would include the school, the church, just that sense of network.

Another interviewee both echoed and developed this by saying:

I tend to look quite systematically, looking at people's support networks, family dynamics, not just what they say to each other but how they actually communicate with each other and just building trusting relationships with them

so they can feel they can talk honestly about what is happening. I do look at development issues.

These two workers are clearly describing a family systems approach, not just how the family functions 'internally' but also how the family relates to other systems 'externally'.

In just under half the interviews, specific mention was made of genograms being drawn up during contacts at initial assessments, usually by nursing staff, which explained the inter-relationship family system (McGoldrick et al. 1999; Parker and Bradley 2003). These were considered by the palliative care social workers to be a helpful tool and although not always complete, they were the bases from which to start their engagement. However, it was not always clear whether the genograms referred to were merely a family tree, e.g. names of parents and children with dates of birth, or a more specific picture of the strengths and weaknesses of family relationships (Thoburn et al.1985). Detailed genograms have been found to help clarify who the ill person and family members are close to, from whom they receive support, their relationships with other agencies, how they coped with previous losses (Oliviere et al. 1998; Stokes 2005), as well as building up a 'story of the illness' (Kissane and Bloch 2002).

What is missing is a clear picture about how workers recognise or assess the level of abuse and neglect of children. One interviewee spoke about looking at children's behaviour, and another the stress and pressure in families. Only one worker specified she would undertake a risk evaluation of a child as part of her assessment, although references were made by other interviewees in different ways about the need to be alert for children being at risk. A third of the survey respondents also referred to children who might be at risk.

Surprisingly, there was little mention of the Assessment Framework in England and Wales, or the Integrated Assessment Framework in Scotland to assess children's

needs<sup>17</sup>. Whilst it would be very unlikely for palliative care social workers to be nominated the lead professional in such an assessment, they may be called upon to make a contribution. Indeed, an awareness of the aspects within the three domains<sup>18</sup> might help workers in making their own initial assessment of a child's vulnerability and needs. One worker in reply to a question about possible use of the Assessment Framework remarked:

I am probably aware of the triangle. I think it is probably in-built. I don't sort of think about it. I am aware of the sort of things and where a child is struggling and you just throw in the illness of one of the parents.

There was, however, an example in the survey where a senior palliative care social worker, in her supervision of another worker, had applied the framework in a specific way:

A colleague reported to me that a father of three revealed to her that he had hit one of his children. He is dying of cancer and lost his wife (and mother of the three children) to cancer also, recently. I explored with the colleague whether the father could identify any possible 'solutions' (approaches) to the increasingly tense family situations. He could, to some degree, but he was reported to be in denial about his prognosis and also not clearly able to think through the children's needs. We explored the child concern model<sup>19</sup> as a possible framework particularly for one of the three who seemed at [the] brink of disengaging from education all together.

Here the worker in supervision was assisting a colleague to think about the child's circumstances in a more structured way. However, being able to apply aspects of the Assessment Framework would be dependent on how much the workers or their supervisors knew about the framework, whether they had received any training on its

<sup>&</sup>lt;sup>17</sup> These frameworks were described in chapter four.

<sup>&</sup>lt;sup>18</sup> The domains are: child's developmental needs; the capacity of the parents or care givers to support their child's development and to respond appropriately; and wider family and environmental factors which may impact on development and parental capacity.

<sup>&</sup>lt;sup>19</sup> The child care concern model is based on the Assessment Framework and defines different levels of vulnerability which should then trigger particular actions by staff in the relevant agencies.

use, how confident they might be about using it and what support they might receive when doing so. Aspects of supervision and training will be explored in chapter eleven.

## **Practice approaches**

During the sharing of practice examples in the interviews, some palliative care social workers were specific about the practice approaches they employed in their contacts with children and families; others less so. Here, the term 'practice approaches' is used to describe approaches, models and methods drawn on a body of theory (Trevithick 2005:79). Those workers who were more specific named the methods and theories they utilised in their work which assisted them in their understanding of situations. Those who were not specific tended to describe what they did but did not attach a 'label' to their work: indicating an implicit use of theories rather than an explicit one which other studies of social work practice have shown (Monroe 1998). This supports Hawkins et al. (2001) who found in their study of social workers' use of language that little use was made of terminology. However, as Sheppard and Ryan (2003:168) point out, just because a social worker refers to a particular theory, it does not necessarily follow that they know what it means. Milner and O'Byrne (2002:69) consider that how any worker uses theory is 'a matter of great uncertainty'. Paley (1987) argues that although there is sometimes an assumption that social workers do not use theory, he believes that it tends to be 'situation specific'. Rather than using an 'empiricist style' where practice could be accounted for theoretically, a more 'contingent' repertoire is used which gives a natural account of their practice. This was certainly the case in almost all of the practice examples and supports Hawkins et al. (2001:2), who argue the way social workers talk about their practice is actually part of their practice. Parton (2003), reporting on a study of social work practice, also highlighted the lack of formal theory demonstrated by practitioners, but commented that it was clear that the workers used their own frameworks. This appeared to be the case in this study.

Those practice approaches that were specifically mentioned included having a taskcentred approach; a psycho-dynamic perspective; crisis intervention; bereavement counselling; and a casework approach. (A brief description of these approaches in this section is given in a glossary in Appendix 10.) One worker spoke of recently attending a cognitive behaviour therapy course. Although the approach to the work described by the palliative care social workers appeared to be psycho-social in nature, it was not described as such by any of the workers. There were those who specifically referred to systems theory. One worker who advocated 'brief solution' approaches commented that:

I am very much a believer in brief solution approaches, so my approach is to go in and do the work and come out again. I am not someone who necessarily wants to have an ongoing relationship; this could be a problem actually. I don't think that part of my role is a dependency relationship.

Her approach was to offer people six sessions of focused counselling which was reviewed before any further action was planned. This was a more unusual approach as for other workers, building a relationship appeared to be the most important feature of their work and this was particularly apparent in their described work with families.

Communication with families is seen to be a key aspect of palliative care and working with families is considered to be an important part of the task of palliative care social worker practice generally (Beresford et al. 2007). Workers in this study displayed skills in family work which included giving consideration to the emotional, physical and financial strains drawing on the families' resources. This included couple counselling particularly in situations regarding the care of the children during times of stress, as was seen in Family A. The literature shows that family meetings, which draw on family systems theory, are recognised to be an important aspect of palliative care (Fineberg 2010) and there were examples in this study of family meetings being held. These were for a specific purpose (for example, the meeting held with Family B) or to resolve more complicated issues. One worker wrote in the survey:

Two children of my patient being cared for by their father (my patient's partner) were often being left with her sister to feed, get to school etc. A family meeting was convened, with the outcome that my patient, supported, tried to elicit her partner's long term intentions towards her and the children. With time this proved he wished little role and certainly no caring responsibility.

Here, an outcome was reached which enabled the mother and sister to make realistic plans for the children's future.

#### 'Memory boxes'

Palliative care social workers gave information and encouragement to parents about how to prepare the children for what was happening through the use of already prepared material, for example, from Winston's Wish<sup>20</sup>. Advice was also given on the writing of letters and cards to the children, the taking and gathering of old photos, and on ten reported occasions, the preparation of memory boxes as in Families B and E. The importance and value of biography and memories as part of the process of grieving has now been recognised (Walter 1996) and is much promoted as a way of helping children pre-and post-bereavement (Stokes 2004; 2005). Helping dying parents (and it was mothers who were talked about in that context) to complete memory boxes or books was undertaken by five palliative care social workers in their practice examples and two in the survey. They were seen by them to be an important aspect of the work and valued by the dying parent. For them it might be described as a 'practice rule' (Sheppard and Ryan 2003:165): collecting memories can help children acquire a realistic sense of the person who has died and so it should be done. However, as I did not specifically ask about memory boxes it is difficult to comment on their overall use, although there were two workers who particularly sounded a note of caution. One spoke of families who had heard about this approach through the Channel 4 series 'The

<sup>&</sup>lt;sup>20</sup> Winston's Wish is the charity for bereaved children which helps children and young people re-adjust to life after the death of a parent or sibling by providing professional assistance before, during and after bereavement. For further details see www.winstonswish.org.uk

Mummy Diaries' aired in November 2007. The programme had followed five families 'facing the unthinkable - life without mum' (Winston's Wish 2007). After the programme was screened, some families had commented that they felt guilty because they could not be like the people on the programme and this had annoyed the worker. Another worker commented:

I have never met a mother who could bring herself to do one [memory box]. Two mums said when I am well enough I want to enjoy being with the children and do things and keep life going and when they feel awful the last thing they can face is the memory box. I think it is too much for some people.

In the examples provided, the emphasis appeared to be on the compilation of memory boxes when a mother was dying. This could reflect on the proportion of women's deaths to men's at this stage of life or to the attitude to fathers generally.

## **Organisational constraints**

In all the practice examples, there seemed to be a high level of activity with the service users and their families. No workers mentioned being burdened with recording or bureaucratic procedures, even though mention was made of minimum administrative support being provided. Nearly all workers were based in the voluntary sector. Their experiences mirror those of other social workers in the voluntary sector, as described in the *Social Workers' Workload Survey* completed on behalf of the Social Work Task Force (Baginsky et al. 2010).

It has been found that palliative care service users value 'a sensitive, participative and responsive model of social work practice' which includes being given time, the social worker being available and accessible, and the provision of continuity and support (Beresford et al. 2007:93-101). In the five examples, these qualities are present, except for the situation of the two children in Family C due to the boundaries placed on the worker's role where continuity could not be provided. The feeling of betraying children

was echoed by other workers in the study who found themselves to be in similar situations. The ability to continue to work with a bereaved parent, but not directly with a bereaved child under the age of 18 years, with whom a positive relationship had already been formed, appeared to go against what children themselves wanted. This placed the workers in a dilemma. Should they follow the 'rules' of the organisation or find ways of circumventing them? There was recognition that resources were limited but those workers who found themselves in some conflict with their organisation were frustrated by the policy. Aldgate and Statham (2001: 93) found that children are clear about the qualities they look for in social workers: reliability, practical help, ability to give support, time to listen and respond, and the ability to see children's lives in the round. If children feel let down by social workers at this crucial time in their lives, what will their image be of social work in the future?

## Impact of the family's immigration status

Cohen (2001:18) put forward the view that everyone who has a professional contact with children and families regarding welfare provision will 'probably be confronted with immigration control issues'. Families in close contact with social workers for any reason may 'coincidentally' be under threat of immigration controls (p.19). This was true for four of the 45 practice examples described in the interviews; there were no examples given in the focus groups; one was given in the survey.

In one family the mother was very ill and there was a 14 year old child who had been excluded from school. The father was waiting for definite leave to remain in this country and the social worker was in the process of writing a supporting letter. In another, it was the mother who had a work permit to come to the UK with her husband and three young children. When the mother became terminally ill she was clear that she wanted the children to remain in this country, but shortly after her death the father, whose status

was not secure, decided to return to their country of origin with the children. The circumstances of the third family have already been described in chapter six, where a father joined his dying wife to look after the children and remain in the UK, but the Home Office refused a visa for him to extend his stay once his wife had died. The children were very distressed and exhibited concerning behaviour. Help did not seem available from the local authority, and the asylum team, who became involved for a short time, withdrew as the father was not an asylum seeker. An appeal had been lodged and the worker had written letters of support to the Home Office. Family E, the fourth family, were asylum seekers who knew they could only remain in the country as long as their mother was alive. The example from the survey concerned an adolescent girl of a dying father and a mother who did not have a right to stay in the UK long-term. The relationship between the girl and parents was problematic, as well as between the girl and her school peers, resulting in her non-attendance at school.

These examples illustrate the additional stress placed on the parents, who were only too aware that their children could only remain in the UK as long as they were alive. The well parents had to cope with their own loss, that of the children, plus all the future uncertainties as well as the bureaucratic procedures of the Home Office. The children too were bewildered about what might be happening and were faced with multiple losses not only of their mother, but of their friends and way of life to which they had grown accustomed. In these circumstances deportation or removal overrides the provision of the Children Act 1989 (Cohen 2001:166). The Court of Appeal in 1998 (Imm AR cited by Cohen 2001:168) stated:

In the field of immigration, particularly in relation to deportation, the interests of the child are not, and cannot be paramount or primary ... if it were otherwise it would be difficult ever to make a deportation decision in relation to children.

The Court added:

The interests of the child are of great importance and must be given separate consideration, including consideration of child's family ties in this country ... and adults' ties with the child must be taken into account in the balancing exercise. However, as Cohen (ibid) points out 'great importance' is 'inadequate'.

The poor physical and financial situation of families in these situations is due mainly to the lack of benefits available to the family on account of their status in the UK (Cohen 2002). Humphries (2004:93), writing about the role of social workers implementing immigration policy, described people subject to immigration controls as 'some of the most oppressed people on the planet'. The experiences of children of refugees and asylum seekers show that they are 'a particularly vulnerable group even compared to other black and ethic minority children' (Dutt and Phillips 2010:357). The workers involved in the above families did not appear to see themselves colluding with the government (Briskman and Cemlyn 2005:721) by putting a 'human face on to inhuman practices' (Unison and BASW Scotland 2006:7) but were endeavouring to safeguard the children in very emotional circumstances. However, there was a sense of 'helplessness' in the face of the Home Office bureaucracy.

# Conclusion

Due to the holistic nature of palliative care, social workers in this service have a remit to consider the needs not only of the patient but also the patient's family which includes the children. Work concerning the care of these children was wide-ranging, although engagement with parents was the main vehicle for promoting children's welfare. Children were not always seen but the main emphasis appeared to be on ensuring that they were informed about what was happening by providing advice and support to parents. A family systems approach was in evidence, however. Direct work with children depended not only on the willingness of the parents and children for this to happen, but

also on agency policy, general circumstances and on the preference and confidence of the worker. However, in some instances, agency policy prevented a minority of workers from providing continuity of support direct to children post-bereavement.

Whilst there was engagement reported with fathers, it was mothers who mainly featured in the examples provided in focus groups, surveys and interviews. In the majority of the situations described it was the mother who was ill reflecting the national statistics. The future care of the children was often a focus of concern, particularly where the dying parent, nearly always the mother, was the lone carer. Whilst many parents were able to make plans, there were others who were unable to do so.

The plight of children in families where the immigration status of the parents was in question was particularly concerning. Not only did the family have to cope with the uncertainties and distress associated with the parent's illness but also the uncertainties and distress of the likelihood of moving to another country against their will. It was apparent that palliative care social workers found working with these children and their families distressing but they saw themselves as striving to do all they could to support and assist them.

If one accepts Searing's view (2003:318) that the essence of good social work is helping, not hindering, and bringing something human to the relationship, then the evidence here goes some way to demonstrate that this is what palliative care social workers strive for and achieve. Because of the time constraints that the illness of the parent can bring, much of the work is intensive and short term in nature. All the practice examples highlighted not only the range of work undertaken by palliative care social workers where there are concerns about the care of children but also the value they put on their engagement. Agencies that had policies restricting direct contact between children and palliative care social workers after the death of a parent caused challenges

for the workers and apparent disappointment for the children. If trust had been established then it is understandable that the children would wish to turn to someone familiar for help and support at such a difficult time. Workers were concerned not to let the children down.

The following chapter will describe and analyse the palliative care social workers' perceived relationship with local authority children's services.

# **Chapter Ten**

# The role of the palliative care social worker: liaising with local authority children's services

Children's services have a different agenda and priorities to us. Survey respondent

## Introduction

In their work with children and families, palliative care social workers identified liaison with other agencies as an important part of their task. Local authority departments were the most frequently mentioned agencies, in particular the child and family teams of the social service departments. There was limited mention of contact with other local authority services, for example education, adult services and housing, in connection with their work with children and families. As well as contact relating to specific families, there was other involvement too, namely in relation to training and the Local Safeguarding Children Board to be discussed in the next chapter. This chapter therefore concentrates on the relationship between palliative care social workers and social workers in child and family teams. The information is gathered from the interviews, focus groups and the survey. I will describe the amount and reasons for contact before exploring communication and making referrals. This is followed by an analysis of the relationships between these different types of workers as perceived by the palliative care social workers together with their suggestions for improving working relationships. This is followed by a discussion of the emerging themes.

## Amount and reasons for contact

All those palliative care social workers who were interviewed and participated in focus groups spoke of having contact with child and family teams. In the survey, however, about three quarters of the respondents reported that they had contact while the

remainder had either no contact or very minimal contact with these teams. One respondent wrote:

I have not needed to contact the teams regarding child protection issues in the 10 years that I have worked here.

The main reasons for contact fell broadly into two broad areas: one was seeking advice and guidance regarding individual family situations and the other involved making referrals. Advice and guidance covered a broad spectrum including guardianship issues and child care resources. Reasons for making referrals were diverse and included issues relating to the future care of children; parents who were unable to physically cope with the care of their children due to illness; behavioural problems in children; inappropriate care of children; and, suspicions of abuse or neglect. There was a minimal number of reports of referrals from the local authority, except in one hospice where there was a bereavement service which was open to referrals from other agencies in the area and no requirement for a previous contact with the hospice.

Contact with local authorities could be a very complex issue. Hospices and hospitals generally serve a wide geographical catchment area. This meant that a palliative care social worker might be involved with more than one local authority as can be seen from data gathered from the interviews in Table 10.1.

Number of local authorities in catchment area	1	2	3	4	5	6	Not given
Number of workers reporting involvement	3	3	3	0	2	2	4

So, for example, there were two palliative care social workers who could be involved with up to six local authorities. However, it was not just a question of negotiating with

more than one local authority; usually there was more than one child and family team in each authority. One palliative care social worker, for example, spoke of there being six teams in one of the local authorities in her area. The general view was that working with more than one local authority presented difficulties and confusion for a palliative care social worker, particularly as the political outlook in each authority could lead to differing policies, attitudes and resources. Palliative care social workers, therefore, encountered varied responses, making it difficult for them to engage in longer term planning with families. This could lead to feelings of confusion. One authority might be very flexible in its approach and prepared to participate in joint working, while another might refuse to consider any involvement unless a child was judged by them to be 'in need' or subject to abuse. One survey respondent illustrated the difficulties:

I work with a number of local authorities. Some are very flexible about when they can offer and understand the need for early intervention and joint working to avoid a crisis later. Others refused to take any cases that aren't child protection or children in need. Difficult to do long-term care arrangements without their involvement. Parent has to become very ill before they'll get involved.

## Communication

It might be reasonable to assume that in order to successfully seek advice or make a referral there should be good communication between parties. Evidence from this study would suggest that good communication often did not exist between palliative care social workers and social workers in child and family teams. Whilst there were examples of palliative care social workers considering that referrals had been taken efficiently and action successfully put in place, these were in the minority. One palliative care social worker commented:

One local authority refused to talk to us even though children were already known to them.

There were difficulties reported in getting through the 'gatekeepers' in the social services when searching for advice and making a referral and this felt less than satisfactory. Before gaining access to duty workers, palliative care social workers had often first to go through call centres, sometimes talking to up to three people before reaching someone who could actually help. One worker had solved the problem by missing out the call centre; over the years she had collected the direct numbers of several local authority social workers. She commented:

If I get through to them and then say, put me through to somebody else - at least I am in the office and not the call centre.

There were other palliative care social workers who also saw a need to build up a web of contacts to bypass official systems. Use was made of past connections, experience gained from previous employment in the local authority, practice teaching links, friendships, and from a previous positive joint working relationship. As the following quote describes:

Fortunately, I have extensive statutory experience or I would have felt very much unsupported by my social service colleagues.

The few palliative care social workers currently employed by a local authority considered they had better links to children's service workers and were also able to access the local authority database if necessary.

The difficulties involved in gaining the attention of the social services were often mentioned and frequently called for perseverance on the part of the palliative care social worker. One worker reported that it took about five months to arrange for someone to collect children from school. She succeeded after completing 'children in need' forms, writing letters, and sending a complaint to the manager and finally the director of the service: 'it was just tearing your hair out sort of stuff'. Another made three referrals over a period of one month regarding a family before an offer to make an assessment was

made. However the case was closed without a social worker meeting the family. This worker commented:

... the statutory services almost didn't understand why I was concerned and I kept trying in lots of different directions to get assistance for this family but feeling like doors were being shut everywhere that I looked.

There were also reports of communication being very one sided. Palliative care social workers would leave messages but the call back rate was poor. Similarly, a lack of feedback about a referral once made was regularly commented on which led to feelings of frustration. 'I have never had an easy or successful interaction' wrote one worker. Summing up her thoughts about contact with the local authority, one worker said:

Things have changed very dramatically over the last three or four years in that staff members in the community in the children's teams have changed incredibly and no continuity ... but if I were only dependent on trying to get some understanding of what to do or where to go from a duty social worker in children's services, we would frankly be up a creek.

## Seeking advice and guidance

The ability to seek advice appeared to depend on the quality of the communication between parties. A palliative care social worker might wish to make a referral but would be unable to do so for a child 'in need' unless parents first gave their permission. On these occasions, palliative care social workers endeavoured to seek advice from social services regarding possible resources so that this could be discussed with the family. However, there were several reports of occasions when social services refused to give advice unless a referral was actually made. One palliative care social worker wrote:

Contact re family with four children and terminally ill father. I was asking what may be available - respite care? family home care? <u>before [writer's emphasis]</u> discussing with Mum. Duty worker refused to discuss resources, said I must make a referral first - I felt Mum unlikely to agree unless she knew what help she might get.

However, there were also reports of advice having been willingly given and this generally appeared to be in situations where there was already a positive connection between the workers. There were also occasions when local authority workers would be supportive and give advice even if they were not able to provide a service themselves. One worker felt when she said she was from a hospice: 'people seemed to go that extra mile'.

#### **Issues around referrals**

Once a referral had been made by palliative care social workers to child and family teams, the latter did not always become involved, especially in situations where children were not seen as being in immediate risk or danger by the local authority staff. This led to feelings of frustration on the part of palliative care social workers who were unhappy about the different standards that seemed to be operating between themselves and the local authority staff. One palliative care social worker commented:

Children in need seem always to take second, third, fourth, fifth place behind child protection.

This issue of thresholds was a frequently reported difficulty. The local authority staff would appear not be concerned about the situation described and would ask palliative care social workers to make contact again when there was a crisis. This was particularly the situation in one family where there were guardianship issues regarding two children whose mother was dying. The palliative care social worker wrote:

Service user had concerns about arranging guardianship for her children after her death and made a referral to social services; one child diagnosed with learning disability and the other's father was abusive. Children's services were reluctant to be involved until her prognosis was such she had a few weeks to live.

It seemed that the local authority staff were 'too busy' to take their complaints seriously and concerns were minimised. One palliative care social worker wrote:

I referred a young child ... for assessment /care as seemed to be neglected (grandmother dying). The team felt there was no need to get involved. I was told they had worse cases to deal with and didn't have the time as nothing specific had happened.

There were times when palliative care social workers were very worried about children's circumstances and felt responsible for them, particularly when a local authority was reluctant to become involved. One described it as feeling 'really scary' when trying to talk to a member of the duty team 'to make them understand what was happening'. Another described how a family's circumstances caused her concern not only because of what was happening but because of her own agency policy which meant that she could only be involved for three months post-bereavement. She did, however, manage to extend this for a couple of months but had to withdraw before any resolution was made. She wrote:

An ill parent died in hospital leaving two teenage daughters in the care of their much older brother. I attempted to get input from the local authority children's services over a four month period when the brother informed me that one daughter was depressed and suicidal and the other was spending nights away from home, smoking dope with friends. While waiting input, I met with the family several times, to try to assist in discussing the tremendous impact of loss of mum, while also trying to engage young people to reduce risk. Children's services never got involved, even when months later the family faced eviction.

There was less certainty about referring a child who was 'in need' rather than a child who was 'at risk'. There was a view that often such decisions were difficult to make, particularly when past experience had led the palliative care social worker to believe that such referrals would not be acted upon. One worker described herself as being 'behaviourally conditioned' by social services and subsequently had referred 'less and less'. Workers would try to find resources in other ways and 'sort things out' themselves.

Referrals regarding older teenagers, particularly those between ages of 16 and 18 years, caused palliative care social workers particular concerns, as it appeared that social services were less inclined to be involved. There was an example of a 16 year old boy who was living with his mother who was dying. She was a lone carer and there were no other relatives available to help. The palliative care social worker was questioning her own level of responsibility in this matter, as social services told her that they 'might find him somewhere for him to live' when his mother died. He could not take over the tenancy of the flat due to his age. The palliative care social worker was concerned that not only would this young man lose his mother but also his family home. There was another example in the survey of a 16 year old girl described in the previous chapter who was not attending school because of relationship issues with her peers and had a poor relationship with her parents. The palliative care social worker was involved with advocacy and negotiation work with school, parents and daughter. Social services were reported to be aware of the situation but considered the girl was old enough to access her own help.

Regardless of previous responses from the local authority, there was a general opinion among palliative care social workers that it was important to make referrals sooner rather than later, recognising that the parents' permission was required before doing so. However, making a referral to the social services where there were concerns about the safety of a child could, it was recognised, damage the relationship between the family and the worker as well as the agency and examples were given where this had been the case. For one worker, having to put a dying woman through a child protection investigation was a difficult decision, but she considered the children's welfare was paramount and made a referral to social services. The patient was subsequently extremely angry and verbally hostile to the worker and refused to have any further contact with the worker and, after discharge, with the in-patient hospice.

There was also some evidence of palliative care social workers 'giving up' on social services because past negative experiences had coloured their decision; they were reluctant to make a referral at all. One palliative care social worker spoke of her lack of trust in the abilities of local authority social workers to understand and work with families where a parent was terminally ill. This made her reluctant to make referrals. Another worker considered that making a referral would give the parents an 'extra burden' because of the uncertainty of the agency becoming involved. Another worker summed up her views by saying:

I was trying to think why you perhaps don't go more to social services and it is because, you know, you are not going to get a response in terms of a preventative, supportive service. You tend to look around and sort things out yourself.

One interviewee described this lack of resources as being the hardest part of being a palliative care social worker. In her view, if practical support could be offered to the family, then all the pre-bereavement work 'which could be such a value to the family and children' could then be undertaken, as parents would have more energy to take part. The worker commented: 'it's like it's not important enough'.

# Relationships with local authority social workers

There was an overall feeling of disappointment in the interviews about the quality of the relationships with local authority social workers based in child and family teams. Opinions ranged from tenuous to very good; excellent to very, very bad. However, it was recognised that it was difficult to form working relationships when there was minimal contact.

Several workers described their relationships with the local authorities in a combative way. One was vigorous in her assertion that as a qualified social worker her assessment was appropriate and that those to whom she was making a referral needed 'to

understand that my assessment is experienced, expert and valid'. Others spoke of their lack of faith in the child and family workers. Those who had previously worked in child protection in a local authority shared examples about their ability to negotiate their way to obtain services and even on occasion to 'manage' local authority staff. The quality of communication and knowledge of the 'system' was seen as having an impact on the nature of the relationship.

There was a view that local authorities appeared to be 'bogged down in all their procedures'. Workers were concerned about the changing standards and thresholds regarding access to children's services with, as they saw it, a lack of emphasis on prevention and services for children in need. They found it difficult to keep up-to-date with the current thinking and felt let down by the lack of support from their social services colleagues. As one palliative care worker expressed it: 'It rather undermines what you are trying to achieve'.

Most of the negative aspects of relationships centred round issues of referrals and communication discussed earlier. Examples were given where palliative care social workers took on the role of mediator between local authority staff and the family, particularly where it appeared to the palliative care social workers that the local authorities were being 'draconian' in their approach. Those palliative care social workers who had experience in local authority child protection work were particularly able to explain to families what was happening and why. One palliative care social worker described how she attempted to make social services act in a more appropriate and sensitive manner by building up trust with the social worker who was under pressure from her manager. At the same time she explained to the family that social services did have legal rights and a duty to care.

It was almost like having a family that was fighting actually - the two parties.

Sometimes a family seemed to fall between services for adults and those for children and families when it came to the allocation of resources. There were reports of disputes about funding and providing services for children when it was the adult who had care needs. One worker spoke of this as:

... the disconnect between adult services and children's services ... many adult social workers these days who seem to not know anything about children's needs - not only not to be very skilled about those things but also to not see it [as] a priority to get skilled up.

Workers gave examples of the divide between the services, particularly where support workers provided by the adult services were not prepared to undertake care of the children; for example, to prepare a meal for them or collect them from school, even though this would assist the terminally ill parent. There were, however, also examples of individual support workers who had undertaken these tasks once they had become involved with the family.

In contrast, there were examples of positive experiences with work in partnership and an acknowledgement of how much some local authority social workers had struggled to provide what children required. However, these amounted to a small minority compared to the total number of experiences shared. One respondent wrote:

We haven't had a bad experience in many years. We think this is because of our commitment to safeguarding children by being involved with joint training. Also two members of our team worked a significant number of years in child protection and so understand policies and procedures of protection agencies. We have established good rapport with social services and our concerns are respected. ... We are included in case conferences, children with additional needs meetings and make contributions to the assessment framework. There we have input to needs assessment and action plans and our contributions are taken seriously.

In such circumstances, joint working was welcomed and, when undertaken, warmly reported, as one worker explained during an interview:

I got a child and family social worker involved and interested - it was fabulous. I was so pleased ... we actually did manage to progress plans for the future and that felt safer to me.

There were two mentions of positive examples of children's social workers attending case conferences for patients, enabling constructive action plans to be made. Often though, joint working meant that the palliative care social worker continued to work with the parent who was dying and the local authority worker with the child. A survey respondent provided an example:

School concerned about a 15 year old boy with behaviour problems whose single parent mother was dying. I liaised with children and family team who had contact in the past. Worker in team became involved and focused on child's needs whilst I continued to support mother. The boy became closely involved in caring for his mother and her death, though sad, had a profound effect and initially he appeared to cope much better than expected by all concerned. Follow up was by child and family team.

There was a further example of co-operation and collaboration when two primary age children had moved into a hospice to be with their dying mother and stayed for several months. Their father had died a few years previously and there was no else who could care for the children whose behaviour had become concerning. The palliative care social worker spoke positively about involving social services and 'getting them on board'. Social services arranged transport to and from school and provided carers for the children as well.

#### Perceptions of role

At various points during the study, palliative care social workers made comparisons between their role and those of social workers employed by local authorities. The narrowness of the role of local authority social workers was particularly highlighted. In

contrast, palliative care social workers considered that they were able to look at the family as a whole. Those workers who had qualified several years ago believed that their role, being generic in nature unlike that of local authority social workers, was more like the one that they had been trained to carry out. One worker observed:

I know my colleagues in the local authority would think we are kind of dinosaurs here really.

In fact, those workers who had been qualified for some time spoke of the need to return to the model before children's services and adults' services were split and two specifically mentioned the return to neighbourhood services. One of them commented:

It is unbelievable that one person in one office is looking after the mother and someone else somewhere the children.

#### Palliative care social workers' perceptions of local authority social workers

There was an understanding that local authority social workers had a vast job and it was the general impression that they were only able to react to child protection situations and to crises. There was very little evidence of preventative work. This was very much a matter of regret and frustration for many palliative care social workers who saw children in need coming way behind child protection. There was, however, recognition that changes in children's services due to re-organisations, inspections, staff suspensions and a large staff turn-over resulted not only in low morale amongst the staff but also a lack of continuity. There was also recognition that local authorities struggled to provide adequate resources and were over-stretched. A worker who had moved from local authority child protection work into palliative care explained how she had viewed the work:

I was aware that I was not practising with children and families in the way I wanted ... I had to seek to work in some other way where I could bring a standard to what I was doing. I thought I was merely working at such a superficial level I just didn't have the time. I didn't get to know the kids. I was being asked to make huge decisions that would have a huge effect on children's lives ... and I just thought it was awful and wasn't the way I wanted to work.

Another worker who had also moved reflected:

I think when you work in child protection it is very much a tick box thing - what are the risks, what is the risk reducing practice, is the child safe, is there a responsible adult around? It is kind of tick box things but not really thinking about the subtleties or the intensity. It may be that dad is around but dad is also losing his wife and will be a single parent soon and working and caring and so emotionally he may not be that available for the children.

One palliative care social worker thought that some children and family workers saw themselves as the ones 'who do the really important work and are the cutting edge' implying that palliative care social workers do lesser work. Another pleaded that social workers generally 'should not fall into the trap of being isolated within their roles', believing that this could easily happen. Change, she believed, could occur and explained:

It requires a lot of insight and a lot of constantly re-examining, reflexive practice, and it takes resources to do that but it is also a cultural shift.

Palliative care social workers recognised that they had more opportunity to talk to people, especially within a hospice and that this was something local authority social workers did not generally have the time to do. They also felt that families trusted them more than the local authority social workers because of their non-statutory basis and the fact that they kept people informed about what was happening, leaving them in control of what they wanted to do. There was also a belief that local authority workers generally only saw people with low incomes or who were in poverty, whereas, in their specialist posts, palliative care social workers had the opportunity of working with people from a range of backgrounds.

Whilst a minority of palliative care social workers themselves admitted to not being upto-date with all the changes in child protection terminology and policy, they were less

tolerant of staff in local authorities who were not clear about the palliative care needs of parents and the emotional needs of their children. There was a perception, often repeated, that social service social workers did not fully understand the impact of bereavement on children and families. There was also an apparent lack of understanding regarding the nature of the illness itself and the needs of families where a parent was dying. There was a failure to recognise the need for a prompt response, whether it was for information or service provision. One respondent described how social services were involved with a family, and had placed three young children with grandparents. The grandmother was later diagnosed with breast cancer and was receiving daily treatment but could not cope with the children and required assistance. The palliative care social worker found that obtaining this was quite difficult.

There were two examples where workers from children's services appeared reluctant to make face-to-face contact with a dying parent, giving the impression that they were unsure of how to approach the task or perhaps unable to cope with what was happening to that person. One palliative care social worker wrote:

Sensed a marked reluctance to see the patient face to face, because of her medical condition.

These difficulties of understanding illness and bereavement were attributed by the workers to lack of experience and a gap in the qualifying and post-qualifying social work training.

There was a sense of palliative care social workers feeling protective about the families they referred to children's services. This was apparent in the way they spoke about certain members of staff. One worker spoke of her lack of trust in the ability of novice field staff taking over work with a family:

Felt children and families social workers very young, inexperienced and harsh. Another worker remarked: 'I thought they would go in and make a hash of it'.

# Palliative care social workers' understanding of how they are perceived by local authority social workers

It was a prevailing belief that the vast majority of local authority workers did not understand the role of the palliative care social worker and had different priorities and concerns. One palliative care social worker in a focus group reported being referred to as the 'Macmillan nurse' at case conferences regarding children, even when they had been introduced as a palliative care social worker. However, it was considered that social work staff in the adult services might have more of an understanding compared to those workers in child and family teams. This was because palliative care social workers had more regular contact with these staff members, particularly when arranging care packages for patients either living, or returning to live, in the community.

#### Improving relationships

In the individual interviews, palliative care social workers were asked how relationships with social work staff in the local authority might be improved. The most favoured approach was having an opportunity to talk and explain their role. This might mean going out to visit social workers in their teams to talk about what they do and share ideas about how they might work together. One worker expressed it succinctly:

They need to learn from us and we need to learn from them. Other suggestions made included meeting with managers of child and family teams and inviting staff to visit the hospice so that, as one palliative care social worker explained:

They can come and experience it and see what families need to juggle with.

Shadowing was another strategy put forward - in both directions. One palliative care social worker thought that spending time with a child and family worker would help her to understand more clearly the various procedures and statutory responsibilities. Where it had taken place, joint working was also seen as productive way of learning about each other's role. However, one worker sadly commented:

Child protection and family teams would often gain a lot from working in conjunction with us, but they seem just not interested.

Some palliative care social workers who, because of the number of authorities they were linked to, struggled to make contacts and thought that concentrating on the largest authority would be a good starting point. One respondent who had attempted this commented:

...getting better as I am proactively finding individuals I can ring who I can talk to. Used to feel isolated.

Networking was therefore seen as beneficial, although there were time constraints that hampered its productiveness. Having an identified person in the local authority as a liaison contact and meeting periodically with that person to keep up to date with developments in practice, was described as a positive experience but not one universally available. Joint training on child protection, children's needs and current thinking was a particularly favoured avenue. A further suggestion was that this should also include staff from adult services. There were questions about the adequacy of qualifying social work training on the subject of bereavement. One worker suggested that the impact of bereavement on life cycle issues should be made more explicit in training. Taking student social workers on placement was believed to be a productive, but slow method, as it would not only help to influence the individual student's thinking, but the learning gained might be shared with fellow students and, when qualified, with staff in the local authority.

# Discussion

Overall, the findings present a rather depressing picture of palliative care social workers' struggle to ensure that children's needs are met. However, the findings are not new. There is much congruence with those from other studies, particularly around issues

connected with communication, the referral process and relationships generally (Birchall and Hallett 1995; Calder and Horwath 1999; Buckley 2003; Horwath and Saunders 2005; Horwath 2007a). Overall these studies considered the views of other professionals, for example, staff from education, health care and the police in their contact with local authority child and family social workers. In this study the views considered are those of social workers in one setting about social workers in another. Consideration will be given later as to why there might be difficulties in single-profession communication and collaboration.

In Horwath and Saunders' (2005) study four issues emerged which appeared to have a negative influence on referrals to local authority workers: lack of communication and feedback; poor continuity of local authority staff; lack of access; and, perceptions of social work practice. Certainly all these issues were present in this study. There were feelings of anger and frustration at the perceived lack of response from local authority workers, which left palliative care social workers worried and anxious about the future care of the children, together with the feeling of letting down the child and family. This lack of communication after making a referral can also leave the referring professionals feeling powerless and unsure as well as concerned about the consequences for the child (Cleaver et al. 2004; Holland 2004). Local intake procedures contributed to the tension due to the inability to talk directly to a child care worker. There was a sense of powerless in many of the situations described in this study by the workers when referrals were 'not accepted'; one worker described it 'like doors being shut everywhere I looked'. This left workers with feelings that their assessments and concerns did not count and that the children with whom they were involved were not 'important' enough to warrant enough concern for services to be provided.

However, local authority staff may hear a message of concern about a child as a demand for them to do something when they lack the resources to do so (Taylor and

Daniel 2005:16). Reder and Duncan (2003:96) drew attention to the 'the complex interplay' involved in communication. It is not surprising that palliative care social workers reported that they were often reluctant to make a referral for a child in need because of a previous lack of response to requests; a view reported in other studies (Buckley 2003; Horwath and Saunders 2005). In contrast, while issues of confidentiality as a barrier to communication have been raised in other inter-agency studies (Darlington et al. 2005; Horwath and Saunders 2005), they were not mentioned in this study.

Palliative care social workers hoped that when they did refer a child 'in need' it would be a catalyst for intervention; nurses had similar views in a study by Fitzsimmons and White (1997). However, this was frequently not the case. The reported attitude of local authority social workers that there had to be a crisis (as if a parent dying was not crisis enough) before any action could be considered, is concerning. Many workers queried how aware local authority workers were about the impact of living with a dying parent on a child. There were examples of the apparent avoidance of involvement with a dying parent. This may be due to a discomfort about death and dying, a lack of confidence, being unclear about the social work response to dying and bereavement (Currer 2002:211) or to a gap in their training referred to earlier.

The issue of 'thresholds' was a recurring one and is replicated in other studies (Buckley 2003; Horwath and Saunders 2005). However, 'thresholds' have no statutory basis (Laming 2009:29). Despite the requirement that local authorities should provide a range and quantity of services appropriate for children in need, this is clearly difficult to achieve. How decisions are made about which children and families should receive a service is not clear. Literature shows that escalating demands and inadequate resources mean that local authority workers ration the number of cases they can accept and work is dominated by child protection (Scott 1997; Buckley 2003). It is apparent that

there is a 'filtering' system in operation (Buckley 2003:123), particularly in relation to child neglect. Buckley argues that child protection workers struggle with this aspect and prioritise situations that are apparently more urgent and possibly 'dramatic' (p.124). Thus a gatekeeping exercise takes place before a family is able to receive services. There appears to be a greater variation in how children who are exposed to adversities outside their carers' control are supported (Srivastava et al. 2005:131). These are 'children in need' whom local authorities are expected to identify and assist (Children Act 1989 s.17). However, there is a wide variation in how or whether assistance is provided and it was this 'bottleneck' (Daniel et al. 2010:254) that caused palliative care social workers so much frustration. Children between the ages of 16 and 18 seemed to be particularly disadvantaged. It was not clear who had ownership of responsibility for such young people, which could lead to workers taking defensive positions (Buckley 2003:188).

Workers reported wishing to discuss a situation before considering making a referral. Given that many of the palliative care social workers had limited experience within their agency of working with families where there were children under 18 years, this would seem to be a reasonable step to take. They did not wish to raise expectations with a family until they were clear what, if anything, might be available. An informal discussion could clarify thinking about the needs of the children. However, there was a reported reluctance on the part of local authority social workers to engage in this way, insisting that a referral be made before any discussion could be had. A joint inspectorate report in England had also found a reluctance to provide advice, guidance and support to agencies raising concerns about children (Department of Health 2002). This refusal is contrary to Government guidance (HM Government 2010).

The relationships as described by palliative care social workers appeared to be very much a 'them' and 'us' situation. There were reports of some local authority workers

giving the impression that they were undertaking the 'important work'. Buckley (2003:53) referred to statutory social workers seeing themselves taking the 'nasty bits' and others doing the 'nice (therapeutic) work'. This 'outside looking in' (Horwath 2007a:1294) perception of others is particularly interesting in terms of workers who are in the same profession. Scott (1997:73) argues that in these situations it is important to consider the 'culture, structure, mandate and imperatives' of the organisation. Local authority social workers, for example, have been reported as wishing to take on a more welfare approach (Spratt 2001) but are constrained from doing so at an organisational level, a view supported by Parton et al. (1997). Social workers, they argue, mainly carry out their duties as defined by their employing organisation and are anxious to avoid being blamed when something goes wrong. The culture of the organisation therefore, has an impact on the different ways issues may be looked at, how problems are reacted to and procedures and rules interpreted (Murphy and Oulds 2000). All this shapes the way professionals work rather than the knowledge, skills and values of the profession (Peck and Dickinson 2008). Banks (2006:122) argues that the social work professional code that describes the duties to service users and the profession should have 'primacy' but agencies require that policy and procedures have precedence. The social worker's role is determined by legal and professional rights and duties, moral duties and social duties (Downie and Loudfoot 1978) as well as procedural rights and duties (Banks 2006). Along with the role come responsibilities to service users, the profession, the agency and society, and these can conflict with each other.

Palliative care social workers are part of much smaller organisations where children are 'secondary' clients (O'Sullivan 1999). They may be seen as more autonomous professionals, operating within a 'professional' model (Banks 2006:138) where their guidance comes more from the professional code of ethics. Local authority workers, on the other hand, are described as following a 'technical-bureaucratic' model, carrying out the tasks and role required by the agency and, as Banks suggests, becoming 'de-

professionalised' in the process. Palliative care social workers therefore display characteristics of 'individualism' (Peck and Dickinson 2008:93) and are able to exercise 'individualised professional discretion' in their decision-making (Carrier and Kendall 1995, cited in Peck and Dickinson 2008:93). But this autonomy was not always welcomed; one worker felt vulnerable in this position as she was 'now more muddied' about her responsibilities compared to her position in the local authority where 'she knew exactly where she was'.

In summary, it is possible to describe palliative care social workers as being able to take a more holistic and generic approach to their work which is in an emotionally charged environment with people from varied socio-economic backgrounds. They have very limited direct access to resources and are able to exercise professional autonomy having fewer organisational constraints on how they operate. Local authority social workers on the other hand, were seen by palliative care social workers to be more narrow in approach, have high workloads, work in a publicly exposed context, be subject to increased bureaucracy, have a less than adequate resource base, work with people in poverty or low incomes and restricted by organisational policies and procedures. In addition, the nature of the work and decisions they have to make can lead to high levels of stress (Stevens and Higgins 2002: Morrison 2007).

Whilst there were many criticisms about the relationship between workers, positive accounts were also reported. Successful joint working was particularly mentioned and although reasons for success were not always made clear, good communication appeared to be a crucial element. Hallett (1995) observes that the chance of success in multidisciplinary work is greatest when personal relationships between professionals are strong. Stevenson (2005:111) calls for improved inter-personal awareness, believing that 'agencies affect but do not determine the process'. Being able to see the perspective of other workers is an important skill (Stanley et al. 2003). However,

bringing about change takes time and effort (Darlington et al. 2005), and is made even more complicated by the number of local authorities and teams with which the palliative care social workers could potentially be involved. Ways of improving relationships suggested by the workers here echo those made in other studies (Horwath 2007a) and found to be effective in others (Murphy and Oulds 2000; Darlington and Feeney 2008). Workers suggested two broad approaches for improving interagency relationships: effective communication both at organisational and individual case level and an improvement in the knowledge base of all workers. The appointment of a 'link' person to promote inter-agency co-operation and to ensure an exchange of communication can only work if consistent and regular efforts are made to maintain the relationships (Buckley 2005:127).

The Local Safeguarding Children Boards in England and Wales and Child Protection Committees in Scotland are responsible for the co-ordination of policies and for promoting and commissioning inter-agency training for personnel in their areas who work with children to meet local needs (Scottish Executive 2005b; HM Government 2010). Making contact and ensuring that they are kept 'in the loop' would enable palliative care social workers to keep up to date with local developments and training initiatives. Palliative care social workers have knowledge and skills they can pass on to local authority social workers in order to encourage a wider awareness of good practice with people who are dying or bereaved (Sheldon 1997; Quinn 1998; Currer 2002). For 'bridges to be built', workers in both services need to work together in a 'spirit of respect, appreciation, and understanding and eagerness to learn' (Tye and Precey 1999:171). Issues relating to training will be discussed in the next chapter, but appropriate joint training can, according to Murphy and Oulds (2000:120), be 'very effective in promoting inter-system collaboration and understanding'. Such training allows participants to learn about each other's 'worlds' and to explore how their systems cross each other. However, as palliative care social workers are likely to be engaged with more than one

authority, understanding organisational strategies and taking part in multiple training events would inevitably take time and effort which they may not have.

From a systems perspective, palliative care social workers can be seen as endeavouring to involve social workers in children's services as part of their action system in order to achieve the goals for the family. There is clear evidence of both collaborative and conflictual relationships, with the latter unfortunately predominating. When a service is not forthcoming, the children's services system becomes a target for change and palliative care social workers strive to exercise influence in order to bring about a resolution. However, the amount of influence palliative care social workers can exert is often not enough. It would seem that they can only draw on their knowledge of and expertise in the effects of pre- and post-bereavement as well as established relationships. Communication patterns appear to be travelling one way - from the palliative care social workers to the local authority.

# Conclusion

Professional behaviour does not exist in an organisational vacuum; the outlook of each organisation sets the priorities. Agency differences can result in the inability of workers to agree on the seriousness, or indeed existence, of a concern about the care of a child. In this study, palliative care social workers and social workers in child and family teams appeared to have unrealistic expectations of each other. Although the palliative care social workers were aware of the pressures that their local authority colleagues were under, they did hope that there could be more of a common understanding regarding the situations that this community of children faced. The repeated negative rebuttals regarding service requests meant that palliative care social workers were extremely reluctant to engage with the local authority, perhaps to the detriment of children and their parents. This study therefore confirmed findings from other studies that inter-

agency co-operation is not a regular feature in everyday practice. There appeared to be a range of impediments to productive, joint work between palliative care social workers and local authority social workers. Occasionally this came from a lack of information about each other's tasks and responsibilities. There was a danger, as one palliative care social worker noted, that the behaviour of professionals resembled in some ways a dysfunctional family. It was, however, heartening to hear about positive experiences. Workers from the two agencies, palliative care social workers believed, could learn from the other. It is hoped that Professor Monro's review on behalf of the Government (Department of Education 2010) into how early intervention and interaction between social work teams and universal child and family services can be improved might shed some light on the way forward.

In the next chapter, I will consider what informs the palliative care social workers' practice with children where there are concerns about their present and/or future care.

# The role of the palliative care social worker: what informs their practice with children and their parents

If social workers are to act clearly, competently and usefully in practical situations, they need to think theoretically. Howe 2008

# Introduction

The previous chapters considered the process of the palliative care social workers' involvement with children and their parents as well as the local authority children's services. In this chapter, I will examine the information gathered mainly from the interviews but also from the focus groups and surveys to draw some understanding of what informs the palliative care social workers' practice in this work. It will include an analysis of their knowledge base, any supervision and training received, and consideration of the influence of previous work and life experiences.

## Knowledge base

Practice wisdom has been defined as 'the accumulated knowledge practitioners are able to bring to the consideration of individual cases and their practice in general' (Sheppard 1995:279). Sheppard suggests that there are three main sources: knowledge gained from everyday life: 'a common sense framework' (Munro 1998:97); from social sciences; and, from the conduct of social work practice. Mixed into this may also be an 'intuitive awareness' as described by Ling and Luker (2000) in their study of health visitors engaged in child protection work, although surprisingly in this study only one worker specifically mentioned having 'gut feelings' about situations but maybe others did not wish to share this 'publicly'.

Although not a direct question raised in the interviews and focus groups, palliative care social workers described in their examples and discussions particular aspects of

knowledge and values that informed their practice when working with children and their parents. In respect of children, this included knowledge in relation to loss and grief and the impact of bereavement; being aware of how children show distress; the concept of resilience; knowledge of child development; children's needs; the role of young carers and the impact this role may have on the children's lives and development. In relation to child protection it seemed that workers were generally aware of the signs and symptoms of abuse but there was no real sense from the data about the depth of knowledge regarding neglect. Carew's (1979:361) study of social work practice found there were few times when responses about practice were overtly based on theoretical knowledge or generalisations from research. Similarly, in this study where theory was mentioned, it was reported in a piecemeal way or provided as a result of my prompting.

One description of work with a family showed how a palliative care social worker used her awareness of child development and children's needs, as well as the impact of terminal illness on the parenting capacity. The worker wrote:

Parent's reduced abilities to do 'active' things with child and sometimes could not be bothered with child. Discussed quality time, utilising network of family/friends to exercise child. Normalised child's anger and suggested safe ways he could work this off under supervision of responsible adult. Gave literature and links to useful organisations and web sites.

This was a response taken from the questionnaire and is a good example of where workers tended to use more official, technical and colloquial language when writing about their practice than talking about it, as referred to in chapter nine.

Workers recognised particular issues in situations where children were being cared for by a lone parent, especially where there were no family members able to care for the children after that parent's death. Knowledge of family dynamics, particularly in the area of management of change, was recognised as important, as was knowledge of

attachment theory. Included here was the impact of estranged relationships after death as well as the impact of the stress caused by the illness and impending death on the family. The legacy of past abuse in the family was considered to be a dynamic in some situations, particularly where this concerned someone who was currently a grandparent and who may have contact with the children.

Knowledge about managing conflict, being able to say 'really difficult things' to families, working in partnership with families and other workers whilst keeping within the boundaries of confidentiality and social work values, featured in the focus group discussions and interviews. Person-centred work that appears to be such a feature of palliative care social work, places importance on the values of unconditional acceptance, non-possessive warmth and empathy (Sudbery 2002). These values, although not specifically mentioned, were evident in the way palliative care social workers described their practice. They appeared to see the whole person. This included having knowledge about and an understanding of other cultures and how different religious beliefs can impact on the care of the dying person and the family. There was evidence of workers endeavouring to practice in a 'culturally safe' way as described by Oliviere (1999), as for example in Family E in chapter nine.

Two palliative care social workers specifically mentioned that a major part of their theoretical underpinning was Maslow's (1970) hierarchy of needs. In their view, basic needs had to be met before achieving other needs of a 'higher order', for example belonging and self-esteem. One worker explained:

You have to attend to the practical safe things first before thinking about counselling because until you have got all of that in place you can't be offering counselling as a solution.

However, Walsh (2005:96) has put forward an alternative view that when a person is dying it may be more relevant to invert the needs. For example, food may no longer be

a motivator due to illness but self-fulfilment may be. The person may need help to reconcile past events and relationships to be able to make sense of their lives.

I did not specifically ask about the participants' knowledge about the relevant child care and child protection legislative framework. There was minimal reference to Children Acts specifically and only very occasional reference to the term 'children in need'. A minority of people both in the focus group and interviews volunteered that they did not consider themselves to be as up-to-date with regard to legislation and policies as they would have liked.

A palliative care worker who was relatively new to the post and to working with children, explained how she had prepared herself for the new tasks:

I have done quite a lot of reading on developmental work and understanding of issues around bereavement and children. As a new worker to that area that is where I focused my learning. I have certainly gone back to child development work, gone back to social work theory again recently to look at movements in that, attachment theory and other issues. I have really tried to underpin what I am trying to do from a theoretical place.

Two other workers spoke of undertaking literature reviews on bereavement and loss, particularly in relation to children, as well as gathering together relevant literature and materials that could be used in direct work with children. Direct reference to research findings was rare. One worker spoke of research on the best predictor of a child coping being the parent's coping ability; another, on the gender link between the parent who was dying and the child, together with the name of the journal; and another that when support is given to a parent the result is better outcomes for the child. However, whilst direct reference was limited it may not reflect the frequency of research on which practice was based. Buckley (2003:68) in her study found little evidence of social workers using theory to inform their practice, particularly in regard to child development and family dynamics. However, findings here present a different view as palliative care

social workers appeared more able to articulate a theoretical base for this aspect of their work. The areas of knowledge that are identified here support the literature discussed in chapter two.

#### Supervision and support

Supervision in social work is generally considered to be essential (Coulshed and Orme 1998). Indeed, the Social Work Task Force (2009:29) stated that it is 'an integral element of social work practice, not an add-on'. It recommends that there should be a 'strong supervision culture' in all organisations that employ social workers, and where a line manager is not a social worker, professional support should be provided. In Lord Laming's (2003:211) view: 'effective supervision is the cornerstone of safe social work practice. There is no substitute for it'. Lishman (1994:39) considers it is different from management accountability and it should be built on a 'professional relationship'. The general consensus is that it should be a reflective analysis (Darlington et al. 2002) with an emphasis on how a worker forms judgements, a consideration of ways 'worker bias' can be reduced (Coulsed and Orme 1998:35) as well as a review of how decisions are made (Milner and O'Byrne 2002:177). Pritchard (1995) adds that supervision should ensure that practice is sensitive to the needs of service users. It should also be an important part of workers taking care of themselves (Hawkins and Shohet 2006). In palliative care there is an additional dimension that should be recognised in supervision: the acknowledgement and management of strong feelings that can occur when working with people who are facing death (Currer 2002:211). It is Chowns' (2005) view that working with children can be as demanding as working with death and dying and so doing both at the same time is likely to be stressful. Supervision should provide an opportunity for these feelings to be recognised and appropriate support given. The Social Work Task Force (2009:29) sums up all these views in its statement on the purpose of supervision:

... to review day to day practice and decision-making, plan their learning and development as professionals, and work through the considerable emotional and personal demands the job often places on them.

The topic of supervision was explored in the focus groups and interviews and it appeared that a variety of models of supervision existed. Everyone received supervision in some form. This could be from an external person (someone not employed in their work place), for example, an experienced social worker, a lecturer in social work, a counsellor or from someone inside the agency, usually the line manager who was either a social worker or someone with a nursing background. One worker described how she had to arrange her own supervision, and the hospice agreed to pay. Out of the 17 palliative care social workers interviewed, 10 received supervision from a social worker. One worker in a focus group spoke very highly of her external supervisor, particularly in relation to work with children and families:

I am really lucky because I have external supervision and the social worker I see has worked in child care for a long time and she is also a qualified counsellor and supervisor – I dread her retiring because it would be jolly hard to find someone who has that breadth of knowledge.

Those palliative care social workers who described their supervision arrangements seemed satisfied with what was provided, although one worker did comment that it was sometimes difficult to get together with the line manager when both were busy. Recognition was given to the importance of sharing their thinking with someone and checking out attitudes to find out if 'they were getting over-involved and not seeing the wood for the trees'. For immediate discussion regarding a child, those with external supervision would turn to their line manager (often a nurse), or the person who drew up their agency's child protection policy, or staff in the local authority children's services. Another source of support was the 'named nurse for child protection'. If there was an issue they were struggling with or for information and advice, contact might be

made with the social services, either with a team leader or a duty team. So, unless the internal supervisor him or herself had a child care background, finding someone who could assist was not always straightforward. The breadth of the job made it difficult to find someone who could sufficiently supervise all aspects.

Palliative care social workers who were located in a palliative care social work team spoke of the importance of peer support, being able to discuss situations in an informal way or at specific case presentation sessions. There were those who were themselves managers of other palliative care social workers in the team and provided supervision. There were particular difficulties for those palliative care social workers who were lone social workers in their agency and had no opportunity to share concerns with social work colleagues and, in these cases, mention was made of contacting colleagues in other palliative care agencies. One lone hospital worker met with a hospice worker for peer support every six weeks and another with a group of hospice social workers every three months. For these lone workers, and indeed for some others, there was a feeling of vulnerability and the lack of backup that a social work team could provide. This vulnerability regarding child care concerns was succinctly described in a focus group by one lone worker:

You know that there is abuse and when you are a lone worker it is hard to see round it until someone outside the system comes in and you are able to talk it through with them and they are able say to you 'Well actually I think it is abuse' and you can say 'Yes I think you are right we need to do something about it'. And sometimes it is quite hard to do, isn't it, when you are stuck in it?

In the survey, palliative care social workers were asked: If you ever have cause to be worried about the care and welfare of a child of a service user, who would you discuss this with?' Out of the 56 responses there were five main groups of people identified; people indicating more than one. The line manager or manager of the team was the predominate group, followed by the staff in the local authority child protection team or

children's services. There was an even spread among the remaining groups of supervisors, colleagues in the multidisciplinary team and social work colleagues in their own agency.

As palliative care social workers cover the whole range of age groups it may be that not every supervisor has an interest and/or expertise in child care matters. If there was no recognition that working with children was part of the palliative care social worker's responsibilities then it would be unlikely to be a topic for supervision. Helm (2010:57) points out that supervisors need to have 'a sound, contemporary, understanding of theory'. Supervision, according to Haringey LSCB (2009:24) in its review into the death of Baby Peter, 'will not improve the quality of the practice unless the manager has competent knowledge and skills which are relevant to the case'. In addition, supervision should provide an opportunity for the worker to reflect on their work and discover new learning that can emerge from difficult situations and be a crucial part of self development (Hawkins and Shohet 2006:5). It should be a vehicle for considering the worker's professional development and what further training might be required to enhance the worker's knowledge base and effectiveness in practice (Lishman 2002:15). It was difficult to gain a clear view in this study of how well supervisors were able to support the palliative care social workers in their work where there were concerns about children.

# Training on safeguarding children

'The recognition of the need for child protection training is not new' (Baverstock et al. 2008:64). There has been a stress on the importance of training in the various inquiries into the deaths of children through abuse over the years (Laming 2003). Statutory guidance on staff development and training now exists (2010:91) which includes 'being able to recognise when a child may require safeguarding, and knowing

what to do in response to concerns about the welfare of a child'. In particular those who work with adults who are carers and who may be asked to contribute to assessments of children in need:

... should have a higher minimum level of expertise: a fuller understanding of how to work together to identify and assess concerns and plan, undertake and review interventions (p.95).

In England and Wales, Local Safeguarding Children Boards (LSCB), and in Scotland, Child Protection Committees, provide a range of inter-agency training. Section 11 of the Children Act 2004 places a duty on NHS Trusts in England to ensure that all staff working with children and families have a regard to the need to safeguard and promote the welfare of children. This includes training. There has been an emphasis on the need for inter-agency training in order to help develop a shared understanding of the roles and responsibilities of the various professionals. This, in turn, it is argued can contribute to more effective and integrated services; improved communication between professionals; effective working relationships; and, sound decision-making (DfES 2006:4.3).

The evidence base for the assumption that inter-agency training enhances interagency working is 'rather slim' (Ogilvie-Whyte 2006; Taylor and Daniel 2006:179). The evidence from a multidisciplinary interagency training in one London Borough (Barker 1998) however, suggests that child protection awareness was raised and communication between agencies improved. The training highlighted the complexities of the work and the demand on professionals, particularly in terms of how they defined need and risk. There has also been a recognition that whilst task and process are important elements in the training, in order to facilitate better working relationships, there should be more emphasis on process (Glennie 2007:174). Evidence from a recent substantial evaluation of inter-agency training to safeguard children in England

(Carpenter et al. 2010) shows that there were positive outcomes in terms of increased knowledge base and 'very substantial improvements' in self-reported understanding of the role of other professionals involved (p.ii). Participants 'very highly valued' the opportunity to learn together.

The emphasis in policy and guidance documents is that all staff who have contact with children should have ongoing training, both single and multi-agency which 'is tailored to address the identified needs of staff in terms of their level in the organisation and stage of professional development' (Department of Health 2004:170). The non-statutory guidance *The Common Core of Skills and Knowledge* (National Collaborating Centre for Women et al. 2010) covers six areas of expertise that everyone who works with children and young people, even as only part of their job, should be expected to have. It is recommended that these should be used in induction, training and staff development.

In this study, the need for training and keeping up to date in child care matters was generally recognised as important, even if could be quite an effort to achieve. Where training was not immediately available, there were palliative care social workers who spoke of gathering information from the Internet or by reading articles and books. The need for good quality training in safeguarding children, including the relevant legislation was considered to be important, although the latter aspect in particular was not always forthcoming and many workers felt out of date.

From the responses in the survey, the amount of training workers had received on safeguarding children ranged from none during the previous two years to two days a year. A quarter of respondents had not received any training on safeguarding children in the past two years. Of those who had, about a quarter had found the training not very supportive and about half generally supportive or very supportive.

Findings from the interviews showed that child protection issues were generally first raised in agency induction training which included terminology, signs of abuse and agency procedures. This training was generally followed by multidisciplinary training provided in-house and there were examples of palliative care social workers being responsible for its planning and delivery. The training was generally mandatory, although one worker commented that in her experience it had become 'a bit jaded' and needed revising. If the in-house training was not provided by a palliative care social worker in a focus group described her experience:

It was health who came in. We had nothing till last year and then somebody from health came in - it would be the child protection officer, Acute Trust - and did some talks. And my understanding is that they should do that every year but I am sure it is over 12 months since the last. And they haven't come back ... it should be mandatory every 12 months, like your fire lectures.

This view that it should be mandatory and within time limits was repeated regularly.

Whilst there were palliative care social workers who acknowledged that accessing training was easy or relatively easy, others found it relatively difficult to do so. There were reports of workers funding their own training. There were workers who could access courses delivered by local authorities and LSCBs. This training was generally provided free of charge. Those courses supplied by the LSCBs in English authorities were singled out for particular mention by some, whereas others were not aware of the Boards' existence. One worker however described how she was the hospice representative on the local safeguarding children's panel and also part of its training pool.

Those who had a link with the Boards spoke positively about being kept up to date and accessing their web sites for information. Gaining access to or having knowledge of

training provided by local authorities was a particular difficulty for some whereas others had positive experiences in this area. There were views that the local authority training departments did not know that palliative care social workers existed, or if they did, did not understand their role and did not include them in their circulation lists.

Various Primary Care Trusts also provided training on child protection which members of the multidisciplinary team were expected to attend. There was a wish expressed that training be more 'joined up' with adult and child and family social workers attending as well as medical and nursing staff. Separate training, it was believed, perpetuates the gaps between the services as well as the gap of understanding. The split between adult and children's services meant that some palliative care social workers missed out on training on safeguarding children. One worker spoke of her training being 'very out of date' due to this split and the non-expectation that palliative care social workers would be involved with children. The emphasis was rather on adult protection. In contrast, there was one worker who described a positive experience when she attended a two-day joint training event between social services and health aimed at people working primarily with adults who may come into contact with children.

It is possible that this view that palliative care social workers' involvement with children is very limited means that the workers themselves and/or, their managers do not seek or are not encouraged to attend safeguarding training. In the survey, there were comments to this effect and one person, employed by the NHS and working in a combined inpatient and day hospice, wrote that she was not encouraged to access such training and that in fact she was prohibited from working directly with children. Another described how she had to specifically request such training and provide a rationale for the request. There were also palliative care social workers who considered that they

Chapter Eleven

needed to be more proactive in finding out what training was available and accessing it themselves.

In the survey, respondents were asked about their future training requirements. About half felt that an annual or regular update would be beneficial and this would reinforce their basic training. Updates on legislation were particularly mentioned, as was information on local policies, procedures and practices along with services available. Four workers commented that they were happy with the present situation already receiving sufficient training, regularly updated, which met their needs. Seven either gave no response or said 'none'. Three asked that the training be palliative care specific. The other responses were wide ranging, for example, how to talk to parents where there appeared to be a real concern, how schools support children facing death and, dynamics of the dying phase on child protection. Interestingly, no mention was made of direct communication with children.

### Other influences on their practice

Those who had previously worked in the local authority, particularly in child and family teams or child protection, found this experience very beneficial in their current posts in terms of their confidence and knowledge. Of the 17 palliative care social workers who were interviewed, seven had local authority experience of child care work; three as generic workers and four in specific child care teams. One worker still described herself as a child care worker. Another explained that frustration was the reason for moving from local authority child care services where she had been well aware that she was not practising with children and families in the way that she wished to because of the 'enormous' case loads.

Chapter Eleven

There was a spread of other previous work experiences with children including daily minding, youth work, residential child care, teaching, and working with children with disabilities. Those who had worked in the mental health field or more general work with adults found their understanding about anxiety, loss and depression relevant and endeavoured to fill gaps by attending relevant courses as well as reading on child care issues.

Those palliative care social workers who had children of their own saw this personal experience as an asset in helping them understand and communicate with children; those who did not spoke about their experiences with children of friends and family in a similar way. There was also mention by some of the losses in their own lives that had helped them have some understanding of others' grief. Workers recognised that their own upbringing and values shaped their current thinking and was summed up by one worker:

I bring my whole self when I am doing this work. I have a great sense of social responsibility and justice and all of those old-fashioned social work values. All spoke in various ways of their commitment to children and their interest in families. One worker explained:

I enjoy working with children and I think my influence comes from the fact that I feel that everybody has the right to have a decent death, to be told the truth, including the children.

### Conclusion

It was apparent, although not always referred to directly, that the practice of palliative care social workers in their involvement with children and families was informed by various factors. There appeared to be a good knowledge base about the impact of bereavement on children and about child development and attachment as well as what constitutes child abuse, although in contrast, neglect was not so frequently referred to.

Chapter Eleven

However, there were workers who felt less confident about recent child-care policies, guidance and legislation. Basic training on child protection, mainly on terminology and recognition, was provided with some palliative care social workers providing this themselves for other staff in their workplaces. Knowledge about and access to training outside the workplace on safeguarding children varied. There was acknowledgement that personal and previous work experiences gave palliative care social workers confidence in their work with children.

As supervision for social workers is now considered to be an essential requirement, it was interesting to note that there were palliative care social workers who did not receive regular supervision from another social worker. This was more frequently the case for the 'lone' practitioners. Having someone to share professional concerns about the care of the children was considered to be important but not always available, which left some workers feeling vulnerable.

# Chapter Twelve Conclusions

Children are precious ... they are entitled not just to the sentiment of adults but a strategy that safeguards them as children and realises their potential.

### Introduction

The aim of this qualitative study was to explore the role of palliative care social workers in safeguarding and promoting the welfare of children of adults who are receiving specialist palliative care. This study was comprised of focus groups, a survey and semistructured interviews involving a sample of palliative care social workers. The purpose of the study was to both draw and analyse the picture of their 'real-life' work with children and their families. I believe that this wide-ranging examination of their involvement has illustrated not only the dedication of the workers involved and the dynamic nature of their task, but also the benefits and difficulties in carrying it out, as well as the 'swampy territory' in which it is situated. This concluding chapter presents the main conclusions and a summary of the key findings. It is followed by a reflection on the systems perspective, the strengths and limitations of the study and a consideration of the implications of the findings for practice and future research.

### Main conclusions

Safeguarding and promoting the welfare of children is not a straightforward task, particularly with regard to children of adults who are receiving specialist palliative care. Palliative care social workers considered it to be part of their role, but generally it was a very small element of their total workload. Whilst the focus of the work is the adult who is terminally ill, in keeping with the holistic nature of palliative care palliative care, social workers in the study were mindful of the impact of the illness on children in the family; reference was made to 'unwitting emotional neglect' of children by their parents.

Engagement with a family could be very short-term in nature due to the timing of referrals and a rapid progression of the illness and cover a wealth of practical and emotional issues. There was a general feeling of working without support; to a lesser extent in relation to their own agency and to a greater extent to local authority children's services. Part of their role in safeguarding children was knowing whether and when to make contact with these children's services. This appeared to centre on how thresholds regarding formal intervention were interpreted by all parties as well as the quality of relationships between them.

Whilst there was evidence of direct work being undertaken by palliative care social workers with children, support was mainly given to the children via their parents. This seemed to depend on the workers' own interpretation of their role, as well as that of their agency and colleagues, plus any previous experience, training and their confidence in this aspect of work.

# **Key findings**

In order to achieve the aim of the study, six key research questions were developed. I will first summarise the findings for each of these questions before presenting four other key findings that came to light during the analysis.

The first question concerned the palliative care social workers' understanding of the term 'safeguarding children'. It transpired that the workers generally saw the term as embracing a wider range of aspects than child protection alone. Certainly the welfare of children concerned them generally and particularly because of the children's emotional needs during their parent's terminal stage of illness. The findings suggest that whilst palliative care social workers are aware of the requirement for them to safeguard and promote the welfare of children, the extent to which this was carried out varied

considerably. Most embraced it enthusiastically and competently, others less so. However, not all agencies were supportive of direct work being undertaken with children; the emphasis being rather on work with parents. These agencies perhaps did not fully understand the implications of the governments' safeguarding policies.

The second question related to how palliative care social workers identify situations where there are concerns about the care of children and how the palliative care teams are involved in that identification. In the main, palliative care social workers as social workers generally, are dependent on others for referrals who may not be child-centred or aware of children's needs. Almost all referrals come from members of the multidisciplinary team. There were some instances, however, when workers did proactively engage with a patient, but this was in a minority of situations and only in inpatient or day hospice care. Palliative care social workers are in a unique position by being in a medically-orientated setting where there is an ethos of holistic care. This involves seeing the family as the 'unit of care' (WHO 2002) which gives 'authority' to the multidisciplinary team to consider the needs of children as well as the ill parent. Unlike other social workers in adult care, being part of such a team means that palliative care social workers are in a position to exert some influence on the 'child awareness' of their professional colleagues and to ensure that children's needs are not lost in the particular concerns about the patient-parent.

In general, palliative care social workers did not have a clear profile of their caseload or pattern of referrals. Compared with the overall number of patients seen by palliative care social workers, the number of those with children under 18 years was reported to be relatively small, although growing. Many palliative care social workers were therefore not working on a regular basis with these families; indeed, for many, their experience was very minimal. Consequently, there was a lack of opportunity to build up knowledge,

understanding and expertise, not only about child-care practice itself but also the underlying legislation, policies and guidance.

The third question inquired into how palliative care social workers go about their work with children and their families. The study established that the work is wide-ranging, although generally with parents rather than directly with children. This was a surprise. Palliative care social workers claimed ownership of a child-centred approach to their work. However, where there were concerns raised about the care of particular children, it appeared that they were not always seen, let alone spoken to by palliative care social workers. The workers did not always deem this to be necessary and work was undertaken through the parents or other health care workers. There were also those who considered that they did not have the skills and/or experience to offer a suitable service to children directly.

Palliative care social workers have a large amount of professional autonomy in how they conduct their work, particularly those who are lone workers in their agency. Many, therefore, are in the position of deciding how much they can or should do themselves in terms of direct working with the children of dying parents. This could be due to inclination, ability, confidence and resources, as well as appropriate support within and outside the agency. What was apparent throughout the study was the amount of time, energy and thought that was expended by workers in their efforts to assist children and their families.

The fourth question centred on the palliative care social workers' view of the quality of their relationship with social workers within the local authority children's services. The findings here confirm those in other studies which have examined relationships between adult care service workers and those in children's services. In the main, inter-agency co-operation was not a regular feature in everyday practice and there was frustration and

disappointment that this should be so. There was a danger, as one palliative care social worker noted, that the behaviour of professionals resembled in some ways a dysfunctional family. Whilst there were pockets of co-operation and productive joint work, these were in the minority.

There appeared to be a range of impediments to productive co-operation between palliative care social workers and local authority social workers in child and family teams. For some palliative care social workers, there was a dread of 'devastating' a family (Buckley 2003:182) by involving child and family local authority services, perhaps indicating a reluctance to take on the social control function of child protection activity. Occasionally difficulties occurred because of a lack of information about each other's tasks and responsibilities. There did not appear to be any sustained attempts by managers in either setting to alter the situation. Individual palliative care social workers had tried but these tended to fall by the wayside due to lack of time and effort. A commonly held view by palliative care social workers was that local authority social workers did not have a thorough enough understanding of death and dying and its impact on families, particularly in relation to children, and hence shied away from involvement. This lack of understanding was considered to be due to deficiencies in contemporary social work qualifying training.

The fifth question referred to the amount of support and training palliative care social workers receive about safeguarding children. Palliative care social workers participated in basic agency training in child protection and some, further inter-agency training in safeguarding children. However, there were those who had difficulty in knowing about and accessing relevant training. In some agencies, the palliative care social worker was the child protection officer and had taken on the responsibility of training other staff members as well as drawing up the agency child protection guidelines and procedures.

The amount and type of professional supervision and support varied: not all had access to supervision from another social work professional.

The final question asked for insights from the views of the palliative care social workers which may improve how children may be safeguarded in the palliative care setting. These are included in the following section on 'Implications of the findings for practice and future research'.

There were four other key findings. First, having a parent who is dying is a stressful life experience for any child. For those children where a parent's immigration status was in question there were additional stresses that challenged not only the family, but also the palliative care social workers in their task of endeavouring to promote the children's welfare.

Second, children of lone parents who were dying could also be placed in an uncertain situation. Not only do they have to cope with the terminal illness of their parent and the subsequent loss, but also their future care may not be known or secured.

Third, there were reports of agencies that either did not encourage or allow palliative care social workers to provide direct support to children after the death of a parent. The policy rather was to provide support to them via the surviving parents. This would seem to suppose that the relationship between the parent and child will be a positive one, which may not always be the case. This was particularly evident in Family C described in chapter nine. Research has established that children welcome continuity of support and this would seem to be particularly important in a time of loss.

Fourth, there is a neglect of consideration of socio-economic issues in much of the literature, especially given that the incidence of cancers among younger people, who

are likely to have young children, is increasing (Cancer Research UK 2009). Work on health inequalities suggest that this increasing incidence will primarily impact on those in the lower socio-economic groups (Bywaters 2009) where children may already be disadvantaged. Indeed, there were examples provided in this study where this was found to be the case. At the time of writing, the economic climate means that public services are being reduced, benefits reviewed and the cost of living rising, all of which will have an impact on all those who are ill, but particularly those in the lower socioeconomic groups.

#### Reflections on the systems perspective

It has been helpful to keep a systems perspective in mind as I journeyed through this study, especially during the data analysis stages. I took the view that 'the meaning of behaviour has to be understood in context' (Oliviere et al. 1998:55) and systemic thinking therefore helped me to consider the effect of the wider systems and influences on the ill person and their world, including the multidisciplinary team of the hospice or hospital. These contexts include those in which the ill person lives: their work-place, friendships, religious group, neighbourhood, and the reaction of all these to the ill person and family. This view particularly assisted me in my contemplation of the children's and parents' experiences, how children come to the attention of palliative care social workers, the workers' subsequent engagement with parents and children, their interaction with local authority social workers in children's services, and finally the influences on their practice.

In particular, when considering the experiences of children and their parents, I was mindful of Rolland's (2005) Family System-Illness Model. He likens the impact of the more serious and debilitating forms of cancer to the addition of a new member to a family, one with 'special needs', that can compete with the needs of the 'real children' for

potentially scarce family resources (p.2591). One way of considering a family's ability to cope, Rolland believes, is the consideration of the impact of past illnesses, losses and prolonged adversities, for example unemployment and asylum seeking, on the individuals and the family as a whole. The family's belief systems will also have a bearing on this, including those about health issues. This historical view would help explain and predict the family's current style of coping, adaptation and 'creation of meaning' (p.2589). Those families with past unresolved issues or dysfunctional patterns of communication are likely to struggle under the strains that a dying parent can bring. Within many of the examples provided by the palliative care social workers there were these different factors present, resulting in stress which impacted on the children and the parents' coping abilities. The illness therefore does not come alone and palliative care social workers appeared to be aware of the multi-layers in families.

It was very apparent, particularly from the narratives, that dying is a 'family affair' (Hudson and Payne 2009). There were changes in the micro-systems (Bronfenbrenner 1979) in the family, for example brothers or sisters took over the parental role for siblings; estranged parents became re-involved with their children; and relatives such as aunts, uncles and grandparents became carers of the children whose parent had died. The quality of the past and present relationships in a family also impacted on the present and future care of the children.

There was evidence that some palliative care social workers generally approached their work in a systemic way, although they did not always state this explicitly. There were those who demonstrated an awareness that systems theory locates the patient and their family within a number of systems that can have a negative or positive impact on individual members. They recognised that societal systems could both have a hindering or enhancing effect on a family and used their knowledge to mobilise resources and support. However, whilst there was evidence that workers did consider the children and

their needs, the study did not look in depth at the assessment process and so it is not possible to judge to what extent workers considered the children from an ecological perspective. Parenting capacity was a consideration but the knowledge that a parent is soon to die is an extra dimension that workers had to take account of. This dimension is very seldom discussed in the literature.

Palliative care social workers do not work in isolation. They are part of the 'changeagent system' (Pincus and Minahan 1973) which employs them. When endeavouring to bring about change within that system, for example in relation to referral practice or ongoing work with children post-bereavement, the workers' continuing relationships with other people in that system and their status will be factors in whether and how this can be achieved. Lone palliative care social workers may feel particularly vulnerable in a medically-orientated setting and this was reported to be so in this study. Pincus and Minahan (1973:247) draw attention to the exercise of influence that underlie the activities of social workers in their engagement with different systems. From the study it was possible to see that the palliative care social workers' bases of influence derive from their knowledge and expertise, their reputation, their status and established relationships. Their status is linked to that of the agency and is considered to be viewed differently by service users to those workers in local authority children's services. The main area where there were difficulties in exercising influence was in relation to social workers in the children's services of the local authorities. Here the relationship could be described in the main as 'conflictual' although there were a few examples of 'collaborative' relationships between workers. Palliative care social workers seemed to generally believe that they were unable to exercise influence in their efforts to secure the support for children and their families that they thought was appropriate.

Palliative care social workers are themselves influenced by their knowledge, value and practice base, their profession, the multidisciplinary team, the agency, and society as

well as the family with whom they are working. Society here includes legislation and government policies and guidance as well as the general population's expectations of social workers and health care workers with regard to the care of children.

### Strengths and limitations of the study

This thesis contributes to the knowledge about how workers in adult-focused care and health services safeguard and promote the welfare of children with whom they have contact in their everyday work. Whilst the experiences and views are from a group of specialist social workers based in a particular setting, there are themes that are relevant to other workers in other settings. It also continues the debate around the extent to which services for adults are child-centred, and how staff may be supported in their task of safeguarding and promoting the welfare of children.

This study extends the knowledge about the practice of palliative care social work in a particular aspect of their work which is currently under-researched. As the experiences, opinions and insights presented are based on those of a limited number of palliative care social workers, it cannot be assumed that the findings are representative of all. Indeed, as people self-selected and volunteered their time, there is an assumption that they had an interest in the topic. However, those who did participate supplied rich data which was gathered in three ways in an effort to widen the information base, which is a strength of the study.

This is a 'one-dimension' study rather than a 'multi-voiced' reconstruction (Guba and Lincoln 1994:115) in the sense that no children or parents were involved in the research process. I am aware that throughout, and in particular in chapter six, I have interpreted children's voices through an adult lens: the palliative care social workers' (often via parents) and then my own. However, I have endeavoured throughout to ensure that, as

far as possible, there has been a child-centred approach on my part and children's circumstances represented. The study also lacks information from the other professionals involved with the palliative care social workers in their duty to safeguard and promote the welfare of children, in particular members of the agency multidisciplinary teams and social workers based in local authority children's services. The roles that these professionals play have been seen in this study to have an influence on the way in which palliative care social workers are able to carry out their role and responsibilities with regard to children of adults receiving palliative care.

### Implications of the findings for practice and future research

If the argument is accepted that children are to be more centrally located in the family then I would support Chowns' (2006:265) recommendation that palliative care should have a culture that is genuinely 'family' rather than 'adult-family-focused'. This means that all the settings that provide palliative care services should acknowledge the presence and involvement of children. If all staff think in this way, it would help to ensure that children in the family are acknowledged from the start and their circumstances and needs taken into account both before and after the death of a parent. Children need to be recognised as social actors with their own views and concerns that may differ from those of their parents. At the same time, as child protection has now widened to encompass not only preventing abuse but also promoting the welfare of children, agencies 'need to find ways to form a protective network around all children' (Taylor and Daniel 2005:299). This is a challenge not only for palliative care social workers but also for all workers across agencies.

Further, in order to undertake direct work with children, palliative care social workers need to be equipped to have 'open conversations' with children to enable their voice to be heard (Oliviere et al. 1998). There are dangers in assuming that the perceptions of

adults accurately and adequately represent those of a child. Workers should also be willing to 'be alongside' children in all circumstances, particularly when they are sad, and acknowledge their own feelings that this generates. If there are gaps in the palliative care social workers' skills leading them to limit the service they offer to families where there are children, there should be mechanisms in place to enable them to be filled. These would include appropriate training, as part of continuing professional development, and ongoing professional support from someone knowledgeable about child care matters. Indeed, the findings suggest that there is a particular need for interagency training whereby social workers in local authority children's services and those in palliative care participate together in exploring ways of working with children who are living with a terminally ill parent. There is a role here for the LCSBs in England and Wales and the Child Protection Committees in Scotland to encourage such training, and to identify those palliative care social workers in their area in order to ensure that they are kept abreast of training opportunities and up-to-date information about safeguarding children.

In their study, Beresford et al. (2007:152) highlighted the problems of access and referral to palliative care social work. They recommended that workers undertake a systematic audit of referrals to their service. I would support that recommendation with the proviso that there are audit categories that include reference to children under 18 in the family and also to lone carers. I would also recommend that the audit be widened to include the number of referrals to the palliative care service where there are children under 18 years in the family. This would allow a picture to emerge regarding the onward referral rate to palliative care social workers. It would also provide palliative care social workers with information to assist them in arguing for specific resources to assist them in their work with children and families, including training and support.

Future research might also benefit from a case study approach to explore the work of one or two palliative care social work units. This should include children and parents, palliative care social workers, palliative care team members, senior management of the agency, local authority social workers in child and family teams, an analysis of documents and observations of team meetings. This would ensure a more 'rounded' picture in order to highlight the dilemmas and tensions as well as positive outcomes. Exploring two different units would allow for comparisons to be made. The Delphi technique might be usefully applied to a part of the research to determine a multiprofessional understanding about safeguarding children in palliative care practice with adults.

There were reported instances of collaborative relationships between palliative care social workers and social workers in the local authority children's services. Research into what makes these collaborative relationships work would be constructive, not only in relation to the work described here, but also to that between other agencies and children's services.

The impact of the immigration status of the family on the children's welfare during preand post-bereavement is a key finding of this study. This aspect has not previously been researched and therefore is 'hidden' from public scrutiny. The amount of distress and stress that children and their surviving parents experience calls for an urgent need for research. Doing so would provide another dimension to the literature on the implications of immigration for children and inform the agencies with whom they are involved.

## **Concluding remarks**

It is my belief that all social work practitioners, in whatever work setting, should be able to recognise direct and indirect indications of a child being in need of help. Combined

with this they should be able to recognise aspects of parental behaviour that might be expected to have a negative impact on their children. This does not mean though that palliative care social workers should become 'child protection officers'. Rather they should consider if a child's needs are being met and, if not, whether their development is being, or is likely to be, impeded as a result. Such an assessment should include what they themselves are able to provide in terms of direct support to children and their parents and what assistance might be required from elsewhere.

This study adds to the limited existing knowledge about the practice of palliative care social work. If the scope of the work is not promoted and acknowledged more widely, then it is likely that it will remain a 'service in the shadows'.

Adcock, M. (2001) The core assessment: how to synthesise information and make judgements. In J. Horwath (ed) *The Child's World. Assessing Children in Need*. London: Jessica Kingsley Publishers.

Addington-Hall, J.M. and McPherson, C.J. (2001) After-death interviews with surrogates/bereaved family members: some issues of validity. *Journal of Pain and Symptom Management*, 22(3), 784-790.

Ajemian, I. (1993) The interdisciplinary team. In D. Doyle, G. Hanks and N. MacDonald (eds) *Oxford Textbook of Palliative Medicine*. Oxford: Oxford University Press.

Alderson, P. (1992) The rights of children and young people. In A. Coote (ed) *The Welfare of Citizens: Developing New Social Rights.* London: Institute for Public Policy Research/Rivers Oram Press.

Alderson, P. (2008) Young Children's Rights: Exploring Beliefs, Principles and Practice (2<sup>nd</sup> ed). London: Jessica Kingsley Publishers.

Aldgate, J. and Statham, D. (2001) *The Children Act Now: Messages from Research.* London: The Stationery Office.

Aldgate, J. and Tunstill, J. (1995) *Making Sense of Section 17: Implementing Services for Children in Need within the Children Act 1989.* London: HMSO.

Aldgate, J. and Tunstill, J. (1996) Making Sense of Section 17. London: HMSO.

Aldridge, J. (2008) All work and no play? Understanding the needs of children and caring responsibilities. *Children and Society*, 22, 253-264.

Aldridge, J. and Becker, S. (1993) Punishing children for caring: the hidden cost of young carers. *Children and Society*, 7(4), 376-387.

Aldridge, J. and Becker, S. (1999) Children as carers: the impact of parental illness and disability on children's caring roles. *Journal of Family Therapy*, 21, 303-320.

Aldridge, J. and Becker, S. (2003) *Children Caring for Parents with Mental Illness. Perspectives of Young Carers, Parents and Professionals.* Bristol: The Policy Press.

Allmark, P. (2002) The ethics of research with children. Nurse Researcher, 10(2), 7-19.

Alston, M. and Bowles, W. (2003) *Research for Social Workers: An Introduction to Methods (2<sup>nd</sup> ed).* London: Routledge.

Altschuler, J. (1997) *Working with Chronic Illness: A Family Approach*. Basingstoke: Macmillan.

Alvord, M.K. and Grados, J.J. (2005) Enhancing resilience in children: a proactive approach. *Professional Psychology: Research and Practice*, 36(3), 238-245.

Anderson, G.R. (1995) *Children and HIV; orphans and victims*. In K.J. Doka (ed) Children Mourning, Mourning Children. Washington: Hospice Foundation of America.

Andrews, M., Squire, C. and Tamboukou, M. (eds) (2008) *Doing Narrative Research*. London: Sage.

Archard, D. and Skivenes, M. (2009) Hearing the child. *Child and Family Social Work,* 14, 391-399.

Arthur, S. and Nazroo, J. (2003) Designing fieldwork strategies. In J. Ritchie, and J. Lewis (eds) *Qualitative Research Practice. A Guide for Social Science Students and Researchers.* London: Sage.

Association of Palliative Care Social Workers (APCSW) (2006) *Introduction to Palliative Care Social Work.* London: Association of Palliative Care Social Workers.

Axford, N. and Little, M. (2004) *Refocusing Children's Services Towards Prevention. Lessons from Literature.* Dartington Research Unit. Available: http://www.education.gov.uk/research/data/uploadfiles/RR510.pdf [Accessed 12 August 2010].

Axford, N. and Little, M. (2006) Refocusing children's services towards prevention. Lessons from literature. *Children and Society*, 20(4), 299-312.

Axford, N., Green, V., Kalsbeck A., Morpeth, L. and Palmer, C. (2009) Measuring children's needs: how are we doing? *Child and Family Social Work*, 14, 243-254.

Baginsky, M., Moriarty, J., Manthorpe, J., Stevens, M., MacInnes, T. and Nagendran, T. (2010) *Social Workers' Workload Survey. Messages from the Frontline.* Department of Children, Schools and Families. Available:

http://www.dcsf.gov.uk/swtf/downloads/SWT%20workload%survey%20%28final%29.pdf [Accessed 24 July 2010].

Balen, R., Blyth, E., Calabretto, H., Fraser, C., Horrocjs, C. and Manby, M. (2006) Involving children in health and social research: 'Human becomings' or 'active beings'? *Childhood*, 13(1), 29-48.

Ball, C. (1998) Regulating child care: from the Children Act 1948 to the present day. *Children and Family Social Work*, 3(3), 163-171.

Banks, S. (2006) *Ethics and Values in Social Work (3<sup>rd</sup> ed)*. Basingstoke: Palgrave Macmillan.

Barker, J. (1998) Extending the scope of child protection training. *Child Abuse Review*, 7(4), 287-293.

Barker, P. (1981) Basic Family Therapy. St. Albans: Granada Publishing.

Barnard, M. (1999) Forbidden questions: drug dependent parents and the welfare of their children. *Addiction*, 94, 1109-1111.

Barnes, J., Kroll, L., Burke, O., Lee, J., Jones, A. and Stein, A. (1998) Communication about parental illness with children who have learning disabilities and behavioural problems: Three case studies. *Child: Care, Health and Development,* 24(6), 441-456.

Baverstock, A., Bartle, D., Boyd, B. and Finlay, F. (2008) Review of child protection training uptake and knowledge of child protection guidelines. *Child Abuse Review*, 17, 64-72.

Beale, E.A., Sivesind, D. and Bruera, E. (2004) Parents dying of cancer and their children. *Palliative and Supportive Care,* 2, 387-393.

Beder, J. (2006) *Hospital Social Work: The Interface of Medicine and Caring.* New York: Routledge.

Beresford, P., Adshead, L. and Croft, S. (2007) *Palliative Care, Social Work and Service Users: Making Life Possible*. London: Jessica Kingsley Publishers.

Beresford, P., Adshead, L. and Croft, S. (2008) 'We don't see her as a social worker': a service user case study of the importance of the social worker's relationship and humanity. *British Journal of Social Work*, 38, 1388-1407.

Berry, M., Charleston, R. and Dawson, K. (2003) Promoting practices in understanding and treating child neglect. *Child and Family Social Work,* 8(1), 13-24.

Bertalanffy, L. von (1971) *General Systems Theory (revised ed).* George Braziller: New York.

Berzoff, J. and Silverman, P.R. (eds) (2004) *Living with Dying: A Handbook for End-of-life Health Care Practitioners*. New York: Columbia University Press.

Biestek, F.P. (1961) The Casework Relationship. London: Allen Unwin.

Birch, M. (1998) Re/constructing research narratives: self and sociological identity in alternative settings. In J. Ribbens and R. Edwards (eds) *Feminist Dilemmas in Qualitative Research*. London: Sage.

Birchall, E. and Hallett, C. (1995) *Working Together in Child Protection Practice*. London: HMSO.

Birley, J. and Morgan, G. (2005) Written words: hidden meanings. In B. Nyatanga and M. Astley-Pepper (eds) *Hidden Aspects of Palliative Care*. London: Quay Books.

Birmingham Safeguarding Children Board (2010) Serious Case Review under Chapter VIII 'Working Together to Safeguard Children' in Respect of the Death of a Child (Case 14). Available: http://www.lscbbirmingham.org.uk/downloads/Case+14.pdf [Accessed 29 July 2010].

Blackburn, C. (1991) *Poverty and Health. Working with Families*. Buckingham: Open University Press.

Blacker, S. and Deveau, C. (2010) Social work and interprofessional collaboration in palliative care. *Progress in Palliative Care*, 18(4), 237-243.

Blewett, J., Lewis, J. and Tunstill, J. (2007) *The Changing Roles and Tasks of Social Work: A Literature Informed Discussion Paper.* London: General Social Care Council.

Bliss, J. (1998) District nurses and social workers' understanding of each other's role. *British Journal of Community Nursing*, 3(7), 330-336.

Bliss, J., Cowley, S. and While, A. (2000) Interprofessional working in palliative care in the community: a review of the literature. *Journal of Interprofessional Care*, 14(3), 281-290.

Bloor, M., Frankland, J., Thomas, M. and Robson, K. (2001) *Focus Groups in Social Research.* London: Sage Publications.

Bonifant, J. (2000) Palliative care - a universal discipline? *Progress in Palliative Care*, 8(6), 351-353.

Borkan, J. (1999) Immersion/Crystallization. In B.F. Crabtree and W.L. Miller (eds) *Doing Qualitative Research (2<sup>nd</sup> ed).* London: Sage.

Bowlby, J. (1951) *Maternal Care and Mental Health.* Geneva: World Health Organisation.

Bowling, A. (2009) Research Methods in Health. Investigating Health and Health Services ( $3^{rd}$  ed). Maidenhead: Open University Press.

Brace, I. (2004) *Questionnaire Design: How to Plan, Structure and Write Survey Material for Effective Market Research.* London: Kogan Page.

Brandsen, C.K. (2005) Social work and end-of-life care: reviewing the past and moving forward. *Journal of Social Work in End-of-Life and Palliative Care*, 1(2), 45-70.

Braun, V. and Clarke, V. (2006) Using thematic analysis in psychology. *Qualitative Research in Psychology*, 3, 77-101.

Brearley, J. (1991) A psycho-dynamic approach to social work. In J. Lishman (ed) *A Handbook of Theory for Practice Teachers in Social Work*. London: Jessica Kingsley Publishers.

Briskman, L. and Cemlyn, S. (2005) Reclaiming humanity for asylum-seekers. A social work response. *International Social Work,* 48(6), 714-724.

Bronfenbrenner, U. (1979) *The Ecology of Human Development: Experiments with Nature and Design*. Cambridge: Harvard University Press.

Bronfenbrenner, U. (1992) Ecological systems theory. In R. Vista (ed) *Six Theories of Child Development: Revised Formulations and Current Issues.* London: Jessica Kingsley.

Bronstein, L.R. (2003) A model for interdisciplinary collaboration. *Social Work*, 48, 297-306.

Brooks, I. (2009) *Negotiating Power, Resistance and Control: Young women's safety in bars, pubs and clubs.* PhD thesis. Stirling University. Available: http://hdl.handle.net/1893/2049 [Accessed 24 June 2010].

Brown, J.B. (1999) The use of focus groups in clinical research. In B.F. Crabtree and W.L. Miller (eds) *Doing Qualitative Research (2<sup>nd</sup> ed)*. London: Sage.

Bryman, A. (2001) Social Research Methods. Oxford: Oxford University Press.

Bryman, A. (2004) Social Research Methods (2<sup>nd</sup> ed). Oxford: Oxford University Press.

Buckley, H. (2003) *Child Protection Work: Beyond the Rhetoric*. London: Jessica Kingsley Publishers.

Buff, R. (2005) Grounded theory: the methodology. In I. Holloway (ed) *Qualitative Research in Health Care.* Maidenhead: Open University Press.

Bugge, K.E., Helseth, S. and Darbyshire, P. (2008) Children's experiences of participation in a family support program when their parent has incurable cancer. *Cancer Nursing*, 31(6), 426-434.

Butler, I. (2002) A code of ethics for social work and social care research. *British Journal of Social Work*, 32, 239-248.

Butler, I. (2006) *What is so special about social work research?* Unpublished paper presented at the Social Work Postgraduate Research Student Workshop, Glasgow, organised by ESRC Researcher Development Initiative and SWAP.

Butler, I. and Buckley, H. (2005) Neglect: no monopoly on expertise. In J. Taylor and B. Daniel (eds) *Child Neglect. Practice Issues for Health and Social Care.* London: Jessica Kingsley Publishers.

Butler, I., Scanlan, L., Robinson, M., Douglas, G. and Murch, M. (2002) Children's involvement in the parents' divorce: implications for practice. *Children and Society*, 16, 89-102.

Butler, I. and Williamson, H. (1994) *Children Speak: Children, Trauma and Social Work.* Harlow: Longman.

Byrne, A., Canavan, J. and Millar, M. (2009) Participatory research and the voicecentred relational method of data analysis: is it worth it? *International Journal of Social Research Methodology*, 12(1), 67-77.

Bywaters, P. (1991) Case finding and screening for social workers in acute general hospitals. *British Journal of Social Work,* 21(1), 19-31.

Bywaters, P. (2008) Learning from experience: developing a research strategy for social work in the UK. *British Journal of Social Work*, 38, 936-952.

Bywaters, P. (2009) Tackling inequalities in health: a global challenge for social work. *British Journal of Social Work*, 39(2), 353-367.

Cadell, S., Johnston, M., Bosma, H. and Wainwright, W. (2010) An overview of contemporary social work. *Progress in Palliative Care*, 18(4), 205-211.

Calder, M. and Horwath, J. (eds) (1999) Working for Children on the Child Protection Register. An Inter-agency Practice Guide. Aldershot: Arena.

Cancer Research UK (2009) *UK Cancer Incidence Statistics by Age.* Available: http://www.info.cancerresearch.org/cancerstats/incidence/age. [Accessed 25 August 2009].

Carew, R. (1979) The place of knowledge in social work activity. *British Journal of Social Work*, 9(3), 349-364.

Carpenter, J., Hackett, S., Patsios, D. and Szilassy, E. (2010) *Outcomes of Interagency Training to Safeguard Children: Final Report to the Department of Children, Schools and Families and the Department of Health.* Available:

http://publications.education.gov.uk/eOrderingDowload/DCSF-RR209.pdf [Accessed 26 August 2010].

Carrier, J. and Kendall, I. (1995) Professionalism and inter-professionalism in health and community care: some theoretical issues. In P. Owens, J. Carrier and J. Horden (eds) *Inter-professional Issues in Community and Primary Health Care*. London: Macmillan.

Cawson, P. (2002) Child Maltreatment in the Family: The Experience of a National Sample of Young People. London: NSPCC.

Cawson, P., Wattam, C., Brooker, S. and Kelly, G. (2000) *Child Maltreatment in the United Kingdom: A Study of the Prevalence of Child Abuse and Neglect.* London: NSPCC.

Charmaz, K. (2006) *Constructing Grounded Theory. A Practical Guide Through Qualitative Analysis.* London: Sage.

Chowns, G. (2005) Swampy ground: brief interventions with families before bereavement. In B. Monroe and F. Klaus (eds) *Brief Interventions with Bereaved Children*. Oxford: Oxford University Press.

Chowns, G. (2006) 'No - you don't know how we feel'. Researching the experience of children facing the life-threatening illness of a parent, through collaborative inquiry. PhD thesis: University of Southampton.

Christ, G.H. (2000) *Healing Children's Grief: Surviving a Parent's Death from Cancer.* New York; Oxford: Oxford University Press.

Christ, G.H. and Christ, A.E. (2006) Current approaches to helping children cope with a parent's terminal illness. *CA: A Cancer Journal for Clinicians,* 56, 197-212.

Christ, G.H., Siegel, K. and Sperber, D. (1994) Impact of parental terminal cancer on adolescents. *American Journal of Orthopsychiatry*, 64(4), 604-613.

Christie, A. (2001) Gendered discourses of welfare: men and social work. In A. Christie (ed) *Men and Social Work: Theories and Practices*. London: Palgrave Macmillan.

Clark, D. (1998a) An annotated bibliography of the publications of Cicely Saunders–1: 1958-67. *Palliative Medicine*, 12, 181-193.

Clark, D. (1998b) Originating a movement: Cicely Saunders and the development of St. Christopher's Hospice 1957-1967. *Mortality*, 3(1), 43-63.

Clark, D. (1999) Cradled to the grave? Terminal care in the United Kingdom, 1948-67. *Mortality*, 4(3), 225-47.

Clark, D. (2007) From margins to centre: a review of the history of palliative care in cancer. *The Lancet Oncology*, 8(5), 430-438.

Clausen, H., Kendall, M., Murray, S., Worth, A., Boyd, K. and Benton, F. (2005) Would palliative care patients benefit from social workers' retaining the traditional 'casework' role rather than working as care managers? A prospective serial qualitative interview study. *British Journal of Social Work*, 35(2), 277-285.

Cleaver, H. (2000) Fostering Family Contact. London: The Stationery Office.

Cleaver, H., Walker, S. and Meadows, P. (2004) *Assessing Children's Needs and Circumstances*. London: Jessica Kingsley Publications.

Coffey, A. and Atkinson, P. (1996) *Making Sense of Qualitative Data. Contemporary Research Strategies.* London: Sage.

Cohen, P., Dizenhuz, I.M. and Winget, C. (1977). Family adaptation to terminal illness and death of a parent. *Social Casework*, 58, 223-229.

Cohen, S. (2001) *Immigration Controls, the Family and the Welfare State*. London: Jessica Kingsley Publishers.

Cohen, S. (2002) Local state of immigration controls. *Critical Social Policy*, 22(3), 518-543.

Collier, F. (2004) *The Victoria Climbié Memorial Address, May 2004*. Available: http://www.baaf.org.uk/info/lpp/pf/vcmemorialaddress.pdf [Accessed 24 June 2007].

Colton, M., Dury, C. and Williams, M.(1995) Children in need: definition, identification and support. *British Journal of Social Work*, 25(6), 711-728.

Compas, B.E., Worsham, N.L., Ey, S. and Howell, D.C. (1996) When Mom or Dad has cancer: II. Coping, cognitive appraisals, and psychological distress in children of cancer patients. *Health Psychology*, 15(3), 167-175.

Corby, B. (2006a) The role of child care social work in supporting families with children in need and providing protective services – past, present and future. *Child Abuse Review,* 15, 159-177.

Corby, B. (2006b) *Child Abuse. Towards a Knowledge Base (3<sup>rd</sup> ed).* Maidenhead. Open University Press.

Corless, I.B. and Nicholas, P.K. (2004) The interdisciplinary team: an oxymoron? In: J. Berzoff and P.R. Silverman (eds) *Living with Dying: A Handbook for End-of-life Health Care Practitioners*. New York: Columbia University Press.

Cort, M.A. (2004) Cultural mistrust and use of hospice care: challenges and remedies. *Journal of Palliative Medicine*, 7(1), 63-71.

Coulshed, V. and Orme, J. (1998) *Social Work Practice: An Introduction (3<sup>rd</sup> ed).* Basingstoke: Palgrave.

Coulshed, V. and Orme, J. (2006) *Social Work Practice: An Introduction (4<sup>th</sup> ed).* Basingstoke: Palgrave Macmillan.

Crabtree, B.F. and Miller, W.L. (eds) (1999) *Doing Qualitative Research (2<sup>nd</sup> ed).* Thousand Oaks: Sage.

Cree, V. (2003) Worries and problems of young carers: issues for mental health. *Child and Family Social Work*, 8, 301-309.

Cree, V. and Davis, A. (2007) Social Work: Voices from the Inside. London: Routledge.

Cree, V., Kay, H. and Tidsall, K. (2002) Research with children: sharing the dilemmas. *Child and Family Social Work*, 7, 47-56.

Crisp, B.R., Anderson, M.R., Orme, J. and Green Lister, P. (2005) *Learning and Teaching in Social Work Education. Textbooks and Frameworks on Assessment.* Social Care Institute for Excellence. Available:

http://www.scie.org.uk/publications/knowledgereviews/kr09.pdf [Accessed 4 April 2010].

Cullen, N. (2005) Interpersonal skills in palliative care social work. In J. Parker (ed) *Aspects of Social Work and Palliative Care*. London: Quay Books.

Currer, C. (2001) *Responding to Grief: Dying, Bereavement and Social Care.* Basingstoke: Palgrave.

Currer, C. (2002) Dying and bereavement. In R. Adams, L. Dominelli and M. Payne (eds) *Critical Practice in Social Work.* Basinstoke: Palgrave.

Daniel, B. (2000) Judgements about parenting: what do social workers think they are doing? *Child Abuse Review*, 9, 91-107.

Daniel, B. (2004) An overview of the Scottish Multidisciplinary Child Protection Review. *Child and Family Social Work*, 9, 247-257.

Daniel, B. (2005) Introduction to issues for health and social care in neglect. In J. Taylor and B. Daniel (eds) *Child Neglect. Practice Issues for Health and Social Care.* London: Jessica Kingsley Publishers.

Daniel, B., Featherstone, B., Hooper, C-A. and Scourfield, J. (2005) Why gender matters for 'Every Child Matters'. *British Journal of Social Work*, 35, 1343-1355.

Daniel, B. and Rioch, C. (2007) Parenting issues and practice in child protection. In K. Wilson and A. James (eds) *The Child Protection Handbook (3<sup>rd</sup> ed)*. London: Elsevier.

Daniel, B. and Taylor, J. (2001) *Engaging with Fathers: Practice Issues for Health and Social Care.* London: Jessica Kingsley.

Daniel, B. and Taylor, J. (2005) Do they care? The role of fathers in cases of child neglect. In J. Taylor and B. Daniel (eds) *Child Neglect. Practice Issues for Health and Social Care.* London: Jessica Kingsley.

Daniel, B. and Taylor, J. (2006) Gender and child neglect: theory, research and policy. *Critical Social Policy*, 26(2), 426-439.

Daniel, B., Taylor, J. and Scott, J. (2010) Recognition of neglect and early response: overview of a systematic review of literature. *Child and Family Social Work*, 15, 248-257.

Daniel, B., Vincent, S. and Ogilvie-Whyte, S. (2007) *A Process Review of the Child Protection Reform Programme.* Edinburgh: Scottish Executive. Available: http://www.vhscotland.org.uk/library/executive/review\_child\_protection\_reform\_prog\_ma r07.pdf [Accessed 6 June 2010].

Daniel, B. and Wassell, S. (2002) *Adolescence: Assessing and Promoting Resilience in Vulnerable Children.* London: Jessica Kingsley Publishers.

Daniel, B., Wassell, S. and Gilligan, R. (2010) *Child Development for Child Care and Protection Workers (2<sup>nd</sup> ed).* London: Jessica Kingsley Publishers.

Darlington, Y. and Feeney, J.A. (2008) Collaboration between mental health and child protection services: Professionals' perceptions of best practice. *Children and Youth Services Review*, 30, 187-198.

Darlington, Y., Feeney, J.A., and Rixon, K. (2005) Interagency collaboration between child protection and mental health services: practices, attitudes and barriers. *Child Abuse and Neglect,* 29, 1085-1098.

Darlington, Y., Osmond, J. and Peile, C. (2002) Child welfare workers' use of theory in working with physical child abuse: implications for professional supervision. *Families in Society*, 83(1), 54-63.

Darlington, Y. and Scott, D. (2002) *Qualitative Research in Practice: Stories from the Field.* Buckingham: Open University Press.

Davies, M. and Connelly, J. (1995) The social worker's role in the hospital: seen through the eyes of other health care professionals. *Health and Social Care in the Community*, 3(5), 301-309.

D'Cruz, H. and Stagnitti, K. (2008) Reconstructing welfare through participatory and child-centred professional practice: a conceptual approach. *Child and Family Social Work*, 13, 156-165.

Dearden, C. and Aldridge, J. (2010). Young carers: needs, rights and assessments. In J. Horwath (ed) *The Child's World. The Comprehensive Guide to Assessing Children in Need (2<sup>nd</sup> ed).* London: Jessica Kingsley Publishers.

Dearden, C. and Becker, S. (2000) *Growing up Caring: Vulnerability and Transitions to Adulthood - Young Carers' Experiences.* Leicester: Youth Work Press.

Dearden, C. and Becker, S. (2001) Young carers: needs, rights and assessments. In J. Horwath (ed) *The Child's World. Assessing Children in Need.* London: Jessica Kingsley.

Dearden, C. and Becker, S. (2004) Young Carers in the UK: The 2004 Report. London: Carers UK.

Deery, R. (2005) An action-research study exploring midwives' support needs and the affect of group clinical supervision. *Midwifery*, 21, 161-176.

Dehlin, L. and Martensson, L. (2009) Adolescents' experiences of a parents' serious illness and death. *Palliative and Supportive Care*, 7, 13-25.

Denscombe, M. (1998) *The Good Research Guide.* Buckingham: Open University Press.

Dent, A. (2005) Theoretical perspectives: linking research and practice. In B. Monroe and F. Klaus (eds) *Brief Interventions with Bereaved Children.* Oxford: Oxford University Press.

Dent, R.K. and Cocker, C. (2005) Serious case reviews: lessons for practice in cases of child neglect. In J. Taylor and B. Daniel (eds) *Child Neglect. Practice Issues for Health and Social Care.* London: Jessica Kingsley Publishers.

Denzin, N.K. and Lincoln, Y.S. (eds) (2000) *Handbook of Qualitative Research (2<sup>nd</sup> ed)*. Thousand Oaks, CA: Sage.

Department for Education (2010) *Munro Review to look at children's social work and frontline child protection practice.* Available:

http://www.education.gov.uk/news/news/munroreview [Accessed 29 July 2010].

Department of Health (1995) *Child Protection: Messages from Research*. London: HMSO.

Department of Health (1999a) *Caring about Carers: A National Strategy for Carers.* London: Stationery Office.

Department of Health (1999b) *Working Together to Safeguard Children: A Guide to Inter-agency Working to Safeguard and Promote the Welfare of Children.* London: HMSO.

Department of Health (2000) *The NHS Cancer Plan: A Plan for Investment, a Plan for Reform.* London: HMSO.

Department of Health, Department for Education and Employment and Home Office (2000) *Framework for the Assessment of Children in Need and Their Families.* London: The Stationery Office.

Department of Health (2002) Safeguarding children: A joint Chief Inspectors' Report on Arrangements to Safeguard Children. London: Department of Health.

Department of Health (2003) *What To Do If You Are Worried a Child Is Being Abused.* London: Department of Health.

Department of Health (2004) National Health Service Framework for Children and Young People and Maternity Services: Core Standards. London: Department of Health.

Department of Health and Social Security (1988) *Working Together: A Guide to Arrangements for Inter-agency Co-operation for the Protection of Children from Abuse.* London: HMSO.

DfES (Department for Education and Skills) (2003) *Every Child Matters*. CM 5860. London: The Stationery Office.

DfES (Department for Education and Skills) (2005) *Statistics of Education - Referrals, Assessments and Children and Young People on Child Protection Registers: Year Ending 31 March 2004.* London: The Stationery Office.

DfES (Department for Education and Skills) (2006) *Working Together to Safeguard Children. A Guide to Inter-agency Working to Promote the Welfare of Children.* London: The Stationery Office.

Diareme, S., Tsiantis, J., Romer, G., Tsalamanios, E., Anasontzi, S., Paliokosta, E. and Kolaitis, G. (2007) Mental health support for children of parents with somatic illness: a review of the theory and intervention concepts. *Families, Systems, & Health*, 25(1), 98-118.

Dix, O. and Glickman, M. (1997) *Working Party on Psychosocial Care. Occasional Paper 13.* London: National Council for Palliative Care.

Donald, T. and Jureidini, J. (2004) Parenting Capacity. Child Abuse Review, 13, 5-17.

Downie, R. and Loudfoot, E. (1978) Aims, skill and role in social work. In N. Timms and D. Watson (eds) *Philosophy in Social Work.* London: Routledge, Kegan and Paul.

Doyle, D., Hanks, G. and Cherney, N. (eds) (2004) *Oxford Textbook of Palliative Care* (3<sup>rd</sup> ed). Oxford: Oxford University Press.

Dunlop, R.J. and Hockley, J.M. (1998) *Hospital-based Palliative Care Teams: The Hospital-hospice Interface (2<sup>nd</sup> ed)*. Oxford: Oxford University Press.

Dunning, S. (2006) As a young child's parent dies: conceptualizing and constructing preventive interventions. *Clinical Social Work Journal*, 34(4), 499-514.

Dutt, R. and Phillips, M. (2010) Assessing the needs of black and ethnic minority children. In J. Horwath (ed) *The Child's World. The Comprehensive Guide to Assessing Children in Need (2<sup>nd</sup> ed).* London: Jessica Kingsley Publishers.

Egnew, T.R. (2004) Foreword. In J. Berzoff and P.R. Silverman (eds) (2004) *Living with Dying: A Handbook for End-of-life Health Care Practitioners.* New York: Columbia University Press.

Everard, S.R. (2005) Spirituality and social work in palliative care. In J. Parker (ed) *Aspects of Social Work and Palliative Care*. London: Quay Books.

Fairtlough, A.C. (2007) Adapting the voice-centred relational method of data analysis: reading trainees' accounts of their learning on a pilot programme for practitioners working with parents. *Learning in Health and Social Care*, 6(1), 2-13.

Falk, H.S. (1997) Interdisciplinary education and implications for social work practice. *Journal of Education for Social Work,* 13, 30-37.

Farber, S., Egnew, T. and Farber, A. (2004) What is a respectful death? In J. Berzoff and P. R. Silverman (eds) *Living with Dying: A Handbook for End-of-life Health Care Practitioners.* New York: Columbia University Press.

Faulkner, A. and Maguire, P. (1994) *Talking to Cancer Patients and their Relatives*. Oxford: Oxford University Press.

Fearnley, R. (2010) Death of a parent and the children's experience: Don't ignore the elephant in the room. *Journal of Interprofessional Care*, Early Online, 1-10.

Featherstone, B. (1997) What has gender got to do with it? *British Journal of Social Work*, 27, 419-433.

Featherstone, B. (2003) Taking fathers seriously. *British Journal of Social Work*, 33(2), 239-254.

Ferguson, H. (2004) *Protecting Children in Time: Child Abuse, Child Protection and the Consequences of Modernity*. Basingstoke: Palgrave Macmillan.

Field, D. (2000) '*What do we mean by 'psychosocial'?' Briefing Paper No.4.* London: National Council for Hospice and Specialist Palliative Care Services.

Fielding, J. (2008) Coding and managing data. In N. Gilbert (ed) *Researching Social Life* (3<sup>rd</sup> ed). London: Sage.

Fineberg, I.C. (2010) Social work perspectives on family communication and family conferences in palliative care. *Progress in Palliative Care*, 18(4), 213-220.

Finlay, L. and Gough, B. (eds) (2003) *Reflexivity: A Practical Guide for Researchers in Health and Social Sciences.* Oxford: Blackwell Science.

Firth, P. (2003) Multi-professional teamwork. In B. Monroe and D. Oliviere (eds) *Patient Participation in Palliative Care. A Voice for the Voiceless.* Oxford: Oxford University Press.

Firth, P. and Anderson, P. (1994) Teamwork with families facing bereavement. *European Journal of Palliative Care*, 1(4), 157-161.

Firth, S. (2001) *Wider Horizons: Care for the Dying in a Multi-cultural Society*. London: NCHSPCS.

Fish, S., Munro, E. and Bairstow, S. (2009) *Learning Together to Safeguard Children: Developing Multi-agency Systems Approach for Case Reviews*. SCIE: London.

Fisher, A. (2009) Motionless Body. In S. Earle, C. Komaromy, and C. Bartholomew (eds) *Death and Dying: A Reader*. London: Sage Publications.

Fitzgerald, D. and Kay, J. (2008) *Working Together in Children's Services*. London: Routledge.

Fitzpatrick, R. and Boulton, M. (1994) Qualitative methods for assessing health care. *Quality in Health Care*, 3(2),107-113.

Fitzsimmons, P. and White, T. (1997) Crossing boundaries. *Journal of Management in Medicine*, 11(2), 96-101.

Flick, U. (2007a) Designing Qualitative Research. Los Angeles: London: Sage.

Flick, U. (2007b) *Managing Quality in Qualitative Research.* Los Angeles: London: Sage.

Foddy, W. (1993) *Constructing Questions for Interviews and Questionnaires. Theory and Practice in Social Research.* Cambridge: Cambridge University Press.

Fook, J. (2002) Social Work: Critical Theory and Practice. London: Sage Publications.

Fook, J. and Gardner, F. (2007) *Practising Critical Reflection: A Resource Handbook.* Maidenhead: Open University Press.

Fox-Harding, L. (1991) Perspectives in Child Care Policy. London: Longman.

Froggatt, K.A. (2001) The analysis of qualitative data: processes and pitfalls. *Palliative Medicine*, 15, 433-438.

Frost, N. and Robinson, M. (2007) Joining up children's services: safeguarding children in multi-disciplinary teams. *Child Abuse Review*, 16, 184-199.

Gabb, J., Balen, R., Gibbs, G., Hall, C. and Teal, A. (2006) Victoria Climbié Inquiry Data Corpus Project: using the Delphi Method in multidisciplinary child protection research. *British Journal of Social Work*, 36, 577-596.

Gates, M.F. and Lackey, N.R. (1998) Youngsters caring for adults with cancer. *Journal of Nursing Scholarship*, 30(1), 11-15.

Gibbs, G. (2007) Analyzing Qualitative Data. The SAGE Qualitative Research Kit Book 6. Sage: London.

Gilgun, J.F. (1994) A case for case studies in social work research. *Social Work,* 39 (4), 371-380.

Gilligan, P. and Manby, M. (2008) The Common Assessment Framework: does the reality match the rhetoric? *Child and Family Social Work*, 13,177-187.

Gilligan, R. (1997) Beyond permanence? The importance of resilience in child placement practice and permanency. *Adoption and Fostering*, 21(1), 12-20.

Giovanni, J.M. and Becerra, R.M. (1979) Defining Child Abuse. New York: Free Press.

Glaser, D. (2002) Emotional abuse and neglect (psychological maltreatment): a conceptual framework. *Child Abuse and Neglect*, 26(6-7), 697-714.

Glennie, S. (2007) Developing interprofessional relationships: tapping the potential of inter-agency training. *Child Abuse Review*, 16, 171-183.

Golden, M.H., Samuels, M.P. and Southall, D.P. (2003) How to distinguish between neglect and deprivational abuse. *Archives of Disease in Childhood*, 88, 105-107.

Grabiak, B.R., Puskar, K.R. and Bender, C.M. (2007) The impact of parental cancer on the adolescent: an analysis of the literature. *Psycho-Oncology*, 16(2), 127-137.

Graham, B. (2000) Developing a Questionnaire. London: Continuum.

Guba, E.G. and Lincoln, Y.S. (1994) Competing paradigms in qualitative research. In N.K. Denzin and Y.S. Lincoln (eds) *Handbook of Qualitative Research*. Thousand Oaks: Sage.

Hall, C. (1997) *Social Work as a Narrative: Storytelling and Persuasion in Professional Texts.* Aldershot: Ashgate Publishing.

Hall, P. (2005) Interprofessional teamwork: professional cultures as barriers. *Journal of Interprofessional Care,* Supplement 1, 188-196.

Hallett, C. (1995) Interagency Cooperation in Child Protection. London: HMSO.

Hallett, C., Murray, C. and Punch, S. (2003) Young people and welfare. Negotiating pathways. In C. Hallett and A. Prout (eds) *Hearing the Voices of Children: Social Policy for a New Century.* London: RoutledgeFalmer.

Hallett, C. and Stevenson, O. (1980) *Child Abuse: Aspects of Multiprofessional Cooperation.* London: George Allen and Unwin.

Hammersley, M. and Atkinson, P. (1995) *Ethnography. Principles in Practice (2<sup>nd</sup> ed).* London: Routledge.

Hammond, H. (2001) *Child protection inquiry into the circumstances surrounding the death of Kennedy McFarlane (17/4/97).* Dumfries and Galloway Child Protection Committee.

Haringey LSCB (2009) *Serious Case Review: Baby Peter. Executive Summary*. London: Haringey Local Safeguarding Children Board.

Harlow, E. (2003) New managerialism, social service departments and social work practice today. *Practice*, 15(2), 29-44.

Harlow, E. and Shardlow, S.M. (2006) Safeguarding children: challenges to the effective operation of core groups. *Child and Family Social Work*, 11, 65-72.

Harrison, L. and Harrington, R. (2001) Adolescents' bereavement experiences. Prevalence, association with depressive symptoms, and use of services. *Journal of Adolescence*, 24(2), 159-69. Hart, C. (1997) *Doing a Literature Review: Releasing the Social Science Research Imagination.* London: Sage.

Hasson, F., Keeney, S. and McKenna, H. (2000) Research guidelines for the Delphi survey technique. *Journal of Advanced Nursing*, 32(4), 1008-1015.

Hawkins, L., Fook, J. and Ryan, M. (2001) Social workers' use of the language of social justice. *British Journal of Social Work*, 31(1), 1-13.

Hawkins, P. and Shohet, R. (2006) *Supervision in the Helping Professions (3<sup>rd</sup> ed).* Oxford: Oxford University Press.

Hearn, F., Jackman, E., Lake, T., Popplestone-Helm, S. and Young, A. (2008) Reemphasising the social side: a new model of care. *European Journal of Palliative Care*, 15(6), 276-278.

Heaven, C.M. and Maguire, P. (1997) Disclosure of concerns by hospice patients and their identification by nurses. *Palliative Medicine*, 11, 283-290.

Heiney, S.P., Bryant, J.H., Walker, S., Parrish, R.S., Provenzano, F.J. and Kelly, K.E. (1997) Impact of parental anxiety on child emotional adjustment when a parent has cancer. *Oncology Nursing Forum*, 24(4), 655-661.

Helm, D. (2010) *Making Sense of Child and Family Assessment*. London: Jessica Kingsley Publishers.

Help the Hospices (2009) *Hospice and Palliative Care Directory. United Kingdom and Ireland 2009-2010.* London: Help the Hospices.

Helseth, S. and Ulfsaet, N. (2005) Parenting experiences during cancer. *Journal of Advanced Nursing*, 52(1), 38-46.

Hendrick, H. (2003) *Historical Dimensions, Contemporary Debate*. Bristol: The Policy Press.

Hepworth, D.H., Rooney, R.H., Rooney, G.D., Strom-Gottfried, K. and Larsen, J. (2010) *Direct Social Work Practice, Theory and Skills (8<sup>th</sup> ed).* Belmont, C.A.: Brooks/Cole.

Hertz, R. (ed) (1997) Reflexivity and Voice. Thousand Oaks: Sage.

Hill, A. (1998) Multi-professional teamwork in a hospital palliative care team. *International Journal of Palliative Nursing*, 4(5), 214-21.

Himebauch, A., Arnold, R.M. and May, C. (2008) Grief in children and developmental concepts of death. *Journal of Palliative Medicine*, 11(2), 242-243.

HM Government (2004) Every Child Matters: Change for Children. London: DfES.

HM Government (2005) An Overview of Cross Government Guidance. London: DfES.

HM Government (2006a) *Common Assessment Framework for Children and Young People: Practitioners' Guide.* London: DfES.

HM Government (2006b) *Common Assessment Framework for Children and Young People: Managers' Guide.* London: DfES.

HM Government (2006c) *Common Assessment Framework for Children and Young People: Supporting tools.* London: DfES.

HM Government (2007) Statutory guidance on making arrangements to safeguard and promote the welfare of children under section 11 of the Children Act. London: DfES.

HM Government (2008) *Carers at the Heart of 21<sup>st</sup> Century, Families and Communities.* The Stationery Office: London.

HM Government (2010) Working Together to Safeguard Children: A guide to interagency working to safeguard and promote the welfare of children. London: DfES. Available: http://publications.education.gov.uk/eOrderingDowload/00305-2010DOM-EN.PDF [Accessed 24 August 2010].

Hodgson, J. (2005) Working together - a multi-disciplinary concern. In J. Parker (ed) *Aspects of Social Work and Palliative Care.* London: Quay.

Hoghughi, M. and Speight, A. (1998) Good enough parenting for all children - a strategy for a healthier society. *Archives of Disease in Childhood*, 78(4), 293-296.

Holland, S. (2004) *Child and Family Assessment in Social Work Practice*. London: Sage.

Holland, S. (2010) Engaging children and their parents in the assessment process. In J. Horwath (ed) *The Child's World. The Comprehensive Guide to Assessing Children in Need (2<sup>nd</sup> ed).* London: Jessica Kingsley Publishers.

Hollis, F. (1964) Casework: A Psychosocial therapy. New York: Random Press.

Holloway, I. (2005) Qualitative writing. In I. Holloway (ed) *Qualitative Research in Health Care.* Maidenhead: Open University Press.

Holloway, I. and Freshwater, D. (2007) *Narrative Research in Nursing*. Oxford: Blackwell Publishing.

Holman, B. (1998) From children's departments to family departments. *Children and Family Social Work*, 3, 205-211.

Horwath, J. (2001) Assessing the world of the child: background and context. In J. Horwath (ed) *The Child's World. Assessing Children in Need.* London: Jessica Kingsley Publishers.

Horwath, J. (2005) Is this child neglect? In J. Taylor and B. Daniel (eds) *Child Neglect. Practice Issues for Health and Social Care.* London: Jessica Kingsley Publishers.

Horwath, J. (2007a) The missing assessment domain: personal, professional and organizational factors influencing professional judgements when identifying and referring child neglect. *British Journal of Social Work*, 37, 1285-1303.

Horwath, J. (2007b) Safeguarding children: the assessment challenges. In J. Wilson and A. James (eds) *The Child Protection Handbook (3^{rd} ed)*. Edinburgh and London: Ballière Tindall.

Horwath, J. (2010) Assessing children in need: background and context. In J. Horwath (ed) *The Child's World. The Comprehensive Guide to Assessing Children in Need (2<sup>nd</sup> ed).* London: Jessica Kingsley Publishers.

Horwath, J. and Saunders, T. (2005) 'Do you see what I see?' Multiprofessional Perspectives on Child Neglect. The North Eastern Board and University of Sheffield. Available: http://www.nehb.ie/nehb/publications/reports/childneg.pdf [Accessed 24 May 2010].

Hothersall, S. J. (2006) *Social Work with Children, Young People and their Families in Scotland.* Exeter: Learning Matters.

Houldin, A.D. and Lewis, F.M. (2006) Salvaging their normal lives: a qualitative study of patients with recently diagnosed advanced colorectal cancer. *Oncology Nursing Forum*, 33(4), 719-724.

Houston, S. (2002) Re-thinking a systemic approach to child welfare: a critical response to the framework for the assessment of children in need and their families. *European Journal of Social Work*, 5(3), 301-322.

Howe, D. (1998) Relationship-based thinking in practice in social work. *Journal of Social Work Practice*, 12(1), 45-46.

Howe, D. (1999) *Attachment Theory, Child Maltreatment and Family Support.* Basingstoke: Macmillan.

Howe, D. (2005) *Child Abuse and Neglect. Attachment, Development and Intervention.* Basingstoke: Palgrave Macmillan.

Howe, D. (2008) Relating theory to practice. In M. Davies (ed) *The Blackwell Companion to Social Work (3<sup>rd</sup> ed).* Oxford: Blackwell Publishing.

Huberman, A. and Miles, M.B. (1998) Data management and analysis methods. In N. Denzin and Y. Lincoln (eds) *Collecting and Interpreting Qualitative Methods*. Thousand Oaks: Sage.

Hudson, B. (2002) Inter-personality in health and social care: the Achilles' heel of partnership? *Journal of Interprofessional Care,* 16(1), 7-17.

Hudson, P. (2003) Focus group interviews: a guide for palliative care researchers and clinicians. *International Journal of Palliative Nursing*, 9(5), 202-207.

Hudson, P. and Payne, S. (eds) (2009) *Family Carers in Palliative Care*. Oxford; Oxford University Press.

Huizinga, G.A., van der Graaf, W.T., Visser, A., Dijkstra, J.S. and Hoekstra-Weebers, J.E.H.M. (2003) Psychosocial consequences for children of a parent with cancer. A pilot study. *Cancer Nursing*, 26(3), 195-202.

Humphreys, C. and Stanley, N. (eds) (2006) *Domestic Violence and Child Protection. Directions for Good Practice.* London: Jessica Kingsley Publishers.

Humphries, B. (2004) An unacceptable role for social work: implementing immigration policy. *British Journal of Social Work*, 34(1), 93-107.

Humphries, B. and Martin, M. (2000) Disrupting ethics in social research. In B. Humphries (ed) *Research in Social Care and Social Welfare. Issues and Debates for Practice.* London: Jessica Kingsley Publishers.

Hurdman, R. (1995) Meeting social needs – the role of the social worker. In J. Robbins and J. Moscrop (eds) *Caring for the Dying Patient and the Family* (3<sup>rd</sup> ed). London: Chapman and Hall.

Jack, G. (1997) Discourses of child protection and child welfare. *British Journal of Social Work*, 27, 659-678.

Jack, G. (2000) Ecological influences on parenting and child development. *British Journal of Social Work*, 30(6), 703-720.

Jack, G. and Gill, O. (2010) The impact of economic factors on parents or caregivers and children. In J. Horwarth (ed) *The Child's World. The Comprehensive Guide to Assessing Children in Need (2<sup>nd</sup> ed).* London: Jessica Kingsley Publishers.

Jackson, S. (2000a) Assessment in Child Care. In M. Davies (ed) *The Blackwell Companion to Social Work.* Oxford: Blackwell Publishing.

Jackson, S. (2000b) Assessment in Community Care. In M. Davies (ed) *The Blackwell Companion to Social Work.* Oxford: Blackwell Publishing.

James, N. (1993) Divisions of emotional labour: disclosure and cancer. In M. Robb, S. Barrett, C. Komaromy and A. Rogers (eds) (2004) *Communication, Relationships and Care: A Reader.* London: Routledge.

Janis, I.L. (1972) Victims of Group Think. Boston, M.A.: Houghton Mifflin.

Jeffrey, D. (2004) Communication with professionals. In D. Doyle, G. Hanks and N. Cherney (eds) *Oxford Textbook of Palliative Care (3<sup>rd</sup> ed).* Oxford: Oxford University Press.

Jewett, C. (1997) *Helping Children Cope with Separation and Loss*. London: Free Association Books.

Joint Chief Inspectors (2005) *The Second Joint Chief Inspectors' Report on Arrangements to Safeguard Children.* London: Ofsted.

Joint Chief Inspectors (2008) *The Third Joint Chief Inspectors' Report on Arrangements to Safeguard Children.* London: Ofsted.

Jones, D.P.H. (2003) Communicating with Vulnerable Children: A Guide for *Practitioners.* London: Gaskell.

Keeney, S., Hasson, F. and McKenna, H. (2006) Consulting the oracle: ten lessons from using the Delphi technique. *Journal of Advanced Nursing*, 53(2), 205-212.

Kendal, M., Harris, F., Boyd, K., Sheikh, A., Murray, S.A., Brown, D., Mallinson, I., Kennedy, N. and Worth, A. (2009) Key challenges and ways forward in researching the 'Good Death'. In S. Earle, C. Komaromy and C. Bartholomew (eds) *Death and Dying: A Reader*. London: Sage Publications.

Kennedy, V.L. (2008) *Children's Experiences of Parental Advanced Cancer*. PhD Thesis: University of Liverpool.

Kennedy, V.L. and Lloyd-Williams, M. (2009) How children cope when a parent has advanced cancer. *Psycho-oncology*, 18, 886-892.

King, D.A. and Quill, T. (2006) Working with families in palliative care: one size does not fit all. *Journal of Palliative Medicine*, 9(3) 704-715.

Kissane, D.W. and Bloch, S. (2002) *Family Focused Grief Therapy*. Maidenhead: Open University Press.

Kissane, D.W., McKenzie, M., Bloch, S., Moskowitz, C., McKenzie, D.P. and O'Neill, I. (2006) Family focused grief therapy: a randomized, controlled trial in palliative care and bereavement. *American Journal of Psychiatry*, 163(7), 1208-1218.

Kitzinger, J. (2005) Focus group research: using group dynamics to explore perceptions, experiences and understandings. In I. Holloway (ed) *Qualitative Research in Health Care*. Maidenhead: Open University Press.

Knapp, E.R. and DelCampo, R.L. (1995) Developing family care plans: a systems perspective for helping hospice families. *The American Journal of Hospice and Palliative Care*, 12, 39-47.

Koffman, J. and Higginson, I.J. (2001) Accounts of carers' satisfaction with health care at the end of life: a comparison of first generation black Caribbeans and white patients with advanced cancer. *Palliative Medicine*, 15(4), 337-345.

Kristjanson, L.J. and Ashcroft, T. (1994) The family's cancer journey: a literature review. *Cancer Nursing*, 17(1), 1-17.

Kroll, B. and Taylor, A. (2003) *Parental Substance Misuse and Child Welfare*. London: Jessica Kingsley Publishers.

Krueger, R.A. (1993) Quality control in focus group research. In D.L. Morgan (ed) *Successful Focus Groups*. Newbury Park: Sage.

Krueger, R.A. and Casey, M.A. (2000) *Focus Groups: A Practical Guide for Applied Research*. Newbury Park, California: Sage.

Kvale, S. (2007) Doing Interviews. The Saga Qualitative Research Kit. London: Sage.

Lackey, N.R. and Gates, M.F. (2001) Adults' recollections of their experiences as young caregivers of family members with chronic physical illness. *Journal of Advanced Nursing*, 34(3), 320-328.

Laidler, P. (1991) Adults and how to become one. Therapy Weekly, 17(35), 4.

Laming, Lord (2003) *The Victoria Climibié Inquiry Report*. Norwich: The Stationery Office.

Laming, Lord (2009) *The Protection of Children in England: A Progress Report*. London: The Stationery Office.

Landry-Dattee, N. and Delaigue-Cosset, M.F. (2001) Support groups for children. *European Journal of Palliative Care*, 8(3), 107-110.

Larkin, P. (2008) Family centred care: psychosocial care for the marginalized. In M. Lloyd-Williams (ed) *Psychosocial Issues in Palliative Care (2<sup>nd</sup> ed).* Oxford: Oxford University Press.

Lee, R.M. (1993) Doing Research on Sensitive Topics. London: Sage.

Legard, R., Keegan, J. and Ward, K. (2003) *In-depth Interviews*. In J. Ritchie and J. Lewis (eds) *Qualitative Research Practice. A Guide for Social Science Students and Researchers.* London: Sage.

Leigh, S. and Miller, C. (2004) Is the third way the best way? Social work intervention with children and families. *Journal of Social Work*, 4(3), 245-267.

Lewis, D. (2003) Design issues. In J. Ritchie and J. Lewis (eds) *Qualitative Research Practice. A Guide for Social Science Students and Researchers.* London: Sage.

Lewis, F.M. (1990) Strengthening family supports. Cancer and the family. *Cancer*, 65, 752-759.

Lewis, F.M. (2007) Editorial. Parental cancer and dependent children: selected issues for further research. *Psycho-Oncology*, 16, 97-98.

Lewis, F.M. and Hammond, M.A. (1992) Psychological adjustment to breast cancer. A longitudinal analysis. *Journal of American Women's Medical Association*, 47,194-200.

Lichtenthal, W.G. and Kissane, D.W. (2008) The management of family conflict in palliative care. *Progress in Palliative Care*, 16(1), 39-45.

Ling, M.S. and Luker, K.A. (2000) Protecting children: intuition and awareness in the work of health visitors. *Journal of Advanced Nursing*, 32(3), 572-579.

Linhorst, D. M. (2002) A review of the use and potential of focus groups in social work research. *Qualitative Social Work*, 1(2), 208-228.

Lishman, J. (1994) Communication in Social Work. Basingstoke: Macmillan.

Lishman, J. (2002) Personal and professional development. In R. Adams, L. Dominelli and M. Payne (eds) *Social Work: Themes, Issues and Critical Debates (2<sup>nd</sup> ed).* Basingstoke: Palgrave.

Litosseliti, L. (2003) Using Focus Groups in Research. London: Continuum.

Lloyd, M. (1997) Dying and bereavement, spirituality and social work in a market economy of welfare. *British Journal of Social Work*, 27, 175-190.

Lloyd-Williams, M., Friedman, T. and Rudd, N. (1999) A survey of psychosocial service provision within hospices. *Palliative Medicine*, 13, 431-2.

London Borough of Brent (1985) A Child in Trust - Report of the Panel of Inquiry into the Circumstances Surrounding the Death of Jasmine Beckford. London: London Borough of Brent.

Luckock, B., Lefevre, M. and Tanner, K. (2007) Teaching and learning communication with children and young people: developing the qualifying social work curriculum in a changing policy context. *Child and Family Social Work*, 12, 192-201.

Macleod, M. (2000) What do children need by way of child protection? Who is to decide? In N. Baldwin (ed) *Protecting Children. Promoting their Rights*. London: Whiting and Birch.

Macpherson, C. (2005) Telling children their ill parent is dying: a study of the factors influencing the well parent. *Mortality*, 10(2), 113-126.

Macpherson, C. (2009) Childhood abuse uncovered in palliative care audit. *Palliative and Supportive Care*, 7, 481-486.

Macpherson, C. (2010) But ... I come from the land of the one liner. *Illness, Crisis, and Loss,* 18(3), 217-227.

Macpherson, C. and Cooke, C. (2003) Pilot of a workbook for children visiting a loved one in a hospice. *International Journal of Palliative Nursing*, 9(9), 397-403.

Maslow, A.H. (1970) *Motivation and Personality (2<sup>nd</sup> ed)*. New York: Harper & Row.

Mason, J. (2002) *Qualitative Researching (2<sup>nd</sup> ed)*. London: Sage.

Mathew, A., Cowley, S., Bliss, J. and Thistlewood, G. (2003) The development of palliative care in national government policy in England, 1986-2000. *Palliative Medicine*, 17, 270-282.

Mauthner, N. and Doucet, A. (1998) Reflections on voice-centred relational method: analysing maternal and domestic voices. In J. Ribbens and R. Edwards (eds) *Feminist Dilemmas in Qualitative Research.* London: Sage.

McGoldrick, M., Gerson, R. and Shellenberger, S. (1999) *Genograms, Assessment and Intervention (2<sup>nd</sup> ed).* New York: W.W. Norton and Co.

Mcleod, E. and Bywaters, P. (1999) Inequalities in physical health. *British Journal of Social Work*, 29(4), 547-565.

McPherson, C.J. and Addington-Hall, J.M. (2004) Evaluating palliative care: bereaved family members' evaluations of patients' pain, anxiety and depression. *Journal of Pain and Symptom Management*, 28(2), 104-113.

Mehta, A., Cohen, S.R. and Chan, L.S. (2009) Palliative care: a need for a family systems approach. *Palliative and Supportive Care*, 7, 235-243.

Meier, D.E. and Beresford, L. (2008) Social workers advocate for a seat at palliative care table. *Journal of Palliative Medicine*, 11(1), 10-14.

Milner, J. and O'Byrne, P. (2002) *Assessment in Social Work (2<sup>nd</sup> ed).* Basingstoke: Palgrave Macmillan.

Moch, S.D. (2000) The researcher experience in health care research. In S.D. Moch and M.F. Gates (eds) *The Researcher Experience in Qualitative Research*. Thousand Oaks: Sage.

Molyneux, J. (2001) Interprofessional teamworking: what makes teams work well? *Journal of Interprofessional Care,* 15(1), 29-35.

Monroe, B. (1990) Supporting children facing bereavement. In C. Saunders (ed) *Hospice and Palliative Care - An Interdisciplinary Approach*. London: Edward Arnold.

Monroe, B. (1998) Social work and palliative care. In D. Doyle, G. Hanks, and N. MacDonald (eds) *Oxford Textbook of Palliative Care* (2<sup>rd</sup> ed). Oxford: Oxford University Press.

Monroe, B. (2004) Social work and palliative care. In D. Doyle, G. Hanks and N. Cherney (eds) *Oxford Textbook of Palliative Care (3<sup>rd</sup> ed)*. Oxford: Oxford University Press.

Monroe, B. and Oliviere, D. (eds) (2007) *Resilience in Palliative Care: Achievement in Adversity*. Oxford: Oxford University Press.

Monroe, B. and Oliviere, D. (2009) Communicating with family carers. In P. Hudson and S. Payne (eds) *Family Carers in Palliative Care*. Oxford: Oxford University Press.

Moorey, S. and Greer, S. (2002) *Cognitive Behaviour Therapy for People with Cancer.* Oxford: Oxford University Press.

Morgan, D.L. (1988) Focus Groups as Qualitative Research. Sage University Paper Series on Qualitative Research Methods Series 16. Newbury Park: Sage.

Morgan, D.L. (1997) Focus Groups as Qualitative Research (2<sup>nd</sup> ed). Sage University Paper Series on Qualitative Research Methods Series 16. Thousand Oaks: Sage.

Morrison, T. (2007) Emotional intelligence, emotion and social work: context, characteristics, complications and contributions. *British Journal of Social Work*, 37, 245-263.

Morrison, T. (2010) Assessing parental motivation for change. In J. Horwath (ed) *The Child's World. The Comprehensive Guide to Assessing Children in Need (2<sup>nd</sup> ed).* London: Jessica Kingsley Publishers.

Moyers, S. and Mason, A. (1995) Identifying standards of parenting. In H. Ward (ed) *Looking After Children: Research into Practice*. London: HMSO.

Mullard, E. (2005) 'Presencing': the unseen therapeutic relationship. In B. Nyatanga and M. Astley-Pepper (eds) *Hidden Aspects of Palliative Care*. London: Quay Books.

Mullender, A., Kelly, L., Hayne, G., Malas, E. and Iman, U. (2002) *Children's Perspectives on Domestic Violence.* London: Routledge.

Munro, E. (1998) Improving social workers' knowledge base in child protection work. *British Journal of Social Work,* 28, 29-105.

Munro, E. (2007) Child Protection. London: Sage.

Munro, E. and Parton, N. (2007) How far is England in the process of introducing a mandatory reporting system? *Child Abuse Review,* 16, 5-16.

Murphy, J. (1992) *British Social Services: The Scottish Dimension*. Edinburgh: Scottish Academic Press.

Murphy, M. and Oulds, G. (2000) Establishing and developing co-operative links between substance misuse and child protection systems. In F. Harbin and M. Murphy (eds) *Substance Misuse and Child Care*. Lyme Regis: Russell House Publishing.

Nadimi, F. and Currow, D.C. (2008) Transitions in parenting responsibilities for sole parents who are dying. *Journal of Palliative Care*, 24(3), 167-172.

Napier, L. (2003) Palliative care social work. In B. Monroe and D. Oliviere (eds) *Patient Participation in Palliative Care. A Voice for the Voiceless.* Oxford: Oxford University Press.

National Collaborating Centre for Women and Children's Health and Children's Workforce Development Council (2010) *The Common Core of Skills and Knowledge*. Available:

http://www.dcsf.gov.uk/everychildmatters/strategy/deliveringservices1/commoncore/com moncoreofskillsandknowledge/ [Accessed 19 June 2010].

NCHSPCS (National Council for Hospice and Specialist Palliative Care Services) (1995) *Specialist Palliative Care: A Statement of Definitions.* London: National Council for Hospice and Specialist Palliative Care Services.

NCHSPCS (National Council for Hospice and Specialist Palliative Care Services) (2000) *The Palliative Care Survey 1999.* London: National Council for Hospice and Specialist Palliative Care Services.

NHS Quality Improvement Scotland (QIS) (2002) *Clinical Standards for Specialist Palliative Care.* Edinburgh: NHS (QIS).

NICE (National Institute for Clinical Excellence) (2004) *Guidance on Cancer Services: Improving Supportive and Palliative Care for Adults with Cancer: The Manual.* London: NICE. Oakley, A. (1981) Interviewing women: a contradiction in terms. In H. Roberts (ed) *Doing Feminist Research*. London: Routledge.

O'Connor, M., Fisher, C. and Guilfoyle, A. (2006) Interdisciplinary teams in palliative care: a critical reflection. *International Journal of Palliative Nursing*, 12(3), 132-137.

Ofsted, Healthcare Commission and HM Inspectorate of Constabulary (2008) *Joint Area Review. Haringey Children's Services Authority Area.* Available: http://www.ofsted.gov.uk/oxcare\_providers/la\_download/(id)/4657/(as)/JAR/jar\_2008\_30 9\_fr.pdf [Accessed 21 July 2009].

Ogilivie-Whyte, S.A. (2006) A Review of Evidence about the Impact of Education and Training in Child Care and Protection on Practice and Client Outcomes. Dundee: SIESWE.

O'Leary, Z. (2004) The Essential Guide to Doing Research. London: Sage.

Oliviere, D. (1999) Culture and ethnicity. *European Journal of Palliative Care,* 6(2), 53-56.

Oliviere, D. (2000) User involvement in palliative care services. *European Journal of Palliative Care*, 8(6), 238-41.

Oliviere, D. (2001) The social worker in palliative care – the 'eccentric role'. *Progress in Palliative Care*, 9, 237-241.

Oliviere, D., Hargreaves, R. and Monroe, B. (1998) *Good Practices in Palliative Care: A Psychosocial Perspective.* Aldershot: Ashgate.

Osborn, T. (2007) The psychosocial impact of parental cancer on children and adolescents: a systematic review. *Psycho-Oncology*, 16(2), 101-126.

O'Sullivan, T. (1999) Decision Making in Social Work. Basingstoke: Palgrave Macmillan.

Paley, J. (1987) Social work and the sociology of knowledge. *British Journal of Social Work*, 17(1), 169-186.

Paliadelis, P.S. (2005) Rural nursing unit managers: education and support for the role. *Rural and Remote Health*, 5(325). (Online), 2005.

Pardoe, J. (1989) The role of the social worker. In J. Robbins (ed) *Caring for the Dying Patient and the Family (2<sup>nd</sup> ed)*. London: Harper and Row.

Parker, J. (ed) (2005) Aspects of Social Work and Palliative Care. London: Quay Books.

Parker, J. and Bradley, G. (2003) *Social Work Practice: Assessment, Planning, Intervention and Review.* Exeter: Learning Matters.

Parker, J. and Bradley, G. (2007) *Social Work Practice: Assessment, Planning, Intervention and Review (2<sup>nd</sup> ed).* Exeter: Learning Matters.

Part, D. and Comben, C. (2007) The NHS research ethics process and social work. *Ethics and Social Welfare*, 1(1), 97-101.

Parton, N. (2003) Rethinking professional practice: the contributions of social constructionism and the feminist 'ethics of care'. *British Journal of Social Work,* 33, 1-16.

Parton, N. (2006) Safeguarding Childhood: Early Intervention and Surveillance in a Late Modern Society. Basingstoke: Palgrave Macmillan.

Parton, N. (2007) Safeguarding children: a socio-historical analysis. In J. Wilson and A. James (eds) *The Child Protection Handbook (3<sup>rd</sup> ed)*. Edinburgh and London: Ballière Tindall.

Parton, N. (2008) Changes in the form of knowledge in social work. From the 'social' to the 'informational'? *British Journal of Social Work*, 38(2), 253-269.

Parton, N. and O'Byrne, P. (2000) *Constructive Social Work: Towards a New Practice.* Basingstoke: Macmillan.

Parton, N., Thorpe, D. and Wattam, C. (1997) *Child Protection: Risk and the Moral Order*. London: Macmillan Press.

Patton, M.Q. (1990) *Qualitative Research and Evaluation Methods (2<sup>rd</sup> ed).* Newbury Park CA: Sage.

Patton, M.Q. (2002) *Qualitative Research and Evaluation Methods (3<sup>rd</sup> ed).* Thousand Oaks CA: Sage.

Payne, G. and Payne, J. (2004) Key Concepts in Social Research. London: Sage.

Payne, M. (1991) *Modern Social Work Theory: A Critical Introduction*. Basingstoke: The Macmillan Press.

Payne, M. (2000) Teamwork in Multi-professional Care. Basingstoke: Macmillan.

Payne, M. (2005) *Modern Social Work Theory (3<sup>rd</sup> ed).* Basingstoke: Palgrave.

Payne, M. (2006) Identity politics in multiprofessional teams. Palliative care social work. *Journal of Social Work*, 6(2), 137-150.

Payne, M. (2007) Know your colleagues: role of social work in end-of-life care. *End-of-life Care*, 1, 69-73.

Payne, M. (2010) Inequalities, end-of-life care and social work. *Progress in Palliative Care*, 18(4), 221-227.

Payne, M. and Haines, R. (2002) The contribution of psychologists to specialist palliative care. *International Journal of Palliative Nursing*, 8(8), 401-406.

Pearce, J. (2003) Parental mental health and child protection-making links through training. *Child Abuse Review*, 12, 114-118.

Peck, D. and Jones, A. (2004) Bureaucratic barriers to research and research training in the NHS. *Clinical Psychology*, 36, 7-10.

Peck, E. and Dickinson, H. (2008) *Managing and Leading in Inter-agency Settings*. Bristol: The Policy Press.

Pincus, A. and Minahan, A. (1973) *Social Work Practice: Model and Method*. Itasca, III: Peacock Publishers.

Platt, D. (2006a) Investigation or initial assessment of child concerns? The impact of the refocusing initiative on social work practice. *British Journal of Social Work*, 36, 267-281.

Platt, D. (2006b) Threshold decisions: how social workers prioritize referrals of child concern. *Child Abuse Review*, 15, 4-18.

Plummer, K. (2001) *Documents of Life 2. An Invitation to a Critical Humanism*. London: Sage.

Pollak, J. and Levy, S. (1989) Countertransference and failure to report child abuse and neglect. *Child Abuse and Neglect*, 13, 515-522.

Powell, C. (2003) The Delphi technique: myths and realities. *Journal of Advanced Nursing*, 41(4), 376-382.

Powell, J. (2002) The changing conditions of social work research. *British Journal of Social Work*, 32, 17-33.

Preston-Shoot, M. and Wigley, A. (2005) Mapping the needs of children in need. *British Journal of Social Work*, 35(2), 255-275.

Pritchard, J. (1995) *Good Practice in Supervision: Statutory and Voluntary Organisations*. London: Jessica Kingsley Publishers.

Pugh, G. (2005) Policies in the UK to promote the well-being of children. In J. Scott and H. Ward (eds) *Safeguarding and Promoting the Well-being of Children, Families and Communities*. London: Jessica Kingsley Publishers.

Quinn, A. (1998) Learning from palliative care: concepts to underpin the transfer of knowledge from specialist palliative care to mainstream social work settings. *Social Work Education*, 17(1), 9-19.

Randhawa, G. and Owens, A. (2004) Palliative care for minority ethnic groups. *European Journal of Palliative Care*, 11(1), 19-22.

Rauch, P.K. and Muriel, A.C. (2004) The importance of parenting concerns among patients with cancer. *Critical Reviews in Oncology/Hematology*, 49, 37-42.

Raymer, M. and Reese, D. (2004) The history of social work in hospice. In J. Berzoff and P.R. Silverman (eds) *Living with Dying: A Handbook for End-of-life Health Care Practitioners*. New York: Columbia University Press.

Reder, P. and Duncan, S. (1999) *Lost Innocents: a Follow Up Study of Fatal Child Abuse.* London: Routledge.

Reder, P. and Duncan, S. (2003) Understanding communication in child protection networks. *Child Abuse Review*, 12(2), 82-100.

Reese, D.J. and Sontag, M-A. (2001) Successful interprofessional collaboration on the hospice team. *Health and Social Work*, 26(3), 167-176.

Reinharz, S. (1997) Who am I? The need for a variety of selves in the field. In R. Hertz (ed) *Reflexivity and Voice*. Thousand Oaks: Sage.

Reith, M. and Payne, M. (2009) *Social Work in End-of-life and Palliative Care*. Bristol: The Policy Press.

Riessman, C.K. (1993) Narrative Analysis. Newbury Park, CA: Sage Publications.

Ritchie, J., Lewis, J. and Elam, G. (2003) Designing and selecting samples. In J. Ritchie and J. Lewis (eds) *Qualitative Research Practice. A Guide for Social Science Students and Researchers.* London: Sage.

Robbins, M. (1997) Assessing needs and effectiveness: is palliative care a special case? In D. Clark, J. Hockley and S. Ahnedzai (eds) *New Themes in Palliative Care.* Buckingham: Open University Press.

Robson, C. (2002) *Real World Research: A Resource for Social Scientists and Practitioner-researchers (2<sup>nd</sup> ed).* Oxford: Blackwell.

Rolland, J.S. (2005) Cancer and the family: an integrative model. *Cancer*, 104(11) (Supplement), 2584-2594.

Rose, W. (2010) The assessment framework. In J. Horwath (ed) *The Child's World. The Comprehensive Guide to Assessing Children in Need (2<sup>nd</sup> ed).* London: Jessica Kingsley Publishers.

Rose, W. and Barnes, J. (2008) *Improving Safeguarding Practice: Study of Serious Case Reviews 2001-2003.* Nottingham: Department for Children, Schools and Families.

Rutter, M. (1996) Children of Sick Parents. Oxford: Oxford University Press.

Saldinger, A., Cain, A.C. and Porterfield, K. (2003) Managing traumatic stress in children anticipating parental death. *Psychiatry*, 66(2), 168-181.

Saldinger, A., Cain, A.C., Porterfield, K. and Lohnes, K. (2004) Facilitating attachment between school-aged children and a dying parent. *Death Studies*, 28, 915-940.

Sarantakos, S. (2005) Social Research (3<sup>rd</sup> ed). Basingstoke: Palgrave Macmillan.

Saunders, C. (ed) (1990) *Hospice and Palliative Care: An Interdisciplinary Approach*. London: Edward Arnold.

Saunders, C. (1993) Foreword. In D. Doyle, G. Hanks and N. MacDonald (eds) *Oxford Textbook of Palliative Medicine.* Oxford: Oxford University Press.

Schmitt, F., Santalahti, P., Saarelainen, S., Savonlahti, E., Romer, G. and Piha, J. (2008) Cancer families with children: factors associated with family functioning - a comparative study in Finland. *Psycho-Oncology*, 17, 363-372.

Schön, D.A. (1983) *The Reflective Practitioner: How Professionals Think in Action*. London: Temple Smith.

SCIE (2005) Research Briefing 11: The Health and Well-being of Young Carers. London: SCIE.

Scott, D. (1997) Inter-agency conflict: an ethnographic study. *Child and Family Social Work,* 2, 73-80.

Scottish Executive (2001) For Scotland's Children: Better Integrated Services. Scottish Executive: Edinburgh.

Scottish Executive (2002) 'It's Everyone's Job to Make Sure I'm Alright'. Report of the Child Protection Audit and Review. Edinburgh: Scottish Executive.

Scottish Executive (2004) 'Getting it Right for Every Child'. Consultation on Review of the Children's Hearings System. Edinburgh: Scottish Executive.

Scottish Executive (2005a) '*Getting it Right for Every Child'. Proposals for Action.* Edinburgh: Scottish Executive.

Scottish Executive (2005b) *Protecting Children and Young People: Child Protection Committees.* Edinburgh: Scottish Executive.

Scottish Executive (2006a) '*Getting it Right for Every Child*' *Implementation Plan.* Edinburgh: Scottish Executive.

Scottish Executive (2006b) *Changing Lives. Report of 21<sup>st</sup> Century Social Work Review.* Edinburgh: Scottish Executive.

Scottish Executive (2007) *Protecting Children and Young People: Interim Guidance for Child Protection Committees for Conducting a Serious Case Review.* Edinburgh: Scottish Executive.

Scottish Executive (2008) *A Guide to Getting it Right for Every Child*. Edinburgh: Scottish Executive.

Scourfield, J. (2003) Gender and Child Protection. Basingstoke: Palgrave Macmillan.

Scourfield, J. (2006) The challenge of engaging fathers in the child protection process. *Critical Social Policy*, 26(2), 440-449.

Scourfield, J. and Maxwell, N. (2010) Social work doctoral students in the UK: a webbased survey and search of the index of theses. *British Journal of Social Work*, 40, 548-566.

Searing, H. (2003) The continuing relevance of casework ideas to long-term child protection work. *Child and Family Social Work,* 8, 311-320.

Segal, J. and Simkins, J. (1996) *Helping Children with III and Disabled Parents: A Guide for Parents and Professionals*. London: Jessica Kingsley Publishers.

Semmens, J. and Peric, J. (1995) Children's experience of a parent's chronic illness and death. *Australian Journal of Advanced Nursing*, 13(2), 30-38.

Semple, C.J. and McCance, J. (2010) Parent's experience of cancer who have young children: a literature review. *Cancer Nurse*, 33(2), 110-118.

Sheehan, D. (2007) Interaction Patterns Between Parents with Advanced Cancer in Hospice and their Adolescent Children. Doctoral dissertation: University of Akron.

Sheldon, F. (1993) Education and training for social workers in palliative care. In D. Doyle, G. Hanks and N. MacDonald (eds) *Oxford Textbook of Palliative Medicine*. Oxford: Oxford University Press.

Sheldon, F. (1997) Psychosocial Palliative Care. Cheltenham: Stanley Thornes.

Sheldon, F. (2000) Dimensions of the role of the social worker in palliative care. *Palliative Medicine*, 14, 491-498.

Sheldon, F. (2001) Social work in palliative care: counselling and communicating. *Progress in Palliative Care*, 9, 242-243.

Sheldon, F. (2008) Social impact of advanced metastatic cancer. In M. Lloyd-Williams (ed) *Psychosocial Issues in Palliative Care (2<sup>nd</sup> ed)*. Oxford: Oxford University Press.

Sheppard, M. (1995) Social work, social science and practice wisdom. *British Journal of Social Work*, 25, 265-293.

Sheppard, M. and Ryan, K. (2003) Practitioners as rule using analysts: a further development of process knowledge in social work. *British Journal of Social Work*, 33(2), 157-176.

Sidebotham, P. and Weeks, M. (2010) Multidisciplinary contributions to assessment of children in need. In J. Horwath (ed) *The Child's World. The Comprehensive Guide to Assessing Children in Need (2<sup>nd</sup> ed).* London: Jessica Kingsley Publishers.

Siegel, K., Mesagno, F.P., Karus, D., Christ, G., Banks, K. and Moynihan, R. (1992) Psychoscial adjustment of children with a terminally ill parent. *Journal of American Academy of Child Adolescent Psychiatry*, 31(2), 327-33.

Siegel, K., Raveis, V.H. and Karus, D. (1996) Patterns of communication with children when a parent has cancer. In L. Baider, C.L. Cooper and A. Kaplan De-Nour (eds) *Cancer in the Family*. Chichester: John Wiley & Sons.

Siegel, K., Raveis, V.H. and Karus, D. (2000) Correlates of self-esteem among children facing the loss of a parent to cancer. In L. Baider, C.L. Cooper and A. Kaplan De-Nour (eds) *Cancer in the Family (2<sup>nd</sup> ed)*. Chichester: John Wiley & Sons.

Silverman, D. (2000) Doing Qualitative Research: A Practical Handbook. London: Sage.

Sim, J. (1998) Collecting and analysing qualitative data: issues raised by the focus group. *Journal of Advanced Nursing*, 28(2), 345-452.

Skinner, T., Hester, M. and Malos, E. (2005) *Researching Gender Violence: Feminist Methodology in Action.* Cullompton: Willan.

Smale, G. and Tuson, G. with Brehal, N. and Marsh, P. (1993) *Empowerment, Assessment, Care Management and the Skilled Worker.* London: NISW and The Stationery Office.

Small, N. (2001) Social work and palliative care: critical commentary. *British Journal of Social Work*, 31, 961-971.

Small, N. (2003) The changing NHS, user involvement, and palliative care. In B. Monroe and D. Oliviere (eds) *Patient Participation in Palliative Care. A Voice for the Voiceless.* Oxford: Oxford University Press.

Smith, A.B., Taylor, N.J. and Tapp, P. (2003) Rethinking children's involvement in decision-making after parental separation. *Childhood*, 10, 201-216.

Smith, J.A., Jarman, M. and Osborn, M. (1999) Doing interpretative phenomenological analysis. In M. Murray and K. Chamberlain (eds) *Qualitative Health Psychology: Theories and Methods.* London: Sage Publications.

Snape, D. and Spencer, L. (2003) The foundations of qualitative research. In J. Ritchie and J. Lewis (eds) *Qualitative Research Practice. A Guide for Social Science Students and Researchers.* London: Sage.

Social Policy Association (2009) *SPA Guidelines on Research Ethics*. Available: http://www.social-policy.org.uk/downloads/SPA\_code\_ethics\_jan09.pdf [Accessed 27 June 2010].

Social Services Inspectorate (1997) *Evaluating Child Protection Services: Findings and Issues.* London: HMSO.

Social Work Task Force (2009) *Building a Safe, Confident Future. Final Report of the Social Work Task Force.* London: Department of Schools and Families. Available: http://publications.dcsf.gov.uk/eOrderingDowload/01114-2009DOM-EN.pdf [Accessed 8 June 2010].

Speck, P. (2006) Conclusion. In P. Speck (ed) *Teamwork in Palliative Care. Fulfilling or Frustrating?* Oxford: Oxford University Press.

Spencer, L., Ritchie, J. and O'Connor, W. (2003) Analysis: practices, principles and processes. In J. Ritchie and J. Lewis (eds) *Qualitative Research Practice. A Guide for Social Science Students and Researchers.* London: Sage.

Spira, M. and Kenemore, E. (2000) Adolescent daughters of mothers with breast cancer: impact and implications. *Clinical Social Work Journal*, 28(2), 183-194.

Spratt, T. (2000) Decision making by senior social workers at point of first referral. *British Journal of Social Work*, 30, 597-616.

Spratt, T. (2001) The influence of child protection orientation on child welfare practice. *British Journal of Social Work*, 31, 933-954.

Srivastava, O.P., Stewart, J., Fountain, R. and Ayre, P. (2005) Common operational approach using the 'Graded Care Profile' in cases of neglect. In J. Taylor and B. Daniel (eds) *Child Neglect. Practice Issues for Health and Social Care.* London: Jessica Kingsley Publishers.

Stafford, A. and Vincent, S. (2008) *Safeguarding and Protecting Children and Young People.* Edinburgh: Dunedin Academic Press.

Stanley, N., Cleaver, H. and Hart, D. (2010) The impact of domestic violence, parental mental health problems, substance misuse and learning disability on parenting capacity. In J. Horwath (ed) *The Child's World. The Comprehensive Guide to Assessing Children in Need (2<sup>nd</sup> ed).* London: Jessica Kingsley Publishers.

Stanley, N., Penhale, B., Riordan, D., Barbour, S. and Holden, S. (2003) Working on the inter-face: identifying professional responses with mental health and child-care needs. *Health and Social Care in the Community*, 11, 208-218.

Statham, J. and Chase, E. (2010) *Childhood Wellbeing: A Brief Overview. Briefing Paper 1.* Childhood Wellbeing Centre. Available: http://www.education.gov.uk/research/data/uploadfiles/FinalChildDFEwebsite.pdf [Accessed 21 September 2010].

Stevens, M. and Higgins, D. (2002) The influence of risk and protective factors in burn out by those who work with maltreated children. *Child Abuse and Neglect*, 11, 313-331.

Stevenson, O. (1998) It was more difficult than we thought: a reflection on 50 years of child welfare practice. *Child and Family Social Work,* 3, 153-161.

Stevenson, O. (2005) Working together in cases of neglect. In J. Taylor and B. Daniel (eds) *Child Neglect. Practice Issues for Health and Social Care.* London: Jessica Kingsley Publishers.

Stevenson, O. (2007) *Neglected Children and Their Families (2<sup>nd</sup> ed).* Oxford: Blackwell Publishers.

Stewart, D.W. and Shamdasani, P.N. (1990) *Focus Groups: Theory and Practice*. Newbury Park: Sage.

Stewart, J. (2001) Is the Delphi technique a qualitative method? *Medical Education*, 35, 922-923.

Stokes, J. (2004) *Then, Now and Always: Supporting Children as They Journey Through Grief. A Guide for Practitioners.* Cheltenham: Winston's Wish.

Stokes, J. (2005) Family assessment. In B. Monroe and F. Kraus (eds) *Brief Interventions with Bereaved Children.* Oxford: Oxford University Press.

Sudbery, J. (2002) Key features of therapeutic social work: the use of the relationship. *Journal of Social Work Practice*, 16(2), 149-162.

Taylor, A. and Kroll, B. (2004) Working with substance misuse: dilemmas for practice. *British Journal of Social Work*, 34(8), 115-1132.

Taylor, J. and Daniel, B. (2005) Neglect in theory and practice: the messages for health and social care. In J. Taylor and B. Daniel (eds) *Child Neglect. Practice Issues for Health and Social Care.* London: Jessica Kingsley Publishers.

Taylor, J. and Daniel, B. (2006) Standards for education and training for interagency working in child protection in the UK. *Nurse Education Today*, 26(3), 179-182.

Thastum, M., Johansen, M.B., Gubba, L., Olesen, L.B. and Romer, G. (2008) Coping, social relations, and communication: A qualitative exploratory study of children of parents with cancer. *Clinical Child Psychology and Psychiatry*, 13(1), 123-138.

The British Association for Counselling and Psychotherapy (2010) http://www.bacp.co.uk/education/whatiscounselling.html [Accessed 15 August 2010].

The Princess Royal Trust for Carers (2006) *Guidance: Delivering Every Child Matters for Young Carers.* Available: http://static.carers.org/files/trust-guidance-delivering-every-child-matters-for-young-carers-2177.doc [Accessed 18 July 2010].

The Scottish Government (2005) *Getting it Right for Every Child. Proposals for Action.* Available: http://www.scotland.gov.uk/Publications/2005/06/20135608/56144 [Accessed 15 August 2010].

The Scottish Government (2009) Changing Professional Practice and Culture to Get It Right for Every Child. An evaluation report of the development and early implementation phases of GIRFEC in Highland 2006-2009. Available:

http://www.scotland.gov.uk/Resource/Doc/292706/0090352.pdf [Accessed 8 August 2010].

The Scottish Government (2010) *Overview of GIRFEC programme*. Available: http://www.scotland.gov.uk/Topics/People/Young-People/childrensservices/girfec/programme.overview [Accessed 8 August 2010].

The Scottish Office (1998) *Protecting Children: A Shared Responsibility. Guidance on Interagency Co-operation.* Edinburgh: HMSO.

Thoburn, J., Lewis, A. and Shemmings, D. (1995) *Paternalism or Partnership? Family Involvement in the Child Protection Process.* London: HMSO.

Thoburn, J., Brandon, M. and Lewis, A. (1997) Need, risk and significant harm. In N. Parton (ed) *Child Protection and Family Support*. London: Routledge.

Thomas, N., Stainton, T., Jackson, S., Cheung, W.Y., Doubtfire, S. and Webb, A. (2003) 'Your friends don't understand': Invisibility and unmet need in the lives of 'young carers'. *Child and Family Social Work*, 8, 35-46.

Thomas, S.J. (1999) *Designing a Survey That Works! A Step-by-step Guide*. Thousand Oaks: Corwin Press.

Thompson, F. and Payne, S. (2000) Bereaved children's questions to a doctor. *Mortality*, 5(1), 74-96.

Thompson, N. (2010) *Theorizing Social Work Practice*. Basingstoke: Palgrave Macmillan.

Thomson, J. (2005) Culture and ethnicity - its relevance for social work practice in palliative care. In J. Parker (ed) *Aspects of Social Work and Palliative Care*. London: Quay Books.

Thorpe, D. and Bilson, A. (1998) From protection to concern: child protection careers without apologies. *Children and Society*, 12, 373-386.

Trevithick, P. (2003) Effective relationship-based practice: a theoretical explanation. *Journal of Social Work Practice*, 17(2), 163-176.

Trevithick, P. (2005) *Social Work Skills: A Practice Handbook (2<sup>nd</sup> ed).* Maidenhead: Open University Press.

Tuck, V. (2000) Links between social deprivation and harm to children. In N. Baldwin (ed) *Protecting Children. Promoting their Rights*. London: Whiting and Birch.

Turner, J., Clavarino, A., Yates, P., Hargreaves, M., Connors, V. and Hausmann, S. (2007) Development of a resource for parents with advanced cancer: what do parents want? *Palliative and Supportive Care*, 5, 135-145.

Turney, D. (2000) The feminizing of neglect. Child and Family Social Work, 5(1), 47-56.

Tye, C. and Precey, G. (1999) Building bridges: the interface between adult mental health and child protection. *Child Abuse Review*, 8(3), 164-171.

Underdown, A. (2002) 'I'm growing up too fast': Messages from young carers. *Children and Society*, 16, 57-60.

Unison and BASW Scotland (2006) *Asylum in Scotland, Child's Welfare Paramount?* Available: http://www.unisonscotland.org.uk/socialwork/asylumbooklet/asylumbooklet.pdf [Accessed 14 June 2010].

United Nations (1989) *The UN Convention on the Rights of the Child*. Geneva: United Nations.

Vachon, M.L. (1993) Emotional problems in palliative medicine: patient, family and professional. In: D. Doyle, G. Hanks and N. Macdonald (eds) *Oxford Textbook of Palliative Medicine*. Oxford: Oxford University Press.

Vincent, S. (2008) *Child Protection Statistics. A UK Comparison. Briefing No. 3.* Centre for UK-wide Learning in Child Protection: Edinburgh. Available: http://www.clicp.ed.ac.uk/publications/briefings/briefings03.pdf [Accessed 29 September 2010].

Visser, A., Huizinga, G.A., van der Graaf, W.T.A., Hoekstra, H.J. and Hoekstra-Weebers, J.E.H.M. (2004) The impact of parental cancer on children and the family: a review of the literature. *Cancer Treatment Reviews*, 30, 683-694. Visser, A., Huizinga, G.A., Hoekstra, H.J., van der Graaf, W.T.A., Klip, E.C., Pras, E. and Hoekstra-Weebers, J.E.H.M. (2005). Emotional and behavioural functioning of children of a parent diagnosed with cancer: A cross-informant perspective. *Psycho-Oncology*, 14, 746-758.

Walker, S. and Thurston, C. (2006) *Safeguarding Children and Young People*. Lyme Regis: Russell House Publishing.

Walsh, H. (2005) Hidden aspects of psychosocial care. In B. Nyatanga and M. Astley-Pepper (eds) *Hidden Aspects of Palliative Care*. London: Quay Books.

Walsh, F. and McGoldrick, M. (1998) A family systems perspective on loss, recovery and resilience. In P. Sutcliffe, G. Tufnell and M. Cornish (eds) *Working with the Dying and Bereaved.* Basingstoke: Macmillan Press.

Walsh-Burke, K. (2004) Assessing mental health risk in end of life care. In J. Berzoff and P.R. Silverman (eds) *Living with Dying: A Handbook for End-of-life Health Care Practitioners.* New York: Columbia University Press.

Walshe, C.E., Caress, A.L., Chew-Graham, C. and Todd, C.J. (2004). Case studies: A research strategy appropriate for palliative care? *Palliative Medicine*, 18, 677-684.

Walter, T. (1994) The Revival of Death. London: Routledge.

Walter, T. (1996) A new model of grief: bereavement and biography. *Mortality*, 1(1), 7-25.

Ward, H. and Rose, W. (2002) *Approaches to Needs Assessment in Children*. Jessica Kingsley Publishers.

Watson, D. and West, J. (2006) *Social Work Process and Practice: Approaches, Knowledge and Skills.* Basingstoke: Palgrave Macmillan.

Wattam, C. and Parton, N. (1999) *Child Sexual Abuse: Responding to the Experience of Children.* Chichester: John Wiley.

Welsh Assembly Government (2006) Safeguarding Children: Working Together Under the Children Act 2004. Cardiff: Welsh Assembly Government.

Wenger, E. (1998) *Communities of Practice: Learning, Meaning and Identity*. Cambridge: Cambridge University Press.

Whittaker, A. (2009) Research Skills for Social Work. Exeter: Learning Matters.

Wilks, T. (2004) The use of vignettes in qualitative research in social work values. *Qualitative Social Work,* 3(1), 78-87.

Willis, L., Peck, M., Sells, S. and Rodabaugh, K.J. (2001) Custody planning: a retrospective review of oncology patients who were single parents. *Journal of Pain and Symptom Management*, 21(5), 380-384.

Winnicott, D.W. (1965) The Family and Individual Development. London: Tavistock.

Winston's Wish (2007) The Mummy Diaries. Available:

http://www.winstonswish.org.uk/page/asp?section=0001000100100015 [Accessed 17 June 2010].

Woodcock, J. (2003) The social work assessment of parenting: an exploration. *British Journal of Social Work*, 33, 87-106.

Worden, J.W. (1996) Children and Grief. When a Parent Dies. New York: Guilford.

World Health Organisation (1990) Cancer Pain Relief and Palliative Care. Technical report series 804. Geneva: World Health Organisation.

World Health Organisation (2002) *National Cancer Control Programmes: Policies and Managerial Guidelines (2<sup>nd</sup> ed).* Geneva: World Health Organisation.

Wright, M. and Leahey, M. (2005) *Nurses and Families. A Guide to Family Assessment and Intervention (4<sup>th</sup> ed).* Philadelphia: F.A. Davis.

Wyness, M. (2006) *Childhood and Society: An Introduction to the Sociology of Childhood.* Basingstoke: Palgrave.

Yin, R.K. (2003) *Case Study Research: Design and Methods (3<sup>rd</sup> ed)*. Thousand Oaks: Sage Publications.

Zaider, T. and Kissaine, D.W. (2007) Resilient families. In B. Monroe and D. Oliviere (eds) *Resilience in Palliative Care*. Oxford: Oxford University Press.

Zaider, T. and Kissaine, D.W. (2009) The assessment and management of family distress during palliative care. *Current Opinion in Supportive and Palliative Care*, 3, 67-71.

## PARTICIPANT INFORMATION SHEET

Focus Groups

## TITLE OF PROJECT

Safeguarding Children: The Role of the Palliative Care Social Worker

#### INVITATION TO TAKE PART IN A RESEARCH STUDY

You are being asked to take part in a research study, which is exploring the role of the palliative care social worker in situations where a main carer has a terminal illness and the child/children in the family may be vulnerable and at risk.

Carole Comben is undertaking this research as a PhD student based at the University of Dundee in the School of Education, Social Work and Community Education. Carole is a qualified social worker with over 30 years of experience in social work education with particular emphasis on child care. The study is expected to be completed by October 2009. The main data collection phase will be taking place between end July 2007 and April 2008 and will involve focus groups, the administration and analysis of questionnaires and semi-structured interviews.

## PURPOSE OF THE RESEARCH STUDY

This study aims to consider the impact of recent policy and legal developments regarding the safeguarding and protection of children and young people on the work of palliative care social workers. Recent UK research, [Beresford et al. (2006) *Palliative Care, Social Work and Service Users.* London: Jessica Kingsley] shows that prior to meeting the palliative care social worker, patients and families tended to have negative views of social workers in general associating them with the removal of children and loss of independence. Once engaged with each other, this view changed and the psychological support role was seen as particularly significant to them. This study will consider whether this therapeutic relationship has any bearing on the decisions made about the care of children in the family of the patient.

Participation in this research would enable there to be a clarification of the role of the palliative care social worker with regard to safeguarding children, an identification of training required to enable them to carry out their role competently and confidently and, the development of good practice guidelines.

#### TIME COMMITMENT

This part of the study will require attendance of a small number of specialist palliative care social workers at focus groups. Each participant is invited to attend one group meeting of approximately 6-8 workers. The maximum time for the group session will be 1.5 hours.

#### TERMINATION OF PARTICIPATION

You may decide to stop being a part of the research study at any time without explanation.

/RISKS

#### RISKS

There are no known risks for you in this study.

#### COST, REIMBURSEMENT AND COMPENSATION

Your participation in this study is voluntary.

#### **CONFIDENTIALITY/ANONYMITY**

The focus groups will be recorded with the consent of the participants. The recording and transcripts will be kept securely within the School of Education, Social Work and Community Education and will be destroyed two years after the end of the project.

The data we collect will not contain any personal information about you except your professional qualifications, place of employment and length of time in this field of work.

No one will link the data you provide to your identity, name or place of work.

The results will be contained in the PhD thesis. An executive summary will be sent to those workers who participate in the focus groups and semi-structured interviews and also to Association of Palliative Care Social Workers for dissemination. It is anticipated that articles will be submitted to the relevant professional journals. Participants will remain anonymous and will not be identified in any way.

#### FOR FURTHER INFORMATION ABOUT THIS RESEARCH STUDY

Carole Comben will be glad to answer your questions about this study at any time and her contact details are given below.

If you want to find out about the final results of this study, you should also contact Carole Comben.

You may contact her at C.R.Comben@dundee.ac.uk or write to:

Carole Comben Post Graduate Research Office OTI Building – Room 107 School of Education, Social Work and Community Education University of Dundee Nethergate Dundee DD1 4HN.

## **FOCUS GROUPS**

## **Discussion guide**

Note that the terms safeguarding and protecting are both used due to the different use of terminology in Scotland and England/Wales.

#### Introduction

Explanation of the purpose of the study and confidentiality aspects.

#### Specific question

1 Each member will be invited to state where they work, how long in post, their employer (e.g. local authority or hospice) and how long in post/this field of work.

#### General questions to group as a whole

- 2 What is your understanding of safeguarding/protecting children in general?
- 3 How do you think this understanding applies to your role as a specialist palliative care social worker?
  - Is this part of your remit?
  - Are there guidelines in your agency on child protection/safeguarding children?
- 4 What training have you received on the topic of safeguarding and protecting children?
  - How easy has it been to access this training?
  - Did you find the training relevant to your position?
  - Can you identify any gaps?
- 5 To whom do you go for supervision/professional support regarding matters in relation to child protection/safeguarding children?
  - What are your views about this?
- 6 How do you identify situations where a child might be at risk?
  - Any examples of situations you are able to share?
- 7 How would you describe your links with your social work colleagues in the relevant local authority child and family teams?

- 8 What is the role of the specialist palliative care team in relation to safeguarding/protecting children?
  - Any examples of dilemmas?
  - Any examples of positive practice?
- 9 Do you think the nature of the illness of the carer may have any influence on your decisions about the child care offered by the family?
  - To whom are you responsible?





DEPARTMENT OF APPLIED SOCIAL SCIENCE

## Safeguarding Children

## The Role of the Palliative Care Social Worker

# Questionnaire

Carole Comben Postgraduate Research Student Department of Applied Social Science Colin Bell Building University of Stirling Stirling FK9 4LA

#### Notes

The research study is exploring the role of the palliative care social worker in situations where an adult service user has a terminal illness and the child/children in his or her family may be vulnerable and at risk. By completing the enclosed questionnaire you will help to clarify this role, identify the support and training required to enable workers to carry out their role competently and confidently, and assist in the development of good practice guidelines.

All the information you are able to give will greatly assist in the mapping of current working practices and identification of any dilemmas. You are not asked to identify yourself in this questionnaire and therefore no one will be able to link the information you provide to your identity, name or place of work.

It is estimated that the questionnaire will take up to 45 minutes to complete.

# Please go through the questions and answer them in the light of your normal practice, perceptions or experience.

When completed, please return the questionnaire in the postage paid, preaddressed envelope by 19 March 2008.

If you have any queries about the questionnaire or the research study please contact Carole Comben at c.r.comben@stir.ac.uk

If you are not currently working with adults in palliative care please tick the box at the bottom of the page and return the questionnaire in the pre-paid envelope provided.

#### Terminology

Please note that the terms 'safeguarding', 'child protection' and 'protecting children' are all used in the questionnaire to reflect the terminology used in the four countries of the UK.

'Child' refers to children and young people up to the age of 18 years.

I am not currently working with adults in palliative care.  $\Box$ 

## **GENERAL INFORMATION**

This section asks for information about yourself and your job.

1	What is your job title?	
2	When did you gain your social work qualification?	
3	How long have you been in your current post?	
4	How long, approximately, have you been working in palliative ca	re social work?
5	Where is the location of your workplace? ( <i>Please tick</i> ) Day hospice In-patient hospice Combined day and in-patient hospice Hospital Community Other please state	
6	Who is your employer? <i>(Please tick)</i> Hospice NHS Local authority Other please state	
7	In which country do you work? <i>(Please tick)</i> England Wales Scotland Northern Ireland	
8	Please indicate your gender. <i>(Please tick)</i> Male Female	

#### YOUR ROLE IN SAFEGUARDING CHILDREN / CHILD PROTECTION

9 What is your role when working with service users who are parents with children?

10 Please describe a recent example where you have initiated discussion with a service user regarding the care/parenting of his or her child/children.

- 11 Please provide a recent example of what you did in each of the following three circumstances, briefly describing how concerns were raised, what happened and how the situation was dealt with.
  - (a) A service user expressed concern about his or her child/children's care.

(b) Another family member expressed concern about the care of the child/children of a service user.

(c) Another professional expressed concern about the care of the child/children of a service user.

12 What influence do you think the nature of the illness of a parent may have on your decisions about the child care offered by that person and the family?

13 Overall, what do you see your role to be in safeguarding children?

#### SUPPORT AND JOINT WORKING

14 If you ever have cause to be worried about the care and welfare of a child of a service user, who would you discuss this with?

15 If you are a member of a Palliative Care Team, how do other members view your role within the team in relation to safeguarding/protecting children?

16 Please describe your contact with social workers in the local authority child and family and child protection teams in your area. (*Where possible, please illustrate your answer with an example of where this contact has had a positive outcome and an example of where the contact has been less successful.*)

17 How easy is it for you to access services for children?

#### TRAINING

18 What training, if any, have you received in the last two years in relation to safeguarding children?

In general, how supportive to you in your role did you find the training to be? (*Please circle number that most closely matches how you feel.*)

Very supportive	12	3	45	6	Not at all supportive
-----------------	----	---	----	---	-----------------------

- 19 How easy is it for you to access training on matters regarding children?
- 20 Please describe any training on safeguarding children/child protection that you believe would be of benefit to you in your work.

#### IN CONCLUSION

21 Overall, how confident do you feel about working in situations regarding safeguarding children/child protection? (*Please circle number that most closely matches how you feel.*)

Not at all confident 1.....2.....3.....4.....5......6 Very confident

What, if anything, would help you to feel more confident?

22 If there is anything else you would like to add regarding the role of the palliative care social worker in relation to safeguarding children, please write it here.

Thank you for taking the time to complete this questionnaire.

Please return the completed questionnaire in the envelope provided.





DEPARTMENT OF APPLIED SOCIAL SCIENCE

14 February 2008

**Dear Member** 

#### Safeguarding Children: The Role of the Palliative Care Social Worker

This letter asks you to participate in a research study. The enclosed questionnaire is designed to obtain your views on your role as a palliative care social worker in situations where an adult service user has a terminal illness and a child/children in his or her family may be vulnerable and at risk. All the information you are able to provide will greatly assist in the mapping of current working practices and identification of any dilemmas. The Association of Palliative Care Social Workers has given permission for the questionnaire to be sent out to all its members and the study has been approved by the University of Stirling Department of Applied Social Science Ethics Committee. Attached is an information sheet giving more details about the study and how the results will be disseminated. If you are not currently working with adults in palliative care, please tick the box on page 2 of the questionnaire and return the questionnaire in the pre-addressed envelope provided.

It is estimated that the questionnaire will take up to 45 minutes to complete. You are not asked to identify yourself in this questionnaire and therefore no one will be able to link the information you provide to your identity, name or place of work. Please complete the questionnaire as fully as possible and return it to me in the postage paid, pre-addressed envelope by 19 March 2008.

The next phase of the research will be individual interviews with a sample of palliative care social workers. These are scheduled to take place between June and August 2008 at a date, time and location to suit each person being interviewed. The interviews will explore in some detail examples of situations where there have been concerns about the vulnerability of children. The emphasis will be on the outcomes for the child, family and worker. If you are interested in taking part in this important part of the research, please complete the enclosed post card and return it separately from the questionnaire – postage is pre-paid. Full information about the interviews will then be sent to you.

Should you require any further information about the study, or if you have any questions, please do not hesitate to contact me at <u>c.r.comben@stir.ac.uk</u> or write to me at the above address.

Thank you in advance for your assistance.

Yours sincerely

Carole Comben

## PARTICIPANT INFORMATION SHEET

Questionnaire

## TITLE OF PROJECT

Safeguarding Children: The Role of the Palliative Care Social Worker

#### INVITATION TO TAKE PART IN A RESEARCH STUDY

You are being asked to take part in a research study, which is exploring the role of the palliative care social worker in situations where an adult service user has a terminal illness and the child/children in his or her family may be vulnerable and at risk.

Carole Comben is undertaking this research as a PhD student based at the University of Stirling in the Department of Applied Social Science. Carole is a qualified social worker with over 30 years of experience in social work education with particular emphasis on child care. The study is expected to be completed by October 2009. The main data collection phase will be taking place between July 2007 and August 2008 and will involve focus groups, the administration and analysis of questionnaires and semi-structured interviews.

#### PURPOSE OF THE RESEARCH STUDY

Palliative care considers the needs not only of the patient but also of the patient's family. This study aims to consider how palliative care social workers engage with families where children and young people may be vulnerable and at risk, exploring those aspects of their role which may have a bearing on decisions to be taken regarding the care of children. It will also consider how recent policy and legal developments regarding the safeguarding and protection of children and young people may impact on the work of palliative care social workers.

By participating in this research you will assist in clarifying the role of palliative care social workers with regard to safeguarding children, in identifying the training and support required to enable them to carry out their role competently and confidently and the development of good practice guidelines.

#### TIME COMMITMENT

The questionnaire will be distributed to all members of the Association of Palliative Care Social Workers at the beginning of February 2008. It is expected that the questionnaire will take up to 45 minutes to complete.

#### **TERMINATION OF PARTICIPATION**

You may decide to stop being a part of the research study at any time without explanation.

**/**RISKS

#### RISKS

There are no known risks for you in this study.

#### COST, REIMBURSEMENT AND COMPENSATION

Your participation in this study is voluntary.

#### **CONFIDENTIALITY/ANONYMITY**

The questionnaire does not ask for any identifying data and therefore no one will be able to link the information you provide to your identity, name or place of work.

The results will be contained in a PhD thesis. An executive summary will be sent to the Association of Palliative Care Social Workers for dissemination to its members. It is anticipated that articles will be submitted to the relevant professional journals. Participants will remain anonymous and will not be identified in any way.

#### FOR FURTHER INFORMATION ABOUT THIS RESEARCH STUDY

Carole Comben will be pleased to answer your questions about this study at any time and her contact details are given below.

If you want to find out about the final results of this study, you should also contact Carole Comben.

You may contact her at <u>c.r.comben@stir.ac.uk</u> or write to:

Carole Comben Post Graduate Student Department of Applied Social Science Colin Bell Building University of Stirling Stirling FK9 4LA.

The Department of Applied Social Science Research Ethics Committee of the University of Stirling has reviewed and approved this research study.





DEPARTMENT OF APPLIED SOCIAL SCIENCE

Dear

#### Safeguarding Children: The role of the palliative care social worker

Thank you for returning the card indicating your interest in taking part in the interviewing phase of the study.

The interviews are planned to take place between June and October 2008. They will explore situations identified by you where you have been engaged with families where there are children and young people. The emphasis will be on a 'solution focus approach' to identify reasons why a particular intervention was successful and to learn from situations where there was a lack of success. The discussion will concentrate on the process of the work and the outcomes for the child, family and yourself. No identifying names will be asked for and no one will link the data you provide to your identity, name or place of work or anyone else involved in the work. It is expected that the interview will last between 1½ and 2 hours. A participant information sheet providing more details is enclosed.

I am planning to interview up to 20 palliative care social workers from a range of work settings and geographical locations. If you are still interested in taking part, please would you complete the attached form and return it to me in the enclosed postage paid envelope as soon as you are able. If you do return a completed form, I will contact you again after 5 June regarding possible interview dates.

Please contact me at the above address or at <u>c.r.comben@stir.ac.uk</u> if you have any queries. I look forward to hearing from you.

Yours sincerely

Carole Comben Post Graduate Research

Enc: Form Postage paid return envelope Participant Information Sheet – Interviews

#### PARTICIPANT INFORMATION SHEET

Interviews

#### TITLE OF PROJECT

Safeguarding Children: The Role of the Palliative Care Social Worker

**RESEARCHER:** Carole Comben

#### INVITATION TO TAKE PART IN A RESEARCH STUDY

You are being asked to take part in a research study, which is exploring the role of the palliative care social worker in situations where the service user has a terminal illness and the child/children in his or her family may be vulnerable and at risk.

I am undertaking this research as a PhD student based at the University of Stirling in the Department of Applied Social Science. I am a qualified social worker with over 30 years of experience in social work education with particular emphasis on child care. The study is expected to be completed by October 2009. The main data collection phase will be taking place between end July 2007 and August 2008 and will involve focus groups, the administration and analysis of questionnaires and semi-structured interviews.

#### PURPOSE OF THE RESEARCH STUDY

Palliative care considers the needs not only of the patient but also of the patient's family. This study aims to consider how palliative care social workers engage with families where there are children and young people, exploring those aspects of their role which may have a bearing on decisions to be taken regarding the care of children. It will also consider how recent policy and legal developments regarding the safeguarding and protection of children and young people may impact on the work of palliative care social workers.

By participating in this research you will assist in clarifying the role of palliative care social workers with regard to safeguarding children, in identifying the training and support required to enable them to carry out their role competently and confidently and the development of good practice guidelines.

#### TIME COMMITMENT

This phase of the research - the semi-structured interviews with a small number of palliative care social workers - will take place between June and August 2008. The interview will take place at a date, time and venue to suit you. It is expected that the interview should last between 1.5 hours and 2 hours.

#### NATURE OF PARTICIPATION

Your participation in this study is voluntary. You may decide to stop being a part of the research study at any time without explanation.

/COST, REIMBURSEMENT AND COMPENSATION

#### COST, REIMBURSEMENT AND COMPENSATION

No funding is available to cover any costs that may be incurred by you during this study.

#### RISKS

There are no known risks for you in this study.

#### **CONFIDENTIALITY/ANONYMITY**

The interviews will be audio recorded with your consent and will be transcribed and analysed by me. The recording and transcripts will be kept securely within the Department of Applied Social Science; the consent forms will also be kept secure, separately from the recording and transcripts. All this material will be destroyed five years after the end of the project. The transcribed data will not contain any personal information about you except your professional qualifications, type of employment and length of time in this field of work. It will not contain any identifying information about service users. No one will link the analysis to your identity, name or place of work or to any others involved in your work. If during the course of the interview it appears that a child under discussion may be at risk and the situation requires intervention, then we will discuss what action should be taken.

The results of the study will be contained in a PhD thesis. An executive summary will be sent to those workers who participate in the focus groups and semi-structured interviews and also to the Association of Palliative Care Social Workers for dissemination to its members. It is anticipated that articles will be submitted to the relevant professional journals. Participants will remain anonymous and will not be identified in any way.

#### COMPLAINTS

If at any time you wish to make a complaint about any aspect of the interview phase of the study, please do so by contacting Professor Brigid Daniel in the first instance at the Department of Applied Social Science - <u>b.m.daniel@stir.ac.uk</u>

#### FOR FURTHER INFORMATION ABOUT THIS RESEARCH STUDY

I will be pleased to answer your questions about this study at any time and my contact details are given below.

If you want to find out about the final results of this study, please contact me at <u>c.r.comben@stir.ac.uk</u> or write to:

Carole Comben Department of Applied Social Science University of Stirling Stirling FK9 4LA.

The Department of Applied Social Science of the University of Stirling Research Ethics Committee has reviewed and approved this research study.

## **INTERVIEW GUIDE**

## 1 Introduction

- Thank you for agreeing to participate
- Explain the purpose of interview
- Go over issues of confidentiality, anonymity
- Permission to audio record
- Consent form to be signed
- Any questions
- Structure of the interview: general contextual information, then the main part of the interview exploring two examples from your practice with families where there were children and young people; and finally some more general questions regarding your work with children.

## 2 General background and contextual information

In order to set the scene it would be helpful to gain a picture of you and your work setting.

#### Personal information

- What is your job title?
- How long have you been in this post?
- How long in palliative care social work?
- How long (approximately) in social work?
- How would you describe your work place e.g. hospice (what type, inpatient)?
- Who is your employer, e.g. hospice, local authority, hospital?
- Are you the only social worker in your work unit?

#### Palliative care team

- Are you a member of a palliative care team?
- If yes, who are the other members of the team?
- How do you as the social worker receive referrals?

PROBE: are you clear about your own role and responsibility and that of others in terms of referral?

 How would you describe the work of your team with regard to families with children under 18?

PROMPT - 'family-focused'?

- How is information generally gathered about the family of the patient?
- How are families identified as being 'at risk'?

PROBE - How does the team perceive children at risk?

#### **3** Practice examples

In this part of the interview, I would like to explore two situations where you had concerns about the welfare of a child/children. One where you believe the intervention and outcome were successful and another where they were less successful. You choose which one you would like to talk about first. I am particularly interested in the process of the work and the outcomes.

- So, please describe a situation where you had concerns about the welfare of a child/children.
- What were your concerns?
- What were these concerns based on?
- How did you carry out an assessment?
- Did you meet with the child?
- How did you feel about the child, the family and their situation?
- What action did you take as a result of this assessment?

PROBE - did the illness of the parent affect your judgement in any way?

PROBE - To what extent did you consider development theories or anything else?

PROBE - To what extent was your decision influenced by past experiences?

- Were other people involved in the assessment/decision making?
  - If so, who and why?
- Can you tell me about the outcomes (as far as you know) for

-the child?

-the parent/s?

-you?

- How would you describe these outcomes?
   successful: unsuccessful
- Thinking back on this situation now, what are your thoughts about the actions taken and the outcomes?

Thank you. I'd now like to ask you about another situation where the outcome was successful / less successful (depending on choice of first example). Repeat questions.

• How typical were these two cases would you say ?

#### Thanks for all that information.

I would now like to move on to your links with the local authority.

## 4 Decision making - referral to local authority

From what you have said and information I have gained generally from the questionnaires and focus groups, workers have differing views about links with the local authority.

• What factors would influence whether or not you would refer a child/family to the local authority child and family team?

PROMPT - Are your decisions to refer made on the basis of previous responses from the local authority?

PROMPT - Are they based on the level of risk?

- Do these factors affect your approach?
- How, in general, would you describe your working relationship with staff in the local authority?

PROBE - Are you able to discuss your concerns about children and families?

(depending on above answer)

• What if anything would help to make these relationships more effective?

Thanks. I would like ask you some questions regarding safeguarding children in general.

## 5 Safeguarding children

• What does the term 'safeguarding children' mean to you?

PROMPT - keeping children safe, legal terminology

- How do you see this as part of your role?
- The term 'the child's welfare is paramount' is a legal term what does this term mean to you in your work as a palliative care social worker?
- The child's views how are these heard?
- The expression 'good enough parenting' is one that is often used by social workers when assessing child welfare situations. What does this expression mean to you?
- Would this understanding vary in circumstances where a parent has a terminal illness?
- •

PROMPT - behaviour of parent due to brain tumour - pace of disease progression - lone parent

- Are there any other factors, personal or professional, that may affect your work with regard to children?
- •

PROMPT - children of own; personal knowledge/experience of terminal illness

• If yes, how do you think this experience influences your work with families and children?

## 6 Finally

Please complete the following sentence:

'I feel I could be more effective in cases of child welfare if.....

• Is there anything else that you would like to add?

Thank you for your time and consideration.

## INFORMED CONSENT FORM

Focus Groups

## TITLE OF PROJECT

## Safeguarding Children: The Role of the Palliative Care Social Worker

This research study is an exploration of the role of the palliative care social worker in situations where a main carer has a terminal illness and the child/children in the family may be vulnerable and at risk.

The study is part of a PhD based at the University of Dundee in the School of Education, Social Work and Community Education. The Research Ethics Committee of the School has reviewed and approved this research study.

By signing below you are agreeing that you have read and understood the Participant Information Sheet and that you agree to take part in this research study.

Carole Comben

Printed name of person obtaining consent

Signature of person obtaining consent

Date

Thank you very much.

## INFORMED CONSENT FORM

Interviews

## TITLE OF PROJECT

## Safeguarding Children: The Role of the Palliative Care Social Worker

This research study is an exploration of the role of the palliative care social worker in situations where a main carer has a terminal illness and the child/children in the family may be vulnerable and at risk.

The study is part of a PhD based at the University of Stirling in the Department of Applied Social Studies. The Research Ethics Committee of the Department has reviewed and approved this research study.

I have read and understood the 'Participant Information Sheet – Interviews' and agree to take part in this research study. I give my consent to the interview being audio-recorded.

Participant's signature

Date

Carole Comben

Printed name of person obtaining consent Signature of person obtaining consent

Thank you very much...

## Data collection and analysis: illustrations of process

In chapter five, I describe how the preliminary analysis of data helped inform subsequent stages of the data collection. Three illustrations of this process are given below which demonstrate how the analysis was built up across the datasets in slightly different ways and how, in illustration two, reference was also made to findings from the literature.

#### **Illustration one**

The preliminary analysis of the three focus groups highlighted the palliative care social workers' relationships with social workers within the local authority children's services. These were reported to be more negative than positive and there were some very strongly held views about, as one worker described it, the *'draconian way'* the services were delivered. The workers were concerned about the changing standards and thresholds regarding access to children's services with, as they saw it, a lack of emphasis on prevention and children in need. Those who had previously worked in child protection shared examples about their ability to better negotiate their way to obtain services. Whilst there were examples of positive experiences of work in partnership and obtaining appropriate services for the family, these were in the minority.

As I saw liaison with local authority children's services to be an important element of the palliative care social workers' safeguarding role, I followed up this aspect in the survey by asking the following:

Please describe your contact with social workers in the local authority child and family and child protection teams in your area. (Where possible, please illustrate your answer with an example of where this contact has had a positive outcome and an example of where the contact has been less successful.)

By asking for examples in this way, I was inviting workers to share positive experiences as well as those of a more negative nature to help me gain a rounded picture. A preliminary analysis showed once again that there was a variety of experiences; a few workers had very minimal contact. There were those who described their contact as '*varied*' and those who were able to give useful examples of a positive outcome but the overall tenor of the experiences was less than positive.

During the interviews, I hoped through the sharing of practice examples and the asking of particular questions to clarify two issues. The first centred around the factors that might influence the palliative care social workers in their decision to refer, or not to refer, a child to the local authority. The preliminary analyses of the focus groups and survey had shown that these factors might be based on previous responses from the local authority. However, I wanted to establish whether there were any other factors and if so, how they affected the approach the workers might take. The second centred around the working relationships between the palliative care social workers and those in the local authority children's services and what, in their opinion, would help to make these relationships more effective. If, as it had appeared so far, these relationships were less than positive, then ideas about how this might be changed would be important information to share with others.

The analysis from all three data sets helped to form a more composite picture of this aspect of the role of the palliative care social worker in safeguarding children and forms the basis of chapter ten: *The role of the palliative care social worker: liaising with local authority children's services.* 

#### Illustration two

One theme that arose from the preliminary analysis of the focus group data was balancing the needs of the dying parent and those of the child. This was vividly illustrated by one worker who said their role in safeguarding children was:

'...to have compassion with the person who is dying but if you take it too far you can miss the child who is at risk - need to protect the child. We have to hold the balance - not be judgmental and protecting children.'

The literature review highlighted difficulties that other workers in adult services experience when the focus of their work is with an adult who is also a parent. I wondered how far the terminal illness of a parent was a factor that might influence the decisions the palliative care social workers might make regarding the care being offered to the child. In the survey, therefore, I asked a specific question: *'What influence do you think the nature of the illness of a parent may have on your decisions about the child care offered by that person and the family?'* The preliminary analysis of this question, as well as that from information gathered from the practice examples given in response to questions 10 and 11, showed that there was a recognition of the dilemma of balancing the needs of the parent and child in what was known to be a short-term situation. There were those workers who clearly stated that the child's welfare was paramount, whilst

others were less categorical. I considered this to be an important aspect of their safeguarding role and sought in the interviews to widen this issue by asking what the expression 'good enough parenting' (a phrase often used by social workers when assessing child welfare situations) meant to them and whether this understanding might vary in circumstances where a parent has a terminal illness. The preliminary analysis showed that giving a clear definition was subjective, difficult to do and that standards differed. There was also recognition that the illness might prevent parents from caring for their children in the same way as they had previously. The analysis from the three datasets appears in chapter seven.

#### **Illustration three**

During the preliminary analysis of the data from the focus groups on the above theme, it was apparent that palliative care social workers found that the medical and nursing staff sometimes had differing opinions about parenting and children's needs. It was reported that some staff were more forgiving of the parent's behaviour because of the illness and there were others who focused their attention on the parent who was dying and did not consider the needs of the children to be a priority. As the analysis from the focus groups had highlighted that nearly all the referrals to a palliative care social worker were made by a third party, usually a member of the nursing staff, I wondered how their attitudes might influence the referral process. There was no direct question about this in the survey as choices had to be made regarding its length and direction. However, question 15 did ask for the workers' opinions about how other members of the palliative care team viewed their role within the team in relation to safeguarding children. A preliminary analysis showed that the workers considered they were seen as a resource in child care matters and that staff members would come to them if they had concerns.

As children can only be safeguarded by the palliative care social worker if they are known to them, I considered that there should be a place within the interview to explore the issue of referrals in some depth. Specific questions were asked about how the workers received referrals, whether there was clarity about their role in this and how information was generally gathered about the family of the patient. The referral process was also described within the practice examples explored later in the interviews. The analyses from the interviews, focus groups and the survey are presented in chapter eight.

## **Glossary of terms: chapter nine**

Included here are brief definitions of terms used in chapter nine. Further information on social work methods can be found in the following texts:

Coulshed, V. and Orme, J. (2006) *Social Work Practice: An Introduction (4<sup>th</sup> ed).* Basingstoke: Palgrave Macmillan.

Lishman, J. (ed) (1991) *Handbook of Theory for Practice Teachers in Social Work*. London: Jessica Kingsley Publishers.

Parker, J. and Bradley, G. (2003) *Social Work Practice: Assessment, Planning, Intervention and Review.* Exeter: Learning Matters.

Watson, D. and West, J. (2006) *Social Work Process and Practice: Approaches, Knowledge and Skills.* Basingstoke: Palgrave Macmillan.

#### **Brief solution approaches**

'A short-term, strengths based method emphasising the utilisation of existing coping mechanisms/skills to resolve new challenges' (Watson and West 2006:53-54).

#### Casework

Refers to work with individuals and/or families. This is a common form of social work practice which involves 'a worker engaging with an individual, couple or a whole family, depending on the circumstances, forming a picture of the situation and attempting to move forward accordingly' (Thompson 2010:41).

#### Cognitive behaviour therapy

Based on learning theory it is a brief and highly structured approach. The aim is to teach coping skills, thus empowering the client and avoiding dependence on the therapist (Coulshed and Orme 1998).

#### Counselling

The process whereby a trained counsellor explores a difficulty a person is having, distress they may be experiencing or the dissatisfaction they have with their life, or loss of a sense of direction or purpose (The British Association for Counselling and Psychotherapy).

#### **Crisis intervention**

'A short term method of intervention utilising a crisis situation as a catalyst for change' (Watson and West 2006:53-54).

#### **Psychosocial approach**

Usually linked to the writings of Florence Hollis (1964), the goal is to assist the person or situation, or both, by reducing the internal and/or external conflict. The focus will be on the behaviour but there is also an attempt to understand and resolve inner conflicts by sustaining and modifying procedures. Sustaining procedures include offering support and building relationships; and modifying procedures aim to reduce 'outer' pressures and enable the client to gain insight (Coulshed and Orme 1998:146-148).

## Psychodynamic perspective

The value of this perspective is to understand human behaviour. Psychodynamic thinking is seen as predominately concerned with certain key relationships: those between self and significant other people; past and present experience; inner and outer reality. The approach has its basis in psychoanalysis and specifically the work of Freud, but incorporates more recent analytic thought (Brearley 1991:49-62).

#### Solution-focused approach

Focuses on a person's strengths and solutions rather than the problem areas (Hepworth et al. 2010).

#### Systems approach

'Considers the impact of social structures, the kinds of help and support provided and access to them have on the ways in which individuals, families, groups and communities respond' (Parker and Bradley 2007:89). [For more information see chapter five].

#### Task centred approach

'A short-term systematic method focused on the contractual completion of achievable and agreed tasks' (Watson and West 2006:53-54).